## UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

## FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 х

#### For the quarterly period ended March 31, 2020

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 

Commission file number: 001-38399

AdaptHealth Corp.

(Exact name of registrant as specified in its charter)

Delaware

(State of Other Jurisdiction of incorporation or Organization)

82-3677704

(I.R.S. Employer Identification No.)

220 West Germantown Pike Suite 250, Plymouth Meeting, PA (Address of principal executive offices)

19462

(Zip code)

Registrant's telephone number, including area code: (610) 630-6357

Securities registered pursuant to Section 12(b) of the Act:

		Name Of Each Exchange
Title of Each Class	Trading Symbol(s)	On Which Registered
Class A Common Stock, par value \$0.0001 per share	АНСО	The Nasdaq Stock Market LLC

#### Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes x No

Indicate by check mark whether the Registrant has submitted electronically; every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.0405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes X No 🗆

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer $\square$	Accelerated filer <b>x</b>	Non-accelerated filer □	Smaller reporting company <b>x</b> Emerging growth company <b>x</b>

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes  $\square$  No **x** 

As of May 1, 2020, there were 45,432,240 shares of the Registrant's Class A Common Stock issued and outstanding and 28,610,250 shares of the Registrant's Class B Common Stock issued and outstanding.

# TABLE OF CONTENTS

Page

## Number PART I FINANCIAL INFORMATION Item 1. Consolidated Interim Financial Statements (Unaudited) Consolidated Balance Sheets as of March 31, 2020 and December 31, 2019 3 Consolidated Statements of Operations for the three months ended March 31, 2020 and 2019 4 Consolidated Statements of Comprehensive Loss for the three months ended March 31, 2020 and 2019 5 Consolidated Statements of Changes in Stockholders' Deficit for the three months ended March 31, 2020 and 2019 6 Consolidated Statements of Cash Flows for the three months ended March 31, 2020 and 2019 7 Notes to Consolidated Interim Financial Statements 8 Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations 26 Item 3. Quantitative and Qualitative Disclosures About Market Risk 38 Item 4. Controls and Procedures 38 PART II OTHER INFORMATION 39 Item 1. Legal Proceedings 39 Item 1A. Risk Factors 39 Item 2. Unregistered Sales of Equity Securities and Use of Proceeds 42 Item 3. Defaults upon Senior Securities 42 Item 4. Mine Safety Disclosure 42 Item 5. Other Information 42 Item 6. Exhibits 42 **Signatures** 46

# CAUTIONARY STATEMENT

In this Quarterly Report on Form 10-Q, including "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Part I Item 2, and the documents incorporated by reference herein, we make forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements relate to expectations for future financial performance, business strategies or expectations for our business. These statements may be preceded by, followed by or include the words "may," "might," "will, "will likely result," "should," "estimate," "plan," "project," "forecast," "intend," "expect," "anticipate," "believe," "seek," "continue," "target" or similar expressions.

These forward-looking statements are based on information available to us as of the date they were made, and involve a number of risks and uncertainties which may cause them to turn out to be wrong. Accordingly, forward-looking statements should not be relied upon as representing our views as of any subsequent date, and we do not undertake any obligation to update forward-looking statements to reflect events or circumstances after the date they were made, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws. As a result of a number of known and unknown risks and uncertainties, our actual results or performance may be materially different from those expressed or implied by these forward-looking statements. Some factors that could cause actual results to differ include:

- the ability to maintain the listing of our Class A Common Stock on Nasdaq;
- competition and the ability of our business to grow and manage growth profitably;
- changes in applicable laws or regulations;
- fluctuations in the U.S. and/or global stock markets;
- the possibility that we may be adversely affected by other economic, business, and/or competitive factors;
- the impact of the recent coronavirus (COVID-19) pandemic and our response to it; and
- other risks and uncertainties set forth in this Form 10-Q, as well as all documents incorporated by reference herein.

## Consolidated Balance Sheets (Unaudited)

	March 31, 2020	D	ecember 31, 2019
Assets			
Current assets:			
Cash and cash equivalents	\$ 48,163,701	\$	76,878,134
Accounts receivable	118,650,575		78,619,230
Inventory	17,967,948		13,239,037
Prepaid and other current assets	 9,759,439		12,678,423
Total current assets	 194,541,663		181,414,824
Equipment and other fixed assets, net	87,300,924		63,559,080
Goodwill	340,806,853		266,790,518
Other assets	5,670,301		6,851,892
Deferred tax assets	 33,518,857		27,505,379
Total Assets	\$ 661,838,598	\$	546,121,693
Liabilities and Stockholders' Deficit			
Current liabilities:			
Accounts payable and accrued expenses	\$ 136,071,887	\$	102,728,093
Current portion of capital lease obligations	20,421,195		19,749,854
Current portion of long-term debt	2,615,705		1,721,132
Contract liabilities	15,584,066		9,556,423
Other liabilities	 16,459,388		17,138,684
Total current liabilities	 191,152,241		150,894,186
Long-term debt, less current portion	 463,552,896		395,111,563
Other long-term liabilities	36,580,187		29,364,151
Total Liabilities	 691,285,324		575,369,900
Commitments and contingencies (note 14)		_	
Stockholders' Deficit			
Class A Common Stock, par value of \$0.0001 per share, 210,000,000 shares authorized; 43,354,251			
and 40,816,292 shares issued and outstanding as of March 31, 2020 and December 31, 2019,			
respectively	4,336		4,082
Class B Common Stock, par value of \$0.0001 per share, 35,000,000 shares authorized; 30,563,799 and			
31,563,799 shares issued and outstanding as of March 31, 2020 and December 31, 2019, respectively	3,056		3,156
Preferred Stock, par value of \$0.0001 per share, 5,000,000 shares authorized; 0 shares issued and			
outstanding as of March 31, 2020 and December 31, 2019			
Additional paid-in capital	21,843,967		11,252,052
Accumulated deficit	(27,367,676)		(27,209,514)
Accumulated other comprehensive (loss) income	 (5,139,138)		1,431,029
Total stockholders' deficit attributable to AdaptHealth Corp.	(10,655,455)		(14,519,195)
Noncontrolling interests in subsidiaries	(18,791,271)		(14,729,012)
Total Stockholders' Deficit	 (29,446,726)		(29,248,207)
Total Liabilities and Stockholders' Deficit	\$ 661,838,598	\$	546,121,693

See accompanying notes to consolidated interim financial statements.

# Consolidated Statements of Operations (Unaudited)

	 Three Months Ended March 31,		
	 2020		2019
Net revenue	\$ 191,439,034	\$	119,498,274
Costs and expenses:			
Cost of net revenue	166,539,690		100,226,876
General and administrative expenses	14,346,919		13,082,631
Depreciation, excluding patient equipment depreciation	1,241,837		840,722
Total costs and expenses	 182,128,446		114,150,229
Operating income	 9,310,588		5,348,045
Interest expense, net	7,938,243		6,260,331
Loss on extinguishment of debt			2,121,451
Income (loss) before income taxes	 1,372,345	_	(3,033,737)
Income tax expense	1,106,722		2,418,441
Net income (loss)	 265,623	_	(5,452,178)
Income attributable to noncontrolling interests	423,785		348,139
Net loss attributable to AdaptHealth Corp.	\$ (158,162)	\$	(5,800,317)
Net loss per common share attributable to AdaptHealth Corp.:			
Basic and diluted	\$ —	\$	(0.42)
Weighted average shares outstanding for net loss attributable to AdaptHealth Corp.:			
Basic and diluted	41,976,560		13,863,570

See accompanying notes to consolidated interim financial statements.

# Consolidated Statements of Comprehensive Loss (Unaudited)

	-	Three Months Ended March 31,				
		2020		2019		
Net income (loss)	\$	265,623	\$	(5,452,178)		
Other comprehensive loss:						
Interest rate swap agreements, inclusive of reclassification adjustment		(11,417,216)				
Comprehensive loss		(11,151,593)		(5,452,178)		
			-			
Income attributable to noncontrolling interests		423,785		348,139		
Comprehensive loss attributable to AdaptHealth Corp.	\$	(11,575,378)	\$	(5,800,317)		

See accompanying notes to consolidated interim financial statements.

# Consolidated Statements of Changes in Stockholders' Deficit (Unaudited)

	Class A Co Shares	mmon Stock Amount	Class B Cor Shares	mmon Stock Amount	Additional paid-in capital	Members' interest	Controlling interest members' deficit	Accumulated deficit	Accumulated other comprehensive income (loss)	Noncontrolling interests in subsidiaries	Total
Balance, December 31, 2019	40,816,292	\$ 4,082	31,563,799	\$ 3,156	\$11,252,052	\$ —	\$ —	\$(27,209,514)	\$ 1,431,029	\$ (14,729,012)	\$(29,248,207)
Issuance of Class A Common Stock for											
an acquisition	386,874	39			6,247,976					_	6,248,015
Exchange of Class B Common Stock											
for Class A Conmon Stock	1,000,000	100	(1,000,000)	(100)	(361,005)	_			—	361,005	
Cashless exercise of warrants	1,092,468	109	—	_	(109)	_	_	—	—	_	—
Equity-based compensation	58,617	6	—	—	2,222,603	—		—	—	—	2,222,609
Net income (loss)								(158,162)		423,785	265,623
Equity activity resulting from Tax Receivable Agreement	_	_	_	_	2,482,450	_	_	_	_	_	2,482,450
Change in fair value of interest rate swaps, inclusive of reclassification adjustment	_	_	_	_	_	_	_	_	(6,570,167)	(4,847,049)	(11,417,216)
Balance, March 31, 2020	43,354,251	\$ 4,336	30,563,799	\$ 3,056	\$21,843,967	\$ _	<u>\$                                    </u>	\$(27,367,676)	\$ (5,139,138)	\$ (18,791,271)	\$(29,446,726)

					Additional		Controlling interest		Accumulated other	Noncontrolling	
		Common tock Amount		Common tock Amount	paid-in capital	Members' interest	members' deficit	Accumulated deficit	comprehensive income	interests in subsidiaries	Total
Balance, December 31, 2018		\$ -		\$ —	\$ -			\$ -	\$ -	\$ 2,865,125	\$ 102,768,658
Issuance of members' interest, net											
of offering costs of \$837,156	_	_	_			19,162,844	_	_			19,162,844
Redenption of members' interest	_					(2,112,500)	(1,600,955)	_			(3,713,455)
Distributions to members						_	(250,000,000)				(250,000,000)
Equity-based compensation	_					5,223,108		_			5,223,108
Net income (loss)	_	_	_	_	_	_	(5,800,317)	_	_	348,139	(5,452,178)
Balance, March 31, 2019		\$ —		\$ —	\$ —	\$ 135,547,633	\$ (270,771,920)	\$	s —	\$ 3,213,264	\$ (132,011,023)

See accompanying notes to consolidated interim financial statements.

# Consolidated Statements of Cash Flows (Unaudited)

	Three Months	Ended	,
	2020		2019
Cash flows from operating activities:	¢ 0.00 000	•	(5.450.180)
Net income (loss)	\$ 265,623	\$	(5,452,178)
Adjustments to reconcile net income (loss) to net cash provided by operating activities:	16 740 154		14.071 (20
Depreciation, including patient equipment depreciation	16,740,154		14,971,628
Equity-based compensation	2,222,609		5,223,108
Deferred income tax	479,092		1,483,430
Change in fair value of interest rate swaps, net of reclassification adjustment	(707,381)		2,702,400
Change in fair value of contingent consideration	(2,000,000)		12 ( 001
Amortization of deferred financing costs	391,591		136,801
Write-off of deferred financing costs	(500 701)		2,121,451
Gain on sale of investment	(590,701)		
Changes in operating assets and liabilities, net of effects from acquisitions:	(20, 157, 570)		(0.(2(.020
Accounts receivable	(20,457,579)		(8,636,030
Inventory	51,528		(161,065
Prepaid and other assets	3,908,676		(172,468
Accounts payable and accrued expenses	24,076,642		4,015,039
Net cash provided by operating activities	24,380,254		16,232,116
Cash flows from investing activities:			
Purchases of equipment and other fixed assets	(7,534,433)		(5,297,940
Proceeds from sale of investment	2,045,701		
Payments for business acquisitions, net of cash acquired	(105,840,930)		(20,881,343)
Net cash used in investing activities	(111,329,662)		(26,179,283)
Cash flows from financing activities:			
Proceeds from borrowings on long-term debt and lines of credit	70,000,000		317,000,000
Repayments on long-term debt and lines of credit	(984,480)		(151,916,121
Payments on capital leases	(10,780,545)		(9,874,276
Proceeds from issuance of promissory note payable	_		100,000,000
Proceeds from issuance of members' interests	—		20,000,000
Payments for equity issuance costs	_		(837,156
Payments of deferred financing costs	_		(9,027,753
Distributions to members	_		(250,000,000
Payment of contingent consideration	_		(12,000,000
Payments for redemption of members' interests	_		(3,713,455
Net cash provided by (used in) financing activities	58,234,975		(368,761
Net decrease in cash and cash equivalents	(28,714,433)		(10,315,928
Cash and cash equivalents at beginning of period	76,878,134		25,185,681
Cash and cash equivalents at end of period	\$ 48,163,701	\$	14,869,753
Supplemental disclosures:	\$ 40,105,701	ψ	14,007,755
Cash paid for interest	\$ 7,704,405	\$	2,829,928
Cash paid for income taxes	2,085,831	Ф	39,244
Noncash investing and financing activities:	2,085,851		39,244
	\$ 9.757.735	¢	0.5(4.44)
Equipment acquired under capital lease obligations		\$	8,564,442
Unpaid equipment and other fixed asset purchases at end of period	7,814,170		11,073,629
Equity consideration issued in connection with an acquisition	6,248,015		1 500 000
Contingent purchase price in connection with an acquisition Seller note issued in connection with an acquisition			1,500,000
Seller note issued in connection with an acquisition See accompanying notes to consolidated interim fi			2,000,000

See accompanying notes to consolidated interim financial statements.

## Notes to Consolidated Interim Financial Statements (Unaudited)

## (1) General Information

AdaptHealth Corp. and subsidiaries (AdaptHealth or the Company), f/k/a DFB Healthcare Acquisitions Corp. (DFB) is a leading provider of home healthcare equipment, medical supplies to the home and related services in the United States. AdaptHealth focuses primarily on providing (i) sleep therapy equipment, supplies and related services (including CPAP and bi-PAP services) to individuals suffering from obstructive sleep apnea (OSA), (ii) home medical equipment (HME) to patients discharged from acute care and other facilities, (iii) oxygen and related chronic therapy services in the home, and (iv) other HME medical devices and supplies on behalf of chronically ill patients with diabetes care, wound care, urological, ostomy and nutritional supply needs. AdaptHealth services beneficiaries of Medicare, Medicaid and commercial insurance payors.

On July 8, 2019, AdaptHealth Holdings LLC (AdaptHealth Holdings) entered into an Agreement and Plan of Merger (the Merger Agreement), as amended on October 15, 2019, with DFB, pursuant to which AdaptHealth Holdings combined with DFB (the Business Combination). The Business Combination closed on November 8, 2019. Refer to Note 3, *Significant Transactions*, for additional information regarding the Business Combination.

Unless the context otherwise requires, "the Company", "we," "us," and "our" refer, for periods prior to the closing of the Business Combination, to AdaptHealth Holdings and its subsidiaries and, for periods upon or after the closing of the Business Combination, to AdaptHealth Corp. and its subsidiaries, including AdaptHealth Holdings and its subsidiaries.

The consolidated interim financial statements are unaudited, but reflect all normal recurring adjustments that are, in the opinion of management, necessary to fairly present the information set forth herein. The interim consolidated financial statements should be read in conjunction with the audited consolidated financial statements and related notes included in the Company's Annual Report on Form 10-K for the year ended December 31, 2019. Interim results are not necessarily indicative of the results for a full year.

There have been no material changes in the Company's significant accounting policies as compared to the significant accounting policies described in the Company's Annual Report on Form 10-K for the year ended December 31, 2019.

#### (a) Basis of Presentation

The consolidated interim financial statements of the Company have been prepared in accordance with accounting principles generally accepted in the United States of America (U.S. GAAP). In the opinion of management, the consolidated interim financial statements include all necessary adjustments for a fair presentation of the financial position and results of operations for the periods presented.

The Business Combination was accounted for as a reverse recapitalization, with DFB treated as the acquired company and AdaptHealth Holdings as the acquirer, for financial reporting purposes. Therefore, the equity structure has been restated to that of the Company.

The Company is an "emerging growth company," as defined in Section 2(a) of the Securities Act of 1933, as amended, (the Securities Act), as modified by the Jumpstart our Business Startups Act of 2012, (the JOBS Act), and it may take advantage of certain exemptions from various reporting requirements that are applicable to other public companies that are not emerging growth companies including, but not limited to, not being required to comply with the auditor attestation requirements of Section 404 of the Sarbanes-Oxley Act, reduced disclosure obligations regarding executive compensation in its periodic reports and proxy statements, and other exemptions.

## Notes to Consolidated Interim Financial Statements (Unaudited)

## (b) Basis of Consolidation

The accompanying consolidated financial statements include the accounts of the Company and its wholly-owned subsidiaries. All intercompany balances and transactions have been eliminated in consolidation.

## (c) Concentration of Credit Risk

Financial instruments which potentially subject the Company to concentrations of credit risk consist principally of cash and trade accounts receivable. The Company maintains its cash in bank deposit accounts, which, at times, may exceed federally insured limits. The Company has not experienced any losses in such accounts and believes it is not exposed to any significant credit risk on cash and cash equivalents.

## (d) Accounting Estimates

The preparation of consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, and the disclosure of contingent assets and liabilities at the date of the consolidated financial statements, and reported amounts of revenues and expenses during the reporting period. Management bases these estimates and assumptions upon historical experience, existing and known circumstances, authoritative accounting pronouncements and other factors that management believes to be reasonable. Significant areas requiring the use of management estimates relate to revenue recognition and the valuation of accounts receivable (implicit price concession), income taxes, contingent consideration, equity-based compensation, interest rate swaps, and long-lived assets, including goodwill. Actual results could differ from those estimates.

#### (e) Business Segment

The Company's chief operating decision-makers are its Chief Executive Officer and President, who make resource allocation decisions and assess performance based on financial information presented on an aggregate basis. There are no segment managers who are held accountable by the chief operating decision-makers, or anyone else, for any planning, strategy and key decision-making regarding operations. The corporate office is responsible for contract negotiation with vendors and payors, corporate compliance with healthcare laws and regulations, and revenue cycle management. Accordingly, the Company has a single reportable segment and operating segment structure.

## (f) Recently Issued Accounting Pronouncements

In February 2016, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) No. 2016-02, *Leases (Topic 842)* (ASU 2016-02), which amended authoritative guidance on accounting for leases. The new provisions require that a lessee of operating leases recognize a liability to make lease payments and a right-of-use asset representing its right to use the underlying asset for the lease term. The Company is required to adopt the new standard for the annual reporting period beginning January 1, 2021, and interim reporting periods beginning January 1, 2022. The adoption of this standard is expected to have a material impact on the Company's financial position. The Company is still evaluating the impact on its results of operations and does not expect the adoption of this standard to have an impact on liquidity.

In January 2017, the FASB issued ASU 2017-04, *Intangibles - Goodwill and Other (ASC Topic 350): Simplifying the Test for Goodwill Impairment*, which will eliminate the requirement to calculate the implied fair value of goodwill, commonly referred to as "Step 2" in the current goodwill impairment test. An entity will still have the option to perform the qualitative assessment for a reporting unit to determine if the quantitative impairment test is necessary. The Company adopted this standard on January 1, 2020, which did not have a material impact on the Company's consolidated financial statements.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

## (2) Revenue Recognition and Accounts Receivable

#### **Revenue Recognition**

The Company generates revenues for services and related products that the Company provides to patients for home medical equipment, related supplies, and other items. The Company's revenues are recognized in the period in which services and related products are provided to customers and are recorded either at a point in time for the sale of supplies and disposables, or over the fixed monthly service period for equipment.

Revenues are recognized when control of the promised good or service is transferred to customers, in an amount that reflects the consideration to which the Company expects to receive from patients or under reimbursement arrangements with Medicare, Medicaid and third-party payors, in exchange for those goods and services.

Sales revenue is recognized upon transfer of control of products or services to customers in an amount that reflects the consideration the Company expects to receive in exchange for those products or services. Revenues for the sale of durable medical equipment and related supplies, including oxygen equipment, ventilators, wheelchairs, hospital beds and infusion pumps, are recognized at the time of delivery.

The Company provides certain equipment to patients which is reimbursed periodically in fixed monthly payments for as long as the patient is using the equipment and medical necessity continues (in certain cases, the fixed monthly payments are capped at a certain amount). The equipment provided to the patient is based upon medical necessity as documented by prescriptions and other documentation received from the patient's physician. The patient generally does not negotiate or have input with respect to the manufacturer or model of the equipment prescribed by their physician and delivered by the Company. Once initial delivery of this equipment is made to the patient for initial setup, a monthly billing process is established based on the initial setup service date. The Company recognizes the fixed monthly revenue ratably over the service period as earned, less estimated adjustments, and defers revenue for the portion of the monthly bill that is unearned. No separate revenue is earned from the initial setup process. Included in fixed monthly revenue are unbilled amounts for which the revenue recognizion criteria had been met as of period-end but were not yet billed to the payor. The estimate of net unbilled fixed monthly revenue recognized is based on historical trends and estimates of future collectability.

The Company disaggregates net revenue from contracts with customers by payor type and by core service lines. The Company believes that disaggregation of net revenue into these categories depicts how the nature, amount, timing and uncertainty of revenue and cash flows are affected by economic factors. The payment terms and conditions within the Company's revenue-generating contracts vary by payor type and payor source.

The composition of net revenue by payor type for the three months ended March 31, 2020 and 2019 are as follows:

	Three Mon Marc	nths Ended h 31,
	2020	2019
Insurance	\$114,450,697	\$ 67,717,162
Government	51,244,994	38,100,765
Patient pay	25,743,343	13,680,347
Net revenue	\$191,439,034	\$119,498,274

#### Notes to Consolidated Interim Financial Statements (Unaudited)

The composition of net revenue by core service lines for the three months ended March 31, 2020 and 2019 are as follows:

	Three Mon Marc	nths Ended h 31,
	2020	2019
Net sales revenue:		
Sleep	\$ 68,893,964	\$ 47,127,169
Supplies to the home	33,338,901	2,028,936
HME	11,579,127	10,489,009
Respiratory	2,768,427	1,279,075
Other	12,393,306	8,031,775
Total net sales revenue	\$ 128,973,725	\$ 68,955,964
Net revenue from fixed monthly equipment reimbursements:		
Sleep	\$ 22,668,559	\$ 18,056,858
HME	12,177,277	10,242,636
Respiratory	25,006,951	20,429,189
Other	2,612,522	1,813,627
Total net revenue from fixed monthly equipment reimbursements	\$ 62,465,309	\$ 50,542,310
Total net revenue:		
Sleep	\$ 91,562,523	\$ 65,184,027
Supplies to the home	33,338,901	2,028,936
HME	23,756,404	20,731,645
Respiratory	27,775,378	21,708,264
Other	15,005,828	9,845,402
Total net revenue	\$ 191,439,034	\$ 119,498,274

## Accounts Receivable

Due to the continuing changes in the healthcare industry and third-party reimbursement environment, certain estimates are required to record accounts receivable at their net realizable values. Inherent in these estimates is the risk that they will have to be revised or updated as additional information becomes available. The complexity of third-party billing arrangements and laws and regulations governing Medicare and Medicaid may result in adjustments to amounts originally recorded.

The Company performs a periodic analysis to review the valuation of accounts receivable and collectability of outstanding balances. Management's evaluation takes into consideration such factors as historical bad debt experience, business and economic conditions, trends in healthcare coverage, other collection indicators and information about specific receivables. The Company's evaluation also considers the age and composition of the outstanding amounts in determining their estimated net realizable value.

Included in accounts receivable are earned but unbilled accounts receivables. Billing delays, ranging from several days to several weeks, can occur due to the Company's policy of compiling required payor specific documentation prior to billing for its services rendered. The Company recorded unbilled revenue of \$18,064,863 and \$8,611,272 as of March 31, 2020 and December 31, 2019, respectively.

## Notes to Consolidated Interim Financial Statements (Unaudited)

## (3) Significant Transactions

## Acquisitions

During the three months ended March 31, 2020 and 2019, the Company made several acquisitions to strengthen its current market share in existing markets or to expand into new markets. The goodwill generated from these acquisitions is attributable to expected growth and cost synergies and the expected contribution of each acquisition to the overall Company strategy and is expected to be deductible for tax purposes. The estimated fair values of the net assets of acquired businesses as described below are subject to change resulting from such items as working capital adjustments post-acquisition. As a result, the acquisition accounting for certain acquired businesses could change in subsequent periods resulting in adjustments to goodwill once finalized.

#### Three Months Ended March 31, 2020

On January 2, 2020, the Company purchased 100% of the equity interests of the Patient Care Solutions business (PCS), a subsidiary of McKesson Corporation. PCS is a home medical equipment supplies business. The Company allocated the consideration paid to the estimated fair values of the net assets acquired on a preliminary basis, including \$16.3 million to accounts receivable, \$0.5 million to equipment and other fixed assets, \$1.4 million to goodwill, and \$3.2 million of net liabilities to other working capital accounts. In addition, on March 2, 2020, the Company purchased certain assets of the durable medical equipment business of Advanced Home Care, Inc. (Advanced). The Company allocated the consideration paid to the estimated fair values of the net assets acquired on a preliminary basis, including \$18.5 million to equipment and other fixed assets, \$38.5 million to goodwill, and \$1.5 million of net assets acquired on a preliminary basis, including \$18.5 million of Advanced also includes a potential contingent payment of up to \$9.0 million based on certain conditions after closing. The Company is in the process of determining the fair value of such contingent payment, and as such an estimated fair value was not included in the consideration paid as part of the Company's preliminary acquisition accounting. The valuation of such payment will be completed during the second quarter of 2020. In addition, during the period, the Company completed acquisitions of certain individually immaterial businesses, the results of which were immaterial to the Company's results for the three months ended March 31, 2020.

The following table summarizes the consideration paid for the acquisitions during the three months ended March 31, 2020:

Cash consideration	\$ 106,178,017
Equity consideration	6,248,015
Deferred payments	14,250
Total	\$ 112,440,282

The Company allocated the consideration paid to the net assets acquired based on their estimated fair values. The Company is still evaluating the fair value of certain assets and liabilities for which provisional amounts were recorded and expects to finalize such evaluation during the second quarter of 2020. Based upon management's evaluation, which is preliminary and subject to completion of working capital and other adjustments, the consideration paid was allocated as follows during the three months ended March 31, 2020:

#### Notes to Consolidated Interim Financial Statements (Unaudited)

Cash	\$ 337,087
Accounts receivable	19,573,766
Inventory	4,780,439
Prepaid and other current assets	1,334,306
Equipment and other fixed assets	24,406,410
Goodwill	74,016,335
Accounts payable and accrued expenses	(6,494,599)
Contract liabilities	(2,467,643)
Unfavorable lease liability	(1,418,931)
Capital lease obligations	(1,626,888)
Net assets acquired	\$ 112,440,282

#### Three Months Ended March 31, 2019

On January 2, 2019, the Company purchased 100% of the equity of Gould's Discount Medical, LLC (Goulds). Goulds is a home medical equipment and supplies business. During the three months ended March 31, 2019, the Company allocated the consideration paid to the estimated fair values of the net assets acquired on a preliminary basis, including \$3.7 million to accounts receivable, \$2.4 million to inventory, \$1.7 million to equipment and other fixed assets, \$18.6 million to goodwill, and \$2.1 million of net liabilities to other working capital accounts. In addition, during the period, the Company completed acquisitions of certain individually immaterial businesses, the results of which were immaterial to the Company's results for the three months ended March 31, 2019.

The following table summarizes the consideration paid for the acquisitions during the three months ended March 31, 2019.

Cash consideration	\$ 21,562,495
Seller note	2,000,000
Estimated contingent consideration	1,500,000
Total	\$ 25,062,495

The Company allocated the consideration paid to the net assets acquired based on their estimated fair values. Based upon management's evaluation, which was preliminary and subject to completion of working capital and other adjustments, the consideration paid was allocated as follows during the three months ended March 31, 2019. The Company finalized the valuation of the fair value of the net assets acquired for these acquisitions during the remainder of 2019.

Cash	\$ 117,000
Accounts receivable	3,691,030
Inventory	2,468,427
Prepaid and other current assets	11,835
Equipment and other fixed assets	1,658,714
Goodwill	19,381,515
Accounts payable and accrued expenses	(2,266,026)
Net assets acquired	\$ 25,062,495

During the three months ended March 31, 2019, the Company received net cash of \$564,152 relating to working capital adjustments associated with businesses that were acquired during 2018 which was recorded as a reduction to goodwill during the period.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

## Pro-Forma Information

The unaudited pro-forma financial information presented below has been prepared by adjusting the historical results of the Company to include the historical results of the significant acquisitions described above. The unaudited pro-forma financial information is presented for illustrative purposes only and may not be indicative of the results of operations that would have actually occurred. In addition, future results may vary significantly from the results reflected in the pro-forma information. The unaudited pro-forma financial information does not reflect the impact of future events that may occur after the acquisitions, such as the impact of cost savings or other synergies that may result from these acquisitions, and does not include interest expense associated with debt incurred to fund the acquisitions.

		Three Months Ended March 31,		
	2020	2019		
Net revenue	\$204,619,618	\$181,724,059		
Operating income (loss)	\$ 9,082,482	\$ (4,746,892)		

The pro-forma operating loss for the three months ended March 31, 2019 is primarily due to operating losses related to PCS.

#### Results of Businesses Acquired

The following table presents the amount of net revenue and operating income (loss) since the respective acquisition dates included in the Company's consolidated statements of operations for the three months ended March 31, 2020 and 2019:

		Three Months Ended March 31,		
	2020	2019		
Net revenue	\$40,724,747	\$ 8,626,378		
Operating income (loss)	\$ (5,559,851)	\$ 1,431,523		

The operating loss for the three months ended March 31, 2020 is primarily due to operating losses related to PCS.

#### **Business Combination**

As discussed in Note 1, *General Information*, on July 8, 2019, AdaptHealth Holdings entered into the Merger Agreement, as amended on October 15, 2019, with DFB, pursuant to which AdaptHealth Holdings combined with DFB. The completion of the Business Combination (the Closing) occurred on November 8, 2019. AdaptHealth Holdings was the accounting acquirer in the Business Combination, which was treated as a reverse recapitalization. Accordingly, for accounting purposes, the merger was treated as the equivalent of AdaptHealth Holdings issuing stock for the net assets of DFB, accompanied by a recapitalization. In connection with the Business Combination, the name of the combined company was changed to AdaptHealth Corp.

Following the Closing of the Business Combination, the holders of Class A Common Stock owned an approximate 56% direct controlling interest, with the remaining 44% direct noncontrolling interest owned by the former owners of AdaptHealth Holdings in the form of common units representing limited liability company interests in AdaptHealth Holdings from and after the Closing (New AdaptHealth Units), which is presented as noncontrolling interest in the consolidated financial statements. These members hold common unit interests of AdaptHealth Holdings and a corresponding number of shares of non-economic Class B Common Stock, which enables the holder to one vote per share. The New AdaptHealth Units and a corresponding number of shares of Class B Common Stock are exchangeable on a one-to-one basis for shares of Class A Common Stock. The holders of New AdaptHealth Units owned an approximate 41% direct noncontrolling economic interest in AdaptHealth Holdings at March 31, 2020. This direct

#### Notes to Consolidated Interim Financial Statements (Unaudited)

noncontrolling interest will continue to decrease as New AdaptHealth Units and a corresponding number of shares of Class B Common Stock are exchanged for shares of Class A Common Stock.

## (4) Equipment and Other Fixed Assets

Equipment and other fixed assets as of March 31, 2020 and December 31, 2019 are as follows:

	March 31,		December 31,
	2020		2019
Patient medical equipment	\$ 132,311,790	\$	112,070,831
Vehicles	6,916,895		4,461,041
Other	18,456,871		15,474,589
	157,685,556		132,006,461
Less accumulated depreciation	(70,384,632)		(68,447,381)
	\$ 87,300,924	\$	63,559,080

## (5) Goodwill

The change in the carrying amount of goodwill for the three months ended March 31, 2020 was as follows:

Balance at December 31, 2019	\$ 266,790,518
Acquired goodwill during the period	 74,016,335
Balance at March 31, 2020	\$ 340,806,853

The Company did not record any goodwill impairment charges during the three months ended March 31, 2020 and 2019.

## (6) Fair Value of Assets and Liabilities

FASB ASC Topic 820, Fair Value Measurements and Disclosures (ASC 820), creates a single definition of fair value, establishes a framework for measuring fair value in U.S. GAAP and expands disclosures about fair value measurements. Assets and liabilities adjusted to fair value in the balance sheet are categorized based upon the level of judgment associated with the inputs used to measure their fair value. Level inputs, as defined by ASC 820, are as follows:

Level input	Input Definition
Level 1	Inputs are unadjusted, quoted prices for identical assets or liabilities in active markets at the
	measurement date.
Level 2	Inputs, other than quoted prices included in Level 1 that are observable for the asset or liability
	through corroboration with market data at the measurement date.
Level 3	Unobservable inputs that reflect management's best estimate of what market participants would
	use in pricing the asset or liability at the measurement date.

The following table presents the valuation of the Company's financial assets and liabilities as of March 31, 2020 and December 31, 2019 measured at fair value on a recurring basis. These estimates are not necessarily indicative of the amounts the Company could ultimately realize.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

	Level 1		Level 2		Level 3 Fai		Fair Value
_		_				_	
\$	4,117,161	\$	—	\$		\$	4,117,161
\$	4,117,161	\$	_	\$	—	\$	4,117,161
\$	—	\$	—	\$	7,675,000	\$	7,675,000
	—		_		5,050,000		5,050,000
	—		5,373,647				5,373,647
	—		13,675,476				13,675,476
\$		\$	19,049,123	\$	12,725,000	\$	31,774,123
	Level 1		Level 2		Level 3		Fair Value
						_	
\$	54,014,591	\$	_	\$		\$	54,014,591
\$	54,014,591	\$		\$		\$	54,014,591
		_				-	
						-	
\$	_	\$	_	\$	4,825,000	\$	4,825,000
\$		\$		\$	4,825,000 9,900,000	\$	4,825,000 9,900,000
\$	=	\$	 2,157,324	\$		\$	
\$		\$	 2,157,324 6,181,964	\$		\$	9,900,000
	\$	\$ 4,117,161 \$ 4,117,161 \$	\$ 4,117,161 \$ \$ 4,117,161 \$ \$ 4,117,161 \$ \$	\$       4,117,161       \$       —         \$       4,117,161       \$       —         \$       4,117,161       \$       —         \$       4,117,161       \$       —         \$       4,117,161       \$       —         \$       4,117,161       \$       —         \$       4,117,161       \$       —         \$       -       \$       -         \$       -       \$       5,373,647         _       -       5,373,647       _         _       -       5,373,647         \$       -       13,675,476       \$         \$       -       \$       19,049,123         Level 1       Level 2       _         \$       54,014,591       \$       —	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

#### Interest Rate Swaps

The Company recognizes its interest rate swaps as either assets or liabilities in the accompanying consolidated balance sheets at fair value. The valuation of these derivative instruments is determined using widely accepted valuation techniques, including discounted cash flow analysis on the expected cash flows of each derivative. This analysis reflects the contractual terms of the derivatives, including the period to maturity, and uses observable market-based inputs, including interest rate curves and implied volatilities. The Company's interest rate swaps held as of March 31, 2020 and December 31, 2019 were classified as Level 2 of the fair value hierarchy. Refer to Note 7, Derivative Instruments and Hedging Activities, for additional information regarding the Company's derivative instruments.

#### Contingent Consideration

The Company estimates the fair value of acquisition-related contingent consideration liabilities by applying the income approach using a probability-weighted discounted cash flow model. This fair value measurement is based on significant inputs not observed in the market and thus represents a Level 3 measurement. Level 3 instruments are valued based on unobservable inputs that are supported by little or no market activity and reflect the Company's own assumptions in measuring fair value. At March 31, 2020, contingent consideration liabilities of \$7,675,000 and \$5,050,000 were included in other current liabilities and other long-term liabilities, respectively, in the accompanying consolidated balance sheets. At December 31, 2019, contingent consideration liabilities of \$4,825,000 and \$9,900,000 were included in other current liabilities and other long-term liabilities, respectively, in the accompanying consolidated balance sheets.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

A reconciliation of the Company's contingent consideration liabilities related to acquisitions for the three months ended March 31, 2020 and 2019 is as follows:

Three Months Ended March 31, 2020 Contingent consideration - Level 3 liabilities	Beginning Balance \$ 14,725,000	Additions \$ —	Payments \$	Change in Fair Value \$ (2,000,000)	Ending Balance \$ 12,725,000
Three Months Ended March 31, 2019 Contingent consideration - Level 3 liabilities	Beginning Balance \$ 15,250,000	Additions \$ 1,500,000	Payments \$ (12,000,000)	Change in Fair Value \$ —	Ending Balance \$ 4,750,000

#### (7) Derivative Instruments and Hedging Activities

The Company records all derivatives on its consolidated balance sheet at fair value. As of March 31, 2020 and December 31, 2019, the Company had outstanding interest rate derivatives with third parties in which the Company pays a fixed interest rate and receives a rate equal to the one-month LIBOR. The notional associated with the swap agreements was \$250,000,000 as of March 31, 2020 and December 31, 2019 and have maturity dates at certain dates through March 2024. Prior to August 22, 2019, the interest rate swap agreements were not designated as cash flow hedging instruments for accounting purposes and accordingly changes in fair value of the interest rate swap agreements were recorded in earnings. On August 22, 2019, the Company designated its swaps as effective cash flow hedges of interest rate risk. Accordingly, subsequent to August 22, 2019, changes in the fair value of the interest rate swaps are recorded as a component of accumulated other comprehensive income (loss) within stockholders' equity and subsequently reclassified into interest expense in the same period during which the hedged transaction affects earnings.

The table below presents the fair value of the Company's derivative financial instruments as well as their classification in the consolidated balance sheets at March 31, 2020 and December 31, 2019:

	As of March 31	, 2020
	Balance Sheet Location	Fair Value Asset (Liability)
Derivatives designated as hedging instruments:		
Interest rate swap agreements	Other current liabilities	\$ (5,373,647)
Interest rate swap agreements	Other long-term liabilities	(13,675,476)
Total derivatives designated as hedging instruments		\$ (19,049,123)
	As of December 3	31, 2019
	Balance Sheet Location	Fair Value Asset (Liability)
Derivatives designated as hedging instruments:		
Interest rate swap agreements	Other current liabilities	\$ (2,157,324)
Interest rate swap agreements	Other long-term liabilities	(6,181,964)
Total derivatives designated as hedging instruments		\$ (8,339,288)

During the three months ended March 31, 2020, as a result of the effect of cash flow hedge accounting, the Company recognized a loss of \$10,709,835 in other comprehensive income (loss) and \$707,381 was reclassified from other comprehensive income (loss) and recognized as a reduction to interest expense, net, in the accompanying consolidated statements of operations. During the three months ended March 31, 2019, as a result of the effect of the Company's derivative financial instruments that were not designated as hedging instruments, the Company recognized \$2,702,400 in interest expense, net in the accompanying consolidated statements of operations.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

## (8) Accounts Payable and Accrued Expenses

Accounts payable and accrued expenses as of March 31, 2020 and December 31, 2019 consisted of the following:

	March 31,	December 31,
	2020	2019
Accounts payable	\$ 105,194,402	\$ 79,237,323
Employee related accruals	9,002,909	12,319,746
Accrued interest	4,672,868	4,021,660
Other	17,201,708	7,149,364
Total	\$ 136,071,887	\$ 102,728,093

#### (9) Debt

The following is a summary of long term-debt as of March 31, 2020 and December 31, 2019:

		March 31, 2020	December 31, 2019
Secured term loans	\$	295,937,500	\$ 246,250,000
Revolving credit facility		32,000,000	12,000,000
Note payable		143,500,000	143,500,000
Other		1,053,205	1,725,185
Unamortized deferred financing fees		(6,322,104)	(6,642,490)
	_	466,168,601	396,832,695
Current portion		(2,615,705)	(1,721,132)
Long-term portion	\$	463,552,896	\$ 395,111,563

In March 2019, the Company entered into several agreements, amendments and new credit facilities (herein after referred to as the March 2019 Recapitalization Transactions). The March 2019 Recapitalization Transactions included \$425,000,000 in new credit facilities, which consisted of a \$300,000,000 Initial Term Loan (Credit Facility Term Loan), \$50,000,000 Delayed Draw Term Loan (Delayed Draw), and \$75,000,000 Revolving Credit Facility (New Revolver), all with maturities in March 2024. In November 2019, the Company amended its credit agreement primarily to (i) increase the amount available under the Delayed Draw to \$100,000,000, and (ii) revise the Consolidated Total Leverage Ratio thresholds and lower the applicable margin to determine the variable quarterly interest rate under the credit agreement. Amounts borrowed under the Credit Facility Term Loan and Delayed Draw bear interest quarterly at variable rates based upon the sum of (a) the LIBOR Rate for such interest period, plus (b) an applicable margin based upon the Company's Consolidated Total Leverage Ratio. The Delayed Draw carries 0.5% of unused fee per annum, and the New Revolver carries 0.5% of unused line fee per annum. Under the credit facility, the Company is subject to various agreements that contain a number of restrictive covenants that, among other things, impose operating and financial restrictions on the Company. Financial covenants include a Consolidated Total Leverage Ratio and a Fixed Charges Coverage Ratio, as defined in the agreement. Additionally, under the terms of the debt amendment, the Company may be required to repay principal based on excess cash flow, as defined.

The proceeds from the March 2019 Recapitalization Transactions were used to (1) repay existing amounts outstanding under the Company's credit facility of \$151,875,000, (2) pay transaction costs, fees and expenses related to the consummation of the transactions contemplated under the agreement (see Note and Unit Purchase Agreement discussed below), (3) pay a \$250,000,000 distribution to AdaptHealth Holdings' members, and (4) redeem certain members' interests, including the cumulative preferred dividends, for \$3,713,455. In addition, the Company paid deferred financing costs of \$9,027,753; amortization of such costs is included in interest expense, net in the accompanying consolidated statements of operations. Further, the Company wrote off deferred financing costs of

## Notes to Consolidated Interim Financial Statements (Unaudited)

\$2,121,451, which is included in loss on extinguishment of debt in the accompanying consolidated statements of operations for the three months ended March 31, 2019.

#### Secured Term Loans

The Credit Facility Term Loan requires quarterly principal repayments beginning June 30, 2019 through December 31, 2023, and the unpaid principal balance is due at maturity in March 2024. In November 2019, the Company repaid \$50,000,000 under the Credit Facility Term Loan; such repayment satisfied the required principal repayments through September 2023. At March 31, 2020, there was \$246,250,000 outstanding under the Credit Facility Term Loan. The interest rate under the Credit Facility Term Loan was 4.35% at March 31, 2020.

The Delayed Draw has an availability period from the first business day immediately following the closing date (March 2019) to the earliest of (a) the Credit Facility Term Loan maturity date, (b) twenty-four months following the closing date, or (c) the date of the termination of the commitment. During the three months ended March 31, 2020, the Company borrowed \$50,000,000 under the Delayed Draw. The borrowing under the Delayed Draw requires quarterly principal repayments of \$312,500 beginning March 31, 2020, through December 31, 2020, quarterly principal repayments of \$625,000 beginning March 31, 2021 through December 31, 2023, and the unpaid principal balance is due at maturity in March 2024. At March 31, 2020, there was \$49,687,500 outstanding under the Delayed Draw. The interest rate under the Delayed Draw was 4.11% at March 31, 2020.

#### Revolving Credit Facility

During the three months ended March 31, 2020, the Company borrowed \$20,000,000 under the New Revolver; such amount was repaid in April 2020. At March 31, 2020, there was \$32,000,000 outstanding under the New Revolver. The interest rate under the New Revolver was 4.35% at March 31, 2020. After consideration of stand-by letters of credit outstanding of \$2,496,518, the remaining maximum borrowings available pursuant to the New Revolver were \$40,503,482 at March 31, 2020.

#### Note Payable

In connection with the March 2019 Recapitalization Transactions, the Company signed a Note and Unit Purchase Agreement with an investor. Pursuant to the agreement, the Company signed a promissory note agreement with a principal amount of \$100,000,000 (the Promissory Note) and the Company also received proceeds of \$20,000,000 for the purchase of members' interests. In connection with the transactions completed as part of the Business Combination, the Promissory Note was replaced with a new amended and restated promissory note with a principal amount of \$100,000,000, and the investor converted certain of its members' interests to a \$43,500,000 promissory note. The new \$100,000,000 promissory note, together with the \$43,500,000 promissory note, are collectively referred to herein as the New Promissory Note. The outstanding principal balance under the New Promissory Note is due on the tenth anniversary of the closing date of the Business Combination and bears interest at the following rates (a) for the period starting on the closing date and ending on the seventh anniversary, a rate of 12% per annum, with 6% payable in cash and 6% Payment in Kind (PIK), and (b) for the period starting on the day after the seventh anniversary of the closing date and ending on the maturity date, a rate equal to the greater of (i) 15% per annum or (ii) the twelvemonth LIBOR plus 12% per annum. The Company has the option to pay the PIK interest in cash under the New Promissory Note, which it did during the three months ended March 31, 2020. If the Company elects to prepay the New Promissory Note prior to the third anniversary of the Closing of the Business Combination, then such prepayment of the outstanding principal and accrued interest will be subject to a make-whole premium equal to 10% of the total amount of outstanding principal and accrued interest through the date of such prepayment. If the Company elects to prepay the New Promissory Note prior to the fourth anniversary but after the third anniversary of the Closing of the Business Combination, then such prepayment of outstanding principal and accrued interest will be subject to a make-whole premium equal to 5% of the total amount of outstanding principal and accrued interest through the date of such prepayment.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

In connection with the Business Combination the investor generated taxable income and a current federal and state income tax liability of approximately \$5,870,000 on the exchange of its members' interests. Under the terms of the Merger Agreement, all investors indemnified the Company for all taxes attributable to periods prior to or on the closing date of the Business Combination. Accordingly, the Company recorded an indemnification asset of such amount, included in Prepaid and other current assets, and a corresponding current liability included in Other liabilities, in the accompanying consolidated balance sheets as of and December 31, 2019. This amount is no longer outstanding as of March 31, 2020.

## (10) Stockholders' Equity

The Closing of the Business Combination occurred on November 8, 2019, refer to Note 3, *Significant Transactions*, for additional details regarding the Business Combination.

#### Warrants

At the Closing of the Business Combination, the Company had 12,666,666 warrants outstanding. Each warrant is exercisable for one share of common stock at a price of \$11.50 per share. The exercise price and number of common stock issuable upon exercise of the warrants may be adjusted in certain circumstances including in the event of a share dividend, or recapitalization, reorganization, merger or consolidation. However, the warrants will not be adjusted for issuance of common stock at a price below its exercise price. During the three months ended March 31, 2020, 3,938,630 warrants were exercised in cashless transactions which resulted in the issuance of 1,092,468 shares of Class A Common Stock. As of March 31, 2020, the Company had 8,728,036 warrants outstanding.

#### Contingent Consideration

Pursuant to the Merger Agreement, the former owners of AdaptHealth Holdings who received Class B Common Stock in connection with the Business Combination are entitled to receive an equity classified earn-out consideration to be paid in the form of New AdaptHealth Units (and a corresponding number of shares of Class B Common Stock) and the former owners of AdaptHealth Holdings who received Class A Common Stock in connection with the Business Combination are entitled to receive earn-out consideration to be paid in the form of Class A Common Stock in connection with the Business Combination are entitled to receive earn-out consideration to be paid in the form of Class A Common Stock, if the 30-day volume-weighted average price of the Company's Class A Common Stock equals or exceeds certain hurdles set forth in the Merger Agreement. The former owners of AdaptHealth Holdings can potentially receive up to an additional 1,000,000 shares in December 2020, 2021 and 2022, for a total of 3,000,000 shares, as a part of the earn-out consideration. As of March 31, 2020, the hurdles have not been met.

#### Equity-based Compensation

#### Incentive Units

AdaptHealth Holdings granted Incentive Units in June 2019 (the 2019 Incentive Units) and in April 2018 (the 2018 Incentive Units) to certain members of management. With respect to the 2019 Incentive Units, 50% of the awards vest in equal annual installments on each of the first, second, third and fourth anniversaries of the Vesting Commencement Date as defined in the agreements (May 20, 2019). The remaining 50% had vesting terms based upon a performance condition. In connection with the Business Combination, the vesting conditions for this portion of the 2019 Incentive Units was changed to vest quarterly during the one-year period subsequent to the Closing of the Business Combination. The grant date fair value of the 2019 Incentive Units, as calculated under an Option Pricing Method, was \$4,511,120, and is being recognized as expense over the employees' requisite service period based on the vesting conditions described above. In conjunction with the March 2019 Recapitalization Transactions, the vesting of certain of the 2018 Incentive Units was accelerated and all holders of the 2018 Incentive Units received an advance for future distribution, which were treated as a modification of the awards for accounting purposes. In conjunction with the Business Combination, the vesting of a majority of the unvested 2018 Incentive Units was accelerated.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

## 2019 Stock Incentive Plan

On November 7, 2019, the stockholders of the Company approved the AdaptHealth Corp. 2019 Stock Incentive Plan (the 2019 Plan), effective upon closing of the Business Combination. In connection with the 2019 Plan, the Company provides equity-based compensation to attract and retain employees while also aligning employees' interest with the interests of its stockholders. The 2019 Plan permits the grant of various equity-based awards to selected employees and directors. The 2019 Plan permits the grant of up to 8,000,000 shares of Class A Common Stock, subject to certain adjustments and limitations.

#### Stock Options

During the fourth quarter of 2019, the Company granted 3,416,666 options to purchase shares of common stock of the Company to certain executive officers that have an exercise price of \$11.50 per share. A portion of the options are eligible to vest on December 31, 2020, 2021 and 2022 based on defined performance conditions, subject to the employees' continuous employment through the applicable vesting date. The grant-date fair value of the awards, using a Black-Scholes option pricing model, was \$7,248,653 and is being recognized as expense on a straight-line basis over the employees' requisite service period subject to management's estimation of the probability of vesting of such awards. The Company has no other options outstanding as of March 31, 2020.

#### Restricted Stock

On March 3, 2020, the Company granted 300,000 shares of restricted stock to an employee in conjunction with an acquisition. Of the total shares granted, 250,000 are eligible to vest based on certain performance conditions, subject to the employee's continuous employment through the applicable vesting date. The remaining 50,000 shares will vest 25% annually on December 31, 2020 through 2023, subject to the employee's continuous employment through the applicable vesting date. The total grant-date fair value of the award was \$4,905,000 and is being recognized as expense on a straight-line basis over the employee's requisite service period subject to management's estimation of the probability of vesting of such awards (as it relates to the performance-based awards).

On March 3, 2020, the Company granted 321,123 shares of restricted stock to various employees. Of the total shares granted, 15,417 shares vested on the grant date, and the remaining shares will vest 25% on each anniversary of the Vesting Commencement Dates (as defined in the agreements), subject to the employees' continuous employment through the applicable vesting date. The grant-date fair value of the awards was \$5,250,361, of which \$252,068 was recognized as expense on the grant date and \$4,998,293 is being recognized as expense on a straight-line basis over the employees' requisite service period.

Activity related to the Company's non-vested restricted stock grants for the three months ended March 31, 2020 is presented below							
	Number of Shares of		Weighted-Average Grant Date Fair Value per				
	Restricted Stock		Share				
Non-vested balance at January 1, 2020	901,250	\$	5.83				
Granted	621,123	\$	16.35				
Vested	(15,417)	\$	16.35				
Forfeited	(20,000)	\$	8.11				
Non-vested balance at March 31, 2020	1,486,956	\$	10.09				

During the three months ended March 31, 2020, the Company recorded equity-based compensation expense of \$2,222,609, of which \$1,671,923 and \$550,686 was included in general and administrative expenses and cost of net revenue, respectively, in the accompanying consolidated statement of operations. During the three months ended March 31, 2019, the Company recorded equity-based compensation expense of \$5,223,108 which is included in general and administrative expenses in the accompanying consolidated statements of operations. The expense during the three

#### Notes to Consolidated Interim Financial Statements (Unaudited)

months ended March 31, 2019 included \$4,894,720 in connection with the acceleration of vesting of certain of the 2018 Incentive Units and the modification of such awards discussed above. At March 31, 2020, there was \$20,025,505 of unrecognized compensation expense related to equity-based compensation awards, which is expected to be recognized over a weighted-average term of 2.8 years. At March 31, 2020, 3,037,761 shares of the Company's Class A Common Stock are available for issuance under the 2019 Plan.

#### (11) Net Income (Loss) Per Common Share

The Business Combination was accounted for as a reverse recapitalization by which AdaptHealth Holdings issued stock for the net assets of the Company accompanied by a recapitalization. Earnings per share has been recast for all historical periods to reflect the Company's capital structure for all comparative periods.

The Company excluded the effect of the warrants, unvested restricted stock, stock options and Class B Common Stock from the computation of diluted net income (loss) per share for the three months ended March 31, 2020 because the effect of including them would be anti-dilutive as a result of the Company being in a net loss position for such period. There were no such items outstanding for the three months ended March 31, 2019.

#### (12) Leases

#### **Capital Leases**

The Company has acquired patient medical equipment and supplies, and office equipment through multiple capital leases. The capital lease obligations represent the present value of minimum lease payments under the respective agreement, payable monthly at various interest rates. Interest expense related to capital leases was \$16,305 and \$35,083 for the three months ended March 31, 2020 and 2019, respectively. As of March 31, 2020, future annual minimum payments required under lease obligations are as follows:

Twelve months ended March 31,	
2021	\$ 20,442,099
2022	215,164
Total	20,657,263
Less amount representing interest	(70,192)
	20,587,071
Current portion	(20,421,195)
Long-term portion	\$ 165,876

At March 31, 2020 and December 31, 2019, equipment under capital leases consisted of patient equipment with a cost basis of approximately \$40,600,000 and \$39,100,000, respectively, and accumulated depreciation of approximately \$12,200,000 and \$11,700,000, respectively. Depreciation expense for equipment purchased under capital leases is primarily included in cost of net revenue in the accompanying consolidated statements of operations.

#### **Operating Leases**

The Company leases its office facilities and office equipment under noncancelable lease agreements which expire at various dates through March 2033. Some of these lease agreements include an option to renew at the end of the term. The Company also leases certain patient medical equipment with such leases set to expire at various dates through November 2021. The Company also leases certain office facilities on a month to month basis. In some instances, the Company is also required to pay its pro rata share of real estate taxes and utility costs in connection with the premises. Some of the leases contain fixed annual increases of minimum rent. Accordingly, the Company recognizes rent expense on a straight-line basis and records the difference between the recognized rent expense and the amount payable under the lease as deferred rent. The deferred rent recorded in accounts payable and accrued expenses on the accompanying



#### Notes to Consolidated Interim Financial Statements (Unaudited)

consolidated balance sheets at March 31, 2020 and December 31, 2019 was \$1,180,874 and \$1,124,702, respectively. The Company recorded rent expense of \$3,505,955 and \$2,351,566 for the three months ended March 31, 2020 and 2019, respectively, which is primarily included in cost of net revenue in the accompanying consolidated statements of operations.

The minimum annual lease commitments under noncancelable leases with initial or remaining terms in excess of one year as of March 31, 2020 are as follows:

Twelve months ended March 31,	
2021	\$ 16,547,233
2022	11,746,784
2023	9,823,064
2024	8,316,930
2025	6,390,029
Thereafter	17,668,659
Total minimum payments required (a)	\$ 70,492,699

(a) Minimum payments have not been reduced by minimum sublease rentals of \$2,712,886 due in the future under noncancelable subleases.

#### (13) Income Taxes

The Company is subject to U.S. federal, state, and local income taxes with respect to its allocable share of any taxable income or loss of AdaptHealth Holdings. AdaptHealth Holdings is treated as a partnership for U.S. income tax purposes and generally does not pay income taxes in most jurisdictions. Instead, AdaptHealth Holdings' taxable income or loss is passed through to its members, including the Company. Additionally, the Company is subject to U.S. federal, state, and local income taxes on the taxable income or loss of the underlying C-corporations in the AdaptHealth group where taxes are paid at the entity level.

For the three months ended March 31, 2020 and 2019, the Company recorded income tax expense of \$1,106,722 and \$2,418,441 respectively.

As of March 31, 2020 and December 31, 2019, the Company had no uncertain tax positions that would require recognition or disclosure in the consolidated interim financial statements.

#### Tax Receivable Agreement

AdaptHealth Corp. is party to a Tax Receivable Agreement (TRA) with certain current and former members of AdaptHealth Holdings. The TRA provides for the payment by AdaptHealth Corp. of 85% of the tax savings, if any, that AdaptHealth Corp. realizes (or is deemed to realize in certain circumstances) as a result of (i) certain increases in tax basis resulting from exchanges of New AdaptHealth Units and shares of Class B Common Stock; (ii) certain tax attributes of the corresponding sellers existing prior to an exchange; (iii) imputed interest deemed to be paid by AdaptHealth Corp. as a result of payments it makes under the TRA; and (iv) certain increases in tax basis resulting from payments AdaptHealth Corp. makes under the TRA.

During the three months ended March 31, 2020, the Company increased its TRA liability through an aggregate \$4.0 million reduction in additional-paid-in capital resulting from additional exchanges of New AdaptHealth Units and shares of Class B Common Stock. Correspondingly, during the three months ended March 31, 2020, the Company increased its deferred tax asset by approximately \$6.5 million through an increase in additional-paid-in-capital resulting from these exchanges and additional increases of AdaptHealth Corp.'s ownership interest in AdaptHealth Holdings.

#### Notes to Consolidated Interim Financial Statements (Unaudited)

At March 31, 2020 and December 31, 2019, the Company had a liability recorded relating to the TRA of approximately \$14,800,000 and \$10,800,000, respectively, which is included in other long-term liabilities in the accompanying consolidated balance sheets.

#### (14) Commitments and Contingencies

In the normal course of business, the Company is subject to loss contingencies, such as legal proceedings and claims arising out of its business that cover a wide range of matters. In accordance with FASB ASC Topic 450, *Accounting for Contingencies*, the Company records accruals for such loss contingencies when it is probable that a liability has been incurred and the amount of loss can be reasonably estimated. Significant judgement is required to determine both probability and the estimated amount. The Company reviews at least quarterly and adjusts accordingly to reflect the impact of negotiations, settlements, rulings, advice of legal coursel, and updated information. At this time, the Company has no accrual related to lawsuits, claims, investigations and proceedings.

In connection with the Company's acquisition of PPS HME Holdings LLC (PPS) in May 2018, the Company assumed a Corporate Integrity Agreement (CIA) at one of PPS' subsidiaries, Braden Partners L.P. d/b/a Pacific Pulmonary Services (BP). The CIA was entered into with the Office of Inspector General of the U.S. Department of Health and Human Services (OIG). The CIA has a five-year term which expires in April 2022. In connection with the acquisition and integration of PPS by AdaptHealth, the OIG confirmed that the requirements of the CIA imposed upon BP would only apply to the operations of BP and therefore no operations of any other AdaptHealth affiliate are subject to the requirements of the CIA following the acquisition.

## (15) Related Party Transactions

The Company has an outstanding note payable with a principal balance of \$143,500,000 with an investor who also has equity ownership in the Company.

The Company and two of its executive officers each owned an equity interest in a vendor of the Company that provides workflow technology services. Each individual's equity ownership was less than 1%. The expense related to this vendor was \$1,456,501 and \$779,110 for the three months ended March 31, 2020 and 2019, respectively. The Company accounted for this investment under the cost method of accounting based on its level of equity ownership. In February 2020, the Company and each executive officer sold their respective equity interest. The Company's investment had a carrying value of \$1,455,000 and the Company received proceeds of \$2,045,701 in connection with the transaction, resulting in a gain of \$590,701 which is included in cost of net revenue in the accompanying consolidated statements of operations for the three months ended March 31, 2020.

The Company and two of its executive officers and shareholders own an equity interest in a vendor of the Company that provides automated order intake software. Each individual's equity ownership is less than 1%. The expense related to this vendor was \$544,449 and \$450,000 for the three months ended March 31, 2020 and 2019, respectively. The Company accounts for this investment under the cost method of accounting based on its level of equity ownership.

## (16) Subsequent Events

In response to the COVID-19 pandemic and the National Emergency Declaration, dated March 13, 2020, the Company activated certain business interruption protocols, including acquisition and distribution of personal protective equipment to its patient-facing employees, accelerated capital expenditures of certain products and relocation of significant portions of its workforce to "work-from-home" status. The Company also increased its cash liquidity by, among other things, seeking recoupable advance payments of approximately \$47 million made available by CMS under the Coronavirus Aid, Relief, and Economic Security Act (the CARES Act) legislation, which was received in April 2020. In addition, in April 2020, the Company received distributions of the CARES Act provider relief funds of approximately \$17 million targeted to offset lost revenue and expenditures incurred in connection with the COVID-19

#### Notes to Consolidated Interim Financial Statements (Unaudited)

pandemic. The provider relief funds are subject to certain restrictions and are subject to recoupment if not used for designated purposes. As permitted under the CARES Act, the Company has also elected to defer certain portions of employer-paid FICA taxes otherwise payable from March 27, 2020 to January 1, 2021, which will be paid in two equal installments on December 31, 2021 and December 31, 2022.

In April 2020 the Company repaid \$20 million of amounts borrowed under the New Revolver upon receipt of the advanced payments from CMS and provider relief funds discussed above.

Subsequent to March 31, 2020, holders of New AdaptHealth Units and Class B Common Stock exchanged 1,953,549 New AdaptHealth Units together with a corresponding number of shares of Class B Common Stock for 1,953,549 shares of Class A Common Stock.

Subsequent to March 31, 2020, 109,983 warrants were exercised for proceeds of \$1,264,804 resulting in the issuance of 109,983 shares of Class A Common Stock.

## Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion should be read in conjunction with AdaptHealth Corp.'s ("AdaptHealth" or the "Company") consolidated interim financial statements and the accompanying notes included in this report. All amounts presented are in accordance with U.S. generally accepted accounting principles ("U.S. GAAP"), except as noted. In addition to historical information, this discussion contains forward-looking statements that involve risks, uncertainties and assumptions that could cause actual results to differ materially from management's expectations.

## AdaptHealth Corp. Overview

AdaptHealth is a leading provider of home healthcare equipment, medical supplies to the home and related services in the United States. The Company focuses primarily on providing (i) sleep therapy equipment, supplies and related services (including CPAP and bi-PAP services) to individuals suffering from obstructive sleep apnea (OSA), (ii) home medical equipment (HME) to patients discharged from acute care and other facilities, (iii) oxygen and related chronic therapy services in the home and (iv) other HME medical devices and supplies on behalf of chronically ill patients with diabetes care, wound care, urological, ostomy and nutritional supply needs. The Company services beneficiaries of Medicare, Medicaid and commercial insurance payors. As of March 31, 2020, AdaptHealth services over approximately 1.6 million patients annually in all 50 states through its network of 220 locations in 38 states. The Company's principal executive offices are located at 220 West Germantown Pike, Suite 250, Plymouth Meeting, Pennsylvania 19462.

#### Trends and Factors Affecting AdaptHealth's Future Performance

Significant trends and factors that AdaptHealth believes may affect its future performance include:

- Home Medical Equipment Growth. According to CMS, the HME industry has grown from \$40 billion in 2010 to \$56 billion in 2018 (representing a 4.3% CAGR), of which AdaptHealth's total addressable market for its sleep therapy, oxygen services, mobility products and hospice HME business lines comprised approximately \$12 billion to \$15 billion in 2018. During that time Medicaid data shows a continued shift of long-term services and supports (LTSS) spending into the home, with 57% of that spending going to home and community-based services in 2016. According to CMS, the HME market is projected to continue to grow at a 6.1% CAGR over the next nine years. As a result of the acquisition of the diabetic, wound care, ostomy and urological supplies business of PCS in January 2020, the Company believes it has more than doubled its addressable market to more than \$25 billion.
- Aging U.S. Population. The population of adults aged 65 and older in the U.S., a significant group of end users of AdaptHealth's products and services, is expected to continue to grow and thus grow AdaptHealth's market opportunity. According to CMS, in the U.S., the population of adults between the ages of 65 and 84 is expected to grow at a 2.5% CAGR through 2030, while the population of adults over 85 is projected to grow at a 2.9% CAGR during that same time period. Not only is the elderly population expected to grow, but they are also expected to make up a larger percentage of the total U.S. population. According to the U.S. Census Bureau, the U.S. geriatric population was approximately 15% of the total population in 2014 and is expected to grow to approximately 24% of the total population by 2060.
- Increasing Prevalence of Chronic Conditions. HME is necessary to help treat significant health issues affecting millions of Americans, such as chronic obstructive pulmonary disease, congestive heart failure, obstructive sleep apnea and diabetes.
- Increasing Prevalence of and Preference for In-Home Treatments. The number of conditions that can be treated in the home continues to grow, with recent additions including chronic wound care, sleep testing, dialysis and chemotherapy. In home care is also increasingly becoming the preferred method of treatment, particularly for the elderly population. According to the AARP Public Policy Institute, 90% of patients over age 55 have indicated a preference to receive care in the home rather than in an institutional setting.

• Home Care is the Lowest Cost Setting. Not only is in-home care typically just as effective as care delivered in an inpatient setting, but it has also proven to be more cost effective. This is especially important within the context of government pressures to lower the cost of care, pushing clinicians to seek care settings that are less costly than hospitals and inpatient facilities. On a daily basis, home healthcare has been estimated by Cain Brothers & Company, LLC to be approximately seven times less expensive than care provided in skilled nursing facilities, the closest acuity site of care. In-home care offers a significant cost reduction opportunity relative to facility-based care without sacrificing quality.

Certain additional items may impact the comparability of the historical results presented below with AdaptHealth's future performance, such as the cost of being a public company. To operate as a public company, AdaptHealth is required to continue to implement changes in certain aspects of its business and develop, manage, and train management level and other employees to comply with ongoing public company requirements, including compliance with Section 404 and the evaluation of the effectiveness of internal controls over financial reporting. AdaptHealth also incurs other expenses as a public company, including expenses associated with public reporting obligations, proxy statements and stockholder meetings, stock exchange fees, transfer agent fees, SEC and Financial Industry Regulatory Authority filing fees and offering expenses.

#### Impact of the COVID-19 Pandemic

AdaptHealth's priorities during the COVID-19 pandemic are protecting the health and safety of its employees (including patientfacing employees providing respiratory and other services), maximizing the availability of its services and products to support patient health needs, and the operational and financial stability of its business.

In response to the COVID-19 pandemic and the National Emergency Declaration, dated March 13, 2020, AdaptHealth activated certain business interruption protocols, including acquisition and distribution of personal protective equipment (PPE) to its patient-facing employees, accelerated capital expenditures of certain products and relocation of significant portions of its workforce to "work-from-home" status. AdaptHealth also increased its cash liquidity by, among other things, seeking recoupable advance payments of approximately \$47 million made available by CMS under the Coronavirus Aid, Relief, and Economic Security Act (the CARES Act) legislation, which was received in April 2020. In addition, in April 2020, AdaptHealth releved distributions of the CARES Act provider relief funds of approximately \$17 million targeted to offset lost revenue and expenditures incurred in connection with the COVID-19 pandemic. The provider relief funds are subject to certain restrictions and are subject to recoupent if not used for designated purposes. As the result of these actions, and the lack of distribution to date of its vendors' ability to supply product despite the COVID-19 pandemic, AdaptHealth has been able to substantially maintain its operations.

While the impact of the COVID-19 pandemic, the National Emergency Declaration and the various state and local government imposed stay-at-home restrictions did not have a material impact on AdaptHealth's consolidated operating results for the three months ended March 31, 2020, AdaptHealth has begun to experience declines in net revenues in certain services associated with elective medical procedures (such as commencement of new CPAP services and medical equipment and orthopedic supply related to facility discharges) and such declines may continue during the duration of the COVID-19 pandemic. In response to these declines, as well as certain over staffing related to recent acquisitions, AdaptHealth conducted a workforce assessment and implemented a reduction in force in April 2020 resulting in the elimination of approximately 6% of its workforce. In connection with the workforce reductions, AdaptHealth will incur a one-time charge for severance and related expenses estimated to be approximately \$1.6 million.

Offsetting these declines in net revenue, AdaptHealth is experiencing an increase in net revenue related to increased demand for certain respiratory products (such as oxygen), increased sales in its resupply businesses (primarily as a result of the increased ability to contact patients at home as a result of state and local government imposed stay-at-home orders) and the one-time sale of certain respiratory equipment (primarily ventilators, bi-level PAP devices and oxygen concentrators) to hospitals and local health agencies. Additionally, suspension of Medicare sequestration through December 31, 2020 (resulting in a 2% increase in Medicare payments to all providers), and recent regulatory guidance from CMS expanding telemedicine and reducing documentation requirements during the emergency period, are expected to result in increased net revenues for certain products and services.

The full extent of the impact of the COVID-19 pandemic on AdaptHealth's business, operations and financial results will depend on numerous evolving factors that it may not be able to accurately predict. For additional information on risk factors that could impact AdaptHealth's results, please refer to "Risk Factors" in Part II, Item 1A of this Form 10-Q.

# Key Components of Operating Results

*Net Revenue.* Net revenue is recorded for services that AdaptHealth provides to patients for home healthcare equipment, medical supplies to the home and related services. AdaptHealth's primary service lines are (i) sleep therapy equipment, supplies and related services (including CPAP and bi-PAP services) to individuals suffering from OSA, (ii) home medical equipment to patients discharged from acute care and other facilities, (iii) oxygen and related chronic therapy services in the home and (iv) other HME medical devices and supplies on behalf of chronically ill patients with diabetes care, wound care, urological, ostomy and nutritional supply needs. Revenues are recorded either (x) at a point in time for the sale of supplies and disposables, or (y) over the service period for equipment rental (including, but not limited to, CPAP machines, hospital beds, wheelchairs and other equipment), at amounts estimated to be received from patients or under reimbursement arrangements with Medicare, Medicaid and other third-party payors, including private insurers.

**Cost of Net Revenue.** Cost of net revenue primarily includes the cost of non-capitalized medical equipment and supplies, distribution expenses, labor costs, facilities rental costs, third-party revenue cycle management costs and depreciation for capitalized patient equipment. Distribution expenses represents the cost incurred to coordinate and deliver products and services to the patients. Included in distribution expenses are leasing, maintenance, licensing and fuel costs for the vehicle fleet; salaries, benefits and other costs related to drivers and dispatch personnel; and amounts paid to couriers.

General and Administrative Expenses. General and administrative expenses consist of corporate support costs including information technology, human resources, finance, contracting, legal, compliance leadership, equity-based compensation, transaction expenses and other administrative costs.

## Depreciation,

*Excluding Patient Equipment Depreciation.* Depreciation expense includes depreciation charges for capital assets other than patient equipment (which is included as part of the cost of net revenue).

## Factors Affecting AdaptHealth's Operating Results

AdaptHealth's operating results and financial performance are influenced by certain unique events during the periods discussed herein, including the following:

#### Acquisitions

AdaptHealth accounts for its acquisitions in accordance with FASB ASC Topic 805, *Business Combinations*, and the operations of the acquired entities are included in the historical results of AdaptHealth for the periods following the closing of the acquisition. The most significant of these acquisitions impacting the comparability of AdaptHealth's operating results in the first quarter of 2020 compared to the first quarter of 2019 were SleepMed Therapies, Inc. ("SleepMed") acquired in July 2019, Choice Medical Healthcare, Inc. ("Choice") acquired in October 2019, the Patient Care Solutions business ("PCS") acquired from McKesson Corporation in January 2020, Healthline Medical Equipment, LLC ("Healthline") acquired in February 2020, and Advanced Home Care, Inc. ("Advanced") acquired in March 2020. Refer to Note 3, *Significant Transactions*, included in our consolidated interim financial statements for the three months ended March 31, 2020 included in this report for additional information regarding AdaptHealth's acquisitions.

## **Debt and Recapitalization**

In March 2019, AdaptHealth restructured its debt borrowings with its bank group. The debt restructuring consisted of \$425 million in credit facilities, which included a \$300 million Initial Term Loan (the "Credit Facility Term

Loan"), \$50 million Delayed Draw Term Loan (the "Delayed Draw"), and \$75 million Revolving Credit Facility (the "New Revolver"), all with maturities in March 2024.

In March 2019, AdaptHealth entered into a Note and Unit Purchase Agreement with an investor. In connection with the agreement, membership interests in AdaptHealth Holdings were purchased for \$20 million, and AdaptHealth also signed a promissory note agreement with a principal amount of \$100 million (the "Promissory Note").

The transactions consummated with respect to the debt restructuring and the Note and Unit Purchase Agreement are hereinafter referred to as the "2019 Recapitalization."

In connection with the closing of the Business Combination, the Company amended its credit facility primarily to (i) increase the amount available under the Delayed Draw from \$50 million to \$100 million, and (ii) revise the Consolidated Total Leverage Ratio thresholds and lower the applicable margin to determine the variable quarterly interest rate under the credit facility. In addition, in November 2019, the Company repaid \$50.0 million under the Credit Facility Term Loan; such repayment satisfied the principal payments required to be paid through September 2023. Further, in connection with the closing of the Business Combination, the Promissory Note was replaced with a new amended and restated promissory note with a principal amount of \$100.0 million. In addition, the investor converted certain of its members' equity interests to a \$43.5 million promissory note. The new \$100.0 million promissory note, together with the \$43.5 million promissory note, are collectively referred to herein as the New Promissory Note. The outstanding principal balance under the New Promissory Note is due on November 8, 2029 and bears interest at the following rates (a) for the period starting on the closing date and ending on the seventh anniversary of the closing date and ending on the maturity date, a rate equal to the greater of (i) 15% per annum or (ii) the twelve-month LIBOR plus 12% per annum. The Company has the option to pay the PIK interest in cash under the New Promissory Note, which it did through March 31, 2020, and thus no amounts were added to the principal balance through that date.

In connection with the acquisition of Advanced in March 2020, the Company borrowed \$50.0 million under the Delayed Draw. Additionally, in March 2020, the Company borrowed \$20.0 million under the New Revolver as a precaution in light of the coronavirus health crisis; such amount was repaid in April 2020.

## Seasonality

AdaptHealth's business is somewhat sensitive to seasonal fluctuations. Its patients are generally responsible for a greater percentage of the cost of their treatment or therapy during the early months of the year due to co-insurance, co-payments and deductibles, and therefore may defer treatment and services of certain therapies until meeting their annual deductibles. In addition, changes to employer insurance coverage often go into effect at the beginning of each calendar year which may impact eligibility requirements and delay or defer treatment. These factors may lead to lower net revenue and cash flow in the early part of the year versus the latter half of the year. Additionally, the increased incidence of respiratory infections during the winter season may result in initiation of additional respiratory services such as oxygen therapy for certain patient populations. AdaptHealth's quarterly operating results may fluctuate significantly in the future depending on these and other factors.

#### **Key Business Metrics**

AdaptHealth focuses on net revenue, EBITDA, Adjusted EBITDA and Adjusted EBITDA less Patient Equipment Capex as it reviews its performance. Total net revenue is comprised of net sales revenue and net revenue from fixed monthly equipment reimbursements less a provision for doubtful accounts and implicit price concessions. Net sales revenue consists of revenue recognized at a point in time for the sale of supplies and disposables. Net revenue from fixed monthly equipment reimbursements consists of revenue recognized over the service period for equipment (including, but not limited to, CPAP machines, hospital beds, wheelchairs and other equipment).

	Three months ended								
		March 31	, 2020	March 31, 2019					
Net Revenue			Revenue		Revenue				
(in thousands, except percentages)		Dollars	Percentage	Dollars	Percentage				
			(Unaudit	ed)					
Net sales revenue:									
Sleep	\$	68,894	36.0 % \$	47,127	39.4 %				
Supplies to the home		33,339	17.4 %	2,029	1.7 %				
HME		11,579	6.0 %	10,489	8.8 %				
Respiratory		2,768	1.4 %	1,279	1.1 %				
Other		12,393	6.5 %	8,032	6.7 %				
Total net sales revenue	\$	128,973	67.3 % \$	68,956	57.7 %				
Net revenue from fixed monthly equipment reimbursements:									
Sleep	\$	22,669	11.8 % \$	18,057	15.1 %				
HME		12,177	6.4 %	10,243	8.6 %				
Respiratory		25,007	13.1 %	20,429	17.1 %				
Other		2,613	1.4 %	1,813	1.5 %				
Total net revenue from fixed monthly equipment reimbursements	\$	62,466	32.7 % \$	50,542	42.3 %				
Total net revenue:									
Sleep	\$	91,563	47.8 % \$	65,184	54.5 %				
Supplies to the home		33,339	17.4 %	2,029	1.7 %				
HME		23,756	12.4 %	20,732	17.4 %				
Respiratory		27,775	14.5 %	21,708	18.2 %				
Other		15,006	7.9 %	9,845	8.2 %				
Total net revenue	\$	191,439	100.0 % \$	119,498	100.0 %				

# **Results of Operations**

# Comparison of Three Months Ended March 31, 2020 and Three Months Ended March 31, 2019.

The following table summarizes AdaptHealth's consolidated results of operations for the three months ended March 31, 2020 and 2019:

	Three Months Ended March 31,					
	202	20	2019			
		Revenue		Revenue	Increase/(I	Decrease)
(in thousands, except percentages)	Dollars	Percentage	Dollars	Percentage	Dollars	Percentage
		(unaudited)				
Net revenue	\$ 191,439	100.0 %	\$ 119,498	100.0 % \$	71,941	60.2 %
Costs and expenses:						
Cost of net revenue	166,540	87.0 %	100,227	83.9 %	66,313	66.2 %
General and administrative expenses	14,347	7.5 %	13,083	10.9 %	1,264	9.7 %
Depreciation, excluding patient equipment						
depreciation	1,241	0.6 %	841	0.7 %	400	47.6 %
Total costs and expenses	182,128	95.1 %	114,151	95.5 %	67,977	59.6 %
Operating income	9,311	4.9 %	5,347	4.5 %	3,964	74.1 %
Interest expense, net	7,938	4.1 %	6,260	5.2 %	1,678	26.8 %
Loss on extinguishment of debt		%	2,121	1.8 %	(2,121)	NM %
Income (loss) before income taxes	1,373	0.8 %	(3,034)	(2.5)%	4,407	(145.3)%
Income tax expense (benefit)	1,107	0.6 %	2,418	2.0 %	(1,311)	NM
Net income (loss)	266	0.2 %	(5,452)	(4.5)%	5,718	(104.9)%
Income attributable to noncontrolling interests	424	0.2 %	348	0.3 %	76	21.8 %
Net loss attributable to AdaptHealth Corp.	\$ (158)	0.0 %	\$ (5,800)	(4.8)% \$	5,642	(97.3)%

*Net Revenue.* Net revenue for the three months ended March 31, 2020 was \$191.4 million compared to \$119.5 million for the three months ended March 31, 2019, an increase of \$71.9 million or 60.2%. The increase in net revenue was driven primarily by acquisitions, which increased revenue by \$57.9 million (including \$33.9 million generated by PCS). The remaining increase in net revenue was primarily attributable to organic growth resulting from stronger CPAP resupply sales and demographic growth in core markets. Net revenue, excluding PCS, was \$157.5 million for the three months ended March 31, 2020. As a result of the coronavirus pandemic, during the three months ended March 31, 2020, the Company experienced reduced demand for products that are related to elective medical services, such as CPAP new starts, orthotics, and certain other HME products, and this trend is expected to remain while the coronavirus crisis continues. However, the Company's CPAP resupply and other supplies business remains healthy, as most patients for that business are in their homes and can be easily contacted to refresh their supplies. Additionally, the coronavirus pandemic has led to an increased demand for respiratory equipment including ventilators and oxygen concentrators.

For the three months ended March 31, 2020, sales revenue (recognized at a point in time) comprised 67% of total net revenue, compared to 58% of total net revenue for the three months ended March 31, 2019. The increase in sales revenue was driven primarily by the PCS acquisition, which is a supplies business, as well as the SleepMed and Choice acquisitions which are primarily CPAP resupply businesses. For the three months ended March 31, 2020, revenue from fixed monthly equipment reimbursements comprised 33% of total net revenue, compared to 42% of total net revenue for the three months ended March 31, 2020, excluding PCS, sales revenue and revenue from fixed monthly equipment reimbursements comprised 60% and 40% of net revenue, respectively.

## Cost of Net Revenue.

The following table summarizes cost of net revenue for the three months ended March 31, 2020 and 2019:

		Т	hree Months End	led March 31,			
	2020 2019						
		Revenue Revenue		Revenue	Increase/	(Decrease)	
(in thousands, except percentages)		Dollars	Percentage	Dollars	Percentage	Dollars	Percentage
				(unau	dited)		
Costs of net revenue:							
Cost of products and supplies	\$	72,003	37.6 %	\$ 34,600	29.0 %	\$ 37,403	108.1 %
Salaries, labor and benefits		54,651	28.5 %	35,618	29.8 %	19,033	53.4 %
Patient equipment depreciation		15,498	8.1 %	14,131	11.8 %	1,367	9.7 %
Rent and occupancy		4,600	2.4 %	3,226	2.7 %	1,374	42.6 %
Other operating expenses		19,336	10.2 %	12,494	10.5 %	6,842	54.8 %
Transaction costs		627	0.3 %	—	%	627	NM %
Equity-based compensation		551	0.3 %	—	%	551	NM %
Severance		365	0.2 %	141	0.1 %	224	158.9 %
Other non-recurring (income) expenses		(1,091)	(0.6)%	17	<u> </u>	(1,108)	NM %
Total cost of net revenue	\$	166,540	87.0 %	\$ 100,227	83.9 %	\$ 66,313	<u>66.2</u> %

Cost of net revenue for the three months ended March 31, 2020 was \$166.5 million compared to \$100.2 million for the three months ended March 31, 2019, an increase of \$66.3 million or 66.2%, which is primarily related to acquisition growth (including \$39.1 million incurred by PCS). Costs of products and supplies increased by \$37.4 million (including \$25.9 million incurred by PCS) as a result of acquisition growth, increased CPAP resupply sales, and to a lesser extent, expenses associated with the coronavirus pandemic, including increased personal protective equipment purchases for patient-facing employees providing respiratory and other services. Salaries, labor and benefits increased by \$19.0 million (including \$7.7 million incurred by PCS) which is primarily related to acquisition growth and increased headcount. The increase in rent and occupancy, other operating expenses, transaction costs and equity-based compensation is related to acquisition growth, primarily from PCS. Cost of net revenue during the three months ended March 31, 2020 included a reduction to expense of \$1.1 million, consisting of a \$2.0 million reduction in the fair value of an earnout liability, a gain of \$0.6 million related to the sale of an investment, offset by a \$1.5 million expense related to the PCS Transiton Services Agreement.

Cost of net revenue was 87.0% of net revenue for the three months ended March 31, 2020. Excluding PCS, cost of net revenue was \$127.4 million, or 80.9% of net revenue, during the three months ended March 31, 2020, compared to 83.9% for the three months ended March 31, 2019. This decrease is primarily due to lower patient equipment depreciation as a percentage of net revenue as a result of a change in product mix as sales revenue was higher in the three months ended March 31, 2020 compared to the comparable period in 2019. Cost of net revenue as a percentage of net revenue for PCS during the three months ended March 31, 2020 was 115.4%, which is primarily due to anticipated restructuring and operating losses incurred during the period as PCS is fully integrated into the Company's technology platform.

*General and Administrative Expenses.* General and administrative expenses for the three months ended March 31, 2020 were \$14.4 million compared to \$13.1 million for the three months ended March 31, 2019, an increase of \$1.3 million or 9.7%. This increase is primarily due to higher labor costs associated with increased headcount, higher professional fees related to acquisition growth and incremental costs associated with operating as a public company, which were offset by a \$3.5 million decrease in equity-based compensation expense. The expense recorded during the three months ended March 31, 2019 included \$4.9 million in connection with the acceleration of vesting and modification of certain equity-based awards. General and administrative expenses as a percentage of net revenue was 7.5% for the three months ended March 31, 2020, compared to 10.9% for the three months ended March 31, 2019. General and administrative expenses during the three months ended March 31, 2020 included total non-recurring charges of \$4.0 million, consisting of \$1.7 million in equity-based compensation expenses for the three months ended March 31, 2019 included total non-recurring charges of \$4.0 million, consisting of \$1.7 million in equity-based compensation expenses. Some and \$2.2 million in transaction costs, and \$0.1 million in severance expenses. General and administrative expenses do compensation expense and \$2.5 million in transaction costs. Excluding the impact of these non-recurring charges, general and administrative expenses during the expenses and \$2.5 million in transaction costs. Excluding the impact of these non-recurring charges, general and administrative expensed to there expenses as a percentage of net revenue was 5.4% and 4.5% for the three months ended March 31, 2020 and 2019, respectively.

Interest Expense. Interest expense for the three months ended March 31, 2020 was \$7.9 million compared to \$6.3 million for the three months ended March 31, 2019. The increase in interest expense was driven by higher long-term debt obligations to fund acquisitions as well as the 2019 Recapitalization. Additionally, during the three months ended March 31, 2019, AdaptHealth recorded non-cash interest expense representing the change in fair value of its interest rate swap agreements of \$2.7 million; such amount would only be paid out if the interest rate swap agreements were terminated. On August 22, 2019, AdaptHealth designated its swaps as effective cash flow hedges. Accordingly, subsequent to August 22, 2019, changes in the fair value of its interest rate swaps are recorded as a component of other comprehensive income (loss) in equity rather than interest expense. As such, there was no non-cash interest expense related to the Company's interest rate swap agreements during the three months ended March 31, 2020.

Loss on Extinguishment of Debt. Loss on extinguishment of debt for the three months ended March 31, 2019 was \$2.1 million which was a result of the write-off of deferred financing costs related to the 2019 Recapitalization. There were no such amounts during the three months ended March 31, 2020.

**Income Tax Expense.** Income tax expense for the three months ended March 31, 2020 was \$1.1 million compared to income tax expense of \$2.4 million for the three months ended March 31, 2019. The decrease in income tax expense was primarily related to decreased pre-tax income associated with the tax paying entities.

## EBITDA, Adjusted EBITDA and Adjusted EBITDA less Patient Equipment Capex

Adjusted EBITDA less Patient Equipment Capex was \$17.5 million for the three months ended March 31, 2020 compared to \$17.0 million for the three months ended March 31, 2019. For the three months ended March 31, 2020, Adjusted EBITDA less Patient Equipment Capex for PCS was a loss of \$4.5 million, and excluding PCS, Adjusted EBITDA less Patient Equipment Capex was \$22.0 million.

Adjusted EBITDA was \$30.5 million for the three months ended March 31, 2020 compared to \$28.2 million for the three months ended March 31, 2019. For the three months ended March 31, 2020, Adjusted EBITDA for PCS was a loss of \$4.5 million, and excluding PCS, Adjusted EBITDA was \$35.0 million.

AdaptHealth uses EBITDA, Adjusted EBITDA and Adjusted EBITDA less Patient Equipment Capex, which are financial measures that are not prepared in accordance with generally accepted accounting principles in the United States, or U.S. GAAP, to analyze its financial results and believes that they are useful to investors, as a supplement to U.S. GAAP measures. In addition, AdaptHealth's ability to incur additional indebtedness and make investments under its existing credit agreement is governed, in part, by its ability to satisfy tests based on a variation of Adjusted EBITDA less Patient Equipment Capex.

AdaptHealth defines EBITDA as net income (loss) attributable to AdaptHealth Corp., plus net income attributable to noncontrolling interests, interest expense (income), income tax expense (benefit), and depreciation.

AdaptHealth defines Adjusted EBITDA as EBITDA (as defined above), plus loss on extinguishment of debt, equity-based compensation expense, transaction costs, severance, and similar items of expense (income).

AdaptHealth defines Adjusted EBITDA less Patient Equipment Capex as Adjusted EBITDA (as defined above) less patient equipment acquired during the period without regard to whether the equipment was purchased or financed through lease transactions.

AdaptHealth believes Adjusted EBITDA less Patient Equipment Capex is useful to investors in evaluating AdaptHealth's financial performance. AdaptHealth's business requires significant investment in equipment purchases to maintain its patient equipment inventory. Some equipment title transfers to patients' ownership after a prescribed number of fixed monthly payments. Equipment that does not transfer wears out or oftentimes is not recovered after a patient's use of the equipment terminates. AdaptHealth uses this metric as the profitability measure in its incentive compensation plans that have a profitability component and to evaluate acquisition opportunities, where it is most often used for purposes of contingent consideration arrangements. In addition, AdaptHealth's debt agreements contain covenants that use a variation of Adjusted EBITDA less Patient Equipment Capex for purposes of determining debt covenant compliance. For purposes of this metric, patient equipment capital expenditure is measured as the value of the patient equipment received during the accounting period without regard to whether the equipment is purchased or financed through lease transactions.

EBITDA, Adjusted EBITDA and Adjusted EBITDA less Patient Equipment Capex should not be considered as measures of financial performance under U.S. GAAP, and the items excluded from EBITDA, Adjusted EBITDA and Adjusted EBITDA less Patient Equipment Capex are significant components in understanding and assessing financial performance. Accordingly, these key business metrics have limitations as an analytical tool. They should not be considered as an alternative to net income or any other performance measures derived in accordance with U.S. GAAP or as an alternative to cash flows from operating activities as a measure of AdaptHealth's liquidity.

The following unaudited table presents the reconciliation of net loss attributable to AdaptHealth, to EBITDA, Adjusted EBITDA and Adjusted EBITDA less Patient Equipment Capex for the three months ended March 31, 2020 and 2019:

	Three Months Ended March 31,			
(in thousands)		2020		2019
	(Unaudited)			I)
Net loss attributable to AdaptHealth Corp.	\$	(158)	\$	(5,800)
Income attributable to noncontrolling interests		424		348
Interest expense excluding change in fair value of interest rate swaps		7,938		3,558
Interest expense - change in fair value of interest rate swaps				2,702
Income tax expense		1,107		2,418
Depreciation, including patient equipment depreciation		16,740		14,972
EBITDA		26,051		18,198
Loss on extinguishment of debt (a)				2,121
Equity-based compensation expense (b)		2,223		5,223
Transaction costs (c)		2,858		2,516
Severance (d)		419		141
Other non-recurring (income) expense (e)		(1,091)		17
Adjusted EBITDA		30,460		28,216
Less: Patient equipment capex (f)		(12,967)		(11,243)
Adjusted EBITDA less Patient Equipment Capex	\$	17,493	\$	16,973

(a) Represents write offs of deferred financing costs related to refinancing of debt.

- (b) Represents amortization of equity-based compensation to employees, including expense resulting from accelerated vesting and modification of certain awards incurred in 2019.
- (c) Represents transaction costs related to acquisitions and the 2019 Recapitalization.
- (d) Represents severance costs related to acquisition integration and internal AdaptHealth restructuring and workforce reduction activities.
- (e) The 2020 period includes a \$2.0 million reduction in the fair value of an earnout liability, a \$0.6 million gain in connection with the sale of a cost method investment, offset by a \$1.5 million expense associated with the PCS Transition Services Agreement.
- (f) Represents the value of the patient equipment obtained during the respective period without regard to whether the equipment is purchased or financed through lease transactions.

## Liquidity and Capital Resources

AdaptHealth's principal sources of liquidity are its operating cash flows, borrowings under its credit agreements and proceeds from equity issuances. AdaptHealth has used these funds to meet its capital requirements, which consist of salaries, labor, benefits and other employee-related costs, product and supply costs, third-party customer service, billing and collections and logistics costs, capital expenditures including patient equipment, acquisitions and debt service. Our future capital expenditure requirements will depend on many factors, including its patient volume and revenue growth rates. AdaptHealth's capital expenditures are made in advance of patients beginning service. Certain operating costs are incurred at the beginning of the equipment service period and during initial patient set up. AdaptHealth may be required to seek additional equity or debt financing in connection with its business growth. In addition, the recent COVID-19 pandemic has caused disruption in the capital markets, which could make financing more difficult and/or expensive. In the event that additional financing is required from outside sources, AdaptHealth may not be able to raise it on acceptable terms or at all. If additional capital is unavailable when desired, AdaptHealth's business, results of operations, and financial condition would be materially and adversely affected. AdaptHealth believes that its expected operating cash flows, together with its existing cash, cash equivalents, and amounts available under its credit facility, will continue to be sufficient to fund its operations and growth strategies for at least the next 12 months.

As of March 31, 2020, AdaptHealth had \$48.2 million of cash and cash equivalents and \$90.5 million available under its credit facility (including \$50.0 million available under the Delayed Draw and \$40.5 million available under its Revolving Credit Facility after consideration of stand-by letters of credit outstanding of \$2.5 million). To supplement its cash liquidity, in April 2020, AdaptHealth received recoupable advance payments of approximately \$47 million which were made available by CMS under the CARES Act. In addition, in April 2020, AdaptHealth received distributions of the CARES Act provider relief funds of approximately \$17 million which are targeted to offset lost revenue and expenditures incurred in connection with the COVID-19 pandemic. The provider relief funds are subject to certain restrictions and are subject to recoupment if not used for designated purposes. In March 2020, AdaptHealth borrowed \$20.0 million under its revolving credit facility as a precaution in light of the coronavirus health crisis and such amount was repaid in April 2020 upon receipt of the advanced payments from CMS and provider relief funds discussed above. Also, as permitted under the CARES Act, AdaptHealth has elected to defer certain portions of employer-paid FICA taxes otherwise payable from March 27, 2020 to January 1, 2021, which will be paid in two equal installments on December 31, 2021 and December 31, 2022.

In March 2019, AdaptHealth restructured its debt borrowings with its bank group. The credit agreement consisted of \$425 million in credit facilities, which included a \$300 million Credit Facility Term Loan, a \$50 million Delayed Draw Term Loan and a \$75 million New Revolver, all with maturities in March 2024. Amounts borrowed under the Credit Facility Term Loan and Delayed Draw Term Loan bear interest quarterly at variable rates based upon the sum of (a) the LIBOR Rate for such interest period, plus (b) an applicable margin based upon the Company's Consolidated Total Leverage Ratio. The Delayed Draw Term Loan carries 0.5% of unused fee per annum, and the New Revolver carries 0.5% of unused line fee per annum. In November 2019, the Company amended its credit facility primarily to (i) increase the amount available under the Delayed Draw Term Loan from \$50 million to \$100 million, and (ii) revise the Consolidated Total Leverage Ratio thresholds and lower the applicable margin to determine the variable quarterly interest rate under the credit facility.

The Credit Facility Term Loan requires quarterly principal repayments beginning June 30, 2019 through December 31, 2023, and the unpaid principal balance is due at maturity in March 2024. In November 2019, the Company repaid \$50.0 million under the Credit Facility Term Loan; such repayment satisfied the required principal repayments through September 2023. At March 31, 2020, AdaptHealth had \$246.3 million outstanding under the Credit Facility Term Loan (4.35% interest rate at March 31, 2020).

The Delayed Draw Term Loan allows up to \$100 million to be drawn in order to fund permitted acquisitions and to pay fees and transaction costs associated with such acquisitions, and has an availability period from the first business day immediately following the closing date (March 2019) to the earliest of (a) the Credit Facility Term Loan maturity date, (b) twenty-four months following the closing date, or (c) the date of the termination of the commitment. During the three months ended March 31, 2020, AdaptHealth borrowed \$50.0 million under the Delayed Draw Term Loan. The borrowing under the Delayed Draw Term Loan requires quarterly principal repayments of \$0.3 million beginning March 31, 2020 through December 31, 2020, quarterly principal repayments of \$0.6 million beginning March 31, 2021 through December 31, 2023, and the unpaid principal balance is due at maturity in March 2024. At March 31, 2020, AdaptHealth had \$49.7 million outstanding under the Delayed Draw Term Loan (4.11% interest rate at March 31, 2020).

The New Revolver allows up to \$75.0 million to be drawn in order to (1) finance working capital, make capital expenditures and for other general corporate purposes in an amount not to exceed \$25.0 million, and (2) finance permitted acquisitions and to pay fees and transaction costs associated with such acquisitions in an amount not to exceed \$50.0 million. As of March 31, 2020, AdaptHealth had \$32.0 million outstanding under the New Revolver. As discussed above, in April 2020, AdaptHealth repaid \$20.0 million under the New Revolver. Amounts outstanding under the New Revolver are due at maturity in March 2024. The interest rate under the New Revolver was 4.35% at March 31, 2020.

Under the credit agreement, AdaptHealth is subject to various agreements that contain a number of restrictive covenants that, among other things, impose operating and financial restrictions on AdaptHealth. Financial covenants include a Consolidated Total Leverage Ratio and a Fixed Charges Coverage Ratio, as defined in the agreement. Additionally, under the terms of the credit agreement, AdaptHealth may be required to repay principal based on excess cash flow, as defined. AdaptHealth was in compliance with all debt covenants as of March 31, 2020.

In March 2019, AdaptHealth signed a Note and Unit Purchase Agreement with an investor. Pursuant to the agreement, AdaptHealth signed a promissory note agreement with a principal amount of \$100.0 million (the "Promissory Note"). In November 2019, in connection with the transactions completed as part of the Business Combination, the Promissory Note was replaced with a new amended and restated promissory note with a principal amount of \$100.0 million, and the investor converted certain of its members' equity interests to a \$43.5 million promissory note. The new \$100.0 million promissory note, together with the \$43.5 million promissory note, are collectively referred to herein as the New Promissory Note. The outstanding principal balance under the New Promissory Note is due on the tenth anniversary of the closing date of the Business Combination and bears interest at the following rates (a) for the period starting on the closing date and ending on the seventh anniversary, a rate of 12% per annum, with 6% payable in cash and 6% Payment in Kind, and (b) for the period starting on the day after the seventh anniversary of the closing date and ending on the maturity date, a rate equal to the greater of (i) 15% per annum or (ii) the twelve-month LIBOR plus 12% per annum AdaptHealth has the option to pay the PIK interest in cash under the New Promissory Note, which it did through March 31, 2020, and thus no amounts were added to the principal balance through that date. If AdaptHealth elects to prepay the New Promissory Note prior to the third anniversary of the Closing of the Business Combination, then such prepayment of the outstanding principal and accrued interest will be subject to a make-whole premium equal to 10% of the total amount of outstanding principal and accrued interest through the date of such prepayment. If AdaptHealth elects to prepay the New Promissory Note prior to the fourth anniversary but after the third anniversary of the Closing of the Business Combination, then such prepayment of outstanding principal and accrued interest will be subject to a make-whole premium equal to 5% of the total amount of outstanding principal and accrued interest through the date of such prepayment.

At March 31, 2020 and December 31, 2019, AdaptHealth had working capital of \$3.4 million and \$30.5 million, respectively. A significant portion of AdaptHealth's assets consists of accounts receivable from third-party payors that are responsible for payment for the equipment and the services that AdaptHealth provides.

Cash Flow. The following table presents selected data from AdaptHealth's consolidated statement of cash flows:

		Three Months Ended March 31,			
<u>(</u> in thousands)	2020 2019		2019		
		(unaudited)			
Net cash provided by operating activities		24,380	\$	16,232	
Net cash used in investing activities		(111,329)		(26,179)	
Net cash provided by (used in) financing activities		58,235		(369)	
Net decrease in cash and cash equivalents		(28,714)		(10,316)	
Cash at beginning of period		76,878		25,186	
Cash at end of period	\$	48,164	\$	14,870	

Net cash provided by operating activities for the three months ended March 31, 2020 was \$24.4 million compared to \$16.2 million for the three months ended March 31, 2019, an increase of \$8.2 million. The increase was the result of a \$5.7 million improvement in net income (loss), a net decrease of \$9.1 million in non-cash charges primarily from depreciation, non-cash interest expense relating to the Company's interest rate swaps, equity-based compensation expense and write-off of deferred financing costs, a change in deferred taxes of \$1.0 million, and a net \$12.6 million increase in cash used resulting from the change in operating assets and liabilities, primarily resulting from the change in accounts receivable and accounts payable and accrued expenses for the period.

Net cash used in investing activities for the three months ended March 31, 2020 was \$111.3 million compared to \$26.2 million for the three months ended March 31, 2019. The use of funds in the three months ended March 31, 2020 consisted of \$105.8 million for acquisitions, primarily from the PCS, Healthline and Advanced acquisitions, \$7.5 million for equipment and other fixed asset purchases, offset by \$2.0 million of cash proceeds from the sale of an investment. The use of funds in the three months ended March 31, 2019 consisted of \$20.9 million for acquisitions, primarily from the Gould's acquisition, and \$5.3 million for equipment and other fixed asset purchases.

Net cash provided by financing activities for the three months ended March 31, 2020 was \$58.2 million compared to net cash used in financing activities of \$0.4 million for the three months ended March 31, 2019. Net cash provided by financing activities for the three months ended March 31, 2020 consisted of \$70.0 million of borrowings from long-term debt and lines of credit, offset by total repayments of \$11.8 million on long-term debt and capital lease obligations. For the three months ended March 31, 2019, net cash used in financing activities was primarily related to the 2019 Recapitalization, and consisted of \$317.0 million of borrowings from long-term debt and lines of credit, \$20.0 million of proceeds from the sale of members' interests, and proceeds of \$100.0 million from the issuance of a note payable, offset by total repayments of \$16.9 million on long-term debt, lines of credit and capital lease obligations, payments of \$9.0 million for debt financing costs, payments of \$0.8 million for equity issuance costs, payment of \$3.7 million for the redemption of members' interests, payment of \$12.0 million for contingent consideration in connection with an acquisition, and distributions to members of \$250.0 million.

#### **Critical Accounting Policies and Significant Estimates**

The discussion and analysis of the Company's financial condition and results of operations is based upon the Company's consolidated financial statements, which have been prepared in accordance with U.S. GAAP. The preparation of the Company's consolidated financial statements requires its management to make estimates and judgments that affect the reported amounts of assets, liabilities, revenue and expenses and related disclosures of contingent assets and liabilities. The Company's management bases its estimates, assumptions and judgments on historical experience and various other factors that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Different assumptions and judgments would change the estimates used in the preparation of the Company's consolidated financial statements which, in turn, could change the results from those reported. In addition, actual results may differ from these estimates and such differences could be material to the Company's financial position and results of operations.

Critical accounting policies and significant estimates are those that the Company's management considers the most important to the portrayal of the Company's financial condition and results of operations because they require management's most difficult, subjective or complex judgments, often as a result of the need to make estimates about the effect of matters that are inherently uncertain. The Company's critical accounting policies and significant estimates in relation to its consolidated financial statements include those related to revenue recognition, accounts receivable, business combinations, and goodwill valuation. There have been no material changes in the Company's critical accounting policies as compared to the critical accounting policies described in the Company's Annual Report on Form 10-K for the year ended December 31, 2019.

#### **Recent Accounting Pronouncements**

Recently issued accounting pronouncements that may be relevant to the Company's operations but have not yet been adopted are outlined in Note 1 (f), *Recently Issued Accounting Pronouncements*, to its consolidated interim financial statements included elsewhere in this report.

#### **Off-Balance Sheet Arrangements**

As of March 31, 2020, the Company did not have any off-balance sheet arrangements, as defined in Item 303(a)(4)(ii) of Regulation S-

#### K.

#### **Commitments and Contingencies**

In the normal course of business, the Company is subject to loss contingencies, such as legal proceedings and claims arising out of its business that cover a wide range of matters. In accordance with the Financial Accounting Standards Board Accounting Standards Codification Topic 450, *Accounting for Contingencies*, the Company records accruals for such loss contingencies when it is probable that a liability has been incurred and the amount of loss can be reasonably estimated. The Company's management believes any liability that may ultimately result from its resolution will not have a material adverse effect on the Company's financial conditions or results of operations.

Other contingencies arising in the normal course of business relate to acquisitions and the related contingent purchase prices and deferred payments.

#### Item 3. Quantitative and Qualitative Disclosures About Market Risk

This item is not applicable to smaller reporting companies.

#### Item 4. Controls and Procedures

#### Evaluation of Disclosure Controls and Procedures

Under the supervision and with the participation of our management, including our principal executive officer and principal financial and accounting officer, we conducted an evaluation of the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act) as of the end of the fiscal quarter ended March 31, 2020. Based on this evaluation, our Chief Executive Officer and Chief Financial Officer have concluded that, during the period covered by this Quarterly Report, our disclosure controls and procedures were not effective due to a material weakness in internal control over financial reporting relating to the timeliness of our review controls over non-routine transactions that was disclosed in Part II, Item 9A of our Annual Report on Form 10-K for the fiscal year ended December 31, 2019 (our "2019 Annual Report"). Notwithstanding the identified material weakness, management, including our principal executive officer and principal financial officer, believes the consolidated financial statements included in this Quarterly Report on Form 10-Q fairly represent in all material respects our financial condition, results of operations and cash flows at and for the periods presented in accordance with U.S. GAAP.

Disclosure controls and procedures are designed to ensure that information required to be disclosed by us in our Exchange Act reports is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including our principal executive officer and principal financial and accounting officer or persons performing similar functions, as appropriate, to allow timely decisions regarding required disclosure.

#### Material Weakness Remediation

Management continues to be actively engaged to take steps to remediate the material weakness noted above, including (1) implementing processes to improve overall efficiency and accuracy of accounting and (2) hiring dedicated and experienced technical resources (including engaging a third-party consultant to assist management) to strengthen its corporate oversight over financial reporting and controls associated with complex accounting matters. While we have made significant progress, the material weakness cannot be considered remediated until the enhanced controls have operated effectively for a sufficient period of time.

#### Changes in Internal Control over Financial Reporting

Except with respect to the changes in connection with the implementation of the initiatives to remediate the material weakness noted above, there were no changes in the Company's internal control over financial reporting that occurred during the fiscal quarter ended March 31, 2020 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting despite the fact that many of our employees are working remotely due to the COVID-19 pandemic. We are continually monitoring and assessing the COVID-19 situation on our internal controls to minimize the impact on their design and operating effectiveness.

#### PART II. OTHER INFORMATION

# Item 1. Legal Proceedings

AdaptHealth is involved in investigations, claims, lawsuits and other proceedings arising in the ordinary course of its business. These matters involve personnel and employment issues, regulatory matters, personal injury, contract and other proceedings arising in the ordinary course of business, which have not resulted in any material losses to date. Although AdaptHealth does not expect the outcome of these proceedings will have a material adverse effect on its financial condition or results of operations, such matters are inherently unpredictable. Therefore, AdaptHealth could incur judgments or enter into settlements or claims that could materially impact its financial condition or results of operations.

In addition, on July 25, 2017, AdaptHealth Holdings was served with a subpoena by the U.S. Attorney's Office for the United States District Court for the Eastern District of Pennsylvania ("EDPA") pursuant to 18 U.S.C. §3486 to produce certain audit records and internal communications regarding ventilator billing. The investigation appears to be focused on billing practices regarding one payor that contracted for bundled payments for certain ventilators. AdaptHealth Holdings has cooperated with investigators and, through agreement with the EDPA, has submitted all information requested. An independent third party was retained by AdaptHealth Holdings that identified overpayments and underpayments for ventilator billings related to the payor, and a remittance was sent to reconcile that account. AdaptHealth Holdings has cooperated and fully complied with the subpoena. On October 3, 2019, AdaptHealth received a follow-up civil investigative demand from the EDPA regarding a document previously produced to the EDPA and patients included in the review by the independent third party. AdaptHealth has responded to the EDPA and supplemented its production as requested. At this time, AdaptHealth Holdings cannot provide any assurance as to whether the EDPA will seek additional information or pursue this matter further.

#### Item 1A. Risk Factors

Factors that could cause our actual results to differ materially from those in this report are any of the risks disclosed in our Annual Report on Form 10-K for the year ended December 31, 2019 filed with the SEC on March 6, 2020. Any of those factors could result in a significant or material adverse effect on our results of operations or financial condition. Additional risk factors not presently known to us or that we currently deem immaterial may also impair our business or results of operations.

The following risk factors are provided to update the risk factors previously disclosed under the heading "Risk Factors" in the Company's Annual Report on Form 10-K for the year ended December 31, 2019.

# The recent coronavirus (COVID-19) pandemic and the global attempt to contain it may harm our business, results of operations and ability to execute on our business plan.

The global spread of the coronavirus (COVID-19) and the various attempts to contain it have created significant volatility, uncertainty and economic disruption. In response to government mandates, health care advisories and otherwise responding to employee, customer and supplier concerns, we have altered certain aspects of our operations. Our workforce has had to spend a significant amount of time working from home, which impacts their productivity. While many of our operations can be performed remotely, there is no guarantee that we will be as effective while working remotely because our team is dispersed, many employees may have additional personal needs to attend to (such as looking after children as a result of school closures or family who become sick), and employees may become sick themselves and be unable to work. Our suppliers and vendors have similarly had their operations altered. To the extent the resulting economic disruption is severe, we could see some vendors go out of business, resulting in supply constraints and increased costs or delays in meeting the needs of our patients.

The full extent to which the COVID-19 pandemic and the various responses to it impacts our business, operations and financial results will depend on numerous other evolving factors that we may not be able to accurately predict, including:

- the duration and scope of the pandemic;
- governmental, business and individuals' actions that have been and continue to be taken in response to the pandemic;
- the availability and cost to access the capital markets;
- our ability to pursue, diligence, finance and integrate acquisitions,
- our ability to comply with financial and operating covenants in our debt and operating lease agreements;
- potential for goodwill impairment charges;
- our ability to comply with the requirements necessary to retain the CARES Act provider relief funds we received;
- the effect on our patients, physician and facility referral sources and demand for and ability to pay for medical services;
- disruptions or restrictions on our employees' ability to travel and to work, including as a result of their health and wellbeing;
- availability of third-party providers to whom we outsource portions of our internal business functions, including billing and administrative functions relating to revenue cycle management; and
- increased cybersecurity risks as a result of remote working condition.

During the COVID-19 crisis, we may not be able to provide the same level of service and products that our patients, physicians and facility referral sources are used to, which could negatively impact their perception of our products or services. Furthermore, given increased government expenditures associated with their COVID-19 response, we could see increased government obligations which could negatively impact our results of operations.

We will continue to actively monitor the issues raised by the COVID-19 pandemic and may take further actions that alter our business operations, as may be required by federal, state, or local authorities, or that we determine are in the best interests of our employees, customers, and stockholders. It is not clear what the potential effects any such alterations or modifications may have on our business, including the effects on our customers, suppliers or vendors, or on our financial results.

The potential effects of COVID-19 could also heighten the risks disclosed in many of our risk factors that are included in Part I, Item 1A, Risk Factors, in our 2019 Annual Report, including as a result of, but not limited to, the factors described above. Because the COVID-19 situation is unprecedented and continuously evolving, the other potential impacts to our risk factors that are further described in our 2019 Annual Report are uncertain. See Item 1A, Risk Factors, in our 2019 Annual Report.

# AdaptHealth's reliance on relatively few suppliers for the majority of its patient service equipment and supplies could adversely affect AdaptHealth's ability to operate.

AdaptHealth currently relies on a relatively small number of suppliers to provide it with the majority of its patient service equipment and supplies. Significant price increases, or disruptions in the ability to obtain such equipment and supplies from existing suppliers, may force AdaptHealth to use alternative suppliers. Additionally, any new excise taxes imposed on manufacturers of certain medical equipment could be passed on to customers, such as AdaptHealth. Such manufacturers may be forced to make other changes to their products or manufacturing processes that are unacceptable to AdaptHealth, resulting in a need to change suppliers. Any change in suppliers AdaptHealth uses could cause delays in the delivery of such products and possible losses in revenue, which could adversely affect AdaptHealth's results of operations. In addition, alternative suppliers may not be available, or may not provide their products and services at similar or favorable prices. If AdaptHealth cannot obtain the patient service equipment and supplies it currently uses, or alternatives at similar or favorable prices, AdaptHealth's ability to provide such products may be severely impacted, which could have an adverse effect on its business, financial condition, results of operations, cash flow, capital resources and liquidity. In December 2019, the novel strain of coronavirus began to impact the population of China, and subsequently the COVID-19 pandemic related to this virus has impacted manufacturing in all of the regions where AdaptHealth's suppliers manufacture their products. While the global closures and limitations on movement related to COVID-19 are expected to be temporary, and while such closures, limitations and related impacts have not materially disrupted AdaptHealth's supply chain, such supply chain disruption remains possible and the financial impact of any such disruption cannot be estimated at this time. Should such closures and limitations on movement continue for an extended period of time, the impact on our supply chain could materially and adversely affect our business and results of operations.

# AdaptHealth's revenue could be impacted by federal and state changes to reimbursement and other aspects of Medicaid and Medicare.

AdaptHealth derived approximately 32% of its revenue for the year ended December 31, 2019 and 27% of its revenue for the three months ended March 31, 2020 from Medicare and various state-based Medicaid programs. These programs are subject to statutory and regulatory changes affecting overall spending, base rates or basis of payment, retroactive rate adjustments, annual caps that limit the amount that can be paid (including deductible and coinsurance amounts) for rehabilitation therapy services rendered to Medicare beneficiaries, administrative or executive orders and government funding restrictions, all of which may materially adversely affect the rates and frequency at which these programs reimburse AdaptHealth. For example, the Medicaid Integrity Contractor program is increasing the scrutiny placed on Medicare and legislation at both the federal and state levels have called for cuts in reimbursement for healthcare providers participating in the Medicaid spending could result in substantial reductions in AdaptHealth's revenue and profitability. Payors may disallow AdaptHealth's requests for reimbursement based on determinations that certain costs are not reimbursable or casonable because either adequate or additional documentation was not provided or because certain services were not covered or considered reasonably necessary. Additionally, revenue from these payors can be retroactively adjusted after a new examination during the claims settlement process or as a result of post payment audits.

As a result of the National Emergency Declaration and pursuant to the provisions of the CARES Act, among other things, CMS has issued regulatory guidance indicating enforcement discretion and flexibility regarding the provisions of items and services by DMEPOS suppliers like AdaptHealth. These provisions have been announced through Frequently Asked Questions, Public Releases, blanket waivers under Section 1135 of the Social Security Act and publication of an Interim Final Rule with Request for Comment on March 13, 2020. These provisions include modifications of various requirements required under CMS regulations and Medicare program rules. However, many of the provisions regarding documentation, coverage and flexibilities remain subject to further guidance and interpretation by CMS and Medicare Administrative Contractors. Due to the speed under which this guidance was issued, CMS or Medicare Administrative Contractors will not further modify or clarify this guidance in a way that affects AdaptHealth's operations or cash flows. Because the guidance issued changes frequently, AdaptHealth may be required to modify its compliance process and operations to remain in compliance with such guidance.

The CARES Act also provides for a temporary suspension of reduced rates for items and services provided by AdaptHealth. Under existing regulations, CMS applies a blended payment rate for DME furnished in rural or noncontiguous competitive bidding areas. Through December 31, 2020, that blended rate would be equal to 50% of the adjusted fee schedule amount (adjusted based on competitively bid prices) and 50% of the unadjusted DMEPOS fee schedule amount. DME furnished in non-rural or contiguous areas would not have been eligible for this blended rate, and instead many suppliers would have experienced a reduced payment rate that reflected competitively bid prices. The CARES Act extends the 50% blended rate for rural / non-contiguous areas through December 31, 2020 or until the end of the emergency period (if longer). The CARES Act also introduces a new blended rate for non-rural / contiguous areas equal to 75% of the adjusted fee schedule amount. While these rate changes may provide temporary financial benefit to AdaptHealth, these effects are temporary for the duration of the rate adjustment.

The CARES Act temporarily suspends the 2% payment adjustment currently applied to all Medicare fee-for-service claims due to sequestration. The suspension is effective for claims with dates of service from May 1 through December 31, 2020. However, CMS and Medicare Administrative Contractors may issue guidance that affects the implementation of this provision in the CARES Act, which may adversely affect AdaptHealth. For example, Medicare Administrative Contractors have interpreted this provision not to apply to capped rental items when the start date for rental equipment is before May 1. Additionally, the impact of the temporary suspension of sequestration for Medicare Advantage may depend on specific AdaptHealth individual contracts with Medicare Advantage Organizations.

## Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

# Item 3. Defaults upon Senior Securities

None.

#### Item 4. Mine Safety Disclosures

Not applicable.

## Item 5. Other Information

None.

#### Item 6. Exhibits

See Exhibit Index for documents filed or furnished herewith and incorporated herein by reference.

# EXHIBIT INDEX

Exhibit Number	Description
2.1	Merger Agreement, dated as of July 8, 2019, by and among the Company, Merger Sub, AdaptHealth Holdings, the Blocker
	Companies, the AdaptHealth Holdings Unitholders' Representative and, solely for the purposes specified therein, the
	Blocker Sellers (incorporated by reference to Exhibit 2.1 of the Company's Current Report on Form 8-K, filed with the SEC
	on November 14, 2019).
2.2	Amendment No. 1 to Merger Agreement, dated as of October 15, 2019, by and among the Company, Merger Sub,
	AdaptHealth Holdings, the Blocker Companies, the AdaptHealth Holdings Unitholders' Representative and, solely for the
	purposes specified therein, the Blocker Sellers (incorporated by reference to Exhibit 2.2 of the Company's Current Report on
2.1	Form 8-K, filed with the SEC on November 14, 2019).
3.1	Second Amended and Restated Certificate of Incorporation (incorporated by reference to Exhibit 3.1 of the Company's Current Report on Form 8-K, filed with the SEC on November 14, 2019).
3.2	Amended and Restated Bylaws (incorporated by reference to Exhibit 3.2 of the Company's Current Report on Form 8-K,
5.2	filed with the SEC on November 14, 2019).
3.3	Certificate of Correction to Second Amended and Restated Certificate of Incorporation (incorporated by reference to Exhibit
5.5	10.3 to the Company's Annual Report on Form 10-K filed with the SEC on March 6, 2020).
4.1	Specimen Common Stock Certificate (incorporated by reference to Exhibit 4.2 of the Company's Registration Statement on
	Form S-1, filed with the SEC on February 13, 2018).
4.2	Registration Rights Agreement, dated as of November 8, 2019, by and among AdaptHealth Holdings, the Company and the
	persons listed in Schedule of Investors therein (incorporated by reference to Exhibit 4.1 of the Company's Current Report
	on Form 8-K, filed with the SEC on November 14, 2019).
4.3	Specimen Warrant Certificate (incorporated by reference to Exhibit 4.3 of the Company's Registration Statement on Form S-
	1, filed with the SEC on February 13, 2018).
4.4	Warrant Agreement, dated as of February 15, 2018, by and between the Company and Continental Stock Transfer & Trust
	Company (incorporated by reference to Exhibit 4.1 of the Company's Current Report on Form 8-K, filed with the SEC on February 22, 2018).
10.1	Exchange Agreement, dated November 8, 2019, by and among AdaptHealth Holdings, the Company and the other persons
10.1	listed therein (incorporated by reference to Exhibit 10.1 of the Company's Current Report on Form 8-K, filed with the SEC on
	November 14, 2019).
10.2	Tax Receivable Agreement, dated November 8, 2019, by and among AdaptHealth Holdings, the Company and the Non-
	Blocker AdaptHealth Members and the Blocker Sellers (incorporated by reference to Exhibit 10.2 of the Company's Current
	Report on Form 8-K, filed with the SEC on November 14, 2019).
10.3	Fifth Amended and Restated Limited Liability Company Agreement of AdaptHealth Holdings, dated as of November 8,
	2019, by and between the Company and the Members named therein (incorporated by reference to Exhibit 10.3 of the
	Company's Current Report on Form 8-K, filed with the SEC on November 14, 2019).
10.4†	Form of Indemnification Agreement (incorporated by reference to Exhibit 10.4 of the Company's Current Report on Form 8-
10.5	K, filed with the SEC on November 14, 2019).
10.5	Board Designee Rights Letter Agreement, dated as of November 8, 2019, by and between the Company, AdaptHealth
	Holdings, BM AH Holdings, LLC, BlueMountain Foinaven Master Fund L.P., BMSB L.P., BlueMountain Fursan Fund L.P.
	and BlueMountain Summit Opportunities Fund II (US) L.P. (incorporated by reference to Exhibit 10.5 of the Company's Current Report on Form 8-K, filed with the SEC on November 14, 2019).
10.6	Third Amended and Restated Credit and Guaranty Agreement, dated as of March 20, 2019, by and among AdaptHealth
10.0	LLC, the guarantors named therein, CIT Finance LLC as administrative agent, and the lenders party thereto (incorporated by
	reference to Exhibit 10.6 of the Company's Current Report on Form 8-K, filed with the SEC on November 14, 2019).

10.7	Amendment No. 1 to Third Amended and Restated Credit and Guaranty Agreement, dated as of August 22, 2019, by and
	among AdaptHealth LLC, the guarantors named therein, CIT Finance LLC as administrative agent, and the lenders party
	thereto (incorporated by reference to Exhibit 10.7 of the Company's Current Report on Form 8-K, filed with the SEC on
	<u>November 14, 2019).</u>
10.8*	Amendment No. 2 to Third Amended and Restated Credit and Guaranty Agreement, dated as of November 8, 2019, by and
	among AdaptHealth LLC, the guarantors named therein, CIT Finance LLC as administrative agent, and the lenders party
	thereto (filed to append Annex A thereto, which was inadvertently excluded from Exhibit 10.8 to the Company's Current
	Report on Form 8-K filed with the SEC on November 14, 2019).
10.9†	Employment Agreement, dated as of March 20, 2019, by and between AdaptHealth Holdings and Luke McGee
	(incorporated by reference to Exhibit 10.9 of the Company's Current Report on Form 8-K, filed with the SEC on November 14, 2019).
10.10†	Employment Agreement, dated as of March 20, 2019, by and between AdaptHealth Holdings and Joshua Parnes
	(incorporated by reference to Exhibit 10.10 of the Company's Current Report on Form 8-K, filed with the SEC on November
	14, 2019).
10.11†	Employment Agreement, dated as of November 10, 2014, by and between AdaptHealth Holdings and Gregg Holst
	(incorporated by reference to Exhibit 10.11 of the Company's Current Report on Form 8-K, filed with the SEC on November
	14.2019).
10.12†	AdaptHealth Corp. 2019 Stock Incentive Plan (incorporated by reference to Exhibit 10.12 of the Company's Current Report
	on Form 8-K, filed with the SEC on November 14, 2019).
10.13†	Form of Restricted Stock Grant Notice and Agreement under the AdaptHealth Corp. 2019 Stock Incentive Plan (incorporated
1	by reference to Exhibit 10.13 of the Company's Current Report on Form 8-K, filed with the SEC on November 14, 2019).
10.14†	Form of Option Grant Notice and Agreement under the AdaptHealth Corp. 2019 Stock Incentive Plan (incorporated by
	reference to Exhibit 10.14 of the Company's Current Report on Form 8-K, filed with the SEC on November 14, 2019).
10.15	Letter Agreement, dated as of February 15, 2018, among the Company, Deerfield/RAB Ventures, LLC, Richard Barasch,
	Christopher Wolfe, Steven Hochberg, Dr. Mohit Kaushal, Dr. Gregory Sorensen and Dr. Susan Weaver (incorporated by
	reference to Exhibit 10.1 to the Company's Current Report on Form 8-K filed with the SEC on February 22, 2018).
10.16	Warrant Purchase Agreement, dated February 15, 2018, between the Registrant and Deerfield/RAB Ventures, LLC
	(incorporated by reference to Exhibit 10.4 to the Company's Current Report on Form 8-K filed with the SEC on February 22,
	2018).
10.17†	AdaptHealth Corp. 2019 Employee Stock Purchase Plan (incorporated by reference to Exhibit 4.7 to the Company's
	Registration Statement on Form S-8 filed with the SEC on January 22, 2020).
10.18+	Securities Purchase Agreement, dated as of November 21, 2019, by and among AdaptHealth LLC, McKesson Medical-
	Surgical, Inc., NRE Holding Corporation and McKesson Patient Care Solutions, Inc. (incorporated by reference to Exhibit
	10.18 to the Company's Annual Report on Form 10-K filed with the SEC on March 6, 2020).
31.1*	Certification of Principal Executive Officer Pursuant to Securities Exchange Act Rules 13a-14(a) and 15(d)-14(a), as adopted
	Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2*	Certification of Principal Financial Officer Pursuant to Securities Exchange Act Rules 13a-14(a) and 15(d)-14(a), as adopted
	Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32**	Certification of Principal Executive Officer and Principal Financial Officer Pursuant to 18 U.S.C. Section 1350, as adopted
	Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS***	XBRL Instance Document
101.SCH***	XBRL Taxonomy Extension Schema Document
101.CAL***	XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF***	XBRL Taxonomy Extension Definition Linkbase Document
101 I A B***	XBRI Taxonomy Extension Label Linkhase Document

101.LAB\*\*\*XBRL Taxonomy Extension Label Linkbase Document101.PRE\*\*\*XBRL Taxonomy Extension Presentation Linkbase Document

Table of Contents

<sup>\*</sup> Filed herewith.

<sup>\*\*</sup> Furnished herewith.

<sup>\*\*\*</sup> XBRL (eXtensible Business Reporting Language) information is furnished and not filed or a part of a registration statement or prospectus for purposes of Sections 11 or 12 of the Securities Act of 1933, is deemed not filed for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, and otherwise is not subject to liability under these sections.

<sup>†</sup> Management contract or compensatory plan or arrangement.

<sup>+</sup> Portions of this exhibit have been omitted in accordance with Item 601(b)(10)(iv) of Regulation S-K.

# SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

May 8, 2020

May 8, 2020

AdaptHealth Corp.

By: <u>/s/ Luke McGee</u> Luke McGee Chief Executive Officer and Director (*Principal Executive Officer*)

By: <u>/s/ Gregg Holst</u> Gregg Holst Chief Financial Officer (*Principal Financial and Accounting Officer*)

#### AMENDMENT NO. 2 TO THIRD AMENDED AND RESTATED CREDIT AND GUARANTY AGREEMENT

THIS AMENDMENT NO. 2 TO THIRD AMENDED AND RESTATED CREDIT AND GUARANTY AGREEMENT (this "<u>Amendment</u>"), effective as of November 8, 2019 (the "<u>Effective Date</u>"), is made by and among **CIT FINANCE LLC** ("<u>Administrative Agent</u>"), each of the financial entities set forth on the signature pages hereto constituting all the Lenders under the Credit Agreement, **ADAPTHEALTH LLC**, a Delaware limited liability company ("<u>Borrower</u>"), and each of the entities set forth on the signature pages hereto as "Guarantors" (the "<u>Guarantors</u>").

#### BACKGROUND STATEMENT

A. Borrower, Guarantors, Administrative Agent and Lenders are parties to that certain Third Amended and Restated Credit and Guaranty Agreement, dated as of March 20, 2019 (as amended by that certain Amendment No. 1 to Third Amended and Restated Credit and Guaranty Agreement dated as of August 22, 2019, as amended hereby, and as the same may be further amended, restated, modified or supplemented, the "<u>Credit Agreement</u>"), pursuant to which the Borrower and Guarantors established certain financing arrangements with Lenders upon the terms and conditions set forth therein. Capitalized terms used herein without definition shall have the meanings given to them in the Credit Agreement.

B. Pursuant to that certain Agreement and Plan of Merger, dated as of July 8, 2019 (as amended by that certain Amendment No. 1 to Agreement and Plan of Merger dated as of October 15, 2019, and as the same may be further amended, restated, supplemented or otherwise modified, the "Merger Agreement"), by and among DFB Healthcare Acquisitions Corp., a Delaware corporation ("DFB Healthcare"), DFB Merger Sub LLC, a Delaware limited liability company ("Merger Sub"), AdaptHealth Holdings LLC, a Delaware limited liability company ("Merger Sub"), AdaptHealth Holdings LLC, a Delaware limited liability company ("Merger Sub"), Merger Sub will merge into AdaptHealth Holdings, with AdaptHealth Holdings surviving as a partially owned subsidiary of DFB Healthcare (collectively, the "SPAC Merger");

C. The closing of the SPAC Merger is conditioned upon, among other things, the approval by the stockholders of DFB Healthcare of a number of proposals including the Business Combination Proposal, the Charter Proposal, the Nasdaq Proposal, the 2019 Plan Proposal and the 2019 ESPP Proposal (each as defined in the Proxy Statement for Special Meeting in Lieu of the 2019 Annual Meeting of Stockholders of DFB Healthcare, and collectively the "<u>Proposals</u>"); and

D. In connection with the SPAC Merger, Administrative Agent, Lenders and the Loan Parties wish to amend certain provisions of the Credit Agreement as set forth herein, which shall become effective in accordance with the terms and conditions set forth below.

#### STATEMENT OF AGREEMENT

The parties hereto, in consideration of the mutual covenants and agreements set forth herein (the receipt and sufficiency of which is hereby acknowledged), agree as follows:

1. <u>Recitals</u>. This Amendment shall constitute a Loan Document and the Recitals set forth above shall be construed as part of this Amendment as if set forth fully in the body of this Amendment.

2. <u>Amendment to Credit Agreement</u>. Upon satisfaction of the conditions set forth in

Section 4 hereof, the Credit Agreement is hereby amended pursuant to the attached Annex A as follows:

(a) <u>Annex A</u>. In <u>Annex A</u> hereto, deletions of text in the Credit Agreement (including, to the extent included in such <u>Annex A</u>, each Schedule or Exhibit to the Credit Agreement) are indicated by <del>struck through text</del>, and insertions of text as amended by this Amendment are indicated by <u>bold, double-</u><u>underlined text</u>.

3. **<u>Representations and Warranties; Covenants</u>**. Each Loan Party hereby represents and warrants as follows:

(a) <u>Bringdown</u>. After giving effect to this Amendment, each of the representations and warranties of the Loan Parties contained in the Credit Agreement and in the other Loan Documents is true and correct in all material respects (<u>provided</u>, that if any representation or warranty is by it terms qualified by concepts of materiality, such representation or warranty shall be true and correct in all respects) on and as of the date hereof with the same effect as if made on and as of the date hereof (except to the extent any such representation or warranty is expressly stated to have been made as of a specific date, in which case such representation or warranty is true and correct as of such date).

(b) <u>No Default</u>. No Default or Event of Default has occurred and is continuing under the Credit Agreement or any of the other Loan Documents, each Loan Party is in compliance with all terms and provisions set forth in the Credit Agreement and the other Loan Documents, and each of the conditions set forth in <u>Section 4</u> of this Amendment has been satisfied.

(c) <u>Enforceability: Non-Contravention</u>. The execution and delivery by each Loan Party of this Amendment and the performance by it of the transactions herein contemplated (i) are and will be within its powers; (ii) have been authorized by all necessary action; (iii) are not and will not be in contravention of any Loan Party's Organization Documents; (iv) are not and will not conflict with or result in any breach or contravention of, or the creation of any Lien under, (A) any Contractual Obligation under any Material Contract to which any Loan Party is a party or (B) any order, injunction, writ or decree of any Governmental Authority or any arbitral award to which any Loan Party or the Property of any Loan Party is subject; (v) will not violate any Law; or (vi) will not result in a limitation on any material licenses, permits or other Governmental Approvals applicable to the business, operations or properties of any Loan Party. This Amendment and all allonges, assignments, instruments, documents, and agreements executed and delivered in connection herewith, are and will be valid, binding, and enforceable against Borrower and each other Loan Party in accordance with their respective terms, except as such enforceability may be limited (x) by general principles of equity and conflicts of laws or (y) by bankruptcy, reorganization, insolvency, moratorium or other laws of general application relating to or affecting the enforcement, of Administrative Agent's rights.

(d) <u>No Conflicts</u>. No consent, approval, authorization or order of, or filing, registration or qualification with, any court or Governmental Authority or third party is required in connection with the execution, delivery or performance by the Loan Parties of this Amendment.

(e) <u>No Material Adverse Effect</u>. No Material Adverse Effect has occurred and is continuing, and the Loan Parties know of no event, condition or state of facts since the date of the Credit Agreement that could reasonably be expected to have a Material Adverse Effect.

(f) <u>Obligations</u>. The execution and delivery of this Amendment does not diminish or reduce

the Loan Parties' obligations under the Loan Documents, except as expressly modified by this Amendment.

(g) <u>No Claims</u>. The Loan Parties have no claims, counterclaims, offsets or defenses to the Loan Documents and the performance of their obligations thereunder, or if a Loan Party has any such claims, counterclaims, offsets, or defenses arising from events occurring on or before the date hereof, whether known or unknown, to the Loan Documents or any transaction related to the Loan Documents, the same are hereby waived, relinquished and released in consideration of Administrative Agent's execution and delivery of this Amendment.

4. **Effectiveness Conditions**. This Amendment shall be effective upon completion of the following conditions precedent (all documents to be in form and substance satisfactory to Administrative Agent and Administrative Agent's coursel):

(a) <u>Executed Amendment and Related Deliverables</u>. Prior to the Effective Date, Administrative Agent shall have received each of the following to the satisfaction of Administrative Agent in its sole discretion:

(i) Executed counterparts of this Amendment and other Loan Documents to be executed as of the Effective Date, each properly executed by a Responsible Officer of the signing Loan Party and each other Person party thereto;

(ii) each agreement, document and instrument set forth on the closing checklist prepared by Administrative Agent's counsel and provided to the Loan Parties, each in form and substance satisfactory to Administrative Agent and such further documents, information, certificates, records and filings as Administrative Agent may reasonably request;

(iii) all representations and warranties of the Loan Parties contained herein shall be true, correct and complete in all material respects (provided, that if any representation or warranty is by its terms qualified by concepts of materiality, such representation shall be true and correct in all respects) as of the Effective Date, except to the extent that any such representation or warranty relates to a specific date in which case such representation or warranty shall be true and correct as of such earlier date (and each Loan Party's delivery of its respective signature hereto shall be deemed to be its certification thereof);

(iv) all costs and expenses referenced in <u>Section 5(d)</u> to this Amendment, and all other fees, expenses and other amounts due and payable under any Loan Document on or prior to the date hereof;

(v) delivery to Administrative Agent of a certificate executed by a Responsible Officer of the Borrower Representative certifying that the conditions herein have been satisfied and that the representations and warranties contained herein are true and correct in all material respects (provided, that if any representation or warranty is by its terms qualified by concepts of materiality, such representation and warranty shall be true and correct in all respects) as of the Effective Date, except to the extent that any such representation or warranty relates to a specific date in which case such representation or warranty shall be true and correct as of such earlier date;

(vi) delivery to Administrative Agent and Lenders all documentation and other information about the Loan Parties required under applicable "know your customer" and anti-money laundering rules and regulations, including the PATRIOT Act, that has been requested by Administrative Agent and the Lenders;

(vii) delivery to Administrative Agent and Lenders of any requested Beneficial Ownership Certification in relation to any Borrower that qualifies as a "legal entity customer" under the Beneficial Ownership Regulation; and

(viii) delivery of a legal opinion in form and substance satisfactory to Administrative Agent.

(b) <u>SPAC Merger Related Deliverables</u>. Administrative Agent shall have received each of the following to the satisfaction of Administrative Agent in its sole discretion:

(i) fully executed and delivered copies of the Merger Agreement and related agreements, in each case acceptable to Administrative Agent in its sole discretion;

(ii) such documents and certifications as Administrative Agent may reasonably require to evidence the approval of the Proposals by the stockholders of DFB Healthcare;

(iii) such documents and certifications as Administrative Agent may reasonably require to evidence the closing of the SPAC Merger; and

(iv) such further documents, information, certificates, records and filings (including those related to the SPAC Merger) as Administrative Agent may reasonably request, including with respect to confirmation of ownership and capital structure of the Loan Parties.

5. <u>Effect of Amendment</u>. From and after the Effective Date, all references to the Credit Agreement set forth in any other Loan Document or other agreement or instrument shall, unless otherwise specifically provided, be references to the Credit Agreement as amended by this Amendment and as may be further amended, modified, restated or supplemented from time to time. This Amendment is limited as specified and shall not constitute or be deemed to constitute an amendment, modification or waiver of any provision of the Credit Agreement or of any other Loan Document except as expressly set forth herein. Except as expressly amended hereby, the Credit Agreement shall remain in full force and effect in accordance with its terms.

(a) <u>Ratification of Loan Documents</u>. Except as expressly set forth herein, all of the terms and conditions of the Credit Agreement and Loan Documents are hereby ratified and confirmed and continue unchanged and in full force and effect.

(b) <u>Governing Law</u>. This Amendment shall be governed by and construed and enforced in accordance with the laws of the State of New York (without regard to the conflicts of law provisions thereof).

(c) <u>Waiver of Trial by Jury</u>. FACH OF THE LOAN PARTIES AND ADMINISTRATIVE AGENT, BY ITS EXECUTION OR ACCEPTANCE OF THIS AMENDMENT, REAFFIRMS ITS WAIVER OF THE RIGHT TO TRIAL BY JURY IN ANY

# ACTION, SUIT OR PROCEEDING OR COUNTERCLAIM OF ANY KIND ARISING OUT OF OR RELATED TO ANY OF THE LOAN DOCUMENTS, THE OBLIGATIONS OR THE COLLATERAL.

(d) Expenses. The Loan Parties agree to pay upon demand all reasonable, documented out-of-pocket costs and expenses of Administrative Agent (including, without limitation, all reasonable Attorney Costs) in connection with the preparation, negotiation, execution and delivery of this Amendment.

(e) Third Parties. No rights are intended to be created hereunder for the benefit of any third party donee, creditor, or incidental beneficiary.

(f) <u>Severability</u>. To the extent any provision of this Amendment is prohibited by or invalid under the applicable law of any jurisdiction, such provision shall be ineffective only to the extent of such prohibition or invalidity and only in any such jurisdiction, without prohibiting or invalidating such provision in any other jurisdiction or the remaining provisions of this Amendment in any jurisdiction.

(g) <u>Successors and Assigns</u>. This Amendment shall be binding upon, inure to the benefit of and be enforceable by the respective successors and assigns of the parties hereto.

(h) <u>Construction</u>. The headings of the various sections and subsections of this Amendment have been inserted for convenience only and shall not in any way affect the meaning or construction of any of the provisions hereof.

(i) <u>Counterparts: Effectiveness</u>. This Amendment may be executed in any number of counterparts and by different parties hereto on separate counterparts, each of which when so executed and delivered shall be an original, but all of which shall together constitute one and the same instrument. This Amendment shall become effective as of the Effective Date upon the execution and delivery of a counterpart hereof by the Loan Parties, Administrative Agent and Lenders, and the satisfaction of the conditions set forth in <u>Section 4</u> hereof. Signatures of the parties to this Amendment transmitted by facsimile or via other electronic format shall be deemed to be their original signatures for all purposes.

## [Signatures on following page.]

In witness whereof, this Amendment has been duly executed as of the day and year first above written.

ADMINISTRATIVE AGENT:

# CIT FINANCE LLC, as Administrative Agent

By: /s/ Andres Alev Name: Andres Alev Title: Director

LENDERS

CIT FINANCE LLC, as a lender

By: /s/ Andres Alev

Name: Andres Alev Title: Director

REGIONS BANK, as a lender

By: /s/ Brian Walsh Name: Brian Walsh Title: Director

SUNTRUST BANK, as a lender

By: /s/ Ben Cumming

Name: Ben Cumming Title: Managing Director CITIZENS BANK, N.A, as a lender

By: /s/ Christopher DeLauro Name: Christopher DeLauro Title: Senior Vice President

PEOPLE'S UNITED BANK, as a lender

By: /s/ Henry L. Petrillo Name: Henry L. Petrillo Title: Senior Vice President

FIFTH THIRD BANK as a lender

By: /s/ John McChesney Name: John McChesney Title: AVP

JP MORGAN CHASE BANK, N.A, as a lender

By: <u>/s/ Kristina Harbison</u> Name: Kristina Harbison Title: Authorized Signer

BANKUNITED, N.A, as a lender

By: /s/ Craig Kinade Name: Craig Kinade Title: Senior Vice President

FIST MIDWEST BANK, as a lender

By: /s/ James A. Good Name: James A. Good Title: Senior Vice President

# U.S. BANK NATIONAL ASSOCIATION, as a lender

By: <u>/s/ Cynthia L. Rogers</u> Name: Cynthia L. Rogers Title: Managing Director

HANCOCK WITNEY BANK, as a lender

By: /s/ Megan Brearey Name: Megan Brearey Title: Senior Vice President

#### WEBSTER BANK, NATIONAL ASSOCIATION as a lender

By: /s/ Melissa Souri Name: Melissa Souri Title: Director PM Sponsor and Specialty Finance

BANCALLIANCE INC, as a lender By: Alliance Partners LLC, its Attorney-in Fact

By: /s/ John Gray Name: John Gray

Title: Executive Vice President

WILLMINGTON SAVINGS FUND SOCIETY, as a lender

By: /s/ James A. Gise Name: James A. Gise Title: Senior Vice President

BORROWER:

GUARANTORS:

## CAPSTAR BANK, as a lender

By: /s/ Mark D. Mattson Name: Mark D. Mattson Title: Executive Vice President

## ADAPTHEALTH LLC,

By: /s/ Luke McGee Name: Luke McGee

Title: Chief Executive Officer

ADAPTHEALTH INTERMEDIATE HOLDCO LLC ADAPTHEALTH - MISSOURI LLC AIR CARE HOME RESPIRATORY, LLC AMERICAN ANCILLARIES, INC. AMERICOAST MARYLAND LLC ASSOCIATED HEALTHCARE SYSTEMS, INC. BENNETT MEDICAL SERVICES LLC BRADEN PARTNERS, L.P. CHOICE MEDICAL HEALTH CARE, LLC CLEARVIEW MEDICAL INCORPORATED CP AP SOLUTIONS, LLC CP AP2ME, INC. FAMILY HOME MEDICAL SUPPLY LLC FIRST CHOICE DME LLC FIRST CHOICE HOME MEDICAL EQUIPMENT, LLC GOULD'S DISCOUNT MEDICAL, LLC HALPRIN, INCORPORATED HEALTH SOLUTIONS LLC HOME MEDICAL EXPRESS, INC. HOME MEDISERVICE, LLC HOMETOWN HOME HEALTH MED STAR SURGICAL & BREATHING EQUIPMENT INC. MED WAY MEDICAL, INC. MEDBRIDGE HOME MEDICAL LLC MED-EQUIP, INC. MEDSTAR HOLDINGS LLC OCEAN HOME HEALTH OF PA LLC OCEAN HOME HEALTH SUPPLY LLC OGLES OXYGEN, LLC ORBIT MEDICAL OF PORTLAND, INC. PALMETTO OXYGEN, LLC

PPS HME HOLDINGS LLC PPS HME LLC ROBERTS HOME MEDICAL, LLC ROYAL DME LLC ROYAL MEDICAL SUPPLY INC. SLEEPEASY THERAPEUTICS, INC. SOUND OXYGEN SERVICE LLC TOTAL RESPIRATORY, LLC TRICOUNTY MEDICAL EQUIPMENT AND SUPPLY,LLC VERUS HEALTHCARE, INC. VERUS HEALTHCARE, LLC

By: /s/ Luke McGee

Name: Luke McGee Title: Chief Executive Officer

Annex A

See attached

# THIRD AMENDED AND RESTATED CREDIT AND GUARANTY AGREEMENT

#### Dated as of March 20, 2019

(as amended by Amendment No. 1 to Third Amended and Restated Credit and Guaranty Agreement dated as of August 22, 2019 and Amendment No. 2 to Third Amended and Restated Credit and Guaranty Agreement dated as of November 8, 2019)

by and among

# ADAPTHEALTH LLC,

# as the Borrower,

CIT FINANCE LLC, as Administrative Agent,

and

# THE OTHER LENDERS PARTY HERETO

# REGIONS BANK SUNTRUST BANK as Documentation Agents

CITIZENS BANK, N.A. JPMORGAN CHASE BANK, N.A. as Co-Syndication Agents

Arranged By:

CIT FINANCE LLC REGIONS BANK SUNTRUST ROBINSON HUMPHREY, INC. as Joint Lead Arrangers and Joint Book Runners

# TABLE OF CONTENTS

ARTICLE	1 DEFINITIONS AND ACCOUNTING TERMS	1
1.01	Defined Terms	1
1.02	Other Interpretive Provisions	<b>47</b> 49
1.03	Accounting Terms.	4 <del>8</del> 50
1.04	Rounding	4 <del>9</del> 50
1.05	Times of Day	4 <del>9</del> 51
1.06	Letter of Credit Amounts	4 <del>9</del> 51
1.07	Financial Covenant Defaults	<mark>49</mark> 51
1.08	Divisions	4950 4951 4951 4951 4951
ARTICLE	2 THE COMMITMENTS AND CREDIT EXTENSIONS	
2.01	Loans.	5051 5052 5254 5860 6062 6365 6465 6465 6566 6566 6566 6668 6668
2.02	Borrowings, Conversions and Continuations of Loans.	<del>50</del> 52
2.03	Letters of Credit and Letter of Credit Fees.	<del>52</del> 54
2.04	Swingline Loans.	<del>58</del> 60
2.05	Prepayments.	<del>60</del> 62
2.06	Termination or Reduction of Total Revolving Commitments	<del>63</del> 65
2.07	Repayment of Loans.	<del>6</del> 4 <u>65</u>
2.08	Interest.	<del>65</del> 66
2.09	Fees.	<del>65</del> 67
2.10	Computation of Interest and Fees	<del>66</del> 68
2.11	Evidence of Debt.	<del>66</del> 68
2.12	Payments Generally.	<del>66</del> 69
2.13	Sharing of Payments	<del>68</del> <u>70</u>
2.14	Handling of Proceeds of Collateral; Cash Dominion; Revolving Loan Account.	<del>68</del> <u>70</u>
2.15	Uncommitted Facilities Increase.	<del>70</del> <u>72</u>
2.16	Defaulting Lenders	<del>73</del> 75
2.17	Refinancing Facilities	<del>75</del> 77
2.18	Amend and Extend Transactions.	<del>70</del> 72 <del>73</del> 75 <del>75</del> 77 <del>77</del> 79
ARTICLE	3 TAXES, YIELD PROTECTION AND ILLEGALITY	<del>78</del> 81
3.01	Taxes	<del>78</del> 81
3.02	Illegality	<del>82</del> 85
3.03	Inability to Determine Rate; Alternate Rate of Interest	7881 7881 8285 8285 8285 8387 8488 8488 8589 8589
3.04	Increased Cost and Reduced Return; Capital Adequacy	<mark>83</mark> 87
3.05	Funding Losses	<del>84</del> 88
3.06	Matters Applicable to all Requests for Compensation.	<del>85</del> 89
3.07	Survival	<del>85</del> 89

Page i

	4 CONDITIONS PRECEDENT TO CREDIT EXTENSIONS	<del>85</del> 89
4.01	Conditions of Initial Credit Extension	<del>85</del> 89
4.02	Conditions to all Credit Extensions	<del>88</del> 93
4.03	Satisfaction of Conditions	<del>90</del> 94
ARTICLE	5 REPRESENTATIONS AND WARRANTIES	<del>90</del> 94
5.01	Existence, Qualification and Power	<del>90</del> 94
5.02	Authorization; No Contravention	<del>90</del> 95
5.03	Governmental Authorization; Other Consents	<del>91</del> <u>95</u>
5.04	Binding Effect	<del>91</del> 95
5.05	Financial Statements; No Material Adverse Effect.	9195 9195 9296 9296 9296 9296 9296 9296 9397
5.06	Litigation	<del>92</del> 96
5.07	No Default	<del>92</del> 96
5.08	Ownership of Property; Liens	<del>92</del> 96
5.09	Environmental Compliance	<del>92</del> 96
5.10	Insurance	<del>93</del> 97
5.11	Taxes	<del>93</del> 97 <del>93</del> 97
5.12	ERISA Compliance	<del>93</del> 97
5.13	Subsidiaries	<del>94</del> 98
5.14	Margin Regulations; Investment Company Act, Use of Proceeds	<del>94</del> <u>98</u>
5.15	Disclosure	<del>94</del> 99
5.16	Compliance with Laws	<del>94</del> 99 <del>95</del> 99 <del>95</del> 99
5.17	Intellectual Property; Licenses, Etc.	<del>95</del> 99
5.18	Broker's Fees	
5.19	Labor Matters	<del>95</del> 100
5.20	Business Locations	<del>95</del> 100
5.21	Perfection of Security Interests in the Collateral	<del>96</del> 100
5.22	Solvency	<del>96</del> 100
5.23	[Reserved]	<del>96</del> 100
5.24	Material Contracts	<del>96</del> 100
5.25	Accounts	<del>96</del> 100
5.26	Holding Company Status	<del>96</del> 100
5.27	Inventory Suppliers	<del>96</del> 100
5.28	Patriot Act	<del>96</del> <u>101</u>
5.29	Regulatory Matters	<del>96</del> <u>101</u>
5.30	Compliance of Products	<del>100</del> 104
5.31	OFAC	<del>103</del> 104
ARTICLE	6 AFFIRMATIVE COVENANTS	<del>103</del> 107
6.01	Financial Statements	<del>103</del> 107
6.02	Certificates; Other Information	<del>104</del> 108
6.03	Notices	<del>105</del> 110
		_

Page ii

6.04	Payment of Obligations: Tax Returns.	<del>107</del> 112
6.05	Preservation of Existence, Material Contracts, Etc.	<del>107</del> 112
6.06	Maintenance of Properties	<del>108</del> 112
6.07	Maintenance of Insurance	<del>108</del> 113
6.08	Compliance with Laws	<del>109</del> 113
6.09	Books and Records	<del>109</del> 114
6.10	Inspection Rights	<del>109</del> 114
6.11	Use of Proceeds	<del>110</del> 114
6.12	Additional Subsidiaries	<del>110</del> 115
6.13	ERISA Compliance	<del>111</del> 115
6.14	Further Assurances	++++115
6.15	Covenant with Respect to Environmental Matters	<del>112</del> 117
6.16	Covenants with Respect to Real Property	<del>113</del> 117
6.17	Lenders Meetings	<del>113</del> 118
6.18	Post-Closing Covenants	<del>113</del> 118
6.19	Qualified ECP Guarantors	<del>114</del> 118
6.20	Interest Rate Protection	<del>114</del> 118
6.21	Covenants Regarding Products and Compliance with Required Permits	<del>114</del> 119
6.22	Healthcare Operations	<del>114</del> 119
6.23	Patriot Act; OFAC	<del>115</del> 119
ARTICLE 7	NEGATIVE COVENANTS	<del>115</del> 120
7.01	Indebtedness	<del>115</del> 120
7.02	Liens	<del>117</del> 122
7.03	Investments	<del>118</del> 123
7.04	Fundamental Changes	<del>120</del> 125
7.05	Dispositions	<del>120</del> 125
7.06	Restricted Payments	<del>121</del> 126
7.07	Change in Nature of Business; Cloud-Based Billing	<del>122</del> 127
7.08	Transactions with Affiliates and Insiders	<del>123</del> 127
7.09	Burdensome Agreements	<del>123</del> 128
7.10	Use of Proceeds	<del>123</del> 128
7.11	Amendments to Certain Agreements	<del>123</del> 128
7.12	Organization Documents; Fiscal Year; Legal Name, State of Formation and Form of Entity	$\frac{124}{128}$
7.13	Ownership of Subsidiaries	<del>124</del> 129
7.14	Sale and Leaseback Transactions	<del>124</del> 129
7.15	Limitations on Holdings	<del>124</del> 129
7.16	Account Control Agreements; Bank Accounts	<del>124</del> 129
7.17	Permits	<del>12</del> 4 <u>129</u>
7.18	Covenants Relating to Excluded Subsidiaries	<del>125</del> 129

# Page iii

ARTICLE 8 I	FINANCIAL COVENANTS	<del>125</del> 130
8.01	Financial Covenants	<del>125</del> 130
ARTICLE 9 I	EVENTS OF DEFAULT AND REMEDIES	<del>126</del> 130
9.01	Events of Default	<del>126</del> 130
9.02	Remedies upon Event of Default	<del>128</del> 133
9.03	[Reserved]	<del>129</del> 134
9.04	Application of Funds	<del>129</del> 134
9.05	Loan Parties Right to Cure	<del>130</del> 135
ARTICLE 10	GUARANTY	<del>131</del> 136
10.01	The Guaranty	<del>131</del> 136
10.02	Obligations Unconditional	<del>132</del> 137
10.03	Reinstatement	<del>133</del> 137
10.04	Waivers	<del>133</del> 138
10.05	Remedies	<del>134</del> 138
10.06	Contribution by Guarantors	<del>134</del> 139
10.07	Guarantee of Payment; Continuing Guarantee	<del>135</del> 139
10.08	Subordination of Other Obligations	<del>135</del> 139
ARTICLE 11	THE ADMINISTRATIVE AGENT	<del>135</del> 140
11.01	Appointment and Authorization of Administrative Agent	<del>135</del> 140
11.02	Delegation of Duties	<del>136</del> 140
11.03	Liability of Administrative Agent	<del>136</del> 141
11.04	Reliance by Administrative Agent	<del>137</del> 141
11.05	Notice of Default	<del>137</del> 142
11.06	Credit Decision; Disclosure of Information by Administrative Agent	<del>137</del> 142
11.07	Indemnification of Administrative Agent	<del>138</del> 142
11.08	Administrative Agent in its Individual Capacity	<del>138</del> 143
11.09	Successor Administrative Agent	<del>139</del> 143
11.10	Administrative Agent May File Proofs of Claim	<del>139</del> 144
11.11	Collateral and Guaranty Matters	<del>140</del> 144
11.12	Other Agents; Arrangers and Managers	141 145
11.13	Additional Secured Parties	<u>141</u> 146
11.14	Exclusive Right to Enforce Rights and Remedies	141146
11.15	Flood Laws	<u>142</u> 146
11.16	Banking Services Obligations/Hedging	<u>142</u> 147
11.17	Certain ERISA Matters	<del>143</del> 147

Page iv

ARTICLE 12	MISCELLANEOUS	<del>144</del> 149
12.01	Amendments, Etc.	<del>144</del> 149
12.02	Notices and Other Communications; Facsimile Copies	<del>146</del> 151
12.03	No Waiver; Cumulative Remedies	<del>148</del> 152
12.04	Attorney Costs, Expenses	<del>148</del> 152
12.05	Indemnification by the Loan Parties	<del>149</del> 153
12.06	Payments Set Aside	<del>150</del> 154
12.07	Successors and Assigns	<del>150</del> 154
12.08	Confidentiality	<del>157</del> 161
12.09	Set-off	<del>158</del> 163
12.10	Interest Rate Limitation	<del>159</del> 163
12.11	Counterparts	<del>159</del> 163
12.12	Integration	<del>159</del> 163
12.13	Survival of Representations and Warranties	<del>159</del> 164
12.14	Severability	<del>159</del> 164
12.15	Replacement of Lenders	<del>160</del> 164
12.16	Governing Law	<del>160</del> 165
12.17	Waiver of Right to Trial by Jury	<del>161</del> 165
12.18	USA Patriot Act Notice	<del>161</del> 166
12.19	Nonliability of Lenders	<del>161</del> 166
12.20	Acknowledgment and Consent to Bail-In of EEA Financial Institutions	<del>163</del> 167
12.21	Effect of Amendment and Restatement	<del>163</del> 167
ARTICLE 13 THE BORRO	APPOINTMENT OF THE BORROWER REPRESENTATIVE; JOINT AND SEVERAL LIABILITY OF WERS	<del>164</del> <u>168</u>
13.01	Borrower Representative	<del>164</del> 168
13.02	Joint and Several Liability of Borrowers	<del>164</del> 168

Page v

# SCHEDULES

1.01(a)	Existing Letters of Credit
1.01(b)	Excluded Accounts
1.01(c)	Inventory Suppliers
2.01	Commitments and Pro Rata Shares
5.10	Insurance
5.13	Capitalization
5.17	IP Rights
5.20(a)	Locations of Real Property
5.20(b)	Locations of Tangible Personal Property
5.20(c)	Locations of Chief Executive Office
<del>5.24</del>	Material Contracts
5.25	Accounts
5.29(a)	Regulatory Matters
5.29(b)	Specific Licensing
5.29(c)	Physician Ownership
5.29(f)(iii)	Healthcare Proceedings
5.30(a)	Compliance of Products
5.30(b)	Required Permits
6.18	Post-Closing Covenants
7.01	Indebtedness Existing on the Closing Date
7.02	Liens Existing on the Closing Date
7.03	Investments Existing on the Closing Date
12.02	Certain Addresses for Notices

# EXHIBITS

A-1	Form of Loan Notice
A-2	Form of Swingline Loan Notice
B-1	Form of Revolving Note
B-2	Form of Term Note
B-3	Form of Swingline Note
С	Form of Compliance Certificate
D	Form of Assignment and Assumption Agreement
E	Form of Solvency Certificate
F-1	Form of U.S. Tax Compliance Certificate
F-2	Form of U.S. Tax Compliance Certificate
F-3	Form of U.S. Tax Compliance Certificate
F-4	Form of U.S. Tax Compliance Certificate
G	Form of Dutch Auction Procedure for Borrower Buy-Backs
Z	Form of Affiliated Lender Assignment and Assumption

Page vi

# THIRD AMENDED AND RESTATED CREDIT AND GUARANTY AGREEMENT

This THIRD AMENDED AND RESTATED CREDIT AND GUARANTY AGREEMENT is entered into as of March 20, 2019, among ADAPTHEALTH LLC (formerly known as QMES LLC), a Delaware limited liability company, and such other Persons joined hereto as a Borrower from time to time (each a "Borrower" and together, the "Borrowers"), the Guarantors (as hereinafter defined) from time to time party hereto, the Lenders (as hereinafter defined) from time to time party hereto, and CIT FINANCE LLC ("CIT"), as Administrative Agent.

The Administrative Agent and certain lenders have previously established credit facilities in favor of Borrowers pursuant to that certain Second Amended and Restated Loan and Security Agreement (as amended by that certain Amendment No. 1 to Second Amended and Restated Credit and Guaranty Agreement, Joinder and Limited Consent dated as of May 17, 2018, Amendment No. 2 to Second Amended and Restated Credit and Guaranty Agreement and Limited Consent dated as of June 28, 2018, Amendment No. 3 to Second Amended and Restated Credit and Guaranty Agreement, and Limited Consent dated as of December 20, 2018, and as further amended, restated, supplemented or otherwise modified form time to time, the "Original Loan Agreement"), dated as of February 16, 2018, by and among Borrowers, the Administrative Agent, and certain lenders party thereto.

The Borrowers have requested that the Administrative Agent and Lenders agree to certain modifications and to provide \$425,000,000 in credit facilities for the purposes set forth herein and the Lenders are willing to do so on the terms and conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

# ARTICLE 1

# **DEFINITIONS AND ACCOUNTING TERMS**

**1.01 Defined Terms**. As used in this Agreement, the following terms shall have the meanings set forth below:

"Acceptable Intercreditor Agreement" means a customary intercreditor agreement, subordination agreement, or other intercreditor agreement in form and substance reasonably acceptable to the Administrative Agent.

"<u>Accounts</u>" means all of the Loan Parties' present and future: (a) accounts (as defined in the UCC); (b) instruments, documents, chattel paper (including electronic chattel paper) (all as defined in the UCC); (c) reserves and credit balances arising in connection with or pursuant to this Agreement; (d) guaranties; (e) other supporting obligations, payment intangibles and letter of credit rights (all as defined in the UCC); (f) property, including notes and deposits, of the Loan Parties' account debtors securing the obligations owed by such account debtors to the Loan Parties; and (g) all proceeds of any of the foregoing.

"<u>Acquisition</u>", by any Person, means the acquisition by such Person, in a single transaction or in a series of related transactions, of (a) all or substantially all of the Property of another Person, (b) all or substantially all of a division or operating group of another Person, or (c) all of the Capital Stock of another Person, in each case whether or not involving a merger or consolidation with such other Person and whether for cash, property, services, assumption of Indebtedness, securities or otherwise.

"AdaptHealth Corp." means AdaptHealth Corp., a Delaware corporation.

"Additional Credit Extension Amendment" means an amendment to this Agreement (which may, at the option of the Administrative Agent, be in the form of an amendment and restatement of this Agreement) providing for any Extended Term Loans and/or Extended Revolving Commitments pursuant to Section 2.18, which shall be consistent with the applicable provisions of this Agreement and otherwise satisfactory to the parties thereto. Each Additional Credit Extension Amendment shall be executed by the Administrative Agent, the L/C Issuer, and/or the Swingline Lender (to the extent Section 2.18 would require the consent of the L/C Issuer and/or the Swingline Lender, respectively, for the amendments effected in such Additional Credit Extension Amendment), the applicable Loan Parties and the other parties specified in Section 2.18 (but not any other Lender). Any Additional Credit Extension Amendment may include conditions for delivery of opinions of counsel and other documentation consistent with the conditions in Section 4.01, all to the extent reasonably requested by the Administrative Agent or the other parties to such Additional Credit Extension Amendment.

"Additional Lender" has the meaning specified in Section 2.15(d).

"Administrative Agent" means CIT in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.

"<u>Administrative Agent's Bank Account</u>" means the Administrative Agent's bank account number 432233505, ABA No. 021000021, account name CIT Finance LLC-Healthcare, at JPMorgan Chase Bank NA in New York, New York, Reference: AdaptHealth LLC.

"<u>Administrative Agent's Office</u>" means the Administrative Agent's address and, as appropriate, account as set forth on <u>Schedule 12.02</u> or such other address or account as the Administrative Agent may from time to time notify the Borrowers and the Lenders.

"Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.

"<u>Affiliate</u>" means, with respect to any Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.

"Affiliated Lender" means, collectively, the Sponsor or BlueMountain or any of their respective Affiliates (other than Holdings and its Subsidiaries).

"Affiliated Lender Assignment and Assumption" has the meaning specified in Section 12.07(i)(ii).

"<u>Agent-Related Persons</u>" means the Administrative Agent, together with its Affiliates, and its Approved Funds, and the officers, directors, employees, agents, advisors, auditors and Controlling Persons and attorneys-in-fact of such Persons, Affiliates and Approved Funds, <u>provided</u>, however, that for the purposes of this Agreement, no Agent-Related Person shall be deemed an Affiliate of the Sponsor, the Borrowers or the Guarantors.

"Aggregate Payments" has the meaning set forth in Section 10.06.

"<u>AHYDO Catch-Up Payment</u>'s means any minimum prepayment or redemption necessary at the end of each accrual period (as determined for purposes of Section 163(i) of the Internal Revenue Code) ending after the fifth anniversary of March 20, 2019, pursuant to the terms of the Preferred Note that is

intended or designed to cause the loan under the Preferred Note not to be treated as an "applicable high yield discount obligation" within the meaning of Section 163(i) of the Internal Revenue Code.

"Agreement" means this Third Amended and Restated Credit and Guaranty Agreement, as amended, modified, restated, supplemented or extended from time to time.

"<u>Applicable Margin</u>" means (a) prior to the Second Amendment Effective Date, the "Applicable Margin" as defined in this Agreement prior to the Second Amendment Effective Date, and (b) from and after the Second Amendment Effective Date, with respect to the Loans and the Letters of Credit, the following percentages per annum, based upon the Consolidated Total Leverage Ratio as set forth in the most recent Compliance Certificate received by the Administrative Agent pursuant to <u>Section 6.02(b)</u>:

	Consolidated		Initial Term Loan Revolving Loans		Swingline Loans	Delayed – Term			
	Total	Letters		Base		Base			Base
Pricing	Leverage	of	LIBOR	Rate	LIBOR	Rate		LIBOR	Rate
Level	Ratio	Credit	Loans	Loans	Loans	Loans		Loans	Loans
1	> 3.00 to 1.00	<del>3.50%</del> 3.	<del>3.50%</del> 3.2	<del>2.50%</del> 2.2	<del>3.50%</del> 3.2	<del>2.50%</del> 2.2	<del>2.50%</del> 2.25	<del>3.50%</del> 3.	<del>2.50%</del> 2.
		25%	5%	5%	5%	5%	%	25%	25%
2	> 2.50 to 1.00	<del>3.00%</del> 2.	<del>3.00%</del> 2.7	<del>2.00%</del> 1.7	<del>3.00%</del> 2.7	<del>2.00%</del> 1.7	<del>2.00%</del> 1.75	<del>3.00%</del> 2.	<del>2.00%</del> 1.
ł	ut $\leq$ 3.00 to 1.00	75%	5%	5%	5%	5%	%	75%	75%
3	> 2.00 to 1.00	<del>2.75%</del> 2.	<del>2.75%</del> 2.5	<del>1.75%</del> 1.5	<del>2.75%</del> 2.5	<del>1.75%</del> 1.5	<del>1.75%</del> 1.50	<del>2.75%</del> 2.	<del>1.75%</del> 1.
ł	$tut \le 2.50 to 1.00$	50%	0%	0%	0%	0%	%	50%	50%
4	> 1.75 to 1.00	<del>2.50%</del> 2.	<del>2.50%</del> 2.2	<del>1.50%</del> 1.2	<del>2.50%</del> 2.2	<del>1.50%</del> 1.2	<del>1.50%</del> 1.25	<del>2.50%</del> 2.	<del>1.50%</del> 1.
bu	< to 2.00 to 1.00	25%	5%	5%	5%	5%	%	25%	25%
5	< to 1.75 to 1.00	2.00%	2.00%	1.00%	2.00%	1.00%	1.00%	2.00%	1.00%

Any increase or decrease in the Applicable Margin resulting from a change in the Consolidated Total Leverage Ratio shall become effective as of the first (1st) Business Day immediately following the date a Compliance Certificate is required to be delivered to the Administrative Agent pursuant to Section 6.02(b), provided, that if a Compliance Certificate is not delivered to the Administrative Agent when due in accordance with such Section, then Pricing Level 1 shall apply as of the first (1st) Business Day after the date on which such a Compliance Certificate was required to have been delivered and shall continue to apply until the first (1st) Business Day immediately following the date a Compliance Certificate is delivered to the Administrative Agent in accordance with Section 6.02(b), whereupon the Applicable Margin shall be adjusted based upon the calculation of the Consolidated Total Leverage Ratio contained in such Compliance Certificate. Notwithstanding the foregoing, the Applicable Margin in effect from the Closing Date through the first (1st) Business Day immediately following the date a Compliance Certificate is required to be delivered to the Administrative Agent pursuant to Section 6.02(b), whereupon the Applicable Margin shall be adjusted based upon the calculation of the Consolidated Total Leverage Ratio contained in such Compliance Certificate. Notwithstanding the foregoing, the Applicable Margin in effect from the Closing Date through the first (1st) Business Day immediately following the date a Compliance Certificate is required to be delivered to the Administrative Agent pursuant to Section 6.02(b) for the Fiscal Quarter ending June 30, 2019 shall be set at Pricing Level 2. In the event that (a) any financial statement delivered pursuant to Section 6.01(a) or Section 6.01(b) or Compliance Certificate delivered pursuant to Section

<u>6.02(b)</u> is shown to be inaccurate and (b) such inaccuracy if corrected, would have led to the application of a higher Applicable Margin for any period (an "<u>Applicable Period</u>") than the Applicable Margin applied for such Applicable Period, then (i) the Borrowers shall immediately deliver to the Administrative Agent a corrected Compliance Certificate for such Applicable Period, (ii) the Applicable Margin for such Applicable Period shall be the Applicable Margin corresponding to the information disclosed on the corrected Compliance Certificate, and (iii) the Borrowers shall immediately pay to the Administrative Agent the accrued additional interest owing as a result of the application of such increased Applicable Margin for such Applicable Period, which payment shall be promptly applied by the Administrative Agent. The rights of the Administrative Agent set forth above shall survive the Termination Date and are in addition to rights of the Administrative Agent and Lenders with respect to <u>Sections 2.08</u> and <u>9.02</u> and other of their respective rights under this Agreement.

"<u>Appropriate Lender</u>" means, at any time, (a) with respect to the Revolving Commitments and the Term Loans, a Lender that has a Commitment with respect thereto or holds a Loan thereunder at such time, as applicable (b) with respect to the Letter of Credit Sublimit, the L/C Issuer and (c) with respect to the Swingline Loan Sublimit, the Swingline Lender.

"<u>Approved Fund</u>" means (a) any Person (other than a natural person) engaged in making, purchasing, holding, or investing in commercial loans and similar extensions of credit and that is advised, administered, or managed by a Lender, an Affiliate of a Lender (or an entity or an Affiliate of an entity that administers, advises or manages a Lender); (b) with respect to any Lender that is an investment fund, any other investment fund that invests in loans and that is advised, administered or managed by the same investment advisor as such Lender or by an Affiliate of such investment advisor; and (c) any third party which provides "warehouse financing" to a Person described in the preceding clause (a) or (b) (and any Person described in said clause (a) or (b) shall also be deemed an Approved Fund with respect to such third party providing such warehouse financing).

"Arranger" means CIT Finance LLC, in its capacity as sole lead arranger and book runner.

"Assignment and Assumption" means an Assignment and Assumption Agreement substantially in the form of Exhibit D.

"Attorney Costs" means and includes all documented and out-of-pocket fees, expenses and disbursements of any law firm or other external counsel.

"<u>Attributable Indebtedness</u>" means, on any date, in respect of any Capital Lease of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP.

"<u>Audited Pre-Closing Financial Statements</u>" means the audited consolidated balance sheet of the Loan Parties and their Subsidiaries for the Fiscal Years ended December 31, 2015, December 31, 2016, and December 31, 2017, and the related consolidated statements of income or operations, shareholders' equity and cash flows of the Loan Parties and their Subsidiaries for such Fiscal Year, including the notes thereto.

"<u>Availability Period</u>" means, with respect to the Revolving Commitments, the period from and including the first (lst) Business Day immediately following the Closing Date to the earliest of (a) the date that is five (5) Business Days before the Revolving Loan Maturity Date, (b) the date of termination of the Revolving Commitments pursuant to <u>Section 2.06</u>, and (c) the date of termination of the commitment of each Lender to make Loans pursuant to <u>Section 9.02</u> and of the obligation of the L/C Issuer (or the Support Provider, as the case may be) to make (or cause to make) L/C Credit Extensions pursuant to <u>Section 9.02</u>.

"Bail-In Action" means the exercise of any Write-Down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"<u>Bail-In Legislation</u>" means with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"Banking Services" mean each and any of the following banking services provided to any Loan Party as follows: (a) commercial credit cards, purchasing cards or other similar charge cards provided by any Person that is a Lender or an Affiliate of a Lender at the time provided (or otherwise introduced by, provided by, and/or such obligations guaranteed by, the Administrative Agent or any Lender), (b) stored value cards provided by any Person that is a Lender or an Affiliate of a Lender at the time provided and (c) treasury management services (including, without limitation, controlled disbursement, automated clearinghouse transactions, return items, overdrafts and interstate depository network services) provided by the Administrative Agent or any of the Administrative Agent's Affiliates.

"Banking Services Obligations" mean any and all obligations of the Loan Parties, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefor) in connection with Banking Services.

"Bankruptcy Code" means Title 11 of the United States Code entitled "Bankruptcy," as now and hereafter in effect, or any successor statute.

"Base Rate" means for any day a fluctuating rate per annum equal to the highest of: (a) the Federal Funds Rate plus 1/2 of 1%; (b) the rate of interest in effect for such day as publicly announced from time to time by JPMorgan Chase Bank as its "prime rate" in effect for such day; or (c) the most recently available LIBO Base Rate (as adjusted by any minimum LIBO Rate floor) plus one percent (1%) provided that for the purpose of this definition, the LIBO Base Rate for any day shall be based on the LIBO Screen Rate. Any change in the "prime rate" announced by JPMorgan Chase Bank shall take effect without notice to the Borrowers at the opening of business on the day specified as the effective date of change in the public announcement or publication of such change. The Base Rate is not necessarily the lowest rate of interest charged by Lenders in connection with extensions of credit. If JPMorgan Chase Bank ceases to announce its "prime rate", the Administrative Agent may select a reasonably comparable index or source to use as the basis for the Base Rate. For the avoidance of doubt, if the Base Rate shall be less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

"Base Rate Loan" means a Loan that accrues interest by reference to the Base Rate in accordance with the terms of this Agreement.

"Beneficial Ownership Certification" means a certification regarding beneficial ownership as required by the Beneficial Ownership Regulation.

## "Beneficial Ownership Regulation" means 31 C.F.R. § 1010.230.

"Benefit Plan" means any of (a) an "employment benefit plan" (as defined in ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in Section 4975 of the Internal Revenue Code or (c) any Person whose assets include (for purposes of ERISA Section 3(42) or otherwise for purposes of Title I of ERISA or Section 4975 of the Internal Revenue Code) the assets of any such "employee benefit plan" or "plan".

"BlueMountain" means investment funds managed by BlueMountain Capital Management LLC.

"Borrower Representative" means AdaptHealth LLC, in its capacity as the borrowing agent on behalf of itself and the Borrowers.

"Borrowers" has the meaning specified in the introductory paragraph hereto, together with all permitted successors and assigns of such Person and any other Person joining this Agreement as a "Borrower" pursuant to Section 6.12 hereof or otherwise.

"Borrowing" means (a) a borrowing consisting of simultaneous Loans (other than Swingline Loans) of the same Type and, in the case of LIBOR Loans, having the same Interest Period made by the Lenders pursuant ts<u>Sections 2.01</u> and 2.02 or (b) a borrowing of a Swingline Loan made by Swingline Lender pursuant to Section 2.04.

"Brightree Collateral Assignment Agreement" shall have the meaning set forth in Section 4.01(j) hereof.

"Brightree Services Agreement" shall mean collectively each of (a) that certain Business Solutions Provider Agreement dated as of October 29, 2009, by and between Royal Homestar LLC and Brightree LLC, (b) that certain Brightree LLC QMES Pricing Agreement dated as of December 1, 2014, between Brightree LLC and Borrower Representative, (c) that certain Brightree LLC Business Solutions Provider Agreement dated as of May 4, 2014, between Brightree LLC and Borrower Representative, and (d) that certain Business Solutions Provider Agreement dated as of September 21, 2005, by and between Brightree LLC and Ocean Home Health Supply LLC.

"Business" or "Businesses" means, at any time, a collective reference to the businesses operated by the Borrowers and their Subsidiaries at such time, including the provision of home health equipment and supplies.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, either New York or in the state where the Administrative Agent's Office is located or, with respect to a Letter of Credit, the state where the L/C Issuer's (or the Support Provider's, as the case may be) office is located and, if such day relates to any LIBOR Loan or any Base Rate Loan bearing interest at a rate based on the LIBO Rate, means any such day meeting the above requirements on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Capital Expenditures" means, with respect to any Person, all expenditures which, in accordance with GAAP, would be required to be capitalized and shown on the balance sheet of such Person, including expenditures in respect of Capital Leases.

"<u>Capital Lease</u>" means, with respect to any Person, any lease of any property (whether real, personal or mixed) by such Person as lessee which would, in accordance with GAAP, be required to be accounted for as a capital lease on the balance sheet of such person.

"<u>Capital Stock</u>" means, with respect to any Person, all of the shares of capital stock of (or other ownership or profit interests in) such Person, all of the warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such person, all of the securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or such other interests), and all of the other ownership or profit interest in

such Person (including partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such shares, warrants, options, rights or other interests are outstanding on any date of determination.

"<u>Cash Collateralize</u>" means to pledge and deposit with or deliver to the Administrative Agent (or with and to a bank designated by the Administrative Agent to be held in a deposit account subject to a control agreement), for the benefit of the Administrative Agent (on behalf of itself, the Support Providers, L/C Issuers and the other Secured Parties), as collateral for the total Letter of Credit Liabilities or other contingent Obligations, cash or deposit account balances pursuant to documentation in form and substance satisfactory to the Administrative Agent, the Support Provider and the L/C Issuer, if the L/C Issuer is a Lender (which documents are hereby consented to by the Lenders). Derivatives of the term Cash Collateralize have corresponding meanings. "Cash Collaterali" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral.

"<u>Cash Control Period</u>" means the period of time commencing upon the occurrence of an Event of Default and the delivery by the Administrative Agent of a notice of control to the applicable financial institution and ending on the earlier of (a) the written waiver of such Event of Default by the Required Lenders, or (b) the Termination Date.

"Cash Equivalents" means, as of any date of determination, (a) securities issued or directly and fully guaranteed or insured by the United States or any agency or instrumentality thereof (provided that the full faith and credit of the United States is pledged in support thereof) having maturities of not more than twelve (12) months from the date of acquisition, (b) Dollar denominated time deposits and certificates of deposit of (i) any Lender, (ii) any domestic commercial bank of recognized standing having capital and surplus in excess of \$500,000,000 or (iii) any bank whose short term commercial paper rating from S&P is at least A1 or the equivalent thereof or from Moody's is at least P1 or the equivalent thereof (any such bank being an "Approved Bank"), in each case with maturities of not more than 270 days from the date of acquisition, (c) commercial paper and variable or fixed rate notes issued by any Approved Bank (or by the parent company thereof) or any variable rate notes issued by, or guaranteed by, any domestic corporation rated A1 (or the equivalent thereof) or better by S&P or P1 (or the equivalent thereof) or better by Moody's and maturing within six (6) months of the date of acquisition, (d) repurchase agreements entered into by any Person with a bank or trust company (including any of the Lenders) or recognized securities dealer having capital and surplus in excess of \$500,000,000 for direct obligations issued by or fully guaranteed by the United States in which such Person shall have a perfected first priority security interest (subject to no other Liens) and having, on the date of purchase thereof, a fair market value of at least one hundred percent (100%) of the amount of the repurchase obligations and (e) Investments, classified in accordance with GAAP as current assets, in money market mutual funds (as defined in Rule 2(a).7 of the Investment Company Act) registered under the Investment Company Act of 1940, as amended, which are administered by reputable financial institutions having capital of at least \$500,000,000 and the portfolios of which are limited to Investments of the character described in the foregoing clauses (a) through (d).

"CFC" means a "controlled foreign corporation" as such term is defined in Section 957 of the Internal Revenue Code.

"<u>Change in Law</u>" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority. For purposes of this Agreement, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, guidelines and directives in connection therewith and (ii) all requests, rules, guidelines or directives promulgated by the Bank for International

Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States regulatory authorities, in each case pursuant to Basel III, shall, in each case, be deemed to have been adopted and gone into effect after the date of this Agreement.

"Change of Control" means, at any time, (ia) the Sponsor and its Affiliates shall eease to beneficially ownany "person" or "group" (as such terms are used in Sections 13(d) and 14( $\overline{d}$ ) of the Exchange Act, but excluding any employee benefit plan of such person or its subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Securities Exchange Act of 1934, except that a person or group shall be deemed to have "beneficial ownership" of all Capital Stock that such person or group has the right to acquire (such right, an "option right"), whether such right is exercisable immediately or only after the passage of time), directly or indirectly, at least eighteen percent (18% of 35% (other than the Non-Blocker AdaptHealth Members) or more of the Capital Stock of Holdings of AdaptHealth Corp. entitled to vote for members of the board of directors or equivalent governing body of AdaptHealth Corp. on a fully-diluted basis, (ii) Management Entity shall cease to beneficially own directly or indirectly, at least 41,252 of the Common Units (as defined in the Fourth Amended and Restated Limited Liability Company Agreement of Holdings) of Holdings, (iii) the Sponsor and its Affiliates (excluding Affiliates that are portfolio companies) and Management Entity (and taking into account all such Capital Stock that such person or group has the right to acquire pursuant to any option right); (b) during any period of twelve (12) consecutive months, a majority of the members of the board of directors or other equivalent governing body of AdaptHealth Corp. cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first day of such period, (ii) whose selection or nomination to that board or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (iii) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body; (c) AdaptHealth Corp. shall cease to beneficially own directly or indirectly, at least thirty-three percent (33%) of the Capital Stock of Holdings on a fully diluted basis; (iv) any Person or "group" (within the meanings of Rules 13d-3 and 13d-5 under the Exchange Act), other than Sponsor, its Affiliates (excluding Affiliates that are portfolio companies), Management Entity, or BlueMountain shall have acquired beneficial ownership of twenty percent (20%) or more on a fully diluted basis of the voting and/or economic interest in the Capital Stock of Holdings; (v) the Sponsor and its Affiliates (excluding Affiliates that are portfolio companies) and Management Entity shall cease to have the power to appoint, remove or replace at least three (3) of the seats on the board of directors (or similar governing body) of Holdings; provided, the number of seats on the board of directors (or similar governing body) of Holdings shall not exceed nine (9); (vi) any Person or "group" (within the meanings of Rules 13d-3 and 13d-5 under the Exchange Act), other than Sponsor, its Affiliates (excluding Affiliates that are portfolio companies) or Management Entity shall have acquired beneficial ownership of and control more than fifty percent (5050.0%) on a fully diluted basis, of the voting and/or economic interestand voting interests in the Capital Stock of Holdings; (viid) Holdings shall cease to own and control one hundred percent (100%), on a fully diluted basis, of the economic and voting interests in the Capital Stock of Intermediate Holdings; or (viiic) Intermediate Holdings shall cease to own and control one hundred percent (100%), on a fully diluted basis, of the economic and voting interests in the Capital Stock of AdaptHealth LLC.For the avoidance of doubt, any transfer of Capital Stock of Holdings to a "Permitted Transferee" as defined in the Amended and Restated Limited Liability Company Agreement of Holdings as in effect on the date hereof, shall be ignored for purposes of determining if any Change of Control has occurred.

"CIT" has the meaning specified in the introductory paragraph hereto.

"Closing Date" means March 20, 2019.

"Closing Date Distributions" has the meaning specified in Section 6.11(a)(i)(z).

"Closing Date Facilities" has the meaning specified in Section 2.15(c).

"<u>Collateral</u>" means, collectively, all real and personal Property (other than Excluded Property) with respect to which Liens in favor of the Administrative Agent are granted (or were intended to be granted) pursuant to and in accordance with the terms of the Collateral Documents.

"<u>Collateral Assignment of Agreement</u>" means any agreement executed by each of the applicable Loan Parties in favor of the Administrative Agent under and pursuant to which such Loan Parties collaterally assign to the Administrative Agent all of their respective rights and remedies to certain agreements, in each case as amended, supplemented, modified, replaced, substituted for or restated from time to time and all exhibits and schedules attached thereto.

"<u>Collateral Documents</u>" means, collectively, the Security Agreement, the Mortgage Instruments, the <u>Landlord Agreements</u>, the Collateral Assignment(s) of Agreement, the Securities Account Control Agreement(s), the Deposit Account Control Agreement(s), the Uncertificated Securities Control Agreement(s) and such other security documents as may be executed and delivered by the Loan Parties pursuant to the terms of <u>Section 6.14</u>.

"<u>Commitment</u>" means, as to each Lender, the Revolving Commitment and/or Term Loan Commitment, as applicable, set forth opposite such Lender's name on <u>Schedule 2.01</u> or in the Register, as applicable, as the same may be reduced or modified at any time and from time to time pursuant to the terms hereof.

"<u>Commodity Exchange Acf</u>" means the Commodity Exchange Act (7 U.S.C. § 1 et seq.) and any successor statute, and any rule, regulation, or order promulgated thereunder, in each case as amended from time to time.

"<u>Competitor</u>" means (i) any Person specifically identified in writing to the Administrative Agent as of the Closing Date and (ii) any other bona fide competitor of Borrower (other than banks, fixed income investors, investment funds or other non-bank lending entities) identified in writing by Borrower Representative after the Closing Date to the Administrative Agent and agreed to by the Lenders at their sole option.

"Compliance Certificate" means a certificate substantially in the form of Exhibit C.

"<u>Connection Income Taxes</u>" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"<u>Consolidated Capital Expenditures</u>" means, for any period, for the Consolidated Group on a consolidated basis, all Capital Expenditures, as determined in accordance with GAAP, provided, that Consolidated Capital Expenditures shall not include (a) expenditures made with proceeds of any Involuntary Disposition to the extent such expenditures are used to purchase Property that is the same as or similar to the Property subject to such Involuntary Disposition or (b) Permitted Acquisitions.

"<u>Consolidated Current Assets</u>" means, as of any date of determination, for the Consolidated Group on a consolidated basis, the total assets that may properly be classified as current assets in conformity with GAAP, excluding cash and Cash Equivalents.

"<u>Consolidated Current Liabilities</u>" means, as of any date of determination, for the Consolidated Group on a consolidated basis, all amounts that would, in conformity with GAAP, be set forth opposite the caption "total current liabilities" (or any like caption) on a balance sheet of the Consolidated Group (or any member thereof), but excluding (i) the current portion of any Funded Indebtedness and (ii) without duplication of clause (i) above, all Loans then outstanding hereunder.

"Consolidated EBITDA' means, for any period for the Consolidated Group on a consolidated basis (without duplication), an amount equal to Consolidated Net Income for such periodplus the following, without duplication, to the extent deducted and not already added back in calculating such Consolidated Net Income: (a) Consolidated Interest Charges for such period, (b) the provision for federal, state, local and foreign income taxes payable by the Consolidated Group for such period, (c) the amount of depreciation and amortization expense for such period, (d) other non-cash charges, expenses or losses (provided, in each case, that if any non-cash charges represent an accrual or reserve for potential cash items in any future period, the cash payment in respect thereof in such future period shall be subtracted from Consolidated EBITDA to such extent), and including amortization of any prepaid cash item that was paid in a prior period, (e) fees, costs and expenses related to or incurred in connection with (i) the Transactions in an amount \$11,566,970.18 and <del>(iithe</del> equal to Second Amendment Transactions in an amount equal to \$18,000,000.00, (ii) preparations for and implementation of compliance with the requirements of the Sarbanes-Oxley Act of 2002 and other Public Company Costs, and (ii) following the Closing Date, any consummated Permitted Acquisition (provided, the amount added back with respect to any individual Permitted Acquisition shall not exceed \$1,500,000(or \$10,000,000 in respect of a single Permitted Acquisition that requires Required Lender consent in accordance with clause (x) of the definition of "Permitted Acquisition"), and the aggregate amount added back pursuant to this clause (de)(iiii) for any four (4) Fiscal Quarter period shall not exceed \$7,500,000 (excluding a single Permitted Acquisition that requires Required Lender consent in accordance with clause (x) of the definition of "Permitted Acquisition")), (f) extraordinary, unusual, and/or non-recurring losses or charges (provided, the aggregate amount added back pursuant to this clause (f) for any four (4) Fiscal Quarter period, when taken together with the aggregate amount added back pursuant to clauses (g) and (h) immediately following, shall not exceed twenty percent (2025%) of Consolidated EBITDA (calculated without giving effect to the add backs permitted pursuant to this clause (f) and clauses (g) and (h) immediately following)), (g) non-recurring cash expenses during such period resulting from restructuring charges, accruals, reserves and business optimization expenses (provided, the aggregate amount added back pursuant to this clause (g) for any four (4) Fiscal Quarter period, when taken together with the aggregate amount added back pursuant to clauses (f) and (h) of this definition, shall not exceed twenty percent (2025%) of Consolidated EBITDA (calculated without giving effect to the add backs permitted pursuant to this clause (g) and clauses (f) and (h) immediately following)), and (h) the amount of "run rate" net cost savings and operating expense reductions, other operating improvements and synergies (calculated on a Pro Forma Basis as though such items had been realized on the first day of such period) as a result of (i) actions taken in connection with any Permitted Acquisition, Investment, Disposition, restructuring or cost savings initiative or (ii) actions which will be taken within twelve (12) months after the date of such Permitted Acquisition, Investment, Disposition, restructuring or cost savings initiative, in each case (A) that are projected by the Borrowers in good faith to be realized as a result of such actions taken or to be taken and (B) net of the amount of actual benefits realized during such period that are otherwise included in the calculation of Consolidated EBITDA from such actions provided, the aggregate amount added back pursuant to this clause (h) for any four (4) Fiscal Quarter period, when taken together with the aggregate amount added back pursuant to clauses (f) and (g) immediately preceding, shall not exceed 2025% of Consolidated EBITDA (calculated without giving effect to the add backs permitted pursuant to this clause (h) and clauses (f) and (g) immediately preceding)) and minus the following to the extent included in calculating such Consolidated Net Income: (w) Consolidated Interest Income, (x) income tax credits (to the extent not netted from income taxes payable), (y) any extraordinary, unusual or non-recurring income receipts or gains (including gains on the sale of assets outside the ordinary course of business) and related tax effects thereon, and (z) other non-cash

income, receipts of gains (excluding any such non-cash item of income to the extent it represents a receipt of cash in any future period), all as determined in accordance with GAAP. Notwithstanding anything herein to the contrary, for purposes of determining Consolidated EBITDA under this Agreement for any period that includes any of the Fiscal Quarters ended March 31, 2018, June 30 2018, September 30, 2018 and December 31, 2018, Consolidated EBITDA for such Fiscal Quarters shall be \$29,408,000 \$28,879,000, \$30,939,000, and \$31,203,000, respectively, in each case, subject to any adjustment set forth above with respect to the Transactions.

"<u>Consolidated Fixed Charges</u>" means, for any period for the Consolidated Group on a consolidated basis, an amount equal to the sum of, without duplication, (a) the cash portion of Consolidated Interest Charges for such period, plus (b) Consolidated Scheduled Funded Debt Payments (excluding any payments in respect of Capital Leases for such period) for such period, plus (c) Restricted Payments paid in cash during such period (including Restricted Payments paid to (i) any holders (other than Loan Parties) of Capital Stock of any non-Wholly Owned Subsidiary and (ii) to BlueMountain (including any of BMSB L.P., BlueMountain Summit Opportunities Fund II (US) L.P., BlueMountain Fursan Fund L.P., and/or BlueMountain Foinaven Master Fund L.P.) to be applied to cash interest payments due and owing under the Preferred Note), all as determined in accordance with GAAP.

"<u>Consolidated Fixed Charges Coverage Rati</u>" means, as of any date of determination, the ratio of (a) Consolidated EBITDA for the period of the four (4) Fiscal Quarters most recently ended less the sum of (i) eighty-five percent (85%) of Capital Expenditures made in such period and (ii) taxes, along with any Tax Distributions, paid in cash for such period, to (b) Consolidated Fixed Charges for such period.

"Consolidated Funded Indebtedness" means Funded Indebtedness of the Consolidated Group on a consolidated basis determined in accordance with GAAP.

"Consolidated Group" means Intermediate Holdings and its Subsidiaries (other than any Excluded Foreign Subsidiary or Excluded Subsidiary).

"<u>Consolidated Interest Charges</u>" means, for any period, the interest expense (including any rent expense for such period under Capital Leases that is treated as interest in accordance with GAAP) of the Consolidated Group for such period with respect to all outstanding Indebtedness of the Consolidated Group (including all commissions, discounts and other fees and charges owed with respect to letters of credit and bankers' acceptance financing and net costs under Hedge Agreements in respect of interest rates to the extent such net costs are allocable to such period in accordance with GAAP), determined on a consolidated basis in accordance with GAAP.

"Consolidated Interest Income" means, for any period, the interest income of the Consolidated Group for such period, determined on a consolidated basis in accordance with GAAP.

"<u>Consolidated Net Income</u>" means, for any period for the Consolidated Group on a consolidated basis, the net income of the Consolidated Group for such period as determined in accordance with GAAP, provided that there shall be excluded from Consolidated Net Income (a) the income (or deficit) of any Person (other than a Subsidiary of a Borrower) in which a Borrower or any of its Subsidiaries has an ownership interest, except to the extent that any such income is actually received by such Borrower or such Subsidiary in the form of cash dividends or similar cash distributions, (b) the undistributed earnings of any Subsidiary of the Borrower Representative to the extent that the declaration of payment or dividends or similar distributions by such Subsidiary is not at the time permitted by the terms of any Contractual Obligation, governing document or Law applicable to such Subsidiary and (c) the income (or deficit) of any Subsidiary of Intermediate Holdings that is not a Loan Party.

"<u>Consolidated Scheduled Funded Debt Payments</u>" means, for any period for the Consolidated Group on a consolidated basis, the sum of all scheduled payments of principal on Consolidated Funded Indebtedness scheduled to be paid during such period, as determined in accordance with GAAP. For purposes of this definition, payments of principal scheduled to be paid (a) shall be determined without giving effect to any reduction of such scheduled payments resulting from the application of any voluntary or mandatory prepayments made during the applicable period, (b) shall be deemed to include the Attributable Indebtedness in respect of Capital Leases, Synthetic Leases and Sale and Leaseback Transactions, (c) shall not include any voluntary or mandatory prepayments made pursuant to <u>Section 2.05</u>, and (d) shall be determined without giving effect to the Following Business Day Convention.

"<u>Consolidated Total Leverage Ratio</u>" means, as of any date of determination, the ratio of (a) Consolidated Funded Indebtedness as of such date less the amount of unrestricted cash and Cash Equivalents on such date (the aggregate amount of which shall not exceed \$25,000,000,000) to (b) Consolidated EBITDA for the period of the four (4) Fiscal Quarters most recently ended.

"<u>Consolidated Unfinanced Capital Expenditure</u>" means any Consolidated Capital Expenditure to the extent not financed with Funded Indebtedness within ninety (90) days of the incurrence of the Capital Expenditure (excluding any Funded Indebtedness included as part of the determination of Revolving Exposure).

"Consolidated Working Capital" means, as at the date of determination, the excess or deficiency of Consolidated Current Assets over Consolidated Current Liabilities.

"<u>Contractual Obligation</u>" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its Property is bound.

"Contributing Guarantors" has the meaning set forth in Section 10.06.

"<u>Control</u>" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "<u>Controlling</u>" and "<u>Controlled</u>" have meanings correlative thereto. Without limiting the generality of the foregoing, a Person shall be deemed to be Controlled by another Person if such other Person possesses, directly or indirectly, power to vote 15% or more of the securities having ordinary voting power for the election of directors, managing general partners or the equivalent.

"Controlled Account" means each bank account that is established by the Borrowers pursuant to Section 2.14(c) of this Agreement.

"<u>Correction</u>" means repair, modification, adjustment, relabeling, destruction or inspection (including patient monitoring) of a product without its physical removal to some other location; or any plan in response to a state or federal notice of violation or deficiency, such as, without limitation, FDA 483 inspection reports, FDA warning letters, and any plans to implement, monitor and audit ongoing compliance with plans of correction.

"Credit Extension" means each of the following: (a) a Borrowing and (b) an L/C Credit Extension.

"Cure Amount" has the meaning set forth in Section 9.05.

"Cure Date" has the meaning set forth in Section 9.05.

"Cure Right" has the meaning set forth in Section 9.05.

"Debt Issuance" means the issuance of any Indebtedness for borrowed money by any Loan Party other than Indebtedness permitted under Section 7.03.

"Debtor Relief Laws" means the Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would be an Event of Default.

"Default Excess" means, with respect to any Defaulting Lender, the excess, if any, of such Defaulting Lender's Pro Rata Share of the aggregate outstanding principal amount of all Revolving Loans (calculated as if all Defaulting Lenders (other than such Defaulting Lender) had funded their respective Pro Rata Shares of all Revolving Loans) over the aggregate outstanding principal amount of all Revolving Loans of such Defaulting Lender.

"Default Rate" means (a) when used with respect to Obligations other than Letter of Credit Fees and LIBOR Loans, an interest rate equal to (i) the Base Rate <u>plus</u> (ii) the Applicable Margin applicable to Base Rate Loans<u>plus</u> (iii) two percent (2%) per annum; (b) when used with respect to a LIBOR Loan, an interest rate equal to (i) the LIBO Rate applicable to such LIBOR Lo<u>phus</u> (ii) the Applicable Margin applicable to LIBOR Loans<u>plus</u> (iii) two percent (2%) per annum; and (c) when used with respect to Letter of Credit Fees, a rate equal to (i) the Applicable Margin then applicable to Letters of Credit<u>plus</u> (ii) two percent (2%) per annum, in all cases to the fullest extent permitted by applicable Laws. Interest accruing at the Default Rate shall be immediately payable upon demand.

"Defaulting Lender" means any Lender that has at any time after the Closing Date (a) defaulted in its obligation under this Agreement to make a Revolving Loan, a Delayed – Draw Term Loan or to fund its participation in any Letter of Credit, Support Agreement or Swingline Loan required to be made or funded by it hereunder within three (3) Business Days of the date when due (unless such failure is the subject of a good faith dispute), (b) failed to pay over to the Administrative Agent or any Lender any other amount required to be paid by it hereunder within three (3) Business Days of the date when due (unless such failure is the subject of a good faith dispute), (c) notified the Administrative Agent or a Loan Party in writing that it does not intend to satisfy any such obligation or has made a public statement to the effect that it does not intend to comply with its funding obligations under this Agreement or under agreements in which it commits to extend credit generally, (d) failed within three (3) Business Days after the request of the Administrative Agent to confirm that it will comply with the terms of this Agreement relating to its obligations to fund prospective Revolving Loans, Delayed – Draw Term Loans and participations in then outstanding Letters of Credit, Support Agreements and Swingline Loans, (e) (i) been (or has a parent company that has been) determined by any Governmental Authority having regulatory authority over such Person or its assets to be insolvent, or the assets or management of which has been taken over by any Governmental Authority, or (ii) become (or has a parent company that has become) the subject of a bankruptcy or insolvency proceeding under any Debtor Relief Laws, unless in the case of any Lender subject to this clause (e), the Borrowers, Administrative Agent, L/C Issuer, Support Provider and Swingline Lender shall each have determined that such Lender intends, and has all approvals required to enable it, to continue to perform its obligations as a Lender hereunder, or (f) become

"Delayed - Draw Term Loan" has the meaning specified in Section 2.01(c).

"<u>Delayed – Draw Term Loan Availability Period</u>" means, with respect to the Delayed – Draw Term Loan Commitments, the period from and including the first (1st) Business Day immediately following the Closing Date to the earliest of (a) the Term Loan Maturity Date, (b) twenty-four (24) months following the Closing Date, and (c) the date of termination of the commitment of each Lender to make Loans pursuant to <u>Section 9.02</u>.

"<u>Delayed – Draw Term Loan Commitment</u>" means, as to each Lender, its obligation to make its portion of the Delayed – Draw Term Loan to the Borrowers pursuant to <u>Section 2.01(c)</u> and the other terms and conditions of this Agreement, in the principal amount set forth opposite such Lender's name on <u>Schedule 2.01</u>, as such amounts may be adjusted from time to time in accordance with this Agreement. The initial aggregate amount of the Delayed – Draw Term Loan Commitments is \$50,000,000100,000,000.

"Deposit Account Control Agreement?" means an agreement among a Loan Party, a depository institution, and the Administrative Agent, which agreement is in a form reasonably acceptable to the Administrative Agent and which provides the Administrative Agent with "control" (as such term is used in Article 9 of the UCC) over the deposit account(s) described therein, as the same may be amended, modified, extended, restated, replaced, or supplemented from time to time, and contains such other terms and conditions as Administrative Agent may reasonably require, including a requirement that such depository institution shall wire, or otherwise transfer, in immediately available funds, on a daily basis during a Cash Control Period, to Administrative Agent's Bank Account all immediately available funds received or deposited into such deposit account.

"Device Application" means a 510(k) premarket notification or premarket approval (PMA) application, as appropriate, as those terms are defined in the FDCA.

"Disposition" means the sale, transfer, license, lease or other disposition (including any Sale and Leaseback Transaction) of any Property by any Loan Party or any Subsidiary (including the Capital Stock of any Subsidiary), including any Sale and Leaseback Transaction and including any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith, provided that the term Disposition shall not include Equity Issuances of Capital Stock by Intermediate Holdings.

"Disqualified Capital Stock" means any Capital Stock which, by its terms (or by the terms of any security or any other Capital Stock into which it is convertible or for which it is exchangeable) or upon the happening of any event or condition, (i) matures or becomes mandatorily redeemable (other than solely for Capital Stock that is not Disqualified Capital Stock) pursuant to a sinking fund obligation or otherwise, except as a result of a customarily defined change of control or asset sale and only so long as any rights of the holders thereof after such change of control or asset sale shall be subject to the prior repayment in full of the Obligations (other than contingent indemnification obligations to the extent no claim giving rise thereto has been asserted) and the termination of all Commitments and Letters of Credit, (ii) becomes redeemable at the option of the holder thereof (other than solely for Capital Stock that is not Disqualified Capital Stock), in whole or in part, (iii) provides for scheduled payments of dividends in cash or (iv) becomes convertible into or exchangeable for indebtedness for borrowed money or any other Disqualified Capital Stock, in whole or in part, in each case on or prior to the date that is ninety-one (91) calendar days after the later of the Revolving Loan Maturity Date and the Term Loan Maturity Date at the time of issuance.

"Disqualified Institution" means, on any date, (a) any Person designated by the Borrower Representative as a "Disqualified Institution" by written notice delivered to the Administrative Agent on or prior to the date hereof and (b) any other Person that is a Competitor of the Borrowers or any of its Subsidiaries, which Person has been designated by the Borrowers as a "Disqualified Institution" by written notice to the Administrative Agent and the Lenders (including by posting such notice on or through the

E-System) not less than ten (10) Business Days prior to such date;<u>provided</u> that "Disqualified Institutions" shall (i) exclude any Person that the Borrowers (or any of them) have (or has) designated as no longer being a "Disqualified Institution" by written notice delivered to the Administrative Agent from time to time and shall exclude any Lender party hereto prior to such tenth (10h) Business Day (and such Lender's Approved Funds and Affiliates shall also be so excluded) and (ii) not apply at any time upon the occurrence and during the continuance of an Event of Default.

"Dollar" and "\$" mean lawful money of the United States.

"Domestic Subsidiary" means any Subsidiary that is organized under the laws of any political subdivision of the United States or any state or commonwealth thereof or under the laws of the District of Columbia.

"E-System" has the meaning specified in Section 12.04.

"Earn-Out Obligations" means, with respect to any Person, "earn-outs" and similar payment obligations of such Person, excluding, for the avoidance of doubt, purchase price holdbacks (whether or not in escrow) to secure seller indemnities.

"<u>EEA Financial Institution</u>" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in any EEA Member Country which is a subsidiary of an institution described in clause (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any member state of the European Union, Iceland, Liechtenstein and Norway.

"EEA Resolution Authority" means any public administrative authority or any Person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"<u>Eligible Assignee</u>" means any Person that meets the requirements to be an assignee under Section 12.08(b)(iii) and (v) (subject to such consents, if any, as may be required under Section 12.08(b)(iii)).

"<u>Environmental Laws</u>" means the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. § 9601<u>et seq</u>.), the Hazardous Materials Transportation Act (49 U.S.C. § 5101<u>et seq</u>.), the Resource Conservation and Recovery Act (42 U.S.C. § 6901, <u>et seq</u>.), the Federal Clean Water Act (33 U.S.C. § 1251<u>et seq</u>.), the Clean Air Act (42 U.S.C. § 7401<u>et seq</u>.), the Toxic Substances Control Act (15 U.S.C. § 2601<u>et seq</u>.), the Safe Drinking Water Act (42 U.S.C. § 300f to 300j-26<u>et seq</u>.), the Oil Pollution Act of 1990 (33 U.S.C. § 2701<u>et seq</u>.) and the Occupational Safety and Health Act (29 U.S.C. § 651<u>et seq</u>.), as such laws may be amended or otherwise modified from time to time, and any other federal, state, local, foreign and other applicable statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions and common law relating to pollution, the protection of the environment, natural resources, human health or the release of any materials into the environment, including those related to Hazardous Materials, hazardous substances or wastes, indoor and outdoor air emissions, soil, groundwater, wastewater, surface water, stormwater, wetlands, sediment and discharges of wastewater to public treatment systems.

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, losses, punitive damages, consequential damages, costs of environmental investigation and remediation, fines, penalties, indemnities or expenses (including all reasonable fees, disbursements and expenses of counsel, experts and consultants)), of the Borrowers or any Subsidiary directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the release or threatened release of any Hazardous Materials into the environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

## "Equity Issuance" has the meaning set forth in Section 2.05(b)(iii).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder, and any successor thereto.

"ERISA Affiliate" means any corporation, trade or business (whether or not incorporated) under common control with a Borrower within the meaning of Section 414(b) or (c) of the Internal Revenue Code (and Sections 414(m) and (o) of the Internal Revenue Code for purposes of provisions relating to Section 412 of the Internal Revenue Code). Any former ERISA Affiliate of a Borrower or any of its Subsidiaries shall continue to be considered an ERISA Affiliate of a Borrower or any of its Subsidiaries within the meaning of this definition with respect to the period such entity was an ERISA Affiliate of a Borrower or any of its Subsidiaries and with respect to liabilities arising after such period for which a Borrower or any of its Subsidiaries could be liable under the Internal Revenue Code or ERISA.

"ERISA Event" means (a) a Reportable Event with respect to a Pension Plan; (b) the failure to meet the minimum funding standard of Section 412 of the Internal Revenue Code with respect to a Pension Plan (whether or not waived in accordance with Section 412(d) of the Internal Revenue Code) or the failure to make a required installment under Section 412(m) of the Internal Revenue Code with respect to a Pension Plan; (c) a withdrawal by a Borrower or any ERISA Affiliate from a Pension Plan subject to Section 4063 of ERISA during a plan year in which it was a substantial employer (as defined in Section 4001(a)(2) of ERISA) or a cessation of operations that is treated as such a withdrawal under Section 4062(e) of ERISA; (d) a complete or partial withdrawal by a Borrower or any ERISA Affiliate from a Multiemployer Plan or notification that a Multiemployer Plan is in reorganization or insolvency pursuant to Section 4241 or 4245 of ERISA; (e) the filing of a notice of intent to terminate, the treatment of a Plan amendment as a termination under Sections 4041 or 4041A of ERISA, or the commencement of proceedings by the PBGC to terminate a Pension Plan or Multiemployer Plan; (f) an event or condition which constitutes grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan or Multiemployer Plan; (g) the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, upon a Borrower or any ERISA Affiliate; (h) the occurrence of an act or omission which could give rise to the imposition on a Borrower or any ERISA Affiliate of fines, penalties, taxes or related charges under Chapter 43 of the Internal Revenue Code or under Section 409, Section 502(c), (i) or (l), or Section 4071 of ERISA in respect of any Plan; (i) the assertion of a material claim (other than routine claims for benefits) against any Plan other than a Multiemployer Plan or the assets thereof, or against the Borrowers or any ERISA Affiliate in connection with any Plan; (j) receipt from the IRS of notice of the failure of any Pension Plan (or any other Plan intended to be qualified under Section 401(a) of the Internal Revenue Code) to qualify under Section 401(a) of the Internal Revenue Code, or the failure of any trust forming part of any Pension Plan to qualify for exemption from taxation under Section 501(a) of the Internal Revenue Code; (k) the imposition of a Lien pursuant to Section 401(a)(29) or 412(n) of the Internal Revenue Code or pursuant to ERISA with respect to any Pension Plan; (1) the commencement of any administrative investigation, audit or other administrative proceeding by the Department of Labor, IRS or other Governmental Authority, including any voluntary

compliance submission through the IRS's Employee Plans Compliance Resolution System or the Department of Labor's Voluntary Fiduciary Correction Program; (m) the occurrence of a non-exempt "prohibited transaction" within the meaning of Section 406 of ERISA or Section 4975 of the Internal Revenue Code; or (n) the receipt by a Borrower or any of its ERISA Affiliates of notice from any Multiemployer Plan that it is in reorganization or insolvency or that it intends to terminate or has terminated.

"ERISA Plan" means (i) any "employee benefit plan" as defined in Section 3(3) of ERISA which is or was sponsored, maintained or contributed to by, or required to be contributed by the Borrowers, any of their Subsidiaries or any of their respective ERISA Affiliates; and (ii) all other employee benefit plans, programs, policies, agreements or arrangements, including any deferred compensation plan, incentive plan, bonus plan or arrangement, stock option plan, stock purchase plan, stock award plan or other equity-based plan, change in control agreement, retention, severance pay plan, dependent care plan, sick leave, disability, death benefit, group insurance, hospitalization, dental, life, any fund, trust or arrangement providing health benefits including multiemployer welfare arrangements, a multiple employer welfare fund or arrangement, cafeteria plan, employee assistance program, scholarship program, employment contract, retention incentive agreement, termination agreement, severance agreement, non-competition agreement, consulting agreement, confidentiality agreement, vacation policy, employee loan, or other similar plan, agreement or arrangement, whether written or oral, funded or unfunded, or actual or contingent which is or was sponsored, maintained or contributed to by, or required to be contributed by, the Borrowers, any of their Subsidiaries or any of their respective ERISA Affiliates.

"EU Bail-In Legislation Schedule means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor Person), as in effect from time to time.

"<u>Eurodollar Reserve Percentage</u>" means, for any day during any Interest Period, the reserve percentage (expressed as a decimal, carried out to five decimal places) in effect on such day, whether or not applicable to any Lender, under regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to eurocurrency funding (currently referred to as "<u>Eurocurrency Liabilities</u>"). The LIBO Rate for each outstanding LIBOR Loan shall be adjusted automatically as of the effective date of any change in the Eurodollar Reserve Percentage.

"Event of Default" has the meaning specified in Section 9.01.

"Excess Cash Flow" means, with respect to the Loan Parties and their Subsidiaries on a consolidated basis, for any period: (a) without duplication, the sum of (i) Consolidated Net Income, (ii) decreases or minus increases (as the case may be) in Consolidated Working Capital, (iii) extraordinary or non-recurring gains or minus extraordinary or non-recurring losses (as the case may be) which are cash items, (iv) non-cash depreciation, non-cash amortization and other non-cash charges; minus (b) without duplication, the sum of: (i) Consolidated Unfinanced Capital Expenditures, (ii) expenses paid in respect of Permitted Acquisitions during such period, (iii) to the extent not taken into account in the calculation of Consolidated Net Income, (x) the principal amortization paid in cash during such period with respect to Capital Leases and (y) the amount of any Restricted Payments actually made to Holdings pursuant to <u>Section 7.06(g)</u> during such period, (v) Earn-Out Obligations paid in cash and (vi) unfinanced consideration paid in cash for Permitted Acquisitions during such period.

"Excess Liquidity" means the sum (a) Revolving Availability plus (b) the average daily balance (based on preceding thirty (30) days) of unrestricted cash (free of Liens or encumbrances except in favor of

the Administrative Agent) on deposit in the Loan Parties' Controlled Accounts located within the United States.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Excluded Account" means any deposit account (i) that is used solely for payment of payroll, bonuses, other compensation and related expenses; (ii) deposit accounts that are zero balance accounts so long as such deposit accounts are swept on a daily basis to a deposit account that otherwise constitutes Excluded Property, an Excluded Account or that is subject to a Deposit Account Control Agreement or similar agreement; or (iii) that maintains a balance on deposit at any time of not more than \$100,000, provided that, (A) the aggregate balance on deposit at any time in all Excluded Accounts described in clause (i) shall not exceed 105% of the amount to be applied for the pay period next ending, (B) the aggregate balance on deposit at any time in all Excluded Accounts described in clause (ii) shall not exceed \$1,000,000 at any time and (C) the Excluded Accounts are listed on Schedule 1.01(b) and maintained with banks reasonably satisfactory to the Administrative Agent.

"<u>Excluded Foreign Subsidiary</u>" means any Subsidiary of Intermediate Holdings (other than any Loan Party) that is, and continues to be, (a) a Foreign Subsidiary that is a CFC, or (b) a Domestic Subsidiary that is (i) a direct or indirect Subsidiary of a Foreign Subsidiary that is a CFC or (ii) a Subsidiary that is treated as a disregarded entity or as a partnership for United States Federal income tax purposes and all or substantially all of whose assets consist of Capital Stock of one or more Foreign Subsidiaries that are CFCs and conducts no material business other than the ownership of such Capital Stock (a "<u>Foreign Subsidiary Holdco</u>").

"Excluded Hedge Obligation" means, with respect to any Loan Party, any Hedge Obligation if, and to the extent that, all or a portion of the Loan Documents to which such Loan Party is party with respect to, or the grant by such Loan Party of a security interest to secure, such Hedge Obligation (or any Guarantee thereof) is or becomes illegal or unlawful under the Commodity Exchange Act or any rule or regulation promulgated thereunder (or the application or official interpretation of any provision thereof) by virtue of such Loan Party's failure for any reason not to constitute an "eligible contract participant" as defined in the Commodity Exchange Act at the time any such Loan Document would otherwise have become effective with respect to such related Hedge Obligation.

"Excluded Property" means, with respect to any Loan Party (a) any owned or leased real or personal Property (other than Capital Stock) which is located outside of the United States unless requested by the Administrative Agent or the Required Lenders, (b) any leasehold interests in Real Property, (c) any personal Property (other than motor vehicles, which are addressed in clause (i) below) in respect of which perfection of a Lien is not either (i) governed by the UCC or (ii) effected by appropriate evidence of the Lien being filed in either the United States Copyright Office or the United States Patent and Trademark Office, unless requested by the Administrative Agent or the Required Lenders, (d) any Property which, subject to the terms of Section 7.01(c), is subject to a Lien of the type described in Section 7.02(h) pursuant to documents which prohibit such Loan Party from granting any other Liens in such Property, (e) other than Accounts, any contract, permit, agreement, lease, license or license agreement covering real or personal property if the grant of a security interest in such contract, permit, agreement, lease, license or license agreement or the asset subject to such contract, permit, agreement, lease, license or license agreement is prohibited by the terms of such contract, permit, agreement, lease, license or license agreement that (x) any such prohibition would not be rendered ineffective pursuant to the UCC or any other applicable law (including, if and when applicable, Debtor Relief Laws) or principles of equity and would require the consent a Person other than a Loan Party or its Affiliates to waive such prohibition and (y) such contract, agreement, lease, license or license

agreement was not entered into in contemplation of circumventing any Loan Party's obligation to pledge such contract or agreement as collateral security, (f) any United States intent-to-use trademark applications to the extent that, and solely during the period in which, the grant of a security interest therein would impair the validity or enforceability of such intent-to-use trademark applications under applicable federal law, provided that upon submission and acceptance by the United States Patent and Trademark Office of an amendment to allege use pursuant to 15 U.S.C. Section 1060(a) (or any successor provision), such intent-to-use trademark application shall not be considered Excluded Property, (g) any Excluded Account, provided that for the avoidance of doubt, any proceeds of Collateral held from time to time in any such Excluded Account shall not cease to be Collateral solely because such proceeds are held in an Excluded Account, (h) any particular asset, if the pledge thereof or the security interest therein is prohibited by applicable Law (but only for so long as such prohibition remains in effect), other than to the extent such prohibition is rendered ineffective under the UCC or other applicable Laws, (i) any motor vehicle or other equipment subject to a certificate of title, but only to the extent that (x) the fair market value thereof is less than \$50,000 individually or \$250,000 in the aggregate, (y) as of the Closing Date, the name of CIT or any of its Affiliates is not notated on the certificate of title and (z) a security interest therein cannot be perfected by the filing of a UCC financing statement, (i) any Capital Stock in joint ventures or any non-Wholly Owned Subsidiary to the extent (i) consent of the minority investor is required pursuant to the Organizational Documents of such Excluded Subsidiary (and such consent was not required in connection with or in contemplation or anticipation of such pledge) and not obtained after using commercially reasonable efforts and (ii) such non-Wholly Owned Subsidiary is an Excluded Subsidiary, (k) any Capital Stock in any Excluded Foreign Subsidiary held by the Loan Parties in excess of sixty-five percent (65%) of the total outstanding voting Capital Stock of such Excluded Foreign Subsidiary, and (I) particular assets if and for so long as, in the reasonable judgment of the Administrative Agent, the Required Lenders and the Borrower Representative (as set forth in a written agreement among the Administrative Agent, the Required Lenders and the Borrower Representative), the cost of obtaining a security interest in such assets exceeds the practical benefits to the Secured Parties afforded thereby; provided, "Excluded Property" shall not include any proceeds, products, substitutions, or replacements of Excluded Property (unless such proceeds, products, substitutions, or replacements would otherwise constitute Excluded Property).

"Excluded Subsidiary" means, as of any date, any Subsidiary of Intermediate Holdings (other than any Loan Party) designated as such by the Borrowers in writing to the Administrative Agent that is, and continues to be, a non-Wholly Owned Subsidiary of Intermediate Holdings, that satisfies the following requirements: (a) the aggregate Consolidated EBITDA attributable to such Person individually does not exceed 5.0% of Consolidated EBITDA of the Loan Parties and their Subsidiaries for the last twelve (12) months ending on the last day of the last month that is also the end of a Fiscal Quarter for which financial statements of the type set forth in <u>Sections 6.01(a)</u> or (b) have been delivered to Administrative Agent and (b) the aggregate Consolidated EBITDA attributable to such Person together with all other Excluded Subsidiaries does not exceed 7.50% of Consolidated EBITDA of the Loan Parties and their Subsidiaries for the last twelve (12) months ending on the last day of the last month that is also the end of a Fiscal Quarter for which financial statements of the type set forth in <u>Section 6.01(a)</u> or (b) have been delivered to the Administrative Agent; provided, to the extent any Excluded Subsidiary ceases to be a joint venture and becomes a Wholly Owned Subsidiary of a Loan Party, then such Subsidiary will be required to become a Loan Party at such time.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an

applicable interest in a Loan or Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the Loan or Commitment (other than pursuant to an assignment request by the Borrower Representative under<u>Section 12.15</u>) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to <u>Section 3.01</u>, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with <u>Section 3.01(f)</u> and (d) any U.S. federal withholding Taxes imposed under FATCA.

"Extended Revolving Commitment" means any Revolving Commitments the maturity of which shall have been extended pursuant to Section 2.18.

"Extended Revolving Loans" means any Loans made pursuant to the Extended Revolving Commitments.

"Extended Term Loans" means any Term Loans the maturity of which shall have been extended pursuant to Section 2.18.

"Extension" has the meaning set forth in Section 2.18(a).

"Extension Offer" has the meaning set forth in Section 2.18(a).

"Extraordinary Receipts" means any cash received by or paid to or for the account of any Loan Party not in the ordinary course of business (and not consisting of proceeds described in any of Section 2.05(b)(ii), (iii) and (iv)) including without limitation amounts received in respect of indemnity obligations of a seller under any agreement governing a Permitted Acquisition, foreign, United States, state or local tax refunds to the extent not included in the calculation of Consolidated EBITDA and pension plan reversions.

"Facilities" means, at any time, the facilities and real properties owned, leased, managed or operated by any Loan Party or any Subsidiary, from which any Loan Party or any Subsidiary provides or furnishes goods or services.

"Fair Share" has the meaning set forth in Section 10.06.

"Fair Share Contribution Amount" has the meaning set forth in Section 10.06.

"<u>FATCA</u>" means Sections 1471 through 1474 of the Internal Revenue Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof, any agreement entered into pursuant to Section 1471(b) of the Internal Revenue Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement entered into in connection with the implementation of such sections of the Internal Revenue Code.

"FDA" means the Food and Drug Administration of the United States of America or any successor entity thereto.

"FDCA" means the Federal Food, Drug, and Cosmetic Act, as amended, 21 U.S.C. Section 301 et seq. and all regulations promulgated thereunder.

"<u>Federal Funds Rate</u>" means, for any day, the rate per annum equal to the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System on such day, as published by the Federal Reserve Bank on the Business Day next succeeding such day, <u>provided</u> that (a) if

such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to the Administrative Agent on such day on such transactions as determined by the Administrative Agent, in its sole discretion.

"Fee Letter" means the letter agreement dated as of February 27, 2019, among the Loan Parties, the Administrative Agent, and the Arranger.

"FIRREA" means the Financial Institutions Reform Recovery and Enforcement Act.

"Fiscal Quarter" means a fiscal quarter of any Fiscal Year.

"<u>Fiscal Year</u>" means the fiscal year of <u>AdaptHealth Corp., Holdings, and</u>the Loan Parties and their Subsidiaries ending on December 31st of each calendar year.

"Flood Hazard Property" means Mortgaged Property in an area designated by the Federal Emergency Management Agency as having special flood and mud slide hazards.

"Flood Laws" has the meaning set forth in Section 11.15.

"Following Business Day Convention" means a contractual provision or provision of applicable Laws pursuant to which a scheduled date for payment or performance of an obligation, which date is not a Business Day, is extended to the first following day that is a Business Day.

"Foreign Lender" means a Lender that is not a U.S. Person.

"Foreign Subsidiary" means any Subsidiary that is not a Domestic Subsidiary.

"Foreign Subsidiary Holdco" has the meaning specified in the definition of Excluded Foreign Subsidiary.

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"Fronting Exposure" means, at any time there is a Defaulting Lender, with respect to the L/C Issuer (or the Support Provider, as the case may be), such Defaulting Lender's Pro Rata Share of the outstanding Letter of Credit Liabilities other than Letter of Credit Liabilities as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof.

"Fronting Fee" has the meaning specified in Section 2.03(c).

"Funded Indebtedness" means, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

(a) all obligations for borrowed money, whether current or long-term (including the Obligations) and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments;

(b) all purchase money indebtedness;

(c) the principal portion of all obligations under conditional sale or other title retention agreements relating to Property purchased by such Person (other than customary reservations or retentions of title under agreements with suppliers entered into in the ordinary course of business);

(d) the maximum amount available to be drawn under letters of credit (including standby and commercial), bankers' acceptances, bank guaranties, surety bonds and similar instruments;

(e) all obligations in respect of the deferred purchase price of Property or services (other than trade accounts payable in the ordinary course of business);

(f) Attributable Indebtedness in respect of Capital Leases;

(g) all preferred stock or other Disqualified Capital Stock providing for mandatory redemptions, sinking fund or like payments prior to the Termination Date;

(h) Earn-Out Obligations if, and only to the extent, such obligation has not been paid in full in cash when initially due and payable;

(i) all indebtedness of the types specified in <u>clauses (a)</u> through (<u>h</u>) above secured by (or for which the holder of such Funded Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on, or payable out of the proceeds of production from, Property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed; and

(j) all Guarantees with respect to indebtedness of the types specified in <u>clauses (a)</u> through (i) above of another Person.

"Funding Guarantor" has the meaning set forth in Section 10.06.

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board, consistently applied and as in effect from time to time.

"Gould's Acquisition" means the acquisition by Borrower of one hundred percent (100%) of the Capital Stock of Gould's Discount Medical, LLC a Kentucky limited liability company, as set forth in the Gould's Purchase Agreement.

"<u>Gould's Purchase Agreement</u>" means the Equity Purchase Agreement dated January 1, 2019, by and among Borrower, as buyer, Edmond L. Gould, Sharon S. Gould, Kenneth C. Gould Irrevocable Trust, and David R. Gould Irrevocable Trust, as selling shareholders, Gould's Discount Medical, LLC, a Kentucky limited liability company, as company, and GDM HoldCo, Inc., a Kentucky corporation, as holdco.

"Gould's Subordinated Note" means that certain Seller Note as described in Section 2.3(a)(ii) of the Gould's Purchase Agreement.

"Government Reimbursement Program" shall mean (i) the Medicare program established under the Title XVIII of the Federal Social Security Act, the Federal Employees Health Benefit Program under 5 U.S.C. §§ 8902 et seq., the TRICARE program established by the Department of Defense under 10 U.S.C. §§ 1071 et seq. or the Civilian Health and Medical Program of the Uniformed Services under 10 U.S.C. §§ 1079 and 1086, (ii) the Medicaid program of any state or the District of Columbia acting pursuant to a

health plan adopted pursuant to Title XIX of the Federal Social Security Act or (iii) any agent, administrator, intermediary or carrier for any of the foregoing.

"Governmental Approvals" means any and all governmental licenses, authorizations, registrations, permits, certificates, franchises, qualifications, accreditations, consents and approvals required under any applicable Law and required in order for any Person to carry on its business as now conducted, of each Governmental Authority issued or required under Laws applicable to the business of any Borrower or any of its Subsidiaries or to the transactions described herein (including any Acquisitions and the Loans made hereunder) or necessary in the sale, furnishing, or delivery of goods or services under Laws applicable to the business of any Borrower or any of its Subsidiaries.

"Governmental Authority" means any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, administrative tribunal, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"<u>Guarantee</u>" means, as to any Person, (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation payable or performable by another Person (the "<u>primary obligor</u>") in any manner, whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other obligation of the payment or performance of such Indebtedness or other obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other obligation of the payment or performance thereof or to protect such oblige against loss in respect thereof (in whole or in part), or (b) any Lien on any assets of such Person securing any Indebtedness or other obligation of any other Person, whether or not such Indebtedness or other obligation is assumed by such Person. The amount of any Guarantee shall be deemed to be an amount equal to the stated or determinable amount of the related primary obligation, or portion thereof, in respect of which such Guarantee is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. The term "Guarantee" as a verb has a corresponding meaning.

"Guaranteed Obligations" has the meaning set forth in Section 10.01.

"Guarantor" means each of Intermediate Holdings, and all future direct or indirect Subsidiaries of Intermediate Holdings (other than the Borrowers, any Excluded Subsidiary and any Excluded Foreign Subsidiary), and any other Person joining this Agreement as a "Guarantor" hereunder from time to time.

"Guaranty" means the guaranty made by each Guarantor in favor of the Administrative Agent, the Lenders and the other Secured Parties pursuant to <u>Article 10</u>.

"<u>Hazardous Materials</u>" means all explosive or radioactive substances or wastes and all hazardous or toxic substances, wastes or other pollutants, including petroleum or petroleum distillates, asbestos or asbestos-containing materials, polychlorinated biphenyls, lead-based paint, toxic mold or fungus, radon gas, infectious or medical wastes and all other substances or wastes of any nature regulated pursuant to any Environmental Law.

"Healthcare Law(s)" shall mean: (a) any and all federal, state and local fraud and abuse laws, including (i) the federal Anti-Kickback Statute (42 U.S.C. § 1320a-7(b)), (ii) the Stark Law (42 U.S.C. § 1395nn and §1395(q)), (iii) the civil False Claims Act (31 U.S.C. § 3729 et seq.), (iv) Sections 1320a-7 and 1320a-7a of Title 42 of the United States Code, and (v) the regulations promulgated pursuant to such statutes; (b) the FDCA; (c) the Health Insurance Portability and Accountability Act of 1996 (Pub. L. No. 104-191) and the regulations promulgated pursuant thereto; (d) laws, rules and regulations governing Medicare and Medicaid; (e) the Medicare Prescription Drug, Improvement, and Modernization Act of 2003 (Pub. L. No. 108-173) and the regulations promulgated pursuant thereto; (f) quality, safety, life safety, and accreditation standards and requirements of all applicable state laws or regulatory bodies; (g) any applicable Law relating to the Loan Parties' ownership, management, or operation of a healthcare facility or business, or assets used in connection therewith; (h) any applicable Law relating to the billing or submission of claims, collection of accounts receivable, underwriting the cost of, or provision of management or administrative services in connection with, any and all of the foregoing, by any Loan Party; (i) the Patient Protection and Affordable Care Act (Pub. L. 111-148) and (j) any and all other applicable healthcare laws, regulations, manual provisions, policies and administrative guidance, each of (a) through (j) as may be amended from time to time.

"Hedge Agreement" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"<u>Hedge Obligations</u>" of a Person means any and all obligations of such Person, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefor), under (a) any and all Hedge Agreements, and (b) any and all cancellations, buy backs, reversals, terminations or assignments of any Hedge Agreement transaction.

"Hedge Party" means any Person that is a counterparty to a Hedge Agreement with a Loan Party.

"<u>HIPAA</u>" shall mean the Health Insurance Portability and Accountability Act of 1996, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.

"<u>HIPAA Complian</u>t" shall mean that such Loan Party is in compliance in all material respects with HIPAA and has adopted and implemented policies and procedures to strive to ensure compliance with and has trained its personnel in the transaction standards, privacy standards and security standards promulgated pursuant to HIPAA.

"Holdings" means AdaptHealth Holdings LLC, a Delaware limited liability company.

"Holdings Operating Agreement" means that certain Fourth Amended and Restated Limited Liability Company Agreement of AdaptHealth Holdings LLC, dated as of March 20, 2019.

"Incremental Equivalent Debt" means a single issuance of Indebtedness incurred by any Borrower or their Wholly Owned Subsidiaries (including Guarantees thereof by any Loan Party) in the form of senior secured or unsecured notes or loans or junior secured or unsecured notes or loans and/or commitments in respect of any of the foregoing issued or implemented in lieu of Incremental Term Loans; provided, that:

(a) Administrative Agent shall have received from Borrower Representative a Pro Forma Compliance Certificate demonstrating that after giving effect on a Pro Forma Basis to the Incremental Equivalent Debt (and giving effect on a Pro Forma Basis to any Acquisition consummated concurrently therewith), the Consolidated Total Leverage Ratio (calculated on a Pro Forma Basis) as of the end of the most recent Test Period would not be greater than 3.00:1.00;

(b) No Default or Event of Default has occurred and is continuing immediately prior to or after giving effect to such Incremental Equivalent Debt;

(c) The Weighted Average Life to Maturity of any Incremental Equivalent Debt shall be no shorter than the remaining Weighted Average Life to Maturity of the outstanding Term Loan(s) funded on the Closing Date;

(d) The final maturity date of any Incremental Equivalent Debt shall be no earlier than the latest maturity date with respect to any Loan hereunder at such time, including the latest maturity date of any Loan or Incremental Facility, including, in each case, giving effect to any Extension with respect thereto;

(e) Subject to clauses (c) and (d), such Incremental Equivalent Debt may otherwise have an amortization schedule as determined by the Borrower Representative and the lenders providing such Incremental Equivalent Debt;

(f) Such Incremental Equivalent Debt shall be subject to an Acceptable Intercreditor Agreement, and such Incremental Equivalent Debt shall be *pari passu* or junior in right of payment and to the extent secured by any Collateral, shall be secured on a *pari passu* or junior in priority to the remaining portion of the Term Loans, Revolving Loans or any Incremental Facilities;

(g) No such Indebtedness may be (i) guaranteed by any Person which is not a Loan Party or (ii) secured by any assets other than the Collateral;

(h) The proceeds of any Incremental Equivalent Debt shall be used onlyto finance a Permitted Acquisition that requires Required Lender consent in accordance with clause (x) of the definition of "Permitted Acquisition" and any related costs and expenses with respect thereto;

(i) All of the representations and warranties contained in the Loan Documents are true and correct in all material respects (or, in all respects, if already qualified by materiality);

(j) Such Incremental Equivalent Debt may have an interest rate and such other pricing as the Borrower Representative in its reasonable discretion finds acceptable; and

(k) The other terms and conditions in respect of such Incremental Equivalent Debt shall not be more restrictive taken as a whole (as determined by the Administrative Agent in its reasonable discretion), than those applicable to the Term Loans, unless otherwise reasonably acceptable to the Required Lenders.

"Incremental Facilities" has the meaning specified in Section 2.15(a).

"Incremental Facility Amendment" has the meaning specified in Section 2.15(d).

"Incremental Facility Closing Date" has the meaning specified in Section 2.15(d).

"Incremental Revolving Commitments" has the meaning specified in Section 2.15(a).

"Incremental Revolving Lender" has the meaning specified in Section 2.15(d).

"Incremental Term Loans" has the meaning specified in Section 2.15(a).

"Indebtedness" means, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

- (a) all Funded Indebtedness;
- (b) the Termination Value of any Secured Hedge Agreement;
- (c) Synthetic Leases, Sale and Leaseback Transactions and Securitization Transactions;
- (d) all obligations in respect of Disqualified Capital Stock; and
- (e) all Guarantees with respect to outstanding indebtedness of the types specified in clauses (b), (c) and (d) above of any other Person.

"Indemnified Liabilities" has the meaning set forth in Section 12.05.

"Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Loan Party under any Loan Document and (b) to the extent not otherwise described in (a), Other Taxes.

"Indemnitees" has the meaning set forth in Section 12.05.

"Information" has the meaning set forth in Section 12.08.

"Initial Term Loan" has the meaning specified in Section 2.01(b).

"Initial Term Loan Commitment" means, as to each Lender, its obligations to make its portion of the Initial Term Loan to the BorrowerBorrowers pursuant to Section 2.01(b) and the other terms and conditions of this Agreement, in the principal amount set forth opposite such Lenders name on Schedule 2.01, as such amounts may be adjusted from time to time, in accordance with this Agreement. The initial aggregate amount of the Initial Term Loan Commitments is \$300,000,000.

"Interest Payment Date" means (a) as to any LIBOR Loan, the last day of each Interest Period applicable to such LIBOR Loan and the Term Loan Maturity Date or Revolving Loan Maturity Date (as applicable)<u>provided</u>, that if any Interest Period for a LIBOR Loan exceeds three (3) months, the respective dates that fall every three (3) months after the beginning of such Interest Period shall also be Interest Payment Dates; (b) as to any Base Rate Loan (other than a Swingline Loan), the last Business Day of each

calendar quarter and the Revolving Loan Maturity Date or Term Loan Maturity Date (as applicable); and (c) with respect to any Swingline Loan, the day that such Loan is required to be repaid.

"Interest Period" means, as to each LIBOR Loan, the period commencing on the date such LIBOR Loan is disbursed or converted to or continued as a LIBOR Loan and ending on the date one (1), two (2), three (3) or six (6) months thereafter or, if approved by all affected Lenders, the Administrative Agent and the Borrowers, twelve (12) months thereafter, as selected by the Borrower Representative in its Loan Notice, provided, that:

(a) any Interest Period that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day;

(b) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period; and

(c) no Interest Period shall extend beyond the Term Loan Maturity Date or Revolving Loan Maturity Date, as applicable.

"<u>Interim Pre-Closing Financial Statement</u>" means, collectively, (i) the unaudited consolidated financial statements of the Loan Parties and their Subsidiaries for the Fiscal Quarters ended March 31, 2018, June 30, 2018, September 30, 2018 and December 31, 2018, and the related consolidated statements of income or operations, shareholders' equity and cash flows, for the Fiscal Quarter ended on that date.

"Intermediate Holdings" means AdaptHealth Intermediate Holdco LLC, a Delaware limited liability company.

"Internal Revenue Code" means the Internal Revenue Code of 1986, as amended, and any successor thereto.

"Internally Generated Cash" means, with respect to any Fiscal Year, cash of the Consolidated Group received in such Fiscal Year and not constituting (a) proceeds of an Equity Issuance quity issuance, (b) proceeds of the incurrence of Indebtedness, (c) proceeds of Dispositions and Involuntary Dispositions, and (d) insurance proceeds in respect of any equipment, fixed assets or real property (including any improvements thereon) to replace or repair such equipment, fixed assets or real property.

"Inventory Suppliers" means the financing companies, lenders and/or leasing companies set forth on Schedule 1.01(c) that provide equipment or inventory financing or leasing services to any Loan Party with an aggregate value of \$250,000 or more.

"<u>Investment</u>" means, as to any Person, any direct or indirect acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of any of the Capital Stock of another Person, (b) a loan, advance or capital contribution to, Guarantee or assumption of debt of, or purchase or other acquisition of any other debt or equity participation or interest in, another Person, or (c) an Acquisition. For purposes of covenant compliance, the amount of any Investment shall be the amount actually invested, without adjustment for subsequent increases or decreases in the value of such Investment or any returns of capital.

"Involuntary Disposition" means any loss of, damage to or destruction of, or any condemnation or other taking for public use of, any Property of any Loan Party.

"IP Rights" has the meaning set forth in Section 5.17.

"IRS" means the United States Internal Revenue Service.

"<u>ISP</u>" means, with respect to any standby Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice (or such later version thereof as may be in effect at the time of issuance).

"Issuer Documents" means with respect to any Letter of Credit, any application for such Letter of Credit and any other document, agreement and instrument entered into by the applicable L/C Issuer and the Borrower Representative on behalf of Borrowers or in favor of such L/C Issuer and relating to any such Letter of Credit.

"<u>Landlord Agreement</u>" means any agreement between the Administrative Agent and the landlord of Real Property occupied by a Loan Party, as tenant, which agreement shall be in form and substance as is satisfactory to the Administrative Agent, as amended, supplemented, modified, replaced, substituted for or restated from time to time and all exhibits and schedules attached thereto.

"Law(s)" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, compacts, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority.

"<u>L/C Credit Extension</u>" means (a) the issuance, extension or increase of a Letter of Credit, and (b) with respect to any Supported Letter of Credit, the entry into any Support Agreement by the Administrative Agent or its Affiliate.

"L/C Issuer" means (a) Bank of America, N.A. or (b) a Revolving Lender willing and able to issue Letters of Credit and acceptable to the Administrative Agent.

"<u>Lender</u>" means each Person identified as a "Lender" on the signature pages hereto and its successors and assigns and, as the context requires, includes the L/C Issuer to the extent it is a party to this Agreement. Unless the context otherwise requires, the term "Lenders" includes the Swingline Lender.

"Lender Letter of Credit" means a Letter of Credit issued by an L/C Issuer that is also, at the time of issuance of such Letter of Credit, a Lender.

"Lender Parties" has the meaning specified in Section 12.07(g).

"Lender Securitization" has the meaning specified in Section 12.07(g).

"Lending Office" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Borrower Representative and the Administrative Agent.

"Letter of Credif' means a standby or documentary (trade) letter of credit issued for the account of any Borrower by an L/C Issuer pursuant to the terms of this Agreement.

"Letter of Credit Fee" has the meaning specified in Section 2.03(c).

"Letter of Credit Liabilities" means, at any time of calculation, the sum of the following (without duplication): (a) the amount then available for drawing under all outstanding Letters of Credit, in each case without regard to whether any conditions to drawing thereunder can then be met, and (without duplication) the amount of any outstanding Support Agreement related to a Letter of Credit, and (b) the aggregate of all Unreimbursed Amounts. For all purposes of this Agreement, if as of any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount that remains available to be drawn. The Letter of Credit Liability of any Revolving Lender at any time shall be equal to its Pro Rata Share of the total Letter of Credit Liabilities at such time.

"Letter of Credit Sublimit" means an amount equal to the lesser of (a) the total Revolving Commitments and (b) \$6,000,000. The Letter of Credit Sublimit is part of, and not in addition to, the total Revolving Commitments.

## "LIBO Base Rate" means,

(1) for any Interest Period with respect to any LIBOR Loan:

(a) the rate per annum equal to the rate determined by the Administrative Agent to be the London Interbank Offered Rate benchmark rate which is calculated and distributed by the ICE Benchmark Administration Data Service ( $\underline{\text{TCE}}^{"}$ ) (or any successor thereto) for deposits in Dollars (for delivery on the first (1st) day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:45 a.m. (London time) (or such other time as confirmed by ICE (or any successor thereto)) two (2) Business Days prior to the first (1st) day of such Interest Period as displayed on such day and time on pages LIBOR01 or LIBOR02 of the Reuters screen that displays such rate (or, in the event such rate does not appear on a Reuters page or screen, on any successor or substitute page on such screen that displays such rate, or on the appropriate page of such other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion) (the "LIBO Screen Rate"), or

(b) if the rate referenced in the preceding clause (a) does not appear through such service or such service shall not be available, the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate which is calculated and distributed daily by ICE (or any successor thereto) as an average ICE Benchmark Administration Limited Interest Settlement Rate for deposits in Dollars (for delivery on the first (1st) day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:45 a.m. (London time) two (2) Business Days prior to the first (1st) day of such Interest Period, or

(c) if the rates referenced in the preceding clauses (a) and (b) are not available, the rate per annum (rounded upward to the next 1/100th of 1%) determined by the Administrative Agent as the rate of interest at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the LIBOR Loan being made, continued or converted by JPMorgan Chase Bank and with a term equivalent to such Interest Period would be offered by JPMorgan Chase Bank's London Branch (or such other major bank as is acceptable to the Administrative Agent if JPMorgan Chase Bank is no longer offering to acquire or allow deposits in the London interbank eurodollar market) to major banks in the London interbank eurodollar market at their request at approximately 11:45 a.m. (London time) two (2) Business Days prior to the first day of such Interest Period.

(2) for any day with respect to an interest rate calculation for a Base Rate Loan:

(a) the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate which is calculated and distributed daily by ICE (or any successor thereto) as an average ICE Benchmark Administration Limited Interest Settlement Rate for deposits in Dollars (for delivery on such day) with a term equivalent to three (3) months, determined as of approximately 11:45 a.m. (London time) two (2) Business Days prior to such day, or

(b) if the rate referenced in the preceding <u>clause (a)</u> does not appear through such service or such service shall not be available, the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate through such other service that displays an average ICE Benchmark Administration Limited Interest Settlement Rate for deposits in Dollars (for delivery on such day) with a term equivalent to three (3) months, determined as of approximately 11:45 a.m. (London time) two (2) Business Days prior to such day, or

(c) if the rates referenced in the preceding <u>clauses (a)</u> and <u>(b)</u> are not available, the rate per annum (rounded upward to the next 1/100th of 1%) determined by the Administrative Agent as the rate of interest at which deposits in Dollars (for delivery on such day in same day funds) with a term equivalent to three months, determined as of approximately 11:00 a.m. (London time) two (2) Business Days prior to such day in the approximate amount of such Base Rate Loan by JPMorgan Chase Bank and with a term equivalent to three (3) months would be offered by JPMorgan Chase Bank's London Branch (or such other major bank as is acceptable to the Administrative Agent if JPMorgan Chase Bank is no longer offering to acquire or allow deposits in the London interbank eurodollar market at their request at approximately 11:45 a.m. (London time) two (2) Business Days prior to such day.

"LIBO Rate" means the greater of: (a) onezero percent (1.00.0%); and (b) (1) for any Interest Period with respect to any LIBOR Loan, a rate per annum determined by the Administrative Agent to be equal to the quotient obtained by dividing (i) the LIBO Base Rate for such LIBOR Loan for such Interest Period by (ii) one minus the Eurodollar Reserve Percentage for such LIBOR Loan for such Interest Period and (2) for any day with respect to any Base Rate Loan bearing interest at a rate based on the LIBO Rate, a rate per annum determined by the Administrative Agent to be equal to the quotient obtained by dividing (i) the LIBO Base Rate Loan for such day by (ii) one minus the Eurodollar Reserve Percentage for such Base Rate Loan for such day by (ii) one minus the Eurodollar Reserve Percentage for such Base Rate Loan for such day.

"LIBO Screen Rate" has the meaning set forth in clause (1)(a) of the definition of LIBO Base Rate.

"<u>LIBOR Loan</u>" means any Loan (other than Swingline Loans and Base Rate Loans bearing interest at a rate based on the LIBO Rate) which accrues interest solely by reference to the LIBO Rate plus the Applicable Margin, in accordance with the terms of this Agreement.

"<u>Lien</u>" means any mortgage, pledge, hypothecation, collateral assignment, deposit arrangement, encumbrance, lien (statutory or other), charge, or preference, priority or other security interest or preferential arrangement of any kind or nature whatsoever (including any conditional sale or other title retention agreement, and any financing lease having substantially the same economic effect as any of the foregoing).

"Loan" means an extension of credit by a Lender to any Borrower under <u>Article 2</u> in the form of an Initial Term Loan, a Delayed – Draw Term Loan, a Revolving Loan and/or a Swingline Loan.

"Loan Documents" means this Agreement, each Note, each Letter of Credit, each Collateral Document, each Request for Credit Extension, each Issuer Document, each Compliance Certificate, the Fee Letter, each Subordination Agreement<u>each Acceptable Intercreditor</u> Agreement, each Secured Hedge

Agreement and each other document, instrument or agreement from time to time executed by any Loan Party or any Subsidiary or any Responsible Officer thereof and delivered in connection with the transactions contemplated by this Agreement; provided, however, that all Secured Hedge Agreements, agreements hereafter executed solely in respect of the Banking Services Obligations and all Issuer Documents executed by or on behalf of any Loan Party and delivered concurrently herewith or at any time hereafter shall not constitute "Loan Documents" solely for the purposes of Sections 9.01(a), (b), (c), (d) and (j) hereof or Section 11.14.

"Loan Notice" means a notice of (a) (i) a Borrowing of Revolving Loans or (ii) a Borrowing of a Delayed – Draw Term Loan, (b) a conversion of Loans from one Type to the other pursuant to Section 2.02(a), or (c) a continuation of LIBOR Loans pursuant to Section 2.02(a), which, if in writing, shall be substantially in the form of Exhibit A-1.

"Loan Parties" means, collectively, each Borrower and each Guarantor party hereto.

"<u>Management Entity</u>" means, collectively, each of Josh Parnes, Luke McGee and any entities wholly owned or otherwise controlled by such individuals, including, without limitation, for so long as such entities are so owned or controlled, Jedi Enterprises LLC, a New York limited liability company and 2321 Capital LLC, a Delaware limited liability company.

"Master Agreement" has the meaning set forth in the definition of "Hedge Agreement."

"<u>Material Adverse Effect</u>" means (a) a material adverse change in, or a material adverse effect upon, the operations, business, properties, assets, liabilities (actual or contingent) or financial condition of the Loan Parties and their Subsidiaries taken as a whole; (b) (i) a material impairment of the ability of any Borrower to perform its obligations under any Loan Document to which it is a party or (ii) a material adverse effect upon the legality, validity, binding effect or enforceability against any Loan Document to which they are party; (c) a material adverse effect upon the legality, validity, perfection or priority of a Lien in favor of the Administrative Agent for the benefit of the Secured Parties on any material portion of the Collateral or a material adverse effect on the use or scope of any material healthcare Permit or the continued participation or the ability to accept or bill for goods or services in the Medicaid, Medicare or other Government Reimbursement Program by any Loan Party in good faith (a "Voluntary Termination").

"<u>Material Contract</u>" means any lease of real or personal property, contract or other arrangement to which any Loan Party or any of its Subsidiaries is a party (other than the Loan Documents), for which breach, nonperformance, cancellation or failure to renew could reasonably be expected to have a Material Adverse Effect.

"<u>Medicaid</u>" shall mean, collectively, the healthcare assistance program established by Title XIX of the Social Security Act (42 U.S.C. §§ 1396 et. seq.) and any statutes succeeding thereto, and all laws, rules, regulations, manuals, orders, guidelines or requirements (whether or not having the force of law) pertaining to such program, in each case as the same may be amended, supplemented or otherwise modified from time to time.

"<u>Medicare</u>" shall mean, collectively, the health insurance program for the aged and disabled established by Title XVIII of the Social Security Act (42 U.S.C. §§ 1395 et. seq.) and any statutes succeeding thereto, and all laws, rules, regulations, manuals, orders or guidelines (whether or not having the

force of law) pertaining to such program, in each case as the same may be amended, supplemented or otherwise modified from time to time.

"Med Way Acquisition" means the acquisition by Med Star Surgical & Breath Inc., a Loan Party, of one hundred percent (100%) of the Capital Stock of Med Way Medical, Inc., a Utah corporation, as set forth in the Med Way Purchase Agreement.

"Med Way Purchase Agreement" means the Stock Purchase Agreement to be entered into by and among Med Star Surgical & Breathing Inc., as buyer, Med Way Medical, Inc., a Utah corporation, as company, and Benjamin Dickman, Matthew Larsen, and David Larsen, as selling shareholders.

"Moody's" means Moody's Investors Service, Inc. and any successor thereto.

"Mortgage Instrument" means a fully executed and notarized mortgage, deed of trust or deed to secure debt, delivered by any Loan Party to the Administrative Agent with respect to Real Property owned or leased by such Loan Party, in each case to secure the payment and performance of the Obligations of such Loan Party. Unless otherwise consented to by the Administrative Agent, each Mortgage Instrument shall grant to the Administrative Agent for the benefit of the Secured Parties a first priority perfected Lien on the Real Property interests described therein.

"Mortgage Supporting Documents" means, with respect to a Mortgage Instrument for a parcel of Mortgaged Property, to the extent requested by Administrative Agent, each of the following:

(a) (i) evidence in form and substance reasonably satisfactory to the Administrative Agent that the recording of counterparts of such Mortgage Instrument in the recording offices specified in such Mortgage Instrument will create a valid and enforceable first priority Lien on the Property described therein in favor of the Administrative Agent (or in favor of such other trustee as may be required or desired under local Law) for the benefit of the Secured Parties, subject only to (A) Permitted Liens and (B) such other Liens as the Administrative Agent may reasonably approve and (ii) an opinion of counsel in each state in which any such Mortgage Instrument is to be recorded in form and substance and from counsel reasonably satisfactory to the Administrative Agent;

(b) (i) mortgagee's title policy (or policies) or marked-up unconditional binder (or binders) for such insurance (or other evidence reasonably acceptable to the Administrative Agent proving ownership thereof) ("<u>Mortgagee's Title Insurance Policy</u>"), dated a date reasonably satisfactory to the Administrative Agent, which shall (A) be in an amount not less than the then recently appraised fair market value (determined by reference to an appraisal) of such parcel of Real Property in form and substance satisfactory to the Administrative Agent, provided, that the amount of such title insurance required pursuant to such Mortgagee's Title Insurance Policy shall not materially exceed such appraised fair market value, (B) be issued at ordinary rates, (C) insure that the Lien granted pursuant to the Mortgage Instrument insured thereby creates a valid first Lien on such parcel of Real Property free and clear of all defects and encumbrances, except for Permitted Liens and for such other defects and encumbrances as may be approved by the Administrative Agent, (D) name the Administrative Agent as the insured thereunder, (E) be in the form of ALTA Loan Policy - 2006 (or such local equivalent thereof as is reasonably satisfactory to the Administrative Agent), (F) contain a comprehensive lender's endorsement (including, but not limited to, a revolving credit endorsement and a floating rate endorsement), (G) be issued by Chicago Title Insurance Company, First American Title Insurance Company, Lawyers Title Insurance Corporation or any other title company reasonably satisfactory to the Administrative Agent (including any such title companies acting as co-insurers) and (H) be otherwise in form and substance reasonably satisfactory to the Administrative Agent and (ii) a copy of all documents referred to, or listed as exceptions to title, in such

title policy (or policies) in each case in form and substance reasonably satisfactory to the Administrative Agent;

(c) in the case of each Mortgaged Property that is leased by any Loan Party, such estoppel letters, consents, Landlord Agreements and waivers from the landlords on such Mortgaged Property (and any mortgagee of such landlord) as may be reasonably required by the Administrative Agent; [Intentionally Omitted];

(d) a current Survey of such parcel of Mortgaged Property certified to and received by (in a manner reasonably satisfactory to each of them) the Administrative Agent and the title insurance company issuing the Mortgagee's Title Insurance Policy for such Mortgage Instrument, dated a date reasonably satisfactory to the Administrative Agent and such title insurance company, by an independent professional licensed land surveyor reasonably satisfactory to the Administrative Agent and such title insurance company;

(e) evidence in form and substance reasonably satisfactory to the Administrative Agent that all premiums in respect of each Mortgagee's Title Insurance Policy, all recording fees and stamp, documentary, intangible or mortgage taxes, if any, in connection with the Mortgage Instrument have been paid;

(f) a Phase I Environmental Site Assessment with respect to such parcel of Mortgaged Property, dated a date reasonably satisfactory to the Administrative Agent showing no material condition of environmental concern or recognized environmental conditions and otherwise in form and substance reasonably satisfactory to the Administrative Agent;

(g) evidence as to (A) whether such Mortgaged Property is a Flood Hazard Property, and (B) if such Mortgaged Property is a Flood Hazard Property, (1) whether the community in which such Mortgaged Property is located is participating in the National Flood Insurance Program, (2) the applicable Loan Party's written acknowledgment of receipt of written notification (a) as to the fact that such Mortgaged Property is a Flood Hazard Property and (b) as to whether the community in which each such Flood Hazard Property is located is participating in the National Flood Insurance Program and (3) copies of insurance policies or certificates of insurance of the Loan Parties and their Subsidiaries which shall evidence flood insurance which satisfies all regulatory requirements for flood insurance and is otherwise satisfactory to the Administrative Agent and naming the Administrative Agent as sole loss payee on behalf of the Secured Parties.

(h) evidence reasonably satisfactory to the Administrative Agent that such Mortgaged Property, and the uses of such Mortgaged Property, are in compliance in all material respects with all applicable zoning laws, regulations, ordinances and requirements (the evidence submitted as to which should include the zoning designation made for such Mortgaged Property, the permitted uses of such Mortgaged Property under such zoning designation and, if available, zoning requirements as to parking, lot size, ingress, egress and building setbacks);

(i) an appraisal of such Mortgaged Property satisfying the requirements of FIRREA; and

(j) such other agreements, documents and instruments in form and substance reasonably satisfactory to the Administrative Agent as the Administrative Agent reasonably deems necessary or appropriate to create, register or otherwise perfect, maintain, evidence the existence, substance, form or validity of, or enforce a valid and enforceable first priority Lien on such parcel of Mortgaged Property in favor of the Administrative Agent for the benefit of the Secured Parties (or in favor

of such other trustee as may be required or desired under local Law) subject only to (i) Permitted Liens and (ii) such other Liens as the Administrative Agent may reasonably approve; <u>provided</u>, notwithstanding anything contained in this Agreement to the contrary, no Mortgage Instrument shall be executed and delivered with respect to any real property unless and until each Lender has received, at least twenty (20) Business Days prior to such execution and delivery, a life loan flood zone determination and such other documents as it may reasonably request to complete its flood insurance due diligence (including those described in the preceding <u>clause (g)</u>) and has confirmed to the Administrative Agent that flood insurance due diligence and flood insurance compliance has been completed to its satisfaction.

"Mortgaged Property" means all Real Property owned by any Loan Party other than Excluded Property.

"<u>Multiemployer Plan</u>" means any employee benefit plan of the type described in Sections 4001(a)(3) or 3(37) of ERISA that is sponsored or maintained by any Borrower or any ERISA Affiliate or to which any Borrower or any ERISA Affiliate makes or is obligated to make contributions, or during the preceding six (6) plan years, has made or been obligated to make contributions.

"<u>Net Cash Proceeds</u>" means the aggregate cash and Cash Equivalents proceeds (including insurance proceeds and condemnation awards) received by any Loan Party or any Subsidiary in respect of any Disposition, Involuntary Disposition, Equity Issuance, or Debt Issuance net of (a) direct third-party costs incurred in connection therewith (including legal, accounting and investment banking fees, and sales commissions payable to third parties unrelated to Loan Parties), (b) taxes paid or payable as a result thereof (after taking into account any available tax credits or deductions and any tax sharing arrangements), and (c) the amount necessary to retire any Indebtedness secured by a Permitted Lien on the related Property that is senior in priority to the Lien of Administrative Agent and is also required to be discharged in connection with such disposition or issuance; it being understood that "Net Cash Proceeds" shall include any cash or Cash Equivalents received upon the sale or other disposition of any non-cash consideration received by any Borrower or any Subsidiary in any Disposition, Involuntary Disposition, Equity Issuance or Debt Issuance.

"Non-Blocker AdaptHealth Members" mean collectively each of (a) 2321 Capital LLC, (b) CFCP LLC, (c) Fresh Pond Investment LLC, (d) Jedi Enterprises, LLC, (e) LBM DME Holdings LLC, (f) Mayaid2001 LLC, (g) McLarty Capital Partners SBIC, L.P., (h) Ocean Rock NJ LLC (held in trust by The Mykonos 2019 NGCG Nevada Trust), (i) Plains Capital LLC (held in trust by The Mykonos 2019 NGCG Nevada Trust), (j) Blue River NJ LLC (held in trust by Still Water Nevada Trust), (k) Quad Cap LLC (held in trust by Still Water Nevada Trust), (l) Quadrant Management, Inc., (m) Verus Equity Holding Company LLC, (n) Verus Note Holding Company LLC, and (o) Wheatfield LLC.

"Note" or "Notes" means each Term Note, each Revolving Note and/or the Swingline Note, individually or collectively, as appropriate.

"Note and Unit Purchase Agreement" means that certain Note and Unit Purchase Agreement dated as of February 27, 2019, by and between Holdings and BM AH Holdings, LLC.

"<u>Notice of L/C Credit Even</u>" means a notice from a Responsible Officer of Borrower Representative to Administrative Agent with respect to any issuance or amendment (including any increase or extension) of a Letter of Credit specifying: (i) the date of issuance or amendment of a Letter of Credit; (ii) the identity of the L/C Issuer with respect to such Letter of Credit; (iii) the expiry date of such Letter of Credit; (iv) the proposed terms of such Letter of Credit (or amendment thereof), including the face amount; and (v) the transactions that are to be supported or financed with such Letter of Credit or increase thereof.

"Obligations" means all advances to, and debts, liabilities, obligations, covenants and duties of, any Loan Party now or hereinafter arising from time to time under this Agreement and any other Loan Document or otherwise with respect to any Loan (including the obligation to pay principal and interest thereon and all fees and other costs and liabilities with respect thereto), Letter of Credit, Support Agreement, Reinbursement Obligation or Unreinbursed Amount, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding. The foregoing shall also include any Banking Services Obligations and any obligations or liabilities of any Loan Party under any Secured Hedge Agreement. Notwithstanding anything to the contrary, the term "Obligations" shall not include, with respect to any Loan Party, any Excluded Hedge Obligation of such Loan Party. On the Closing Date, all existing "Obligations" under, and as defined in, the Original Loan Agreement shall be deemed to be Obligations hereunder.

"OFAC" means The Office of Foreign Assets Control of the U.S. Department of the Treasury.

"Organization Documents" means, (a) with respect to any corporation, the charter, certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization or organization of such entity.

"Original Loan Agreement" has the meaning specified in the recitals.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan or Loan Document).

"<u>Other Taxes</u>" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to <u>Section 12.15</u>).

"Parent" means Access Point Medical LLC, a Delaware limited liability company, Blue River NJ, LLC, a New Jersey limited liability company, Ocean Rock NJ LLC, a New Jersey limited liability company, Jedi Enterprises LLC, a New York limited liability company, 2321 Capital LLC, a Delaware limited liability company, and MCP QMES Corp. a Delaware corporation, BM AH Holdings, LLC, a Delaware limited liability company, and the other holders of approximately 2.0% of the Capital Stock of Holdings as of the date hereof (each a "Parent" and collectively the "Parents").

"Participant" has the meaning set forth in Section 12.07(d).

"Participant Register" has the meaning specified in Section 12.07(d).

"Participation Agreement" has the meaning set forth in Section 6.08.

"Patriot Act" has the meaning specified in Section 5.28.

"Payor" means the party primarily obligated to pay an Account.

"PBGC" means the Pension Benefit Guaranty Corporation or any successor thereto.

"<u>Pension Plan</u>" means any ERISA Plan, other than a Multiemployer Plan, that is subject to Title IV of ERISA and is sponsored or maintained by any Borrower or any ERISA Affiliate or to which any Borrower or any ERISA Affiliate contributes or has an obligation to contribute, or with respect to which any Borrower or any ERISA Affiliate may have any liability or in the case of a multiple employer or other plan described in Section 4064(a) of ERISA, has made contributions at any time during the immediately preceding six (6) plan years.

"Permits" shall mean all accreditations, certifications, provider or supplier numbers, registrations, certificates of authority, certificates of need, certificates of reimbursement, variances, qualifications, filings, consents, governmental licenses, authorizations, supplier numbers, registrations, permits, device authorizations and approvals, certificates, franchises, qualifications, accreditations, consents, approvals, listings, certificates, product clearances or approvals, marketing authorizations and all other licenses, authorizations, registrations, permits, consents and approvals or exemptions thereto required in connection with the conduct of any Loan Party's or any Subsidiary's business or to comply with any applicable Laws, including, without limitation, (a) any "Permits" (as defined prior to this clause (a)) issued or required under Healthcare Laws applicable to the business of any Loan Party or any of its Subsidiaries or necessary in the possession, ownership, warehousing, marketing, promoting, sale, leasing, labeling, furnishing, distribution or delivery of goods or services under Healthcare Laws applicable to the business (b) any "Permits" (as defined prior to clause (a)) issued by any Person from which any Loan Party has, as of the Closing Date, received an accreditation, any of the above issued or required under Healthcare Laws applicable to the ownership or operation of any business location of a Loan Party, and (c) establishment registrations, device listings, Investigational Device Exemptions (IDEs), 510(k) exemptions, 510(k) clearances, and PMA approvals, as those terms are defined in the FDCA and implementing regulations, and those issued by state governments, environmental protection agency permits, and any and all licenses, patents, trademarks and other intellectual property rights necessary for the conduct of any Loan Party's or Subsidiary's business.

"Permitted Acquisitions" means Investments consisting of an Acquisition by any Loan Party or any Subsidiary, provided that (i) the line of business of any acquired Person shall be substantially similar to, or reasonably ancillary, complementary or related to, or a reasonable extension, development or expansion of, the businesses conducted by Borrowers and their respective Subsidiaries on the Closing Date (as determined by the Borrowers in their reasonable discretion), (ii) the Property acquired (or the Property of the Person acquired) in such Acquisition is used or useful in substantially the same line of business as the Borrowers and their respective Subsidiaries were engaged in on the Closing Date, or in a business reasonably ancillary, complementary or related to, or a reasonable extension, development or expansion of, such business, and the Property is located (or if an Acquisition of Capital Stock, the Acquisition target is formed or otherwise organized) in the United States, (ii) the Administrative Agent shall have received not less than ten (10) days (or such shorter period as agreed to by the Administrative Agent in its sole discretion) prior notice of such Acquisition, which notice shall contain a summary, in reasonable detail, of the acquisition terms and conditions, including price, and Borrowers' projections prepared in connection with such Acquisition, (iv) at or priorthe Loan Parties shall take or cause to be taken with respect to the elosing acquisition of such Permitted Acquisition, the Administrative Agent shall be granted a first priority perfected Lien (subject to Permitted Liens) in the assets and Capital Stock (other than Excluded Property)

of such Acquisition target or Subsidiary and such Acquisition target or Subsidiary shall join this Agreement and the other Loan Documents as a Loan Party pursuant to the terms of Section 6.12, provided that nothing in this definition shall permit in any event the Acquisition of, or any Investment in respect of, an Excluded Subsidiary or an Excluded Foreign Subsidiary other than within the limits set forth in clause (x) belowing new Subsidiary, the actions required to be taken under Section 6.12 and Section 6.14, as applicable, (v) in the case of an Acquisition of the Capital Stock of another Person, the board of directors (or other comparable governing body) of such other Person shall have duly approved such Acquisition and such Acquisition shall not be hostile, (vi) the Borrower Representative shall have delivered to the Administrative Agent a Pro Forma Compliance Certificate demonstrating that, upon giving effect to such Acquisition and the incurrence of any Indebtedness related to such Acquisition on a Pro Forma Basis, (A) the Loan Parties would be in compliance with the financial covenants set forth in<u>Article 8</u> as of the most recent Fiscal Quarter for which the Borrowers have delivered financial statements pursuant to Section 6.01(a) or Section 6.01(b), as applicable, (B) the Borrowers' Consolidated Total Leverage Ratio does not exceed 2.703.00:1.00 as of the most recent Fiscal Quarter for which the Borrowers have delivered financial statements pursuant to Section 6.01(a) or Section 6.01(b), as applicable, and (C) no Default or Event of Default exists or would be caused by such Acquisition or the incurrence of any related Indebtedness, (vii) the representations and warranties made by the Loan Parties in each Loan Document shall be true and correct in all material respects (or, in all respects, if already qualified by materiality) at and as if made as of the date of such Acquisition (after giving effect thereto) except to the extent such representations and warranties expressly relate to an earlier date, (viii) immediately after giving effect to such Acquisition, Excess Liquidity shall be at least \$10,000,000[intentionally omitted], (ix) the Borrower Representative shall have delivered to the Administrative Agent a Collateral Assignment of Agreement with respect to the purchase agreement governing the Acquisition unless the same is freely assignable to Administrative Agent for the benefit of the Secured Parties, (x) the Total Consideration paid by the Loan Parties and any Subsidiaryin the aggregate prior to the Termination Date with respect to any individual Acquisition shall not exceed \$100,000,000, provided that300,000,000 without receiving prior written consent from the Required Lenders in accordance with Section 12.01, (xi) the Total Consideration reasonably allocable to all Acquisitions of, or any other Investments in respect of, all Excluded Foreign Subsidiaries shall not exceed \$5,000,000 in the aggregate following the Second Amendment Effective Date andprior to the Termination Date, (xi) with respect to the maximum, potential Earn-Out Obligations (if any) payable under any circumstance in connection with such Acquisition, such Earn-Out Obligations shall not exceed thirty-five percent (35.0%) of the Total Consideration paid in cash at closing for such Acquisitions and shall, at the request of the Administrative Agent, be subject to a Subordination Agreement acceptable to the Administrative Agent in its sole discretion (xii) the business and assets acquired by a Loan Party in such Acquisition shall be free and clear of all Liens (other than Permitted Liens);ii) [intentionally omitted], (xiii) except in the case of a Permitted PAP Acquisition, the acquisition target, division or line of business shall have Consolidated EBITDA (calculated for the twelve (12) consecutive calendar months most recently ended) equal to or greater than -\$1,000,000 (or less than -\$1,000,000 with prior written consent of the Administrative Agent), (xiv) except in the case of a Permitted PAP Acquisition, the Borrower Representative shall have delivered to the Administrative Agent historical financial information (including income statements, balance sheets and cash flows) covering at least the three (3) most recently ended Fiscal Years ended at least ninety (90) days prior to the proposed acquisition for which financial statements have been prepared for such acquisition target, division or line of business to be so acquired prior to the effective date of the acquisition or the entire financial history for such acquisition target, division or line of business to be so acquired, whichever period is shorter, (xv) except in the case of a Permitted PAP Acquisition, the Borrower Representative shall have delivered such financial and other information (including, with respect to any acquisition target with Consolidated EBITDA less than \$0, information to be provided by Borrowers demonstrating steps Borrowers intend to take (and a timeline with respect thereto) in order for such entity to have Consolidated EBITDA greater than \$0), together with, to the extent Total Consideration for such Acquisition exceeds \$7,500,00020,000 (or greater than \$20,000,000 with prior written consent of the Administrative Agent), a quality of earnings report and other third-party reports obtained by a Loan Party or

its Affiliates in connection with such Acquisition, in each case in form and results acceptable to Administrative Agent, with respect to each Person or any division or line of business, (xvi) each Loan Party shall have provided the Administrative Agent copies of all material notifications, reports, submissions, certifications, correspondence and other documents with respect to such acquired Person's initial or continued enrollment in any Government Reimbursement Program, (xvii) to the extent such Acquisition is with respect to a non-Wholly Owned Subsidiary (other than an Excluded Subsidiary), after giving effect to such Acquisition the aggregate Consolidated EBITDA attributable to all non-Wholly Owned Subsidiaries (other than those designated as Excluded Subsidiaries) does not exceed 7.50% of Consolidated EBITDA of the Loan Parties and their Subsidiaries for the last twelve (12) months ending on the last day of the last month that is also the end of a Fiscal Quarter for which financial statements of the type set forth in Section 6.01(a) or (b) have been delivered to the Administrative Agent, and (xviii) the Borrower Representative shall have delivered to the Administrative Agent (A) as soon as available, executed counterparts of the respective agreements, documents or instruments pursuant to which such Acquisition is to be consummated (including any related management, non-compete, employment, option or other material agreements), any schedules to such agreements, documents or instruments and all other material ancillary agreements, instruments and documents to be executed or delivered in connection therewith, and (B) to the extent required under the related acquisition agreement, all consents and approvals from applicable Governmental Authorities and Persons provided, further, to the extent that payment for such Acquisition is made solely from Net Cash Proceeds from the issuance or sale of Qualified Capital Stock of AdaptHealth Corp. (other than Cure Amounts), Borrower and the Loan Parties shall only be required to be in compliance with clauses (iii), (v), (vi), (vii) and (xvii) of this definition for such Acquisition to be deemed a Permitted Acquisition.

"Permitted Liens" means, at any time, Liens in respect of Property of the Loan Parties and their Subsidiaries permitted to exist at such time pursuant to the terms of Section 7.02.

"<u>Permitted PAP Acquisition</u>" means the Acquisition of the positive airway supplies business assets of any Person, consisting of inventory, patient files and referral source contact information, provided that (i) the purchase price thereof is less than or equal to \$1,200,000 and at least 100 active accounts are acquired thereby for each \$12,000 of consideration paid.

"Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"Plan" has the meaning set forth in Section 12.07(i)(iv)(B).

"Preferred Note" means collectively, each of the following: (a) that certain Promissory Note dated as of the date hereof in the amount of \$55,713,800 issued by Holdings to BMSB L.P. as purchaser, (b) that certain Promissory Note dated as of the date hereof in the amount of \$3,964,700 issued by Holdings to BlueMountain Summit Opportunities Fund II (US) L.P. as purchaser, (c) that certain Promissory Note dated as of the date hereof in the amount of \$22,190,200 issued by Holdings to BlueMountain Fursan Fund L.P. as purchaser and (d) that certain Promissory Note dated as of the date hereof in the amount of \$18,131,300 issued by Holdings to BlueMountain Foinaven Master Fund L.P. as purchaser.

"Preferred Note Cash Interest Payments" means payments of interest pursuant to Section 2(b)(i) of the Preferred Note.

"<u>Preferred Note PIK Interest Payment</u>" means payments of interest pursuant to Section 2(b)(ii) of the Preferred Note that Holdings has elected to pay in cash rather than pay in kind and increase the principal amount of the Preferred Note; <u>provided</u>, to the extent that any interest is elected by Holdings to be paid in kind pursuant to Section 2(b)(ii) of the Preferred Note, such shall be principal under the Preferred Note and any payment thereof shall not constitute a Preferred Note PIK Interest Payment.

"Proceedings" means any actual or threatened civil, equitable or criminal proceeding litigation, action, suit, claim, investigation (governmental or judicial or otherwise), dispute indictment or prosecution, pleading, demand or the imposition of any fine or penalty or similar matter.

"Products" means any devices or products manufactured, sold, leased, rented, developed, tested or marketed by any Loan Party or any of its Subsidiaries.

"Pro Forma Basis" means, for purposes of calculating any financial covenant (including for purposes of determining the Applicable Margin), that any Disposition (other than any voluntary Disposition described inclause (a), (b), (c) or (d) of Section 7.05), Involuntary Disposition (other than an Involuntary Disposition, the proceeds of which are intended to be and are reinvested with the time periods provided in Section 2.05(b)(ii)), any Restricted Payment, any Investment, any Disposition that results in a Loan Party or a Subsidiary ceasing to be a Subsidiary of Intermediate Holdings or any incurrence or assumption of Indebtedness shall be deemed to have occurred as of the first (1st) day of the four (4) Fiscal Quarter period most recently ended prior to the date of such transaction for which the Borrowers have delivered financial statements pursuant to Sections 6.01(a), or 6.01 (b) or 6.01(d). In connection with the foregoing, (a) with respect to any such Disposition or Involuntary Disposition (i) income statement and cash flow statement items (whether positive or negative) attributable to the Property or Person disposed of or so designated shall be excluded to the extent relating to any period occurring prior to the date of such transaction and (ii) Indebtedness which is retired shall be excluded and deemed to have been retired as of the first (1st) day of the applicable period and (b) with respect to any Investment (i) income statement items attributable to the Person or Property acquired or so designated shall be included to the extent relating to any period applicable in such calculations to the extent (A) such items are supported by financial statements or other information reasonably satisfactory to the Administrative Agent and Required Lenders and (B) such items are not otherwise included in such income statement items for the Loan Parties and their respective Subsidiaries in accordance with GAAP or in accordance with any defined terms set forth in Section 1.01; and (ii) any Indebtedness incurred or assumed by any Loan Party or any Subsidiary (including the Person or Property acquired) in connection with such transaction and any Indebtedness of the Person or Property acquired which is not retired in connection with such transaction (A) shall be deemed to have been incurred as of the first (1st) day of the applicable period and (B) if such Indebtedness has a floating or formula rate, shall have an implied rate of interest for the applicable period for purposes of this definition determined by utilizing the rate which is or would be in effect with respect to such Indebtedness as at the relevant date of determination.

"<u>Pro Forma Compliance Certificate</u>" means a certificate of a Responsible Officer of the Borrower Representative containing reasonably detailed calculations of the financial covenants set forth in <u>Article 8</u> as of the most recent Fiscal Quarter end for which the Loan Parties have delivered financial statements pursuant to <u>Section 6.01(a)</u> or <u>Section 6.01(b)</u> after giving effect to the applicable transaction on a Pro Forma Basis.

"<u>Properly Contested</u>" means, with respect to any obligation of a Loan Party, (a) the obligation is subject to a bona fide dispute regarding amount or the Loan Party's liability to pay; (b) the obligation is being contested in good faith by appropriate proceedings promptly instituted and diligently pursued; (c) appropriate reserves have been established in accordance with GAAP; (d) non-payment could not reasonably be expected to result in a Material Adverse Effect, nor result in forfeiture or sale of any assets of such Loan Party pending resolution of such contest proceedings and the payment of any liabilities resulting therefrom; (e) no Lien (other than a Permitted Lien) is imposed on assets of such Loan Party; and (f) if the obligation results from entry of a judgment or other order, such judgment or order is stayed pending appeal or other judicial review.

"Property" means any interest of any kind in any property or asset, whether real, personal or mixed, or tangible or intangible, including Capital Stock.

"Proposal Letter" means the Proposal Letter dated as of February 27, 2019, issued by CIT Bank, N.A. and accepted by the Borrower.

"<u>Pro Rata Share</u>" means, with respect to any Lender at any time, (a) with respect to such Lender's Revolving Commitment, Letter of Credit Liabilities and Swingline Exposures at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the Revolving Commitment of such Lender at such time and the denominator of which is the amount of the total Revolving Commitments at such time, <u>provided</u> that if commitments of each Lender to make Revolving Loans have been terminated pursuant to <u>Section 9.02</u>, then the Pro Rata Share of each Lender shall be determined based on the Pro Rata Share of such Lender's Revolving Exposure, and (b) with respect to the outstanding Term Loan(s) at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the principal amount of the outstanding Term Loan held by such Lender at such time and the denominator of which is the aggregate outstanding principal amount of the Term Loan(s) held by all Term Loan Lenders at such time. The initial Pro Rata Share of each Lender is set forth opposite the name of such Lender on<u>Schedule 2.01</u> or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"<u>PTE</u>" means a prohibited transaction class exemption issued by the U.S. Department of Labor, as any such exemption may be amended from time to time.

"Public Company Costs" means costs relating to compliance with the provisions of the Securities Act and the Exchange Act, in each case as applicable to companies with equity or debt securities held by the public, the rules of national securities exchange companies with listed equity or debt securities, directors' compensation, fees, indemnities and expense reimbursement, costs relating to investor relations, shareholder meetings and reports to shareholders.

"Qualified Capital Stock" of any Person means any Capital Stock of such Person that is not Disqualified Capital Stock.

"Qualified ECP Guarantor" means in respect of any Hedge Obligation, each Loan Party that has total assets exceeding 10,000,000 at the time the relevant guarantee or grant of the relevant security interest becomes effective with respect to such Hedge Obligation or such other Person as constitutes an "eligible contract participant" under the Commodity Exchange Act or any regulations promulgated thereunder and can cause another Person to qualify as an "eligible contract participant" at such time by entering into a keepwell under Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.

"Rating Agencies" has the meaning set forth in Section 12.08(b).

"Real Property" means the real estate listed on Schedule 5.20(a), and any other real estate owned or leased after the Closing Date.

"Recall" means a Person's Removal or Correction of a marketed product that the FDA considers to be in violation of the laws it administers and against which the FDA would initiate legal action, e.g., seizure.

"Recipient" means (a) the Administrative Agent, (b) any Lender, (c) any L/C Issuer and (d) any Support Provider, as applicable.

"Refinancing Facility" has the meaning set forth in Section 2.17.

"Refinancing Revolving Facility" has the meaning set forth in Section 2.17.

"Refinancing Revolving Loans" has the meaning set forth in Section 2.17.

"Refinancing Term Facility" has the meaning set forth in Section 2.17.

"Refinancing Term Loans" has the meaning set forth in Section 2.17.

"Register" has the meaning set forth in Section 12.07(c).

"Registrar" has the meaning set forth in Section 12.07(c).

"<u>Regulation U</u>" and "<u>Regulation X</u>" mean, respectively, Regulations U and X of the Board of Governors of the Federal Reserve System or any successor, as the same may be amended or supplemented from time to time.

"Reimbursement Loan" has the meaning given to such term in Section 2.03(d)(ii).

"<u>Reimbursement Obligation</u>" means the Borrowers' obligation to immediately reimburse or pay all Unreimbursed Amounts with respect to all Letters of Credit and Support Agreements, as more fully described in <u>Section 2.03(d)</u>.

"Removal" means the physical removal of a device from its point of use to some other location for repair, modification, adjustment, relabeling, destruction, or inspection.

"Reportable Event" means any of the events set forth in Section 4043(c) of ERISA, other than events for which the thirty-day notice period has been waived.

"Representatives" has the meaning set forth in Section 12.08(b).

"Request for Credit Extension" means (a) with respect to a Borrowing, a Loan Notice or Swingline Loan Notice, as the case may be, and (b) with respect to an L/C Credit Extension, a Notice of L/C Credit Event.

"<u>Required Lenders</u>" means (a) at any time there are two or fewer non-Defaulting Lenders, Lenders holding in the aggregate one hundred percent (100%) of (i) the Revolving Commitments and the outstanding Term Loan(s) or (ii) if the Revolving Commitments have been terminated, the Revolving Exposures and the outstanding Term Loan(s) or (b) at any other time, Lenders holding in the aggregate more than fifty percent (50%) of (i) the Revolving Commitments and the outstanding Term Loan(s) or (ii) if the Revolving Commitments have been terminated, the Revolving Exposures and outstanding Term Loan(s); provided, however, as to the threshold requirements disclosed above in clause (b), the relevant Required Lenders threshold requirements may not be satisfied solely by a Lender and other Lenders which are Affiliates of such Lender, if at such time there is another Lender which is not an Affiliate of such Lender. The Revolving Commitments (or, if the Revolving Commitments have terminated, the Revolving Exposure) and the outstanding Term Loan(s) held or deemed held by any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders, including any minimum head-count.

"Required Permit" means a Permit (a) required under applicable Laws applicable to the business of any Loan Party or any of its Subsidiaries or necessary in the manufacturing, importing, exporting,

possession, ownership, warehousing, marketing, promoting, sale, leasing, labeling, furnishing, distribution or delivery of goods or services under applicable Laws applicable to the business of any Loan Party or any of its Subsidiaries or any Device Application (including without limitation, at any point in time, all licenses, approvals and permits issued by the FDA or any other applicable Governmental Authority necessary for the testing, manufacture, marketing or sale of any Product by any Loan Party or any of its Subsidiaries as such activities are being conducted by such Person with respect to such Product at such time), and (b) issued by any Person from which any Loan Party or any of its Subsidiaries has received an accreditation. This shall include all state government registrations and certifications applicable to the operation of the business of any Loan Party or any of its Subsidiaries which are necessary for the manufacture or repair of respiratory therapy healthcare products and related Products.

"<u>Required Revolving Lenders</u>" means, (a) at any time there are two or fewer Revolving Lenders who are non-Defaulting Lenders, Revolving Lenders holding in the aggregate one hundred percent (100%) of (i) the Revolving Commitments or (ii) if the Revolving Commitments have been terminated, the Revolving Exposures, or (b) at any other time, Revolving Lenders holding in the aggregate more than fifty percent (50%) of (i) the Revolving Commitments, or (ii) if the Revolving Commitments have been terminated, the Revolving Exposures; <u>provided</u>, <u>however</u>, as to the threshold requirements described above in <u>clause (b)</u>, the relevant Required Revolving Lenders threshold may not be satisfied solely by a Revolving Lender and other Revolving Lenders which are Affiliates of such Revolving Lender if at such time there is another Revolving Lender which is not an Affiliate of such Revolving Lender. The Revolving Commitments (or, if the Revolving Commitments have terminated, the Revolving Exposure) held or deemed held by any Defaulting Lender shall be excluded for purposes of making a determination of Required Revolving Lenders, including any minimum head-count.

"<u>Required Term Loan Lenders</u>" means, (a) at any other time there are two or fewer Term Loan Lenders who are non-Defaulting Lenders, Term Loan Lenders holding in the aggregate one hundred percent (100%) of the outstanding Term Loan(s), or (b) at any other time, Term Loan Lenders holding in the aggregate more than fifty percent (50%) of outstanding Term Loan(s); <u>provided</u>, <u>however</u>, as to the threshold requirements described above in <u>clause (b)</u>, the relevant Required Term Loan Lenders threshold requirement may not be satisfied solely by a Term Loan Lender and other Term Loan Lenders which are Affiliates of such Term Loan Lender, if at such time there is another Term Loan Lender which is not an Affiliate of such Term Loan Lender. The outstanding Term Loan(s) held or deemed held by any Defaulting Lender shall be excluded for purposes of making a determination of Required Term Loan Lenders, including any minimum head-count.

"<u>Responsible Officer</u>" means the chief executive officer, president, chief financial officer or treasurer of a Loan Party. Any document delivered hereunder that is executed by a Responsible Officer of a Loan Party shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of such Loan Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Loan Party.

"Restricted Payment" means (a) any dividend or other distribution, direct or indirect, on account of any shares (or equivalent) of any class of Capital Stock of any Loan Party or any of its Subsidiaries, now or hereafter outstanding, (b) any redemption, retirement, sinking fund or similar payment, purchase or other acquisition for value, direct or indirect, of any shares (or equivalent) of any class of Capital Stock of any Loan Party or any of its Subsidiaries, now or hereafter outstanding, (c) any payment made to retire, or to obtain the surrender of, any outstanding warrants, options or other rights to acquire shares of any class of Capital Stock of any Loan Party or any of its Subsidiaries, now or hereafter outstanding, (d) any payment from any Loan Party to Holdings not expressly permitted by<u>Section 7.06</u>, and (e) the payment by any Loan Party or any of its Subsidiaries of any management, advisory or consulting fee to the Sponsor or an Affiliate of the Sponsor or the payment of any extraordinary salary, bonus or other form of compensation to any

Person who is directly or indirectly a significant partner, shareholder, owner or executive officer of the Sponsor or Affiliate of the Sponsor, or any such Person, including, without limitation, pursuant to any management fee agreements.

"<u>Revolving Availability</u>" means, at any time beginning on the first (1st) Business Day following the Closing Date and ending five (5) Business Days prior to the Revolving Loan Maturity Date, an amount equal to (a) the total Revolving Commitments less (b) the total Revolving Exposures at such time, <u>provided</u>, that Revolving Availability shall equal zero Dollars (\$0) while any Default or Event of Default exists and remains outstanding.

"Revolving Commitment" means, as to each Lender, its obligation to (a) make Revolving Loans to the Borrowers pursuant to <u>Sections</u> 2.01 and 2.03 (for Reimbursement Loans) and (b) and to acquire participations in Letter of Credit Liabilities and Swingline Loans pursuant to <u>Section 2.03</u> and <u>Section 2.04</u>, respectively, in an aggregate principal amount at any one time outstanding not to exceed the amount set forth opposite such Lender's name on <u>Schedule 2.01</u> or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable, as such amount may be adjusted from time to time in accordance with this Agreement. The initial aggregate amount of the Revolving Commitments of all Revolving Lenders is \$75,000,000.

"Revolving Commitment Increase" has the meaning specified in Section 2.15(a).

"<u>Revolving Exposure</u>" means, with respect to any Lender at any time, the sum of (a) the outstanding principal amount of such Lender's Revolving Loans, (b) its Pro Rata Share of outstanding Letter of Credit Liabilities and (c) its Swingline Exposure at such time.

"<u>Revolving Lenders</u>" means, as of any date of determination, Lenders having a Revolving Commitment, or after the Revolving Commitments have terminated, Lenders holding any portion of the outstanding Revolving Loan.

"Revolving Loan" has the meaning specified in Section 2.01(a).

"<u>Revolving Loan Account</u>" means the loan account on the Administrative Agent's books, in the name of the Borrower Representative on behalf of the Borrowers, in which the Borrowers will be charged with all Obligations when due or incurred by the Administrative Agent or any Lender.

"Revolving Loan Maturity Date" means March 20, 2024.

"Revolving Note" has the meaning specified in Section 2.11(a).

"S&P" means Standard & Poor's Ratings Services, a division of The McGraw Hill Companies, Inc. and any successor thereto.

"Sale and Leaseback Transaction" means, with respect to any Loan Party or any Subsidiary, any arrangement, directly or indirectly, with any Person whereby such Loan Party or such Subsidiary shall sell or transfer any property, real or personal, used or useful in its business, whether now owned or hereafter acquired, and thereafter rent or lease such property or other property that it intends to use for substantially the same purpose or purposes as the property being sold or transferred.

"<u>Sanctioned Entity</u>" means (a) a country or a government of a country, (b) an agency of the government of a country, (c) an organization directly or indirectly controlled by a country or its government, or (d) a Person resident in or determined to be resident in a country, in each case, that is subject

to a country sanctions program administered and enforced by the U.S. Department of State, OFAC, or other relevant United States sanctions authority.

"Sanctioned Person" means a Person named on the OFAC-maintained list of "Specially Designated Nationals" (as defined by OFAC) or a Person otherwise subject to or the target of any Sanctions.

"Sanctions" means the country or list based economic and trade sanctions administered and enforced by OFAC, the U.S. Department of State or other relevant United States sanctions authority.

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"Second Amendment" means Amendment No. 2 to Third Amended and Restated Credit and Guaranty Agreement dated as of November 8, 2019, by and among the Borrowers, the other Loan Parties party thereto, the Administrative Agent, and the Lenders.

"Second Amendment Effective Date" means November 8, 2019.

"Second Amendment Transactions" means, collectively, (a) the execution, delivery and performance by the Loan Parties of the Loan Documents dated as of the Second Amendment Effective Date to which they are a party and (b) the transactions contemplated pursuant to that certain Merger Agreement (as defined in the Second Amendment), the Proxy Statement (Schedule 14A) filed with the SEC on October 23, 2019, and related documents.

"Secured Hedge Agreement" means any Hedge Agreement required or permitted by this Agreement that is entered into by and between a Loan Party and a Secured Hedge Provider (whether entered into before or after the Closing Date).

"Secured Hedge Provider" means (a) the Administrative Agent or any of its Affiliates (or any Person who was an Affiliate of the Administrative Agent at the time such Person entered into a Secured Hedge Agreement), and (b) any other Lender or Affiliate of a Lender (or any Person who was a Lender or an Affiliate of a Lender at the time such Person entered into a Secured Hedge Agreement), in each case which is approved in writing by the Administrative Agent as a Secured Hedge Provider, each in their capacity as a counterparty to a Secured Hedge Agreement; provided, in the case of a Secured Hedge Agreement with a Person who is no longer a Lender (or Affiliate of a Lender), such Person shall be considered a Secured Hedge Provider only through the stated termination date (without extension or renewal) of such Secured Hedge Agreement and provided further that for any of the foregoing to be included as a "Secured Hedge Agreement" on any date of determination by the Administrative Agent, the applicable Hedge Party (other than the Administrative Agent, or an Affiliate of the Administrative Agent) must have delivered a written designation notice, on the form proscribed by Administrative Agent, to the Administrative Agent prior to such date of determination.

"Secured Parties" means, collectively, the Administrative Agent, an Affiliate of the Administrative Agent who provides Banking Services, the Arranger, the Book Runner, the Lenders, an Affiliate of a Lender who provides Banking Services, the Support Provider, the L/C Issuer (solely to the extent such L/C Issuer also is the Administrative Agent or a Lender), and the Secured Hedge Providers.

"Securities Account Control Agreement" means an agreement, among a Loan Party, a securities intermediary, and the Administrative Agent, which agreement is in a form reasonably acceptable to the Administrative Agent and which provides the Administrative Agent with "control" (as such term is used in

Articles 8 and 9 of the UCC) over the securities account(s) described therein, as the same may be as amended, modified, extended, restated, replaced, or supplemented from time to time.

# "Securities Act" means the Securities Act of 1933 and the rules and regulations of the SEC promulgated thereunder, as amended.

"Securitization Transaction" means any financing transaction or series of financing transactions (including factoring arrangements) pursuant to which any Borrower or any Subsidiary may sell, convey or otherwise transfer, or grant a security interest in, accounts, payments, receivables, rights to future lease payments or residuals or similar rights to payment to a special purpose subsidiary or affiliate of any Person.

"Security Agreement" means the Third Amended and Restated Security and Pledge Agreement dated as of the Closing Date executed in favor of the Administrative Agent by each of the Loan Parties which is a party thereto, as the same may be as amended, modified, extended, restated, replaced or supplemented from time to time.

"Solvent" means, with respect to any Person on a particular date, that on such date (a) the fair value of the assets of such Person exceed its liabilities, including contingent liabilities, (b) the present fair saleable value of the assets of such Person is not less than the amount that will be required to pay the probable liabilities of such Person or its debts as they become absolute and matured, (c) the remaining capital of such Person is not unreasonably small to conduct its business, and (d) such Person will not have incurred debts and does not have the present intent to incur debts, beyond its ability to pay such debts as they mature. In computing the amount of contingent liabilities of any Person on any date, such liabilities shall be computed at the amount that, in the judgment of the Administrative Agent, in light of all facts and circumstances existing at such time, represents the amount of such liabilities that reasonably can be expected to become actual or matured liabilities.

#### "Specified Equity Contribution" has the meaning set forth in Section 9.05.

"Specified Laws" means all applicable Laws relating to the operation of private label and other medical device product distributions, and the possession, control, warehousing, marketing, sale, lease, rental, and distribution of medical devices, including without limitation, the FDCA, Current Good Manufacturing Practices (CGMP) requirements of the Quality System regulation for medical devices, as specified in Title 21, Code of Federal Regulations, Part 820 (21 C.F.R. 820), the Occupational Health and Safety Act (29 U.S.C. § 651 et seq.), any laws pertaining to the storage and disposal of biomedical and other hazardous waste, and any implementing regulations to any of the foregoing or other applicable state or federal laws. This shall include all guidelines and standards established by state government agencies for the manufacture or repair of respiratory therapy healthcare products and related Products.

#### "Sponsor" means Quadrant Management, Inc.

"Subordination Agreement(s)" means (a) an agreement (in form and substance satisfactory to the Administrative Agent) among any Loan Party, a subordinating creditor of such Loan Party and the Administrative Agent, on behalf of the Secured Parties, pursuant to which, among other things, (i) Indebtedness is subordinated to the prior payment and satisfaction of the Obligations and (ii) the subordinating creditor agrees not to require, accept or maintain any Lien(s) on any assets of the Loan Parties and their Subsidiaries, except as the Administrative Agent may expressly permit hereunder, and (b) any note, indenture, note purchase agreement or similar instrument or agreement, pursuant to which the indebtedness evidenced thereby or issued thereunder is subordinated to the Obligations by the express terms of such note, indenture, note purchase agreement or similar instrument or agreement, in each case in form and substance satisfactory to the Administrative Agent.

"Subpoena" has the meaning specified in Section 6.03(1).

"<u>Subsidiary</u>" of a Person means a corporation, partnership, limited liability company or other business entity of which a majority of the shares of Capital Stock having ordinary voting power for the election of directors or other governing body (other than Capital Stock having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly or indirectly, through one or more intermediaries, or both, by such Person. Unless otherwise specified, all references herein to a "Subsidiary" or to "Subsidiaries" shall refer to a Subsidiary or Subsidiaries of the Loan Parties.

"Support Agreement" means a guaranty, reimbursement agreement or other arrangement or agreement whereby a Support Provider agrees to guaranty or otherwise provide for the reimbursement of drawings under a Letter of Credit on behalf of the BorrowerBorrowers or another party obligated to make such reimbursement.

"Support Provider" means the Administrative Agent or one of its Affiliates who agrees (in its sole discretion) to provide a Support Agreement.

"Supported Letter of Credit" means a Letter of Credit issued by an L/C Issuer in reliance on one or more Support Agreements.

"Survey" means a survey certified to the Administrative Agent and the Title Insurance Company in a manner reasonably satisfactory to each of the Administrative Agent and the Title Insurance Company, dated a date reasonably satisfactory to each of the Administrative Agent and the Title Insurance Company by an independent professional licensed land surveyor, which surveys shall be sufficient to delete any standard printed survey exception contained in the applicable title policy and be made in accordance with the Minimum Standard Detail Requirements for Land Title Surveys jointly established and adopted by the American Land Title Association and the American Congress on Surveying and Mapping in 2005 with all items from Table A thereof completed, except for Nos. 5 and 12.

"Swingline Exposure" means, at any time, the aggregate principal amount of all Swingline Loans outstanding at such time. The Swingline Exposure of any Lender at any time shall be its Pro Rata Share of the total Swingline Exposure at such time.

"Swingline Lender" means CIT, in its capacity as lender of Swingline Loans hereunder.

"Swingline Loan" means a Loan made pursuant to Section 2.04.

"Swingline Loan Notice" means a notice of a Borrowing of Swingline Loans which, if in writing, shall be substantially in the form of Exhibit A-2.

"Swingline Loan Sublimit" means an amount equal to the lesser of (a) the total Revolving Commitments and (b) \$5,000,000. The Swingline Loan Sublimit is part of, and not in addition to, the total Revolving Commitments.

"Swingline Note" has the meaning specified in Section 2.11.

"Synthetic Lease" means any synthetic lease, tax retention operating lease, off-balance sheet loan or similar off-balance sheet financing arrangement whereby the arrangement is considered borrowed money indebtedness for tax purposes but is classified as an operating lease or does not otherwise appear on a balance sheet under GAAP.

"<u>Tax Distributions</u>" means, for any taxable year for which Holdings is treated under the Internal Revenue Code as a partnership for income tax purposes or otherwise similarly disregarded under the Internal Revenue Code for income tax purposes, dividends and/or distributions paid by Holdings to its owners in an amount not to exceed the product of (i) taxable income related to such owners' ownership interest in Holdings <u>multiplied by</u> (ii) the sum of the highest marginal individual federal and state income tax rates in any state in which any owner resides which were applicable in such taxable year.

"Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Term Loan Commitment" means the Initial Term Loan Commitment and the Delayed - Draw Term Loan Commitment.

"Term Loan Increase" has the meaning specified in Section 2.15(a).

"Term Loan Lender" means, as of any date of determination, (a) any Lender holding an Initial Term Loan Commitment or any portion of the then-outstanding Initial Term Loan and (b) any Lender holding a Delayed – Draw Term Loan Commitment or any portion of the then outstanding Delayed – Draw Term Loan.

"Term Loan Maturity Date" means March 20, 2024.

"Term Loans" has the meaning specified in Section 2.01(c).

"Term Note" has the meaning set forth in Section 2.11.

"Termination Date" means the date that (a) all Obligations (other than contingent obligations in respect of Letters of Credit, Secured Hedge Agreements and Banking Services Obligations) have been paid in full in cash, (b) no commitments or other obligations of any Lender to provide funds to the Borrowers remain outstanding, (c) no Lender Letter of Credit or Supported Letter of Credit remains outstanding (or, to the extent outstanding, such Letters of Credit have been Cash Collateralized as provided in<u>Section 2.03(g)</u>), (d) to the extent requested by the Administrative Agent, receipt by the Secured Parties of customary liability releases from the Loan Parties in form and substance reasonably acceptable to the Administrative Agent and (e) all contingent obligations have been cash collateralized with Administrative Agent in a manner and amounts reasonably acceptable to Administrative Agent.

"<u>Termination Value</u>" means, in respect of any one or more Hedge Agreements, after taking into account the effect of any legally enforceable netting agreement relating to such Hedge Agreements, (a) for any date on or after the date such Hedge Agreements have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Hedge Agreements, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Hedge Agreements (which may include a Lender or any Affiliate of a Lender).

"<u>Test Period</u>" means, as of any date of determination, the most recently completed four (4) consecutive Fiscal Quarters of Intermediate Holdings ending on or prior to such date for which financial statements have been or are required to be delivered pursuant to <u>Section 6.01(a)</u> or <u>Section 6.01(b)</u>.

"<u>Third Party Payor</u>" means a Government Reimbursement Program including Medicare, Medicaid, TRICARE, and other state or federal health care program, Blue Cross and/or Blue Shield, private insurers, managed care plans and any other person or entity which presently or in the future maintains Third Party Payor Programs.

"<u>Third Party Payor Programs</u>" means all payment and reimbursement programs sponsored by a Third Party Payor, in which a Loan Party participates.

"Title Insurance Company" means a title insurance company satisfactory to the Administrative Agent.

"<u>Total Consideration</u>" means, with respect to any Acquisition, all cash and non-cash consideration, including the amount of Indebtedness assumed by the buyer and the amount of Indebtedness evidenced by notes issued by the buyer to the seller, the maximum amount payable in connection with any deferred purchase price obligation (including any Earn-Out Obligation) and the value of any Capital Stock of any Loan Party issued to the seller in connection with such Acquisition.

"Transaction Documents" means the Note and Unit Purchase Agreement, the Preferred Note, Holdings Operating Agreement and any documents entered into in connection therewith.

"Transactions" means the transactions contemplated under the Transaction Documents.

"Transport and Disposal Agreement" means an agreement for the transport and disposal of hazardous wastes in accordance with all applicable Laws, as the same may be amended, supplemented, restated, modified or replaced from time to time.

"TRICARE" means the program administered pursuant to 10 U.S.C. Section 1071 et. seq., Sections 1320a-7 and 1320a-7a of Title 42 of the United States Code and the regulations promulgated pursuant to such statutes.

"Type" means, with respect to any Loan, its character as a Base Rate Loan or a LIBOR Loan.

"UCC" means the Uniform Commercial Code as in effect in any applicable jurisdiction.

"<u>UCP</u>" means, with respect to any commercial Letter of Credit, the "Uniform Customs and Practice for Documentary Credits", as most recently published by the International Chamber of Commerce.

"Uncertificated Securities Control Agreement" means an agreement in form and substance satisfactory to the Administrative Agent among the Administrative Agent, a Loan Party and a pledgor of uncertificated securities which provides the Administrative Agent with "control" (as such term is used in Articles 8 and 9 of the UCC) of such uncertificated securities (as defined in the UCC).

"<u>Unfunded Pension Liability</u>" means the excess of a Pension Plan's benefit liabilities under Section 4001(a)(16) of ERISA, over the current value of that Pension Plan's assets, determined in accordance with the assumptions used for funding the Pension Plan pursuant to <u>Sections 412 and 430</u> of the Internal Revenue Code for the applicable plan year.

"<u>United States</u>" and "<u>U.S.</u>" mean the United States of America.

"<u>Unreimbursed Amount</u>" means the amount of any drawing under a Letter of Credit or payment under a Support Agreement which has not yet been reimbursed by the Borrowers (through direct payment or by the making of a Revolving Loan).

"U.S. Person" means any Person that is a "United States Person" as defined in Section 7701(a)(30) of the Internal Revenue Code.

"U.S. Tax Compliance Certificate" has the meaning assigned to such term in Section 3.01(f).

"Verus Merger" has the meaning set forth in the definition of "Verus Merger Agreement".

"Verus Merger Agreement" means that certain Agreement and Plan of Merger, dated as of May 17, 2018 (the "Verus Merger Agreement"), by and among Holdings, as purchaser, Medstar Surgical & Breathing Equipment, Inc., a New York corporation and indirect wholly owned subsidiary of Holdings ("Merger Sub Parent"), Medstar Acquisition Co, Inc., a Delaware corporation and wholly owned subsidiary of Merger Sub Parent (the "Verus Merger Sub"), Verus Healthcare, Inc., a Delaware corporation ("Verus Company"), the shareholders of Verus Company signatory thereto and Verus Note Holding Company, LLC, a Tennessee limited liability company, in its capacity as the Shareholder Representative, pursuant to which Verus Merger Sub will merge (the "Verus Merger") with and into Verus Company with (a) the purchase price being paid by Holdings as set forth therein and (b) Verus Company surviving as a Wholly Owned Subsidiary of Merger Sub Parent.

"Voluntary Termination" shall have the meaning set forth in the definition of "Material Adverse Effect".

"<u>Weighted Average Life to Maturity</u>" means, when applied to any Indebtedness as of any date of determination, the number of years obtained by dividing: (a) the sum of the products obtained by multiplying (i) the amount of each then remaining installment, sinking fund, serial maturity or other required payments of principal, including payment at final maturity, in respect thereof, by (ii) the number of years (calculated to the nearest one-twelfth) that will elapse between such date and the making of such payment by (b) the then outstanding principal amount of such Indebtedness.

"Wholly Owned Subsidiary" means any Person one hundred percent (100%) of whose Capital Stock is at the time owned by a Loan Party directly or indirectly through other Persons one hundred percent (100%) of whose Capital Stock is at the time owned, directly or indirectly, by such a Loan Party.

"Withholding Agent" means any Loan Party and the Administrative Agent.

"<u>Write-Down and Conversion Powers</u>" means, with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which Write-Down and Conversion Powers are described in the EU Bail-In Legislation Schedule.

**1.02** <u>Other Interpretive Provisions</u> With reference to this Agreement and each other Loan Document, unless otherwise specified herein or in such other Loan Document:

(a) The definitions of terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "<u>include</u>," "<u>includes</u>" and "<u>including</u>" shall be deemed to be followed by the phrase "<u>without limitation</u>." The word "<u>will</u>" shall be construed to have the same meaning and effect as the word "<u>shall</u>." Unless the context requires otherwise, (i) any definition of or

reference to any agreement, instrument or other document (including any Organization Document) shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein or in any other Loan Document), (ii) any reference herein to any Person shall be construed to include such Person's successors and assigns, (iii) the words "herein," "hereof" and "hereunder," and words of similar import when used in any Loan Document, shall be construed to refer to such Loan Document in its entirety and not to any particular provision thereof, (iv) all references in a Loan Document to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, the Loan Document in which such references appear, (v) any reference to any law shall include all statutory and regulatory provisions consolidating, amending, replacing or interpreting such law and any reference to any law or regulation shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time to time, and (vi) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all real and personal property and tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

(b) In the computation of periods of time from a specified date to a later specified date, the word "<u>from</u>" means "<u>from</u> and <u>including</u>," the words "<u>to</u>" and "<u>until</u>" each mean "<u>to but excluding</u>," and the word "<u>through</u>" means "<u>to and including</u>."

(c) Section headings herein and in the other Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Loan Document.

# 1.03 Accounting Terms.

(a) Except as otherwise specifically prescribed herein, all accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Audited Pre-Closing Financial Statements. Notwithstanding any other provision contained herein, any obligations relating to a lease that was accounted for by such Person as an operating lease as of the Closing Date and any similar lease entered into after the Closing Date by such Person shall be accounted for as obligations relating to an operating lease and not as obligations relating to a Capital Lease; provided, however, that the Borrower Representative may elect, with notice to the Administrative Agent to treat operating leases as Capital Leases in accordance with GAAP as in effect from time to time and, upon such election, and upon any subsequent change to GAAP therefor, the parties will enter into negotiations in good faith in an effort to preserve the original intent of the financial covenants set forth herein (it being understood and agreed that the treatment of operating leases be interpreted on the basis of GAAP as in effect on the Closing Date until such election shall have been withdrawn or such provision amended in accordance herewith).

(b) Together with each Compliance Certificate, the Borrower Representative will provide a written summary of any changes in GAAP that materially impact the calculation of the financial covenants in <u>Article 8</u> contained in such Compliance Certificate. If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and any of the Borrowers, the Administrative Agent or the Required Lenders shall so request, the Administrative Agent, the Lenders and the Borrower Representative on behalf of the Borrowers shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP, provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Borrower Representative shall provide to the

Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

(c) Notwithstanding the above, all calculations of the financial covenants in <u>Article 8</u> (including for purposes of determining compliance with such financial covenants) shall be made on a Pro Forma Basis; however, no Specified Equity Contribution shall be included in any calculation made hereunder on a Pro Forma Basis. Further, no Specified Equity Contribution shall be included in the calculation of Excess Cash Flow.

(d) All financial statements delivered hereunder shall be prepared without giving effect to any election under Statement of Financial Accounting Standards Accounting Standards Codification No. 825 – Financial Instruments, or any successor thereto (including pursuant to the Accounting Standards Codification) (or any similar accounting principle) permitting a Person to value its financial liabilities at the fair value thereof.

**1.04** <u>Rounding</u>. Any financial ratios required to be maintained by the Borrowers pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

**1.05** <u>Times of Day</u>. Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).

**1.06** <u>Letter of Credit Amounts</u>. Unless otherwise specified, all references herein to the amount of a Letter of Credit at any time shall be deemed to mean the maximum face amount of such Letter of Credit after giving effect to all increases thereof contemplated by such Letter of Credit or the Issuer Document related thereto, whether or not such maximum face amount is in effect at such time.

**1.07** <u>Financial Covenant Defaults</u>. For the purposes of <u>Sections 9.01</u> and <u>4.02</u>, a breach of a covenant contained in <u>Section 8.01</u> shall be deemed to have occurred as of any date of determination by the Administrative Agent and as of the last day of any specified period of measurement regardless of whether or when the financial statements reflecting such breach are delivered to the Administrative Agent.

**1.08** <u>Divisions</u>. For all purposes under the Loan Documents, in connection with any division or plan of division under Delaware law (or any comparable event under a different jurisdiction's laws): (a) if any asset, right, obligation or liability of any Person becomes the asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) if any new Person comes into existence, such new Person shall be deemed to have been organized on the first date of its existence by the holders of its Capital Stock at such time.

#### **ARTICLE 2**

# THE COMMITMENTS AND CREDIT EXTENSIONS

#### 2.01 <u>Loans</u>.

(a) <u>Revolving Loans</u>. Subject to the terms and conditions set forth herein, each Revolving Lender severally (and neither jointly nor jointly and severally) agrees to make loans to the Borrower Representative on behalf of the Borrowers (each such loan a '<u>Revolving Loan</u>'') in Dollars from time to time on any Business Day during the Availability Period in an aggregate amount not to exceed at

any time outstanding the amount of such Revolving Lender's Revolving Commitment, provided, that after giving effect to any Borrowing of Revolving Loans, (i) the total Revolving Exposure of all Revolving Lenders shall not exceed the total Revolving Commitment, and (ii) the total Revolving Exposure of each Revolving Lender shall not exceed such Revolving Lender's Revolving Commitment, and (iii) the total Revolving Exposure of all Revolving Lenders to be used for working capital, to make Capital Expenditures, and for other general corporate purposes (other than Permitted Acquisitions and fees and transaction costs associated with Permitted Acquisitions) shall not exceed \$25,000,000. Within the limits of each Revolving Lender's Revolving Commitment, and subject to the other terms and conditions hereof, the Borrower Representative on behalf of the Borrowers may borrow under this Section 2.01(a), prepay under Section 2.05, and re-borrow under this Section 2.01(a). The Revolving Loans may be Base Rate Loans or LIBOR Loans, as further provided herein. No Borrowing of Revolving Loans shall be made on the Closing Date.

(b) <u>Initial Term Loan</u> Subject to the terms and conditions set forth herein, each Term Loan Lender severally agrees to fund its Pro Rata Share of a term loan to the Borrower Representative on behalf of the Borrowers (the <u>Initial Term Loan</u>") on the Closing Date in an aggregate amount not to exceed such Term Loan Lender's Initial Term Loan Commitment, provided, that after giving effect to any Borrowing under the Initial Term Loan, the outstanding amount of the Initial Term Loan shall not exceed the total Initial Term Loan Commitments. Amounts repaid or prepaid on the Initial Term Loan may not be re-borrowed. The Initial Term Loan on the Closing Date shall be Base Rate Loans, as further provided hereinprovided, however, all Borrowings of the Initial Term Loan on the Closing Date shall be Base Rate Loans. Any portion of the Initial Term Loan that is repaid or prepaid may not be re-borrowed. Upon making the Initial Term Loan, each Term Loan Commitment shall be reduced to zero.

(c) <u>Delayed – Draw Term Loan</u> Subject to the terms and conditions set forth herein, each Term Loan Lender severally agrees to fund its Pro Rata Share of a term loan to the Borrower Representative on behalf of the Borrowers (the <u>Delayed – Draw Term Loan</u>," and together with the Initial Term Loan, the "<u>Term Loans</u>") during the Delayed – Draw Term Loan Availability Period in an aggregate amount not to exceed such Term Loan Lender's Delayed – Draw Term Loan Commitment;<u>provided</u>, that after giving effect to any Borrowing under the Delayed – Draw Term Loan, the outstanding amount of the Delayed – Draw Term Loan shall not exceed the total Delayed – Draw Term Loan Commitments. Amounts repaid or prepaid on the Delayed – Draw Term Loan may not be re-borrowed. The Delayed – Draw Term Loan may consist of Base Rate Loans or LIBOR Loans, as further provided herein. No more than eight (8) Delayed - Draw Term Loan Borrowings may be made during the Delayed - Draw Term Loan Availability Period. Each Borrowing of Delayed - Draw Term Loans shall be in a principal amount of \$2,000,000 or a whole multiple of \$100,000 in excess thereof and may include funding for one or more Permitted Acquisitions occurring on or prior to the date thereof.

# 2.02 Borrowings, Conversions and Continuations of Loans.

(a) Each Borrowing, each conversion of Loans from one Type to the other, and each continuation of LIBOR Loans shall be made upon the Borrower Representative's irrevocable notice (and if in writing, in the form of the Loan Notice) to the Administrative Agent, which may be delivered by telephone or e-mail request (or such other means as may be agreed upon by the Administrative Agent in its sole discretion). Each such notice must be received by the Administrative Agent not later than 10:00 a.m. (i) three (3) Business Days prior to the requested date of any Borrowing of, conversion to or continuation of LIBOR Loans or of any conversion of LIBOR Loans to Base Rate Loans, and (ii) one (1) Business Day prior to the requested date of any Borrowing of Base Rate Loans (or any conversion to Base Rate Loans). Each telephonic notice by the Borrower Representative pursuant to this<u>Section 2.02(a)</u> must be confirmed promptly by delivery to the Administrative Agent of a written Loan Notice, appropriately completed and

executed by a Responsible Officer of the Borrower Representative. Subject to<u>Section 2.03(d)</u> with respect to Reimbursement Loans, each Borrowing (subject to <u>Section 2.01(c)</u> for Delayed - Draw Term Loans) of, conversion to or continuation of LIBOR Loans shall be in a principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof. Each Borrowing (subject to <u>Section 2.01(c)</u> for Delayed -Draw Term Loans) of or conversion to Base Rate Loans shall be in a principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof. Each Loan Notice pursuant to this <u>Section 2.02(a)</u> (whether telephonic or written) shall specify (i) whether the Borrower Representative is requesting a Borrowing, a conversion of Loans from one Type to the other, or a continuation of LIBOR Loans, (ii) the requested date of the Borrowing, conversion or continuation, as the case may be (which shall be a Business Day), (iii) the principal amount of Loans to be borrowed, converted or continued, (iv) the Type of Loans to be borrower or to which existing Loans are to be converted, and (v) if applicable, the duration of the Interest Period with respect thereto. If the Borrower Representative fails to specify a Type of Loan in a Loan Notice or if the Borrower Representative fails to give a timely notice requesting a conversion or continuation, then the applicable Loans shall be made as, or converted to, Base Rate Loans. Any such automatic conversion to Base Rate Loans shall be effective as of the last day of the Interest Period then in effect with respect to the applicable LIBOR Loans. If the Borrower Representative requests a Borrowing of, conversion to, or continuation of LIBOR Loans in any such Loan Notice, but fails to specify an Interest Period, it will be deemed to have specified an Interest Period of one (1) month.

(b) Following receipt of a Loan Notice, the Administrative Agent shall promptly notify each Lender of the amount of its Pro Rata Share of the applicable Loans, and if no timely notice of a conversion or continuation is provided by the Borrower Representative as required by Section 2.02(a) with respect to any continuation of a LIBOR Loan, the Administrative Agent shall notify each Lender of the details of any automatic conversion of such LIBOR Loans to Base Rate Loans as described in the preceding subsection. In the case of a Borrowing, each Lender shall make the amount of its Loan available to the Administrative Agent in immediately available funds at the Administrative Agent's Office not later than 1:00 p.m. on the Business Day specified in the applicable Loan Notice without setoff, defense, counterclaim or claims in recoupment. Upon satisfaction of the conditions set forth in Section 4.02 (and, if such Borrowing is the initial Credit Extension, Section 4.01 and if such Borrower Representative in like funds as received by the Administrative Agent by wire transfer of such funds, in each case in accordance with instructions provided to (and reasonably acceptable to) the Administrative Agent by the Borrower Representative, provided, that if, on the date of a Borrowing of Revolving Loans, there are Unreimbursed Amounts outstanding, then the proceeds of such Borrowing shall be applied, first, to the payment in full of any such Unreimbursed Amounts, and second, to the Borrower Representative as provided above.

(c) Except as otherwise provided herein, a LIBOR Loan may be continued or converted only on the last day of the Interest Period for such LIBOR Loan. During the existence of an Event of Default, no Loans may be requested as, converted to or continued as LIBOR Loans without the consent of the Administrative Agent or Required Lenders, and the Administrative Agent or Required Lenders may demand that any or all of the then outstanding LIBOR Loans be converted immediately to Base Rate Loans.

(d) The Administrative Agent shall promptly notify the Borrower Representative and the Lenders of the interest rate applicable to any Interest Period for LIBOR Loans upon determination of such interest rate. The determination of the LIBO Rate by the Administrative Agent shall be conclusive in the absence of manifest error.

After giving effect to all Borrowings, all conversions of Loans from one Type to the other, and all continuations of (e) Loans as the same Type, there shall not be more than seven (7) Interest Periods in effect with respect to outstanding Loans.

Section 2.04.

Notwithstanding the foregoing, this Section 2.02 shall not apply to Swingline Loans except as otherwise required by (f)

#### 2.03 Letters of Credit and Letter of Credit Fees.

Provider or Lenders in good faith deems material to it;

#### Letter of Credit. (a)

(ii)

(i) On any Business Day that is at least thirty-one (31) days prior to the Revolving Loan Maturity Date the Revolving Commitment may be used by Borrowers, in addition to the making of Revolving Loans hereunder, for the issuance or arrangement of Letters of Credit and of Support Agreements related thereto pursuant to the terms, and subject to the conditions, set forth herein.

hereunder if:

No Letter of Credit or Support Agreement shall be issued, arranged, increased, amended or extended

(A) such issuance, increase, amendment or extension would violate or be prohibited or enjoined by applicable Law or any decree request or directive of any Governmental Authority or would subject the L/C Issuer, Support Provider or the Lenders to any restriction, reserve or capital requirement not in effect on the Closing Date, or would impose any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which the L/C Issuer, Support

any Lender is at such time a Defaulting Lender, unless the L/C Issuer (or Support Provider) has (B) entered into arrangements, including the delivery of Cash Collateral, satisfactory to the L/C Issuer (or such Support Provider) (in its sole discretion) with the Borrowers or such Lender to eliminate the L/C Issuer's (or such Support Provider's) actual or potential Fronting Exposure (after giving effect to Section 2.16(d)) with respect to the Defaulting Lender arising from either the Letter of Credit then proposed to be issued or that Letter of Credit and all other Letter of Credit Liabilities as to which the L/C Issuer (or such Support Provider) has actual or potential Fronting Exposure, as it may elect in its sole discretion;

the Administrative Agent determines that one or more applicable conditions contained in Article 4 (C) has not been satisfied; or

(D) after giving effect to such issuance, increase or extension, (x) the aggregate Letter of Credit Liabilities under all Letters of Credit exceed the Letter of Credit Sublimit, (y) the total Revolving Exposure of all Lenders exceeds the aggregate Revolving Commitments of all Lenders, or (z) the Revolving Exposure of any Lender exceeds such Lender's Revolving Commitment.

Additionally, no Letter of Credit shall be amended (including any increase in its amount or extension of its term) if such Letter of Credit in its amended form would not be permitted under the terms hereof or if the beneficiary of such Letter of Credit does not accept the proposed amendment to such Letter of Credit.

(iii) Each Letter of Credit shall expire by its terms within one (1) year after the date of issuance and in any event at least thirty (30) days prior to the Revolving Loan Maturity Date. Notwithstanding the foregoing, a Letter of Credit may provide for automatic extensions of its expiry date for one or more successive one (1) year periods, <u>provided</u> that (A) no renewal term may extend the term of the Letter of Credit to a date that is later than the thirtieth (30th) day prior to the Revolving Loan Maturity Date, (B) the L/C Issuer that issued such Letter of Credit has the right (either on its own initiative or at the direction of the Administrative Agent or Support Provider issuing a Support Agreement with respect thereto) not to extend such expiry date and to terminate such Letter of Credit on each such annual expiration date with the giving of notice and (C) no such extension shall be permitted (and the Administrative Agent may notify the L/C Issuer not to so extend such Letter of Credit) if the L/C Issuer, the Support Provider, or the Administrative Agent has determined that such Letter of Credit would not be permitted in its revised (as extended) form under the terms hereof, or the Administrative Agent has determined that one or more of the applicable conditions specified in Article 4 or in this Section 2.03 for Letter of Credit issuance is not then satisfied. Each letter of credit issued or renewed by the L/C Issuer on account of this Agreement or any Support Agreement, and each Support Agreement delivered by a Support Agreement, as applicable, issued, renewed or delivered in full compliance with this Agreement for all purposes hereunder.

(iv) Nothing in this Agreement (other than as provided in Section 2.03(a) as to the L/C Issuer or the Support Provider) shall be construed to obligate any Lender, the Administrative Agent or its Affiliates to arrange, issue, increase the amount of or extend the expiry date of any Letter of Credit or Support Agreement, which act or acts, if any, shall be subject to agreements to be entered into from time to time between the applicable Borrower and such Person.

# (b) <u>Letter of Credit Procedure</u>.

(i) Should a Borrower wish to have a Letter of Credit issued or an existing Letter of Credit amended (including any increase in the amount thereof or extension of the expiry date thereof), Borrower Representative shall deliver to the Administrative Agent a Notice of L/C Credit Event at least five (5) Business Days before the proposed date of issuance or amendment.

(ii) Each L/C Issuer that is a party to this Agreement shall give the Administrative Agent prompt written notice (and Borrowers shall cause each L/C Issuer not a party to this Agreement to give the Administrative Agent prompt written notice) of each issuance or amendment of a Letter of Credit, each payment made by such L/C Issuer in respect of such Letter of Credit issued by it, and any other information requested by the Administrative Agent with respect to such Letter of Credit or amendment.

(c) Letter of Credit Fee Borrowers shall also pay directly to each L/C Issuer (or if applicable, the Support Provider) for its own account a fronting fee with respect to each Letter of Credit issued (or guaranteed) by it equal to 0.125% per annum of the daily maximum amount then available to be drawn under such Letter of Credit (determined without regard to whether any conditions to drawing could then be met) (the "Fronting Fee"). Borrowers shall pay to Administrative Agent, for the benefit of the Revolving Lenders, a letter of credit fee with respect to the Letter of Credit Liabilities for each Letter of Credit, computed for each day from the date of issuance of such Letter of Credit to the date that is the last day a drawing is available under such Letter of Credit, at a rate per annum equal to the Applicable Margin then applicable to Letters of Credit (the 'Letter of Credit Fee") times the daily maximum amount available to be drawn under such Letter of Credit (whether or not such maximum amount is then in effect under such Letter of Credit); provided, however, any Letter of Credit Fees otherwise payable for the account of a

Defaulting Lender with respect to any Letter of Credit as to which such Defaulting Lender has not provided Cash Collateral satisfactory to the L/C Issuer (or the Support Provider, as the case may be) pursuant to this<u>Section 2.03</u> shall be payable, to the maximum extent permitted by applicable Law, to the other Revolving Lenders in accordance with the upward adjustments in their respective Pro Rata Share allocable to such Letter of Credit pursuant to <u>Section 2.16(d)</u>, with the balance of such fee, if any, payable to the L/C Issuer (or the Support Provider, as the case may be) for its own account. Fronting Fees and Letter of Credit Fees shall be (i) computed on a quarterly basis in arrears and (ii) due and payable on the last Business Day of each March, June, September and December, commencing with the first such date to occur after the issuance of such Letter of Credit, on the Revolving Loan Maturity Date and thereafter on demand. If there is any change in the Applicable Margin separately for each period during such quarter that such Applicable Margin was in effect. Notwithstanding anything to the contrary contained herein, while an Event of Default exists, all Letter of Credit Fees shall accrue at the Default Rate to the extent applicable pursuant to <u>Section 2.08(b)</u>. In addition, Borrowers shall pay promptly to the L/C Issuer (or reimburse the Support Provider for) any fronting or other fees, costs or expenses that it may charge in connection with any Letter of Credit.

## (d) <u>Reimbursement Obligations of Borrowers, Reimbursement Loans and Lender Participations</u>.

(i) If an L/C Issuer shall make a payment under a Letter of Credit or a Support Provider shall make a payment under a related Support Agreement, L/C Issuer or the Support Provider, as applicable, shall notify the Borrower Representative thereof. Not later than 11:00 a.m. on the date of any payment by the L/C Issuer under a Letter of Credit or by the Support Provider under a Support Agreement, so long as the Borrower Representative has received telephonic notice of such payment prior to 10:00 a.m. on such date, and otherwise on the following Business Day, the applicable Borrowers shall promptly (but in any event on the same day) satisfy its Reimbursement Obligation by paying to the L/C Issuer or Support Provider (or to the Administrative Agent for the account of the L/C Issuer or Support Provider), as applicable, the full outstanding amount of such Unreimbursed Amount. Such Borrower shall also pay interest, on demand, on all Unreimbursed Amounts for each day until such Unreimbursed Amount is satisfied at a rate per annum equal to the sum of two percent (2%) plus the interest rate applicable to Revolving Loans (which are Base Rate Loans) for such day.

(ii) If any Borrower fails to pay its Reimbursement Obligation when due, the Borrower Representative shall be deemed to have immediately requested that Revolving Lenders make a Revolving Loan (a "Reimbursement Loan"), which shall be a Base Rate Loan, in a principal amount equal to the amount of such Unreimbursed Amount, the proceeds of which shall be applied to satisfy such Reimbursement Obligation. Administrative Agent shall promptly notify Revolving Lenders of any such deemed request and each Revolving Lender shall make available to Administrative Agent not later than 12:00 p.m. on the Business Day following such notification from Administrative Agent such Revolving Lender's Pro Rata Share of such Revolving Loan. Each Revolving Lender hereby absolutely and unconditionally agrees to fund such Revolving Lender's Pro Rata Share of the Reimbursement Loan, unaffected by any circumstance whatsoever, including without limitation (A) the occurrence and continuance of a Default or an Event of Default (but the funding of such a Revolving Loan shall not act as a cure or waiver of any Default or Event of Default other than the non-payment of such Unreimbursed Amount), (B) the fact that, whether before or after giving effect to the making of any such Revolving Loan, the Revolving Exposure exceeds or will exceed the Revolving Commitment, (C) the surrender or impairment of any security for the performance or observance of any of the terms of this Agreement, or (D) the failure of any

condition in <u>Article 4</u> to have been satisfied. Administrative Agent shall apply the gross proceeds of each such Revolving Loan in satisfaction of such Borrowers' Reimbursement Obligation.

Concurrently with the issuance of each Support Agreement and Letter of Credit, each such Revolving Lender (iii) shall be deemed to have purchased and received, without recourse or warranty, an undivided interest and participation, to the extent of such Lender's Pro Rata Share of the Revolving Commitment, in and to the liabilities and obligations in respect of such Letters of Credit and Support Agreements and the corresponding Reimbursement Obligations and Unreimbursed Amounts which may arise therefrom. Such Lenders' participation obligation shall be absolute and unconditional and shall not be affected by any circumstances whatsoever. If, notwithstanding the provision of Section 2.03(d)(i) or (ii) above, any portion of an Unreimbursed Amount remains outstanding (whether due to Borrowers failing to honor their Reimbursement Obligation, or if a Reimbursement Loan cannot for any reason be made, or otherwise) or if any reimbursement received by Support Provider or any L/C Issuer from any Borrower is or must be returned or rescinded upon or during any bankruptcy or reorganization of any Loan Party or otherwise (including any Revolving Loan made pursuant to Section 2.03(d)(ii)), each Revolving Lender shall be irrevocably and unconditionally obligated to fund its participation in such Unreimbursed Amount by paying to Administrative Agent for the account of the Support Provider or L/C Issuer, as applicable, its Pro Rata Share of such Unreimbursed Amount. To the extent any such Revolving Lender shall not have made such amount available to Administrative Agent, as applicable, by 12:00 p.m. on the Business Day on which such Lender receives such notice from Administrative Agent, (A) such Lender shall pay interest on such amount to Administrative Agent on demand accruing daily at the Federal Funds Rate, for the first three (3) days following such Lender's receipt of such notice, and thereafter at the Base Rateplus the Applicable Margin in respect of Revolving Loans that are Base Rate Loans and (B) the Administrative Agent may apply any subsequent payment that such Lender otherwise is entitled to receive under this Agreement to the satisfaction of such Lender's obligation. Any Revolving Lender's failure to fund its participation amount shall not relieve any other Lender of its obligation hereunder to fund such participation, but no Revolving Lender shall be responsible for the failure of any other Lender to fund its participation.

(iv) Notwithstanding the foregoing, payment of any such Lender's participation described in <u>Section 2.03(d)(iii)</u> above, and further disbursement of such payment to the L/C Issuer or Support Provider, shall in no way extinguish the Borrowers' related Reimbursement Obligation and any such Reimbursement Obligation not paid by Borrower or refinanced by Reimbursement Loans shall be due and payable on demand together with interest as described in <u>Section 2.03(d)(i)</u>.

#### (e) <u>Repayment to Lenders</u>.

(i) Until a Lender funds its Reimbursement Loan or participation pursuant to Section 2.03(d), interest with respect to any Unreimbursed Amount shall be for the account of the L/C Issuer or Support Provider, as the case may be. Once the Administrative Agent has received from any Lender such Lender's portion of the Reimbursement Loan or participation, the Administrative Agent shall distribute to such Lender (in the same funds as those received by the Administrative Agent, and whether such funds are directly from the Borrowers or otherwise, including proceeds of cash collateral applied thereto by the Administrative Agent in respect of such Unreimbursed Amount or Reimbursement Loan, plus any interest received by the Administrative Agent which have accrued on such Unreimbursed Amount or Reimbursement Loan for the period after such Lender funded such participation or Reimbursement Loan.

(ii) If any payment received by the Administrative Agent pursuant to <u>Section 2.03(d)</u> is required to be returned under any circumstances (including pursuant to any settlement entered into by the L/C Issuer or the Support Provider, as the case may be, in its discretion), each Revolving Lender shall pay to the Administrative Agent for its own account or for the account of the L/C Issuer or Support Provider, as the case may be, its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect.

(f) <u>Absolute Obligations</u> The obligations of each Borrower to pay its Reimbursement Obligations and its obligation to repay the Reimbursement Loans and the obligations of the Lenders to fund their portion of Reimbursement Loans or participations under <u>Section 2.03(d)</u> shall be absolute, unconditional and irrevocable, and shall be performed strictly in accordance with the terms of this Agreement, under all circumstances whatsoever, including the following:

(i) any lack of validity or enforceability of, or any amendment or waiver of or any consent to departure from, any Letter of Credit, Support Agreement or any related document;

(ii) the existence of any claim, set-off, defense or other right which any Person may have at any time against the beneficiary of any Letter of Credit, the L/C Issuer (including any claim for improper payment), Support Provider, Administrative Agent, any Lender or any other Person, whether in connection with any Loan Document or any unrelated transaction, provided that nothing herein shall prevent the assertion of any such claim by separate suit or compulsory counterclaim;

(iii) any statement or any other document presented under any Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect whatsoever other than in respect of the gross negligence or willful misconduct of the L/C Issuer as determined by a non appealable decision of a court of competent jurisdiction;

(iv) any affiliation between the L/C Issuer, the Administrative Agent and/or the Support Providers; or

(v) to the extent permitted under applicable law, any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, that might, but for the provisions of this Section, constitute a legal or equitable discharge of, or provide a right of setoff against, the Borrowers' Obligations hereunder.

(g) <u>Deposit Obligations of Borrowers</u> Upon the request of the Administrative Agent or the L/C Issuer (or the Support Provider, as the case may be), (x) if the L/C Issuer (or the Support Provider, as the case may be) has honored any full or partial drawing request under any Letter of Credit (as if any Support Provider has made a payment under a Support Agreement) and such drawing (or payment) has resulted in any Unreimbursed Amounts or (y) in the event any Letters of Credit, Support Agreements or Unreimbursed Amounts are outstanding at the time that Borrowers prepay or are required to repay the Obligations or the Revolving Commitment is terminated, Borrowers shall (i) Cash Collateralize one hundred and five percent (105%) of the aggregate outstanding Letter of Credit Liabilities and such Cash Collateral shall be available to Administrative Agent, for its benefit and the benefit of issuers of Lender Letters of Credit and Support Providers, to reimburse payments of drafts drawn under such Letters of Credit and pay any fees and expenses related thereto and (ii) prepay the fee payable under <u>Section 2.03(e)</u> with respect to such Letters of Credit for the full remaining terms of such Letters of Credit. At any time that

there shall exist a Defaulting Lender, promptly upon the request of the Administrative Agent, the L/C Issuer (or the Support Provider, as the case may be) or the Swingline Lender, the Borrowers shall deliver to the Administrative Agent Cash Collateral in an amount sufficient to cover all Fronting Exposure (after giving effect to Section 2.16(d) and any Cash Collateral provided by the Defaulting Lender). Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under this Section 2.03 in respect of Letters of Credit shall be held and applied in satisfaction of the specific Letter of Credit Liabilities, obligations to fund participations therein (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) and other obligations for which the Cash Collateral was so provided, prior to any other application of such property as may be provided herein. Cash Collateral (or the appropriate portion thereof) provided to reduce Fronting Exposure or other obligations shall be released promptly following (i) the elimination of the applicable Fronting Exposure or other obligations giving rise thereto (including by the termination of Defaulting Lender status of the applicable Lender) or (ii) the Administrative Agent's good faith determination that there exists excess Cash Collateral; provided, however, (x) that Cash Collateral furnished by or on behalf of a Loan Party shall not be released during the continuance of a Default or an Event of Default and (y) the Person providing Cash Collateral and the L/C Issuer (or the Support Provider, as the case may be) may agree that Cash Collateral shall not be released but instead held to support future anticipated Fronting Exposure or other obligations. Each Borrower hereby grants to the Administrative Agent, for the benefit of the Administrative Agent and the other Secured Parties, a security interest in all such cash, deposit accounts and all balances therein pledged, deposited with or delivered to the Administrative Agent and all proceeds of the foregoing. Cash Collateral shall be maintained in blocked, non-interest bearing deposit accounts at a bank designated by the Administrative Agent.

(h) <u>Applicability of ISP and UCP</u>. Unless otherwise expressly set forth in the applicable Letter of Credit, (i) the rules of ISP shall apply to each standby Letter of Credit and (ii) the rules of the UCP shall apply to each commercial Letter of Credit.

(i) <u>Letters of Credit Issued for Subsidiaries</u> Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, a Subsidiary of the Borrowers, the Borrowers shall be obligated to pay any Unreimbursed Amount. The Borrowers hereby acknowledge that the issuance of Letters of Credit or any Support Agreement for the account of any such Subsidiary inures to the benefit of the Borrowers and that the Borrowers' business derives substantial benefits from the businesses of such Subsidiaries.

(j) <u>Role of L/C Issuer and Others, Conflicts</u> (i) The L/C Issuer, its correspondents, participants or assignees, the Administrative Agent, the Support Providers and the Agent-Related Persons (collectively, the <u>Released Persons</u>") shall have no responsibility to obtain any document (other than the L/C Issuer's obligation to obtain any sight draft, certificates and documents expressly required by the Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. No Released Person shall be liable to any Loan Party or any Lender for (A) any action taken or omitted in connection herewith at the request or with the approval of the Lenders or the Required Lenders, as applicable; (B) any action taken or omitted in the absence of gross negligence or willful misconduct of such Released Person as determined by a final non appealable decision of a court of competent jurisdiction; or (C) the due execution, effectiveness, validity or enforceability of any Issuer Document or any other document or instrument related to any Letter of Credit or Support Agreement. Each Loan Party and their Subsidiaries hereby assume all risks of the acts or omissions of any beneficiary or transferee with respect to the use of any Letter of Credit. No Released Person shall be liable to responsible for any of the matters described in <u>Section 2.03(f)</u>, provided, that anything in such clauses to the contrary notwithstanding, the Borrowers may have a claim against the L/C Issuer, and the L/C Issuer may be liable to the Borrowers, to the extent, but only to the extent, of any direct, as opposed to consequential, punitive or exemplary, damages suffered by the Borrowers which the

Borrowers prove were caused by the L/C Issuer's willful misconduct or gross negligence or the L/C Issuer's willful failure to pay under any Letter of Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Letter of Credit, each as determined by a final non appealable decision of a court of competent jurisdiction. In furtherance and not in limitation of the foregoing, the L/C Issuer may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and the L/C Issuer shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

(ii) In the event of any conflict between the terms hereof and the terms of any Issuer Documents, the terms hereof shall control.

(iii) The failure of the L/C Issuer to agree to or to conform with the terms of this Agreement (particularly if the L/C Issuer is not a party to this Agreement) shall in no way limit the obligations of the Loan Parties hereunder or subject the Administrative Agent, Support Providers or any Agent-Related Person to any liability.

## 2.04 Swingline Loans.

Swingline Facility. Subject to all of the terms and conditions of this Agreement (including the applicable conditions set (a) forth in Article 4), the Swingline Lender may, in its sole discretion and in reliance upon the representations and warranties of the Borrowers set forth herein and the agreements of the other Lenders set forth in Sections 2.04(c) and 2.04(d), make Swingline Loans to the Borrower Representative on behalf of the Borrowers, from time to time during the Availability Period, for the purposes identified in Section 6.11, notwithstanding the fact that the aggregate amount of the outstanding Swingline Loans, when added to the Swingline Lender's Pro Rata Share of the outstanding Revolving Loans and Letter of Credit Liabilities, from time to time may exceed the amount of such Lender's Revolving Commitment; provided, that, after giving effect to any Borrowing of a Swingline Loan, (i) the total Revolving Exposures shall not exceed the total Revolving Commitments, (ii) the Revolving Exposure of any Lender shall not exceed such Lender's Revolving Commitment (except that in the case of the Swingline Lender, the Swingline Lender's Revolving Exposure (excluding all Swingline Exposure) plus the principal balance of all outstanding and the proposed Swingline Loans shall not exceed the sum of such Revolving Exposure (other than Swingline Exposure) plus the Swingline Loan Sublimit), and (iii) the total Swingline Exposure shall not exceed the Swingline Loan Sublimit. Each Swingline Loan shall be a Base Rate Loan. No Swingline Loan shall be used for the purpose of funding the payment of the principal of any other Swingline Loan. Immediately upon the making of a Swingline Loan, each Revolving Lender shall be deemed to have purchased, and hereby irrevocably and unconditionally agrees to purchase, from the Swingline Lender, a risk participation in such Swingline Loan in an amount equal to the product obtained by multiplying such Lender's Pro Rata Share by the amount of such Swingline Loan. Swingline Loans may be prepaid and reborrowed from time to time during the Availability Period. All Swingline Loans shall be paid in full no later than the earlier of the tenth (10th) Business Day following the Borrowing of such Swingline Loan and the Revolving Loan Maturity Date.

(b) <u>Funding Procedures for Swingline Loans</u> In order to request a Swingline Loan, the Borrower Representative shall give to the Administrative Agent and Swingline Lender a Swingline Loan Notice (or telephonic notice to be confirmed promptly with a Swingline Loan Notice) of a proposed Borrowing consisting of a Swingline Loan, specifying the amount of the requested Swingline Loan, not later than 10:00 a.m. on the Business Day of the proposed Borrowing. Subject to the foregoing, on the Business Day of the proposed Borrowing, the Swingline Lender may make the proceeds of the requested Swingline Loan available to the Borrower Representative by crediting an account of the Borrower

Representative that has been designated for such purpose in writing by the Borrower Representative to the Swingline Lender. Each Swingline Loan shall be in a minimum amount of \$500,000 and in integral multiples of \$100,000 in excess of that amount.

(c) <u>Repayment of Swingline Loans with Revolving Loans</u> Regardless of whether the conditions set forth in <u>Sections 4.01</u> and <u>4.02</u> have been or are capable of being satisfied and without limiting the requirement of the Borrowers to repay the Swingline Loans as set forth in the last sentence of <u>Section 2.04(a)</u>, on any Business Day the Swingline Lender may, in its sole discretion, give notice to the Lenders that some part or all of the outstanding Swingline Loans are to be repaid on the next succeeding Business Day with a Borrowing of Revolving Loans constituting Base Rate Loans made pursuant to <u>Section 2.01</u> (but not subject to the minimum borrowing requirements of <u>Section 2.02</u>) in the same manner and with the same force and effect as if the Borrower Representative had submitted a Loan Notice therefor pursuant to <u>Section 2.02</u>. Swingline Lender hereby agrees that it shall request such a settlement of all of the outstanding Swingline Loans from Revolving Lenders at least once every ten (10) Business Days. Each Lender holding a Revolving Commitment shall make the amount of its Revolving Loan available to the Administrative Agent, in immediately available funds, at Administrative Agent's Office, not later than 11:00 a.m. on the applicable funding date. The Administrative Agent shall make the proceeds of such Revolving Loans available to the Swingline Lender on such funding date by causing an amount of immediately available funds equal to the proceeds of all such Revolving Loans received by the Administrative Agent to be credited to an account of the Swingline Lender at such office of the Administrative Agent or shall make such proceeds available to the Swingline Lender in such other manner as shall be satisfactory to the Administrative Agent and the Swingline Lender.

(d) <u>Participations in Swingline Loans</u> If for any reason a requested Borrowing of Revolving Loans pursuant to<u>Section</u> 2.04(c) is not or cannot be effected, the Revolving Lenders will, as of the date such proposed Borrowing otherwise would have occurred but adjusted for any payments received in respect of such Swingline Loan(s) by or for the account of the Borrowers on or after such date and prior to such purchase, immediately fund their respective participations in the outstanding Swingline Loans as necessary to cause such Lenders to share in such Swingline Loan(s) proportionately in accordance with their respective Pro Rata Shares. Whenever, at any time after any Revolving Lender has funded its purchase of a participating interest in a Swingline Loan, the Swingline Lender receives any payment on account thereof, the Swingline Lender will distribute to such Lender its proportionate share of such amount (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's participating interest was outstanding and funded), provided, that in the event any such payment received by the Swingline Lender is subsequently set aside or is required to be refunded, returned or repaid, such Lender will repay to the Swingline Lender its proportionate share thereof.

(e) <u>Failure to Pay by Lenders</u> If any Lender shall fail to perform its obligation to make a Revolving Loan pursuant to <u>Section 2.04(c)</u> or to fund its purchase of a participation in Swingline Loans pursuant to <u>Section 2.04(d)</u>, the amount in default shall bear interest for each day from the day such amount is payable until fully paid at a rate per annum equal to (i) for the first three (3) days, the Federal Funds Rate and (ii) thereafter, the Base Rate plus the Applicable Margin in respect of Revolving Loans which are Base Rate Loans, and such obligation may be satisfied by application by the Administrative Agent (for the account of the Swingline Lender) of any payment that such Lender otherwise is entitled to receive under this Agreement. Pending repayment, each such advance shall be secured by such Lender's participation interest, if any, in the Swingline Loans and any security therefor, and the Swingline Lender shall be subrogated to such Lender's rights hereunder in respect thereof.

(f) <u>Lenders' Obligations Absolute</u>. The obligation of each Lender to make Revolving Loans pursuant to <u>Section 2.04(c)</u> and to purchase participations in Swingline Loans pursuant to <u>Section 2.04(d)</u> shall be unconditional and irrevocable, shall not be subject to any qualification or exception

whatsoever, shall be made in accordance with the terms and conditions of this Agreement under all circumstances and shall be binding in accordance with the terms and conditions of this Agreement under all circumstances, including the following circumstances: (i) any lack of validity or enforceability of this Agreement, any of the other Loan Documents or any other instrument, document or agreement relating to the transactions that are the subject thereof; (ii) the existence of any claim, defense, set-off or other right that any Borrower, any Guarantor or any Lender may have at any time against any Agent-Related Person, the Swingline Lender, any other Lender, or any other Person, whether in connection with this Agreement, the transactions contemplated herein or any related transactions; (iii) the surrender or impairment of any security for the performance or observance of any of the terms of this Agreement; (iv) the occurrence or continuance of any Default or failure of any condition in <u>Article 4</u> to have been satisfied upon funding the Swingline Loan or thereafter; (v) any adverse change in the condition (financial or other) of any Borrower or any Guarantor; or (vi) any other reason.

## 2.05 Prepayments.

## (a) <u>Voluntary Prepayments of Loans</u>.

Revolving Loans and Term Loans Subject to the limitations set forth in this Section 2.05(a), the Borrowers (i) may, upon notice from the Borrower Representative to the Administrative Agent, at any time or from time to time voluntarily prepay Revolving Loans and the Term Loans in whole or in part without premium or penalty, provided that (i) such notice must be received by the Administrative Agent not later than 10:00 a.m. (A) three (3) Business Days prior to any date of prepayment of LIBOR Loans, and (B) one (1) Business Day prior to the date of prepayment of Base Rate Loans; (ii) any such prepayment of LIBOR Loans shall be in a principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof (or, if less, the entire principal amount thereof then outstanding) and (iii) any such prepayment of Base Rate Loans shall be in a principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof (or, if less, the entire principal amount thereof then outstanding). Each such notice shall specify the date and amount of such prepayment and the Type(s) of Loans to be prepaid. The Administrative Agent will promptly notify each Lender of its receipt of each such notice, and of the amount of such Lender's Pro Rata Share of such prepayment. If such notice is delivered by the Borrowers, the Borrowers shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein, unless such notice is made in connection with the prepayment in full of all Loans and the termination of all commitments under this Agreement, in which case no prepayment shall be required hereunder if the condition to such commitment termination is not satisfied as contemplated by Section 2.06. Any prepayment of a LIBOR Loan shall be accompanied by all accrued interest thereon, together with any additional amounts required pursuant to Section 3.05. Notwithstanding the foregoing, the Borrowers may not voluntarily prepay any Loans that are LIBOR Loans unless such Loans are prepaid at the end of the applicable Interest Period or unless the Borrowers pay all breakage costs associated with such prepayment as provided in Section 3.05 hereof.

(ii) <u>Application of Voluntary Prepayments of Revolving Loans and Term Loans</u> Any voluntary prepayment of the Term Loans shall be applied at the direction of Borrower (or in the absence of direction from Borrower, in the direct order of maturity). Prepayments of Revolving Loans pursuant to this Section 2.05(a) shall not reduce the total Revolving Commitments. Each such prepayment shall be applied to the Loans of the applicable Lenders in accordance with their respective Pro Rata Shares.

(iii) <u>Prepayment of Swingline Loans</u>. The Borrowers may prepay Swingline Loans, in whole or in part, at any time and from time to time. The Borrowers shall, prior to or

contemporaneously with making any such prepayment, give the Swingline Lender such notice of prepayment (written notice or telephonic notice confirmed in writing to the Swingline Lender) as is sufficient to enable the Swingline Lender to apply such prepayment properly to the repayment of Swingline Loans.

#### (b) <u>Mandatory Prepayments of Loans</u>.

(i) <u>Total Revolving Exposure</u>. If, for any reason (a) the total Revolving Exposures at any time exceed the total Revolving Commitments then in effect, (b) the aggregate Letter of Credit Liabilities exceed the Letter of Credit Sublimit, or (c) the total Swingline Exposure exceeds the Swingline Loan Sublimit, the Borrowers shall immediately prepay the Swingline Loans, prepay the Revolving Loans and/or Cash Collateralize the Letter of Credit Liabilities, as applicable, in an aggregate amount equal to any such excess (each such prepayment to be applied as set forth in clause (vii) below).

Dispositions and Involuntary Dispositions. The Borrowers shall prepay the Term Loan and thereafter, all other (ii) Obligations in an aggregate amount equal to one hundred percent (100%) of the Net Cash Proceeds of any Disposition or Involuntary Disposition (each such prepayment to be applied as set forth in <u>clause (vii)</u> below) excluding the proceeds of any Dispositions less than \$2,000,000 in the aggregate in any Fiscal Year and any voluntary Disposition described in clauses (a), (b) or (c) of Section 7.05. Notwithstanding the foregoing, if at the time of the receipt of such Net Cash Proceeds no Default or Event of Default has occurred and is continuing and the Borrower Representative delivers to the Administrative Agent a certificate, executed by the Borrower Representative's chief financial officer, that it intends within 365 days after receipt thereof to use all of such Net Cash Proceeds either to purchase assets used in the ordinary course of business of the Borrowers and their respective Subsidiaries (other than current assets, as defined in accordance with GAAP) or to make Capital Expenditures, the Borrower Borrowers may use such Net Cash Proceeds in the manner set forth in such certificate, provided, that, (A) such Net Cash Proceeds shall be held in a Controlled Account until such time as such Net Cash Proceeds are used to purchase such assets or to make such Capital Expenditures or applied to the Obligations upon the occurrence of any Event of Default, as the case may be, (B) the aggregate amount of such Net Cash Proceeds so used and not subject to prepayment under this clause (ii) of this Section 2.05(b) shall not exceed \$2,000,000 in any Fiscal Year plus any insurance proceeds received in respect of any Disposition, (C) any assets so acquired shall be subject to the security interests under the Collateral Documents with not less than the same priority as the assets subject to such Disposition or Involuntary Disposition, and (D) any such Net Cash Proceeds not so used or committed to such use pursuant to a binding agreement on or before the earliest of the following dates shall promptly (but in any event within two (2) Business Days after such date) be applied as a prepayment in accordance with clause (vii) below: (1) the date that is 365 days (or 540 days, if committed to such use pursuant to a binding agreement that was entered into on or before the 365th day after receipt of such proceeds and notice of such agreement has been delivered to the Administrative Agent) after receipt thereof, (2) the date that is five (5) Business Days after the date on which the Borrower Representative shall have notified the Administrative Agent of the Borrowers' determination not to purchase such replacement assets with such Net Cash Proceeds, (3) the date on which an Event of Default set forth in Section 9.01(a) occurs, and (4) the date that is five (5) Business Days after the delivery of a notice by the Administrative Agent or the Required Lenders to the Borrower Representative during the continuance of any other Event of Default.

(iii) <u>Equity Issuances Immediately upon receipt by Intermediate Holdings of Net Cash Proceeds from a capital</u> contribution to, or the issuance of any Capital Stock of, Intermediate Holdings or any of its Subsidiaries (other than any issuance of Capital Stock (A)

pursuant to the exercise of options or warrants, (B) pursuant to the conversion of any debt securities to equity or the conversion of any elass equity securities to any other elass of equity securities (other than Disqualified Capital Stock), (C) as consideration for orto finance a Permitted Acquisition or permitted Capital Expenditures, (D) to officers and employees pursuant to employee benefit or incentive plans or other similar arrangements, (E) Capital Stock of Intermediate Holdings issued to Holdings or any other Person holding Capital Stock of Intermediate Holdings as of the Closing Date, and (F) Capital Stock issued to another Loan Party) (unless, in each case, made in connection with a Specified Equity Contribution, which shall in all events be subject to the prepayment requirements of this paragraph) (an "Equity Issuance"), the Borrowers shall prepay the Term Loans and thereafter, all other Obligations in an aggregate amount equal to one hundred percent (100%) of such Net Cash Proceeds (or in the case of a Specified Equity Contribution, one hundred percent (100%) of such Specified Equity Contribution) (such prepayment to be applied as set forth in elause (vii) below). [Intentionally Omitted].

(iv) <u>Debt Issuances</u>. Immediately upon receipt by any Loan Party or any Subsidiary of the Net Cash Proceeds of any Debt Issuance, the Borrowers shall prepay the Term Loans and thereafter, all other Obligations in an aggregate amount equal to one hundred percent (100%) of all such Net Cash Proceeds (such prepayment to be applied as set forth in <u>clause (vii)</u> below).

Excess Cash Flow. On the date that is fifteen (15) days after the date on which the annual financial (v) statements are required to be delivered pursuant to Section 6.01(a) for each Fiscal Year (commencing with the Fiscal Year ending December 31, 2019), the Borrowers shall prepay the Term Loans and thereafter, all other Obligations in an aggregate amount equal to fifty percent (50%) (provided, (A) if the Consolidated Total Leverage Ratio was less than 2.00:1.00 but greater than or equal to 1.50:1.00 as of the end of such Fiscal Year as reflected in and properly calculated by the Compliance Certificate for such Fiscal Year, reducing to twenty five percent (25%) for such Fiscal Year and (B) if the Consolidated Total Leverage Ratio was less than 1.50:1.00 as of the end of such Fiscal Year as reflected in and properly calculated by the Compliance Certificate for such Fiscal Year, reducing to zero percent (0%) for such Fiscal Year) of Excess Cash Flow for such Fiscal Year less voluntary principal prepayments under the Loans (other than in respect of the Revolving Loans to the extent not accompanied by an equivalent permanent reduction of the Revolving Commitment) made with Internally Generated Cash. Simultaneously with the delivery by the Loan Parties of the financial statements required to be delivered pursuant to Section 6.01(a) for each Fiscal Year, the Loan Parties shall deliver to the Administrative Agent a calculation (in such detail as the Administrative Agent may reasonably require) of the Excess Cash Flow for such Fiscal Year. Each prepayment with respect to Excess Cash Flow shall be accompanied by a certificate executed by the Borrower Representative's chief financial officer certifying the manner in which Excess Cash Flow and the resulting prepayment were calculated, which certificate shall be in form, substance and detail reasonably satisfactory to the Administrative Agent and shall be applied as set forth in <u>clause (vii)</u> below.

(v i) <u>Extraordinary Receipts</u> Upon receipt by any Loan Party of any Extraordinary Receipts, the BorrowerBorrowers shall prepay the Term Loans and thereafter, all other Obligations in an aggregate amount equal to one hundred percent (100%) of such Extraordinary Receipts and shall be applied as set forth in clause (vii) below.

(vii) <u>Application of Mandatory Prepayments</u> All amounts required to be paid pursuant to this <u>Section 2.05(b)</u> shall be applied as follows:

(A) with respect to all amounts prepaid pursuant to <u>Section 2.05(b)(i)</u>, (x) to the Swingline Loans, to the full extent thereof, (y) after all Swingline Loans have been repaid, to the Revolving Loans to the full extent thereof and, (z) after all Swingline Loans and Revolving Loans have been repaid, to Cash Collateralize any Letter of Credit Liabilities; and

(B) with respect to all amounts prepaid pursuant to <u>Sections 2.05(b)(ii)</u> through (vi), <u>first</u> to the Term Loans (applied in direct order of maturity for the four (4) scheduled principal amortization payments immediately following such prepayment and thereafter pro rata to the remaining principal amortization payments of the Term Loan, including the payment due on the Term Loan Maturity Date), <u>second</u> to the Swingline Loans, <u>third</u> to the Revolving Loans and <u>fourth</u> to Cash Collateralize the Letter of Credit Liabilities.

(C) Within the parameters of the applications set forth above, prepayments shall be applied first to Base Rate Loans and then to LIBOR Loans in direct order of Interest Period maturities. Prepayments of the Revolving Loans pursuant to <u>Sections 2.05(b)(ii)</u> through (vi) shall reduce the total Revolving Commitments. All prepayments under this <u>Section 2.05(b)</u> shall be subject to <u>Section 3.05</u>, and shall be without premium or penalty, and shall be accompanied by a payment of all interest accrued on the principal amount prepaid through the date of prepayment.

**2.06** Termination or Reduction of Total Revolving Commitments. The Borrowers may, upon prior written notice from the Borrower Representative to the Administrative Agent, terminate the total Revolving Commitments or from time to time permanently reduce the total Revolving Commitments; provided, however, that (a) any such notice shall be received by the Administrative Agent not later than 11:00 a.m. (i) ten (10) Business Days prior to the date of termination or (ii) three (3) Business Days prior to the date of reduction, (b) any such partial reduction shall be in an aggregate amount of \$500,000 or any whole multiple of \$100,000 in excess thereof, (c) after giving effect to any reduction of the total Revolving Commitments, the total Revolving Commitments shall not be less than (i) the total Revolving Exposures or (ii) the sum of the Letter of Credit Sublimit and the Swingline Loan Sublimit, (d) any termination of the total Revolving Commitments shall be accompanied by a prepayment in full of all Revolving Loans and all Letter of Credit Liabilities shall be Cash Collateralized in accordance with the terms of Section 2.03(g), and (e) the Administrative Agent will promptly notify the Revolving Lenders of any such notice of termination or reduction of the total Revolving Commitments. Any reduction of the total Revolving Commitments shall be applied to the Revolving Commitment of each Lender according to its Pro Rata Share. All fees accrued with respect thereto until the effective date of any termination of the total Revolving Commitments shall be paid on the effective date of such termination or reduction. Any notice of termination of the total Revolving Commitments delivered by the Borrower Representative pursuant to this Section 2.06 shall be irrevocable.

# 2.07 <u>Repayment of Loans</u>.

(a) <u>Revolving Loans</u>. On the Revolving Loan Maturity Date, the Borrowers shall repay to the Administrative Agent, for the ratable benefit of the Lenders, the aggregate principal amount of all Revolving Loans outstanding on such date.

(b) <u>Term Loans</u>. (i) The Borrowers shall pay the principal amount of the Initial Term Loan in consecutive quarterly installments in the aggregate amounts set forth below commencing with June 30, 2019:

Date	<u>Term Loan Installment</u>
June 30, 2019	\$1,875,000
September 30, 2019	\$1,875,000
December 31, 2019	\$1,875,000
March 31, 2020	\$1,875,000
June 30, 2020	\$1,875,000
September 30, 2020	\$1,875,000
December 31, 2020	\$1,875,000
March 31, 2021	\$1,875,000
June 30, 2021	\$3,750,000
September 30, 2021	\$3,750,000
December 31, 2021	\$3,750,000
March 31, 2022	\$3,750,000
June 30, 2022	\$3,750,000
September 30, 2022	\$3,750,000
December 31, 2022	\$3,750,000
March 31, 2023	\$3,750,000
June 30, 2023	\$3,750,000
September 30, 2023	\$3,750,000
December 31, 2023	\$3,750,000
Term Loan Maturity Date	The unpaid principal amount of the Initial Term Loan then outstanding

(ii) After the initial incurrence of the Delayed – Draw Term Loan, the principal amount of the Delayed – Draw Term Loan will amortize (and be payable by the Borrowers) in equal quarterly installments of two and one-half percent (2.50%) per annum of the sum of the initial principal amounts of each Delayed – Draw Term Loan theretofore incurred until the second anniversary of the Closing Date when such amortization rate will increase to five percent (5.0%) per annum of the sum of the sum of the initial principal amounts of each Delayed – Draw Term Loan theretofore incurred until the second anniversary of each Delayed – Draw Term Loan theretofore incurred with the principal amounts of each Delayed – Draw Term Loan theretofore incurred with the principal balance payable in full by Borrowers on the Term Loan Maturity Date.

(c) <u>Swingline Loans</u>. The Borrowers shall repay each Swingline Loan as provided in <u>Section 2.04(a)</u>.

#### 2.08 Interest.

(a) Subject to the provisions of <u>subsection (b)</u> below, (i) each LIBOR Loan shall bear interest on the outstanding principal amount thereof for each Interest Period at a rate per annum equal to the sum of (A) the LIBO Rate for such Interest Periodus (B) the Applicable Margin; and (ii) each Base Rate Loan bear interest on the outstanding principal amount thereof from the applicable borrowing or conversion date at a rate per annum equal to the (A) Base Rate <u>plus</u> (B) the Applicable Margin.

(b) After the occurrence and during the continuation of an Event of Default, at the election of the Required Lenders (unless an Event of Default exists pursuant to <u>Section 9.01(a)</u>, <u>Section 9.01(f)</u> or <u>Section 9.01(g)</u>, in which event such an election shall be deemed to have automatically occurred

without any further action of the Required Lenders), the Borrowers shall pay interest on the principal amount of all outstanding Loans and any interest payments thereon not paid when due and any fees and other amounts then due and payable hereunder or under any other Loan Document at a rate per annum equal to the Default Rate to the fullest extent permitted by applicable Laws, commencing upon the occurrence of such Event of Default, notwithstanding when such election is made.

(c) Interest on each Loan shall be due and payable in arrears on each Interest Payment Date applicable thereto and at such other times as may be specified herein. Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law whether or not allowed in such proceeding to the fullest extent permitted by Law.

# 2.09 Fees.

(a) <u>Unused Revolving Commitments Fee</u> The Borrowers shall pay, or cause to be paid, to the Administrative Agent for the account of each Revolving Lender in accordance with its Pro Rata Share, an unused fee equal to the product of (i) one-half of one percent (0.50%) per annum (<u>provided</u>, such percentage shall be reduced to one-quarter of one percent (0.25%) per annum so long as the Consolidated Total Leverage Ratio as of the end of the most recent Test Period is less than or equal to 2.50:1.00 as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to <u>Section 6.01(a)</u> or <u>Section 6.01(b)</u>, as applicable) <u>times</u> (ii) the average daily amount by which the total Revolving Commitments exceed the sum of (y) the total outstanding amount of Revolving Loans (excluding Swingline Loans) and (z) the total Letter of Credit Liabilities. This unused fee shall accrue at all times during the Availability Period, including at any time during which one or more of the conditions in <u>Article 4</u> is not met, and shall be due and payable quarterly in arrears on the last day of March, June, September and December, commencing with the first such date to occur after the Closing Date, and on the Revolving Loan Maturity Date; <u>provided</u> that no such unused fee shall accrue on the Revolving Commitment of a Defaulting Lender so long as such Lender shall be a Defaulting Lender (not including any portion thereof reallocated to non-Defaulting Lenders pursuant to <u>Section 2.16(d)</u> hereof).

(b) <u>Unused Delayed – Draw Term Loan Commitments Fee</u> The Borrowers shall pay, or cause to be paid, to the Administrative Agent, for the account of each Term Loan Lender with a Delayed – Draw Term Loan Commitment, in accordance with its Pro Rata Share, an unused fee equal to the product of (i) one-half of one percent (0.50%) per annum (provided, such percentage shall be reduced to one-quarter of one percent (0.25%) per annum so long as the Consolidated Total Leverage Ratio as of the end of the most recent Test Period is less than or equal to 2.50:1.00 as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to Section 6.01(a) or Section 6.01(b), as applicable) times (ii) the average daily amount by which the total Delayed – Draw Term Loan Commitments exceed the total outstanding amount of Delayed – Draw Term Loans. This unused fee shall accrue at all times during the Delayed – Draw Term Loan Availability Period, including at any time during which one or more of the conditions in the <u>Article 4</u> is not met, and shall be due and payable quarterly in arrears on the last day of March, June, September and December, commencing with the first such date to occur after the Closing Date, and on the Term Loan Maturity Date; provided that no such unused fee shall accrue on the Delayed – Draw Term Loan Commitment of a Defaulting Lender so long as such Lender shall be a Defaulting Lender.

(c) <u>Other Fees</u> The Borrowers shall pay the fees in the amounts and at the times specified in the Fee Letter, and in furtherance of the foregoing, the Borrowers hereby assume the obligations of each Person (other than a Secured Party) arising under the Fee Letter. Such fees shall be fully earned when paid and shall be non-refundable for any reason whatsoever.

**2.10** <u>Computation of Interest and Fees</u> All computations of fees and interest shall be made on the basis of a 360-day year and actual days elapsed, except that interest computed by reference to <u>clause (b)</u> of the definition of Base Rate shall be computed on the basis of a year of 365 days (or 366 days in a leap year). Interest shall accrue on each Loan for the day on which the Loan is made, and shall not accrue on a Loan, or any portion thereof, for the day on which the Loan or such portion is paid.

# 2.11 Evidence of Debt.

(a) The Credit Extensions made by each Lender shall be evidenced by one or more accounts or records maintained by such Lender and by the Administrative Agent in the ordinary course of business. The accounts or records maintained by the Administrative Agent and each Lender with respect to this Agreement shall be conclusive absent manifest error of the amount of the Credit Extensions made by the Lenders to the Borrowers and the interest and payments thereon with respect to this Agreement and the other Loan Documents. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of the Borrowers hereunder to pay any amount owing with respect to the Obligations. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of the Administrative Agent in respect of such matters, the accounts and records of the Administrative Agent shall execute and deliver to such Lender (through the Administrative Agent) a promissory note, which shall evidence such Lender's Loans in addition to such accounts or records. Each such promissory note shall (i) in the case of Revolving Loans, be substantially in the form of Exhibit B-1 (a "Revolving Note"), and (ii) in the case of a Term Loan, be substantially in the form of Exhibit B-2 (a "Term Note"), and (iii) in the case of Swingline Loans, be substantially in the form of Exhibit B-3 (the "Swingline Note"). Each Lender may attach schedules to its Note and endorse thereon the date, Type (if applicable), amount and maturity of its Loans and payments with respect thereto, but any failure to do so shall not limit or otherwise affect the Borrowers' Obligations hereunder.

(b) In addition to the accounts and records referred to in <u>subsection (a)</u>, each Lender and the Administrative Agent shall maintain in accordance with its usual practice accounts or records evidencing the purchases and sales by such Lender of participations in Letters of Credit and Swingline

Loans. In the event of any conflict between the accounts and records maintained by the Administrative Agent and the accounts and records of any Lender in respect of such matters, the accounts and records of the Administrative Agent shall control in the absence of manifest error.

## 2.12 <u>Payments Generally</u>.

(a) All payments to be made by the Borrowers of principal, interest, fees and other Obligations shall be absolute and unconditional and shall be made without condition or deduction for any counterclaim, defense, recoupment, setoff or rescission. Except as otherwise expressly provided herein, all payments by the Borrowers hereunder shall be made to the Administrative Agent, for the account of the respective Lenders to which such payment is owed, at the Administrative Agent's Office (or, in the case of Swingline Loans and if so directed in writing by Swingline Lender, delivered directly to the Swingline Lender) in Dollars and in immediately available funds not later than 12:00 p.m. on the date specified herein. The Administrative Agent will promptly distribute to each Lender its Pro Rata Share (or other applicable share as provided herein) of such payment in like funds as received by wire transfer to such Lender's Lending Office. All payments received by the Administrative Agent after 12:00 p.m. shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue.

(b) Subject to the definition of "Interest Period", if any payment to be made by the Borrowers shall come due on a day other than a Business Day, payment shall be made on the next following Business Day, and such extension of time shall be reflected in computing interest or fees, as the case may be.

(c) Unless any Borrower or any Lender has notified the Administrative Agent, prior to the date any payment is required to be made by it to the Administrative Agent hereunder, that any Borrower or such Lender, as the case may be, will not make such payment, the Administrative Agent may assume that the Borrowers or such Lender, as the case may be, has timely made such payment and may (but shall not be so required to), in reliance thereon, make available a corresponding amount to the Person entitled thereto. If and to the extent that such payment was not in fact made to the Administrative Agent in immediately available funds, then:

(i) if any Borrower failed to make such payment, each Lender shall forthwith on demand repay to the Administrative Agent the portion of such assumed payment that was made available to such Lender in immediately available funds, together with interest thereon in respect of each day from and including the date such amount was made available by the Administrative Agent to such Lender to the date such amount is repaid to the Administrative Agent in immediately available funds at the Federal Funds Rate from time to time in effect; and

(ii) if any Lender failed to make such payment, such Lender shall forthwith on demand pay to the Administrative Agent the amount thereof, in immediately available funds, together with interest thereon for the period from the date such amount was made available by the Administrative Agent to the Borrowers to the date such amount is recovered by the Administrative Agent (the "<u>Compensation Period</u>") at a rate per annum equal to (i) for the first three (3) days, the Federal Funds Rate and (ii) thereafter, the Base Rate plus the Applicable Margin in respect of Revolving Loans which are Base Rate Loans. If such Lender pays such amount to the Administrative Agent, then such amount shall constitute such Lender's Loan included in the applicable Borrowing. If such Lender does not pay such amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent may make a demand therefor upon the Borrowers, and the Borrowers shall pay such amount to the Administrative Agent, together with interest thereon for the Compensation Period at a rate per annum equal to the rate of interest applicable to the applicable Borrowing. Nothing herein shall be deemed to relieve any Lender from

its obligation to fulfill its Revolving Commitment or to prejudice any rights which the Administrative Agent or the Borrowers may have against any Lender as a result of any default by such Lender hereunder.

A notice of the Administrative Agent to any Lender or the Borrower Representative with respect to any amount owing under this subsection (c) shall be conclusive, absent manifest error.

(d) The obligations of the Lenders hereunder to make Loans are several and not joint. The failure of any Lender to make any Loan or to fund any participation on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its Loan or purchase its participation.

Sharing of Payments. If, other than as provided elsewhere in this Agreement, in the Fee Letter, in any Assignment and 2.13 Assumption permitted hereunder or in any participation agreement with a Participant permitted hereunder, any Lender shall obtain on account of the Loans made by it, or Letter of Credit Liabilities or Swingline Exposures held by it, any payment (whether voluntary, involuntary, through the exercise of any right of set-off, or otherwise) in excess of its Pro Rata Share (or other share contemplated hereunder) thereof, such Lender shall immediately (a) notify the Administrative Agent of such fact, and (b) purchase from the other Lenders such participations in the Loans made by them, and/or such subparticipations in Letter of Credit Liabilities and Swingline Exposures held by them, as shall be necessary to cause such purchasing Lender to share the excess payment in respect of such Loans or such participations, as the case may be, pro rata with each of them, provided, that if all or any portion of such excess payment is thereafter recovered from the purchasing Lender under any of the circumstances described in Section 12.06 (including pursuant to any settlement entered into by the purchasing Lender in its discretion), such purchase shall to that extent be rescinded and each other Lender shall repay to the purchasing Lender the purchase price paid therefor, together with an amount equal to such paying Lender's ratable share (according to the proportion of (i) the amount of such paying Lender's required repayment to (ii) the total amount so recovered from the purchasing Lender) of any interest or other amount paid or payable by the purchasing Lender in respect of the total amount so recovered, without further interest thereon. The Borrowers agree that any Lender so purchasing a participation from another Lender may, to the fullest extent permitted by law, exercise all its rights of payment (including the right of set-off, but subject to Section 12.09) with respect to such participation as fully as if such Lender were the direct creditor of the Borrowers in the amount of such participation. The Administrative Agent will keep records (which shall be conclusive and binding in the absence of manifest error) of participations purchased under this Section 2.13 and will in each case notify the Lenders following any such purchases or repayments. Each Lender that purchases a participation pursuant to this Section shall from and after such purchase have the right to give all notices, requests, demands, directions and other communications under this Agreement with respect to the portion of the Obligations purchased to the same extent as though the purchasing Lender were the original owner of the Obligations purchased.

# 2.14 Handling of Proceeds of Collateral; Cash Dominion; Revolving Loan Account.

(a) <u>Collection of Accounts and Other Proceeds</u> The Borrowers, at their expense, will enforce and collect payments and other amounts owing on all Accounts in the ordinary course of the Borrowers' business subject to the terms hereof. The Borrowers shall deposit payments on all Accounts directly to one or more Controlled Accounts. Notwithstanding the foregoing, should any Borrower ever receive any payment on an Account or other proceeds of the sale of Collateral, including checks, cash, receipts from credit card sales and receipts, notes or other instruments or property with respect to any Collateral, such Borrower agrees to hold such proceeds separate from such Borrower's other property and funds, and to deposit such proceeds directly into the bank account(s) maintained pursuant to this subsection within three (3) Business Days.

(b) <u>Transfer of Funds</u>. During a Cash Control Period, the Administrative Agent shall have the right, at the Administrative Agent's election in its sole discretion, to require that funds remaining on deposit in any Controlled Account be transferred to the Administrative Agent's Bank Account on each Business Day, and the Borrowers shall take all actions required by the Administrative Agent or by any bank at which any Controlled Account is maintained in order to effectuate the transfer of funds in this manner. All amounts so received will, for purposes of calculating Revolving Availability and interest, be credited to the Revolving Loan Account on the date of deposit in the Administrative Agent's Bank Account. No checks, drafts or other instruments received by the Administrative Agent shall constitute final payment to the Administrative Agent unless and until such instruments have actually been collected.

(c) <u>New Bank Accounts</u>. Each Borrower agrees not to open any new bank account into which proceeds of Collateral are to be delivered or deposited unless concurrently with the opening of such bank account (unless such account constitutes an Excluded Account), the Borrowers enter into a Deposit Account Control Agreement with respect to such bank account; <u>provided</u>, that if any new bank account is acquired in connection with a Permitted Acquisition, such Borrower shall not be required to enter into such Deposit Account Control Agreement until such time as is required pursuant to <u>Sections 6.12</u> or <u>6.14</u> hereof (or such longer period as may be approved in writing by the Administrative Agent at its sole option). Upon compliance with the terms set forth above, such bank account shall constitute a Controlled Account for purposes of this Agreement. Notwithstanding anything to the contrary in this <u>Section 2.14</u>, the Borrowers may maintain one or more accounts constituting Excluded Accounts, <u>provided</u> that if such account ceases to be an Excluded Account, (i) Borrowers shall cause such account to be a Controlled Account within thirty (30) days of such event and (ii) pending such account becoming a Controlled Account, Borrowers shall cause the daily transfer of funds in such account into another Controlled Account.

(d) <u>Collective Borrowing Arrangement</u> The Borrowers have informed the Administrative Agent that: (i) in order to increase the efficiency, profitability and productivity of each Borrower, the Borrower Representative has established a centralized cash management system for the Borrowers that entails, in part, central disbursement and operating accounts for each of the Borrowers in which the Borrower Representative provides the working capital needs of each of the other Borrowers and manages and timely pays the accounts payable of each of the other Borrowers; (ii) the Borrower Representative further enhances the operating efficiencies of the other Borrowers by purchasing, or causing to be purchased, in the Borrower Representative's name for its account, all or substantially all materials, supplies, inventory and services required by the other Borrowers, resulting in a reduction in operating costs of the other Borrowers; and (iii) all of the Borrowers presently engage in an integrated operation that requires financing on an integrated basis, and each Borrower expects to benefit from the continued successful performance of such integrated operations. Therefore, in order to best utilize the borrowing powers of the Borrowers in the most effective and cost efficient manner and to avoid adverse effects on the operating efficiencies of each Borrower and the existing back office practices of the Borrowers, each Borrower has requested that all Revolving Loans, the Term Loans, the Swingline Loans be disbursed, and Letters of Credit be issued, solely upon the request of the Borrower Representative and to bank accounts managed solely by the Borrower Representative, it being the intent and desire of the Borrowers that the Borrower Representative manage for the benefit of each Borrower Representative and usage of such funds.

(e) <u>Revolving Loan Account</u> The Administrative Agent may charge the Revolving Loan Account for all loans and advances made by the Administrative Agent and the Lenders to the Borrower Representative, or otherwise for the Revolving Loan Account, and for any other Obligations, including out-of-pocket expenses of the Administrative Agent, when due and payable hereunder. Interest on the Revolving Loans shall be paid as set forth in <u>Section 2.08</u> hereto. In no event shall prior recourse to any Account or other security granted to or by the Borrowers be a prerequisite to the Administrative Agent's or the Lenders' rights to demand payment of any of the Obligations. In addition, neither the

Administrative Agent nor any Lender shall have any obligation whatsoever to perform in any respect any Borrower's contracts or obligations relating to the Accounts.

#### 2.15 <u>Uncommitted Facilities Increase</u>.

At any time and from time to time prior to twelve (12) months before the Term Loan Maturity Date, subject to the (a) terms and conditions set forth herein and subject to the consent of each Lender, the Borrower Representative may, by notice to the Administrative Agent (whereupon the Administrative Agent shall promptly deliver a copy to each of the Lenders), request to add (x) one or more increases of the existing Term Loans and/or one or more new tranches of term loans to made available under this Agreement (a "Term Loan Increase? or the "Incremental Term Loans") or (y) one or more increases in the Revolving Commitments (a "Revolving Commitment Increase" or the "Incremental Revolving Commitments"; together with the Incremental Term Loans, individually an "Incremental Facility" and collectively the "Incremental Facilities"), or both; provided that (i) at the time of each such request and upon the effectiveness of each Incremental Facility Amendment (as defined below), no Default or Event of Default has occurred and is continuing or shall result therefrom; (ii) the final maturity date of any Incremental Revolving Commitments shall be no earlier than the Revolving Loan Maturity Date, and the final maturity date of any Incremental Term Loans shall be no earlier than the latest Term Loan Maturity Date; (iii) the Weighted Average Life to Maturity of any Incremental Term Loans shall be no shorter than the remaining Weighted Average Life to Maturity of the outstanding Term Loan(s) funded on the Closing Date, and in the case of Incremental Revolving Commitments, such Incremental Facility shall require no scheduled amortization or mandatory commitment reduction prior to the Revolving Loan Maturity Date; (iv) in no event shall the Incremental Term Loans be permitted to be voluntarily or mandatorily prepaid prior to repayment of all Term Loans, unless accompanied by ratable prepayment of Term Loans and any Incremental Revolving Commitments will be subject to the same pro rata borrowing, Letter of Credit participations and prepayment and commitment reduction provisions as the existing Revolving Commitments; (v) Administrative Agent shall have received from Borrower Representative a Pro Forma Compliance Certificate demonstrating the Loan Parties will be in compliance on a Pro Forma Basis with the financial covenants set forth in Article 8 for the most recently completed four (4) Fiscal Quarter period, after giving effect to the Indebtedness incurred under such Incremental Facility, assuming in the case of any Incremental Revolving Commitments such Incremental Revolving Commitments were fully drawn as of the date of measurement; (vi) Administrative Agent shall have received from Borrower Representative a Pro Forma Compliance Certificate demonstrating that after giving effect on a Pro Forma Basis to the Indebtedness incurred under such Incremental Facility (and giving effect on a Pro Forma Basis to any Acquisition consummated concurrently therewith and with all other appropriate adjustments on a Pro Forma Basis as if, if applicable, such Incremental Revolving Commitments were fully drawn as of the date of measurement), (x) the Consolidated Total Leverage Ratio (calculated on a Pro Forma Basis) as of the end of the most recent Test Period would not be greater than 3.00:1.00 and (y) the Consolidated Total Leverage Ratio (calculated on a Pro Forma Basis) as of the end of the most recent Test Period would not be greater than 0.25:1.00 less than the maximum Consolidated Total Leverage Ratio permitted under Section 8.01 at the end of the next measurement date, (vii) after giving effect to the incurrence of such Incremental Facility (but disregarding any proceeds thereof), the Excess Liquidity shall not be less than \$10,000,000, (vii) [intentionally omitted]; (viii) any such Incremental Facility shall have the same Loan Parties obligated in respect of all other Obligations, shall bepari passu with all other Loans and secured on a pari passu basis by the same Collateral securing the existing Obligations; (ix) the proceeds of any Incremental Facility shall be used only to finance a Permitted Acquisition hereunder; (x) Administrative Agent shall have received (x) a satisfactory legal opinion from counsel covering matters reasonably acceptable to Administrative Agent and (y) all Organization Documents reasonably requested by Administrative Agent; (xi) all of the representations and warranties contained in the Loan Documents are true and correct in all material respects (or, in all respects, if already qualified by materiality); and (yxii) the other terms and conditions in respect thereof shall be not more restrictive taken as a whole (as determined by all of the Lenders), than those

applicable to the Revolving Exposure or the Term Loans, or both, as applicable, except (A) if the applicable Lenders holding such Revolving Exposure or Term Loans, or both, as applicable, also receive the benefit of any more restrictive terms, (B) the more restrictive terms are not effective until after the Revolving Loan Maturity Date or the Term Loan Maturity Date, as the case may be, then in effect or (C) otherwise as reasonably satisfactory to all of the Lenders. Borrower Representative may seek commitments for the Incremental Facilities from the existing Lenders, or if the Administrative Agent consents, Additional Lenders who will become Lenders in connection therewith (and meet the requirements of an Eligible Assignee), and for the avoidance of doubt, no Secured Party shall have any obligation to provide or arrange any Incremental Facility or commitment related thereto.

(b) Notwithstanding anything to contrary herein, the aggregate principal amount of all Incremental Facilities (including commitments therefor) shall not exceed \$100,000,000, of which Revolving Commitment Increases shall not exceed \$15,000,000, and not more than three (3) Incremental Facilities may be requested hereunder. Each Incremental Facility shall be in an integral multiple of \$500,000 and be in an aggregate principal amount that is not less than \$10,000,000; provided that such amount may be less than the applicable minimum amount if such amount represents all the remaining availability hereunder as set forth above.

(c) The terms, provisions and documentation of the Incremental Term Loans and Incremental Revolving Commitments shall be on the same terms and conditions of this Agreement, except as agreed (unless otherwise set forth in this <u>Section 2.15</u>) among the Borrowers, each Lender and the applicable Additional Lender(s) providing such Incremental Term Loans and Incremental Revolving Commitments and except that:

if the Applicable Margin (which, for such purposes only, shall be deemed to include all upfront or similar fees (i) or original issue discount (with original issue discount being equated to interest based on an assumed four-year life to maturity or, if shorter, the actual Weighted Average Life to Maturity) payable to all Additional Lenders providing such Incremental Facility (but excluding the portion of structuring, arrangement, commitment or similar fees not shared with all such Additional Lenders in connection therewith)) relating to any Incremental Facility exceeds the then Applicable Margin (which, for such purposes only, shall be deemed to include all upfront or similar fees or original issue discount (with original issue discount being equated to interest based on an assumed four-year life to maturity or, if shorter, the actual Weighted Average Life to Maturity) payable to all Lenders providing the Term Loans and Revolving Commitments extended on the Closing Date (the 'Closing Date Facilities') (but excluding structuring, arrangement, commitment or similar fees not shared with such Term Loan Lenders in connection therewith)) immediately prior to the effectiveness of the applicable Incremental Facility Amendment by more than one-half of one percent (0.50%), the Applicable Margin relating to the Closing Date Facilities shall be adjusted to be equal to the Applicable Margin (which, for such purposes only, shall be deemed to include all upfront or similar fees or original issue discount (with original issue discount being equated to interest based on an assumed four-year life to maturity or, if shorter, the actual Weighted Average Life to Maturity) payable to all Additional Lenders providing such Incremental Facilities (but excluding structuring, arrangement, commitment or similar fees not shared with all such Additional Lenders in connection therewith)) relating to such Incremental Facilities minus one-half of one percent (0.50%); provided that, (x) in the case of this clause (i), if an Incremental Facility includes an interest rate floor greater than the applicable interest rate floor for the Closing Date Facilities, such differential between interest rate floors shall be equated to the Applicable Margin for purposes of determining whether an increase to the Applicable Margin under the Closing Date Facilities shall be required, but only to the extent an increase in the interest rate floor in the Closing Date Facilities would cause an increase in the interest rate then in effect thereunder, and in such case the interest rate floor (but not the Applicable Margin) applicable to the Initial Term

Loans shall be increased to the extent of such differential between interest rate floors and (y) any amendments to the Applicable Margin relating to the Closing Date Facilities that became effective subsequent to the Closing Date but prior to the time of such Incremental Facility shall also be included in the foregoing calculations,

(ii) the amortization requirements for such Incremental Term Loans may differ, so long as any Incremental Term Loans shall not have a Weighted Average Life to Maturity that is shorter than the Weighted Average Life to Maturity of the thenremaining Term Loans part of the Closing Date Facilities (without giving effect to any prepayments), and in no event shall such Incremental Term Loans be permitted to be voluntarily or mandatorily prepaid prior to the repayment in full of all Term Loans borrowed on the Closing Date, unless accompanied by a ratable prepayment of the Term Loans borrowed on the Closing Date,

(iii) any Incremental Revolving Commitments will be subject to the same pro rata borrowing, Letter of Credit participations and prepayment and commitment reduction provisions as the existing Revolving Commitments, and

(iv) such Incremental Facilities (a) shall not be secured by any Lien on any asset of any Person that does not also secure the then outstanding Closing Date Facilities or (b) shall not be guaranteed by any Person other than a Loan Party under the then outstanding Closing Date Facilities.

(d) Each notice from the Borrowers pursuant to this <u>Section 2.15</u> shall set forth the requested amount and proposed terms of the relevant Incremental Term Loans and/or Incremental Revolving Commitments, as applicable. If the Administrative Agent does not receive within a time period proscribed by the Administrative Agent sufficient commitments from existing Lenders to effectuate the Incremental Term Loans and/or the Incremental Revolving Commitments, as applicable, any additional bank, financial institution, existing Lender or other Person that constitutes an entity of the type that would be an Eligible Assignee electing to extend Incremental Term Loans or Incremental Revolving Commitments shall be determined by the BorrowerBorrowers (subject to the consent of (y) the Administrative Agent and (z) the L/C Issuer (or if applicable, the Support Provider) and the Swingline Lender solely with respect to Incremental Revolving Commitments) (any such bank, financial institution, existing Lender or other Person being called an "Additional Lender") and, if not already a Lender, shall become a Lender under this Agreement pursuant to an amendment (an "Incremental Facility Amendment") to this Agreement and, as appropriate, the other Loan Documents, executed by (in form and substance mutually acceptable to each of) the Loan Parties, each Lender, such Additional Lender, the Administrative Agent, and in the case of any Incremental Revolving Commitments, each L/C Issuer (or if applicable, the Support Provider) and Swingline Lender. For the avoidance of doubt, no L/C Issuer (or if applicable, Support Provider) and Swingline Lender is required to act as such for any additional Revolving Commitments unless they so consent. No Incremental Facility Amendment.

Upon each increase in the Revolving Commitments pursuant to this <u>Section 2.15</u>, each Revolving Lender immediately prior to such increase will automatically and without further action be deemed to have assigned to each Lender providing a portion of the Incremental Revolving Commitment (each an <u>Incremental Revolving Lender</u>) in respect of such increase, and each such Incremental Revolving Lender will automatically and without further act be deemed to have assumed, a portion of such Revolving Lender's participations hereunder in outstanding Letters of Credit and Swingline Loans such that, after giving effect to each such deemed assignment and assumption of participations, the percentage of the aggregate outstanding (i) participations hereunder in Letters of Credit and (ii) participations hereunder in Swingline Loans held by each Revolving Lender (including each such Incremental Revolving Lender) will

equal the percentage of the aggregate Revolving Commitments of all Revolving Lenders represented by such Revolving Lender's Revolving Commitment. The Administrative Agent and the Lenders hereby agree that the minimum borrowings, pro rata borrowing and pro rata payment requirements contained elsewhere in this Agreement shall not apply to the transactions effected pursuant to the immediately preceding sentence. Commitments in respect of any Incremental Term Loans or Incremental Revolving Commitments shall become Commitments under this Agreement. The effectiveness of any Incremental Facility Amendment shall, unless otherwise agreed to by the Administrative Agent, each Lender and the Additional Lenders, be subject to the satisfaction on the date thereof (each, an <u>Incremental Facility Closing Date</u>) of each of the conditions set forth in <u>Section 4.01</u> (it being understood that (x) all references to "the date of such Credit Extension" in <u>Section 4.01</u> shall be deemed to refer to the Incremental Facility Closing Date and (y) the Incremental Facility Closing Date shall be deemed to be the initial Credit Extension).

(e) No Incremental Term Loans or Incremental Revolving Commitments may be made by Sponsor, any Loan Party or any of their respective Affiliates or Subsidiaries.

**2.16** <u>Defaulting Lenders</u>. If any Lender becomes a Defaulting Lender, then the following provisions shall apply for so long as such Lender is a Defaulting Lender:

(a) (i) such Defaulting Lender's Revolving Commitment and outstanding Revolving Loans shall be excluded for purposes of calculating the fee payable to Revolving Lenders in respect of <u>Section 2.09(a)</u>, and such Defaulting Lender shall not be entitled to receive any fee pursuant to <u>Section 2.09(a)</u> with respect to such Defaulting Lender's Revolving Commitment or Revolving Loans (in each case not including any fee in connection with any portion of such Defaulting Lender's Revolving Commitment that has been reallocated to non-Defaulting Lenders pursuant to <u>Section 2.16(d)</u> hereof) and (ii) such Defaulting Lender's Delayed – Draw Term Loan Commitment and outstanding Delayed - Draw Term Loans shall be excluded for purposes of calculating the fee payable to the Term Loan Lenders in respect of <u>Section 2.09(b)</u> and such Defaulting Lender shall not be entitled to receive any fee pursuant to <u>Section 2.09(b)</u> with respect to such Defaulting Lender Delayed – Draw Term Loan Commitment or Delayed – Draw Term Loans.

(b) the Revolving Commitments and Loans of such Defaulting Lender shall not be included in determining whether all Lenders or the Required Lenders, Required Revolving Lenders or Required Term Loan Lenders have taken or may take any action hereunder (including any consent to any amendment or waiver pursuant to <u>Section 12.01</u>).

(c) in the event a Defaulting Lender has defaulted on its obligation to fund any Revolving Loan, or purchase any participation pursuant to <u>Section 2.03(d)</u> or <u>Section 2.04(d)</u> hereof, until such time as the Default Excess with respect to such Defaulting Lender has been reduced to zero, any prepayments or repayments on account of the Revolving Loans or participations purchased pursuant to <u>Section 2.04(d)</u>, in each case to the extent they would be otherwise be payable to such Defaulting Lender, shall be applied first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; <u>second</u>, to the payment on a <u>pro rata</u> basis of any amounts owing by such Defaulting Lender to the L/C Issuer, Support Provider or Swingline Lender hereunder<u>third</u>, to Cash Collateralize the L/C Issuer's (or the Support Provider's, as the case may be) Fronting Exposure with respect to such Defaulting Lender; <u>fourth</u>, as the Borrowers may request (so long as no Default or Event of Default exists), to the funding of any Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; <u>fifth</u>, if so determined by the Administrative Agent and the Borrowers, to be held in a deposit account and released pro rata in order to (x) satisfy such Defaulting Lender's potential future funding obligations with respect to Loans under this Agreement and (y) Cash Collateralize the L/C Issuer's (or the Support Provider's, as the

case may be) future Fronting Exposure with respect to such Defaulting Lender with respect to future Letters of Credit issued under this Agreement; <u>sixth</u>, to the payment of any amounts owing to the Lenders, the L/C Issuer, the Support Provider or Swingline Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender, the L/C Issuer, the Support Provider or Swingline Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; <u>seventh</u>, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrowers as a result of any judgment of a court of competent jurisdiction obtained by the Borrowers against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and <u>eighth</u>, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided, <u>that</u>, if (x) such payment is a payment of the principal amount of any Loans or Letter of Credit Liabilities in respect of which such Defaulting Lender has not fully funded its appropriate share and (y) such Loans or Letter of Credit Liabilities were made at a time when the conditions set forth in<u>Section 4.02</u> and, if applicable, <u>Section 4.02(e)</u> were satisfied or waived, such payment shall be applied solely to pay the Loans of, or Letter of Credit Liabilities owed to, such Defaulting Lender. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this<u>Section 2.16(c)</u> shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

Lender then:

(d)

If any Swingline Loans or Letter of Credit Liabilities are outstanding at the time a Lender becomes a Defaulting

(i) so long as no Default or Event of Default then exists, all or any part of such Swingline Loans and Letter of Credit Liabilities shall be reallocated among the non-Defaulting Lenders of Revolving Loans in accordance with their respective Pro Rata Shares of the total Revolving Commitments (calculated without regard to such Defaulting Lender's Revolving Commitments), provided that no Revolving Lender's Revolving Exposure shall exceed its Revolving Commitment;

(ii) if the reallocation described in <u>paragraph (i)</u> above cannot, or can only partially, be effected, the Borrowers shall within one (1) Business Day following notice by the Administrative Agent (A) first, prepay the amount of the Swingline Loans equal to Defaulting Lender's Pro Rata Share thereof after giving effect to any partial reallocation pursuant to <u>paragraph</u> (i) above and (B) second, Cash Collateralize such Defaulting Lender's Pro Rata Share of Letter of Credit Liabilities (after giving effect to any partial reallocation pursuant to <u>paragraph</u> (i) above) in accordance with the procedures set forth in <u>Section 2.03(g)</u> and for so long as any such Obligations related to Letters of Credit are outstanding;

(iii) if the Borrowers Cash Collateralize any portion of such Defaulting Lender's Pro Rata Share of Letter of Credit Obligations pursuant to this <u>Section 2.16(d)</u>, the Borrowers shall not be required to pay any fees to such Defaulting Lender pursuant to <u>Section 2.03(c)</u> with respect to the portion of such Defaulting Lender's Pro Rata Share of Letter of Credit Obligations which have been Cash Collateralized (and the Defaulting Lender shall not be entitled to receive any such fees);

(iv) if the Defaulting Lender's Pro Rata Share of Letter of Credit Obligations are reallocated pursuant to this <u>Section 2.16(d)</u>, then the letter of credit fees payable to the non-Defaulting Lenders pursuant to <u>Section 2.03(c)</u> shall be adjusted accordingly; and

(v) if any Defaulting Lender's Pro Rata Share of Letter of Credit Liabilities is not Cash Collateralized or reallocated pursuant to this <u>Section 2.16(d)</u>, then without prejudice to any rights or remedies of the applicable Support Provider or L/C Issuer hereunder, all letter of credit fees payable under <u>Section 2.03(c)</u> with respect to such Defaulting Lender's Pro Rata Share of Letter of Credit Liabilities shall be payable to the L/C Issuer or if applicable, the Support Provider.

(e) So long as any Lender is a Defaulting Lender, the Swingline Lender shall not be required to fund any Swingline Loan, and no L/C Issuer or Support Provider shall be required to issue, extend or increase any Letter of Credit or Support Agreement, in each case unless it is reasonably satisfied that the related exposure will be one hundred percent (100%) covered by the Revolving Commitments of the non-Defaulting Lenders and/or Cash Collateral will be provided by the Borrowers in accordance with <u>Section 2.03(g)</u>, and participating interests in any such newly issued, extended or increased Letter of Credit or Support Agreement or newly made Swingline Loan shall be allocated among non-Defaulting Lenders in a manner consistent with <u>Section 2.16(d)(i)</u> (and Defaulting Lenders shall not participate therein).

(f) No reallocation permitted pursuant to <u>Section 2.16(d)</u> shall constitute a waiver or release of any claim of any party hereunder against a Defaulting Lender arising from that Lender having become a Defaulting Lender, including any claim of a non-Defaulting Lender as a result of such non-Defaulting Lender's increased exposure following such reallocation.

(g) In the event that the Administrative Agent, the L/C Issuer, the Support Provider and the Swingline Lender each agrees in writing that a Defaulting Lender has adequately remedied all matters which caused such Lender to become a Defaulting Lender, then the Pro Rata Shares of Swingline Loans and Letter of Credit Obligations of the Revolving Lenders shall be readjusted to reflect the inclusion of such Lender's Revolving Commitment and on such date such Lender shall purchase at par such of the Revolving Loans of the other Lenders (other than Swingline Loans) or participations in the Revolving Loans as the Administrative Agent shall determine may be necessary in order for such Lender to hold such Revolving Loans or participations in accordance with its Pro Rata Share<u>provided</u>, that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrowers while that Lender was a Defaulting Lender; <u>provided</u>, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender having been a Defaulting Lender.

(h) The rights and remedies with respect to a Defaulting Lender under this <u>Section 2.16</u> are in addition to any other rights and remedies which the Borrower, the Administrative Agent, the L/C Issuer, the Support Provider or the Swingline Lender, as applicable, may have against such Defaulting Lender.

2.17 <u>Refinancing Facilities</u>. After the Closing Date, the Borrowers may obtain from any Lender (but with the consent of the Administrative Agent (not to be unreasonably withheld or delayed)) to refinance all or any portion of the applicable Loans or Commitments then outstanding under this Agreement (which for purposes of this <u>Section 2.17</u> will be deemed to include any then outstanding Refinancing Facilities), one or more new senior secured first lien term facilities (each, a "<u>Refinancing Term Facility</u>" and the term loans made pursuant to a Refinancing Term Facility, "<u>Refinancing Term Loans</u>") or, in the case of a refinancing and/or replacement of the Revolving Commitments or Revolving Loans, new revolving credit facilities (each, a "<u>Refinancing Revolving Facility</u>" and, together with any Refinancing Term Facility, a "<u>Refinancing Facilities</u>", and the revolving loans made pursuant to a Refinancing Revolving Facility. The "<u>Refinancing Facilities</u>", and the revolving loans made pursuant to a Refinancing Revolving Facility.

(a) the Refinancing Term Loans or Refinancing Revolving Loans, as applicable, will be *pari passu* in right of payment and be secured by the Collateral on a *pari passu* basis with the remaining portion of the Term Loans, Revolving Loans and Revolving Commitments;

(b) with respect to any Refinancing Term Facility, such Refinancing Term Facility shall not (i) have a maturity date that is earlier than ninety-one (91) days after the maturity date of the Term Loans being refinanced by such Refinancing Term Facility or (ii) have a shorter Weighted Average Life to Maturity than the Term Loans being refinanced by such Refinancing Term Facility, and in no event shall the Refinancing Term Facility be permitted to be voluntarily or mandatorily prepaid prior to repayment of all existing Term Loans that survive the initial funding of such Refinancing Term Facility, unless accompanied by ratable prepayment of all Term Loans;

(c) with respect to any Refinancing Revolving Facility, (i) such Refinancing Revolving Facility shall not have a maturity date that is earlier than the maturity date of the Revolving Loans or Revolving Commitments being refinanced by such Refinancing Revolving Facility, (ii) such Refinancing Revolving Facility shall require no scheduled amortization or mandatory commitment reduction prior to the maturity date of any existing Revolving Commitments that survive the initial funding of the Refinancing Revolving Facility and (iii) any Refinancing Revolving Facility will be subject to the same pro rata (or in the case of prepayment, pro rata or less than pro rata, but not greater than pro rata) borrowing, Letter of Credit participations, Swingline Loan participations and prepayment and Commitment reduction provisions as the existing Revolving Commitments that survive the initial funding of the Refinancing Facility (except to the extent applicable only to periods after the latest final maturity date of the relevant Loans or Commitments existing at the time of such refinancing or replacement);

(d) such Refinancing Facility shall have pricing (including interest, fees and premiums), optional prepayment and redemption terms as may be agreed to by the Borrowers and the Lenders party thereto, except as provided herein;

(e) such Refinancing Facility shall not be secured by any assets other than the Collateral;

(f) if any such Refinancing Facility is guaranteed, it shall not be guaranteed by any Person other than the Guarantors;

(g) if any such Refinancing Facility will provide for the issuance or extension of Letters of Credit or the making of Swingline Loans, then the L/C Issuer (or if applicable, Support Provider) and the Swingline Lender, respectively, shall have consented to such Refinancing Facility;

(h) the other terms (excluding those referenced in clauses (a) through (g) above and excluding pricing, fee and prepayment or redemption provisions) of such Refinancing Facility shall be substantially identical to, or (taken as a whole) no more favorable to the Lenders providing such Refinancing Facility than those applicable to the Loans or Commitments being refinanced or replaced (except for covenants or other provisions applicable only to periods after the latest final maturity date of the relevant Loans or Commitments existing at the time of such refinancing or replacement);

(i) the aggregate principal amount of any Refinancing Facility shall not exceed the aggregate principal amount of the Loans and Commitments being refinanced or replaced therewith, plus reasonable and customary interest, premiums, fees and expenses; and

(j) any Refinancing Facility held by any Affiliated Lender shall be subject to the same restrictions applicable to assignments to such Persons as are set forth in <u>Section 12.07</u> (including voting

restrictions, the prohibition on holding Revolving Commitments and an aggregate cap on the amount of Term Loans held by such Person).

Notwithstanding anything to the contrary contained in this Agreement, any Lender may exchange, continue or rollover all or a portion of its Loans in connection with any Refinancing Facility permitted by the terms of this Agreement, pursuant to a cashless settlement mechanism approved by the Borrowers, the Administrative Agent and such Lender.

# 2.18 <u>Amend and Extend Transactions</u>.

(a) The Borrowers may, by written notice to the Administrative Agent from time to time, request an extension (each, an "Extension") of the Revolving Loan Maturity Date and/or the Term Loan Maturity Date to the extended maturity date specified in such request. Such notice shall set forth (i) the amount of the Revolving Commitments and/or Term Loans to be extended (which shall be in minimum increments of \$100,000 and a minimum amount of \$1,000,000), and (ii) the date on which such Extension is requested to become effective (which shall be not less than ten (10) Business Days nor more than sixty (60) days after the date of such requested Extension (or such longer or shorter periods as the Administrative Agent shall agree)). Each Appropriate Lender shall be offered (an "Extension Offer") an opportunity to participate in such Extension on a pro rata basis and on the same terms and conditions as each other Appropriate Lender pursuant to procedures established by, or reasonably acceptable to, the Administrative Agent. Any Lender approached to participate in such Extension may elect or decline, in its sole discretion, to participate in such Extension. If the aggregate principal amount of Revolving Commitments or Term Loans (calculated on the face amount thereof) in respect of which Appropriate Lenders shall have accepted the relevant Extension Offer shall exceed the maximum aggregate principal amount of Revolving Commitments or Term Loans, as applicable, of Appropriate Lenders shall be extended ratably up to such maximum amount based on the respective principal amounts (but not to exceed actual holdings of record) with respect to which such Appropriate Lenders have accepted such Extension Offer.

(b) It shall be a condition precedent to the effectiveness of any Extension that (i) no Default or Event of Default shall have occurred and be continuing immediately prior to and immediately after giving effect to such Extension, (ii) the representations and warranties of the Borrowers and each other Loan Party contained in<u>Article 2</u> and <u>Article 5</u> or any other Loan Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct in all material respects (provided, if any representation or warranty is by its terms qualified by concepts of materiality, such representation and warranties specifically refer to an earlier date, in which case they shall be true and correct in all material respects (provided, if any representation or warranty is by its terms qualified by concepts of materiality, such representation or warranty is by its terms qualified by concepts (provided, if any representation or warranty is by its terms qualified by concepts (provided, if any representation or warranty is by its terms qualified by concepts of material respects) as of such earlier date, (iii) the L/C Issuer and the Swingline Lender shall have consented to any Extension of the Revolving Commitments to the extent that such Extension provides for the issuance of Letters of Credit or making of Swingline Loans at any time during the extended period and (iv) the terms of such Extended Revolving Commitments and Extended Term Loans shall comply with Section 2.18(c).

(c) The terms of each Extension shall be determined by the Borrowers and the applicable extending Lenders and be set forth in an Additional Credit Extension Amendment; provided, that, (i) the final maturity date of any Extended Revolving Commitment or Extended Term Loan shall be no earlier than the Revolving Loan Maturity Date or the Term Loan Maturity Date, respectively, (ii)(A) there shall be no scheduled amortization of the Extended Revolving Commitments and (B) the Weighted

Average Life to Maturity of the Extended Term Loans shall be no shorter than the remaining Weighted Average Life to Maturity of the Term Loans, (iii) the Extended Revolving Loans and the Extended Term Loans will rank pari passu in right of payment and with respect to security with the Revolving Loans and the Term Loans being extended and the borrower and guarantors of the Extended Revolving Commitments or Extended Term Loans, as applicable, shall be the same as the borrower and guarantors with respect to the Revolving Loans or applicable Term Loans being extended Revolving Commitments (iv) the interest rate margin, rate floors, fees, original issue discounts and premiums applicable to any Extended Revolving Commitments (and the Extended Revolving Loans thereunder) and Extended Term Loans shall be determined by the BorrowerBorrowers and the applicable extending Lenders and (v) to the extent the terms of the Extended Revolving Commitments or Extended Term Loans are inconsistent with the terms set forth herein (except as set forth in clause (i) through (iv) above), such terms shall be reasonably satisfactory to the Administrative Agent.

(d) In connection with any Extension, the Borrowers, the Administrative Agent and each applicable extending Lender shall execute and deliver to the Administrative Agent an Additional Credit Extension Amendment and such other documentation as the Administrative Agent shall reasonably specify to evidence such Extension. The Administrative Agent shall promptly notify each Lender as to the effectiveness of each Extension. Notwithstanding anything herein to the contrary, any Additional Credit Extension Amendment may, without the consent of any other Lender, effect such amendments to this Agreement and the other Loan Documents as may be necessary or appropriate (but only to such extent), in the reasonable opinion of the Administrative Agent and the Borrowers, to implement the terms of any such Extension Offer, including any amendments necessary to establish Extended Revolving Commitments or Extended Term Loans as a new tranche of revolving commitments or term loans, as applicable, and such other technical amendments as may be necessary or appropriate in the reasonable opinion of the Administrative Agent and the Borrowers are new tranche of revolving commitments or term loans, as applicable, and such other technical amendments as may be necessary or appropriate in the reasonable opinion of the Administrative Agent and the other contraction of such new tranche (including to preserve the pro rata treatment of the extended and non-extended tranches and to provide for the reallocation of any obligations under Swingline Loans or Letters of Credit (including Letter of Credit Liabilities) upon the expiration or termination of the commitments under any tranche), in each case on terms consistent with this Section 2.18).

2.19 Acknowledgement Regarding Any Supporting QFCs To the extent that the Loan Documents provide support, through a guarantee or otherwise, for Hedge Agreements or any other agreement or instrument that is a QFC (such support, "QFC Credit Support" and each such QFC a "Supported QFC"), the parties acknowledge and agree as follows with respect to the resolution power of the Federal Deposit Insurance Corporation under the Federal Deposit Insurance Act and Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act (together with the regulations promulgated thereunder, the "U.S. Special Resolution Regimes") in respect of such Supported QFC and QFC Credit Support (with the provisions below applicable notwithstanding that the Loan Documents and any Supported QFC may in fact be stated to be governed by the laws of the State of New York and/or of the United State or any other state of the United States):

(a) In the event a Covered Entity that is party to a Supported QFC (each, a "Covered Party") becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer of such Supported QFC and the benefit of such QFC Credit Support (and any interest and obligation in or under such Supported QFC and such QFC Credit Support, and any rights in property securing such Supported QFC or such QFC Credit Support) from such Covered Party will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if the Supported QFC and such QFC Credit Support (and any such interest, obligation and rights in property) were governed by the laws of the United States or a state of the United States. In the event a Covered Party or a BHC Act Affiliate of a Covered Party becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under the Loan Documents that might otherwise apply to such Supported QFC or any QFC Credit Support

that may be exercised against such Covered Party are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if the Supported QFC and the Loan Documents were governed by the laws of the United Sates or a state of the United States. Without limitation of the foregoing, it is understood and agreed that rights and remedies of the parties with respect to a Defaulting Lender shall in no event affect the rights of any Covered Party with respect to a Supported QFC or any QFC Credit Support.

(b) As used in this Section 2.19, the following terms have the following meanings:

"BHC Act Affiliate" of a party means an "affiliate" (as such term is defined under, and interpreted in accordance with, 12 U.S.C. 1841(i)) of such party.

"Covered Entity" means any of the following:

- (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or
- (iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).

"Default Right" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F. R. §§ 252.81, 472 or 382.1, as applicable.

"QFC" has the meaning assigned to the term "qualified financial contract" in, and shall be interpreted in accordance with, 12 U.S.C. 5390(c)(8)(D).

#### ARTICLE 3 TAXES, YIELD PROTECTION AND ILLEGALITY

### 3.01 <u>Taxes</u>.

(a) <u>Payments Free of Taxes</u>. Any and all payments by or on account of any obligation of any Loan Party under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by applicable Law. If any applicable Law (as determined in the good faith discretion of an applicable Withholding Agent) requires the deduction or withholding of any Tax from any such payment by a Withholding Agent, then the applicable Withholding Agent shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable Law and, if such Tax is an Indemnified Tax, then the sum payable by the applicable Loan Party shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.

(b) <u>Payment of Other Taxes by the Loan Parties</u> The Loan Parties shall timely pay to the relevant Governmental Authority in accordance with applicable Law, or at the option of the Administrative Agent timely reimburse it for the payment of, any Other Taxes.

(c) <u>Indemnification by the Loan Parties</u> The Loan Parties shall jointly and severally indemnify each Recipient, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower Representative by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.

(d) Indemnification by the Lenders Each Lender shall severally indemnify the Administrative Agent, within ten (10) days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that any Loan Party has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of the Loan Parties to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of <u>Section 12.07(d)</u> relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with any Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this paragraph (d).

(e) <u>Evidence of Payments</u> As soon as practicable after any payment of Taxes by any Loan Party to a Governmental Authority pursuant to this <u>Section 3.01</u>, such Loan Party shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.

#### (f) <u>Status of Lenders</u>.

(i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to the Borrower Representative and the Administrative Agent, at the time or times reasonably requested by the Borrower Representative or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Borrower Representative or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by the Borrower Representative or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by the Borrower Representative or the Administrative Agent as will enable the Borrower Representative or the Administrative Agent as will enable the Borrower Representative or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section 3.01(f)(ii)(A). (ii)(B) and (ii)(D) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.

(ii) Without limiting the generality of the foregoing,

(A) any Lender that is a U.S. Person shall deliver to the Borrower Representative and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower Representative or the Administrative Agent), executed originals of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding tax;

(B) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower Representative and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower Representative or the Administrative Agent), whichever of the following is applicable:

(1) in the case of a Foreign Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Loan Document, executed originals of IRS Form W-8BEN or IRS Form W-8BEN-E, as applicable, establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under any Loan Document, IRS Form W-8BEN or IRS Form W-8BEN-E, as applicable, establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;

(2) executed originals of IRS Form W-8ECI;

(3) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the Internal Revenue Code, (x) a certificate substantially in the form of <u>Exhibit F-1</u> to the effect that such Foreign Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Internal Revenue Code, a "10 percent shareholder" of a Borrower within the meaning of Section 881(c)(3)(B) of the Internal Revenue Code, or a "controlled foreign corporation" described in Section 881(c)(3)(C) of the Internal Revenue Code (a "<u>U.S. Tax Compliance Certificate</u>") and (y) executed originals of IRS Form W-8BEN or IRS Form W-8BEN-E, as applicable; or

(4) to the extent a Foreign Lender is not the beneficial owner, executed originals of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN or IRS Form W-8BEN-E, as applicable, a U.S. Tax Compliance Certificate substantially in the form of <u>Exhibit F-2</u> or <u>Exhibit F-3</u>, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; <u>provided</u> that if the Foreign Lender is a partnership and one or more direct or indirect partners of such Foreign Lender are claiming the portfolio interest exemption, such Foreign Lender may provide a U.S. Tax Compliance Certificate substantially in the form of <u>Exhibit F-4</u> on behalf of each such direct and indirect partner;

(C) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower Representative and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower Representative or the Administrative Agent), executed originals of any other form prescribed by applicable Law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable Law to permit the Borrower Representative or the Administrative Agent to determine the withholding or deduction required to be made; and

(D) if a payment made to a Lender under any Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Internal Revenue Code, as applicable), such Lender shall deliver to the Borrower Representative and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by the Borrower Representative or the Administrative Agent such documentation prescribed by applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Internal Revenue Code) and such additional documentation reasonably requested by the Borrower Representative or the Administrative Agent as may be necessary for the Borrower Representative and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the Borrower Representative and the Administrative Agent in writing of its legal inability to do so.

(g) <u>Treatment of Certain Refunds</u> If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this <u>Section 3.01</u> (including by the payment of additional amounts pursuant to this <u>Section 3.01</u>), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this Section with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnifying party, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this paragraph (g) (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) in the event that such indemnified party is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this paragraph (g), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this paragraph (g) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This paragraph shall not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.

(h) <u>Survival</u>. Each party's obligations under this <u>Section 3.01</u> shall survive the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.

**3.02** Illegality. If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable Lending Office to make, maintain or fund LIBOR Loans, or to determine or charge interest rates based upon the LIBO Rate, or any Governmental Authority has imposed material restrictions on the authority of such Lender to purchase or sell, or take deposits of Dollars in the London interbank market, then, on notice thereof by such Lender to the Borrowers through the Administrative Agent, any obligation of such Lender to make or continue LIBOR Loans or to convert Base Rate Loans to LIBOR Loans shall be suspended until such Lender notifies the Administrative Agent and the Borrowers that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Borrowers shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay or, if applicable, convert all LIBOR Loans of such Lender to Base Rate Loans, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such LIBOR Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such LIBOR Loans. Upon any such prepayment or conversion, the Borrowers shall also pay accrued interest on the amount so prepaid or converted. Each Lender agrees to use reasonable efforts consistent with legal and regulatory requirements to designate a different Lending Office if such designation will avoid the need for such notice and will not, in the good faith judgment of such Lender, otherwise be disadvantageous to such Lender or cost any additional amount.

# 3.03 <u>Inability to Determine Rate; Alternate Rate of Interest</u> (a) If prior to the commencement of any Interest Period for a Borrowing of LIBOR Loans:

(a) (i) If, at any time, the Administrative Agent determines (which determination shall be conclusive absent manifest error) that, or, solely with respect to the circumstances described in clause (b) of this Section 3.03, the Required Lenders notify the Administrative Agent (with a copy to the Borrower Representative) that the Required Lenders have determined, that for any reason (i) adequate and reasonable means do not exist for ascertainingdetermining the LIBO Base Rateor the LIBO Rate, as applicable (including, without limitation, because the LIBO Screen Rate is not available or published on a current basis) for suchany requested Interest Period or with respect to a proposed LIBOR Loa(ii) the Administrative Agent is advised by the Required Lenders that (including because the applicable LIBO Base Rate is not available or published on a current basis), (ii) the LIBO Base Rateor the LIBO Rate, as applicable, for such any requested Interest Period or base Rate is not available or published on a current basis) for such any requested for such any requested Interest Period will with respect to a proposed LIBOR Loan doesnot adequately and fairly reflect the cost to such the Lenders (or Lender) of making or maintaining their Loans (or its Loan) at the LIBO Base Rate or the LIBO Rate, as applicable for Loan, (ii) deposits in Dollars (in the applicable amounts) are not being offered to the Administrative Agent in the London Interbank Offered Rate market for any requested Interest Period;

then the Administrative Agent shall give notice thereof to with respect to a proposed LIBOR Loan, or (iv) the making or funding of LIBOR Loans has become impracticable, the Administrative Agent will promptly notify the Borrower Representative and the Lenders by telephone or telecopy as promptly as practicable thereafter and all Lenders. Thereafter, (x) the obligation of the Lenders to make or maintain LIBOR Loans shall be suspended and (y) the LIBO Base Rate shall no longer be utilized in determining the Base Rate, in each case, until the Administrative Agent notice. Upon receipt of any such notice, the Borrower Representative and the Lenders that the circumstances giving rise to such notice no longer exist, (A) anymay revoke any pending request for a Borrowing of, conversion to or continuation of LIBOR Loans or, failing that, will be deemed to have converted such request pursuant to Section 2.02 to convert any Loan to

LIBOR Loans, or continue any LIBOR Loans as LIBOR Loans shall be ineffective and (B) if any request pursuant <u>Soction 2.02</u> requests a borrowing of a LIBOR Loan, such borrowing shall be made as into a request for a Borrowing of Base Rate Loan Loans in the amount specified therein.

# (b) If, at any time :

(i) the Administrative Agent determines (which determination shall be conclusive absent manifest error) that (i1) any of the circumstances set forthdescribed in clause (i), (ii), (iii) or (iv) of Section 3.03(a)(i) have arisen and such circumstances are unlikely to be temporary or (ii2) the circumstances set forth in clause (a)U.S. dollar-denominated syndicated credit facilities are being executed or amended to incorporate or adopt a new alternative interest rate to replace the LIBO Base Rate;

(ii) a public statement or publication of information has been made (iA) have not arisen but the by or on behalf of the administrator of the LIBO Base Rate, the regulatory supervisor for the administrator of the LIBO Base Rate, the regulatory supervisor for the administrator of the LIBO Base Rate, a resolution authority with jurisdiction over the administrator for the LIBO Base Rate, a resolution authority over the administrator for the LIBO Base Rate, in each case which states that the administrator of the LIBO Base Rate has ceased or will cease to provide the LIBO Base Rate permanently or indefinitely, provided that, at the time of the statement or publication, there is no successor administrator that will continue to provide the LIBO Base Rate, or (B) by the regulatory supervisor for the administrator of the LIBO Base Rate or anyGovernmental Authority having jurisdiction over the Administrative Agent has made a public statement identifying a specific date after which announcing that the LIBOSereenBase Rate shallis no longer representative or may no longer be used for determining interest rates for loans,; or

(iii) the Required Lenders notify the Administrative Agent, with a copy to the Borrower, that:

(A) the circumstance described in clause (ii) of Section 3.03 has arisen and such circumstance is unlikely to be temporary; or

(B) the Required Lenders have determined that U.S. dollar-denominated syndicated credit facilities are being executed or amended to incorporate or adopt a new alternative interest rate to replace the LIBO Base Rate;

then the Administrative Agent and the Borrower Representative shall endeavor to establishagree upon an alternate rate of interest to the LIBO Base Rate that gives and agree on the replacement spread(s) and floor(s) (if any) applicable thereto, giving due consideration to the any evolving or then -prevailing market convention for determining a replacement rate of interest or spread for similar U.S. Dollar denominated syndicated loans in the United Statescredit facilities (which replacement spread may be a positive or negative value or zero) or any selection or recommendation of a replacement rate and/or spread or the mechanism for determining such a rate or spread by the relevant Governmental Authority in effect at such time, and shall enter into an amendment to this Agreement to reflect such alternate rate of interest, replacement spread(s) and floor(s) (if any) applicable thereto and such other related changes to this Agreement as may be applicable (but for the avoidance of doubtappropriate; provided that such amendment shall provide that if such alternate rate of interest shall at any time be less than zero, such related changes rate shall not include a reduction of the Applicable Marginbe zero for purposes of this Agreement. Such alternate rate of interest may include the forward-looking term rate based on the secured

Notwithstanding anything to the contrary in Section 12.01, such amendment shall become effective without any further (c) action or consent of any other party to this Agreement so long as the Administrative Agent shall not have received, within five (5) on or before the fifth Business Days of Day after the date notice of such alternate rate of interest is provided to the Administrative Agent shall have notified all Lenders of such proposed amendment, a written notice from the Required Lenders stating that such Required Lenders object to such amendment. Until an If no alternate rate of interest shall be determined agreed upon and implemented pursuant to an amendment to this Agreement in accordance with this elause (b) (but, in Section 3.03, and one or more of the ease of circumstances or events described in clause (iib) of the first sentence of this elause Section 3.03 have arisen or occurred, the Administrative Agent will promptly so notify the Borrower Representative and each Lender. Thereafter, (bx), only the obligation of the Lenders to make or maintain LIBOR Loans shall be suspended to the extent of the LIBO Sereen Rate for such frected LIBOR Loans or Interest Period is not available or published at such time on a current basis), (x) any request pursuant to Section 2.02 to convert any Loan to LIBOR Loans, or continue any LIBOR Loans as LIBOR Loans shall be ineffective and (y) if Periods), and (y) the LIBO Base Rate component shall no longer be utilized in determining the Base Rate. Upon receipt of such notice, the Borrower Representative may revoke any pending request pursuant to Section 2.02 requests a borrowing for a Borrowing of, conversion to or continuation of a-LIBORLoan, such borrowing shall be made as a Base Rate Loan; provided that, if such alternate rate of interest shall be less than zero, such rate shall be deemed to be zero for the purposes of Loans (to the extent of the affected LIBOR Loans or Interest Periods) or, failing that, will be deemed to have converted such request into a request for a Borrowing of Base Rate Loans (subject to the foregoing clause (y)) in the amount specified therein.

(d) In connection with the implementation of an alternate rate of interest to the LIBO Base Rate and/or replacement spread(s) applicable thereto, the Administrative Agent will have the right to make any technical, administrative or operational changes (including changes to the definition of Base Rate, the definition of Interest Period, timing and frequency of determining rates and making payments of interest and other administrative matters) that the Administrative Agent decides in its reasonable discretion are appropriate to reflect the adoption and implementation of such alternate rate of interest and/or replacement spread(s) and to permit the administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such alternate rate of interest and/or replacement spread(s) exists, in such other manner of administration as the Administrative Agent decides in consultation with the Borrower Representative is reasonably necessary in connection with the administration of this Agreement) from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such changes will become effective without any further action or consent of any other party to this Agreement.

#### 3.04 Increased Cost and Reduced Return; Capital Adequacy.

(a) If any Change in Law shall:

(i) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge, liquidity requirement or other similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender (except any reserve requirement reflected in the LIBO Rate) or the L/C Issuer or Support Provider;

(ii) subject any Lender or the L/C Issuer or Support Provider to any Tax of any kind whatsoever with respect to this Agreement, any Letter of Credit or Support Agreement, any participation in a Letter of Credit or any LIBOR Loan made by it, or change the basis of taxation of payments to such Lender or the L/C Issuer or Support Provider in respect thereof (other than (A) Indemnified Taxes, (B) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes and (C) Connection Income Taxes); or

(iii) impose on any Lender, the Support Provider or the L/C Issuer or the London interbank market any other condition, cost or expense affecting this Agreement or LIBOR Loans made by such Lender or any Letter of Credit or Support Agreement, or participation therein;

and the result of any of the foregoing shall be to increase the cost to such Lender of making or maintaining any LIBOR Loan (or of maintaining its obligation to make any such Loan), or to increase the cost to such Lender or the Support Provider or L/C Issuer of participating in, issuing or maintaining any Lender Letter of Credit or continuing its obligation under any Support Agreement (or of maintaining its obligation to participate in or to issue any Letter of Credit), or to reduce the amount of any sum received or receivable by such Lender or the L/C Issuer hereunder (whether of principal, interest or any other amount), in each case by an amount deemed by such Lender to be material, then, upon request of such Lender, Support Provider, or the L/C Issuer (with a copy to the Administrative Agent), the Borrowers will pay to such Lender, Support Provider, or L/C Issuer, as the case may be, such additional amount or amounts as will compensate such Lender, Support Provider or the L/C Issuer and the case incurred or reduction suffered.

(b) Without duplication of amounts payable in <u>paragraph (a)</u> above, if any Lender determines that any Change in Law affecting such Lender or any lending office of such Lender or Lender's holding company, if any, regarding capital or liquidity requirements has or would have the effect of reducing the rate of return on such Lender's capital or on the capital of such Lender's holding company, if any, as a consequence of its obligations hereunder or under or in respect of any Letter of Credit or Support Agreement, to a level below that which such Lender or such Lender's holding company could have achieved but for such Change in Law (taking into consideration such Lender's policies and the policies of such Lender's holding company with respect to capital or liquidity adequacy), then from time to time the Borrowers will pay to such Lender such additional amount or amounts as will compensate such Lender or such Lender's holding company for any such reduction suffered.

**3.05 Funding Losses**. Upon demand of any Lender (with a copy to the Administrative Agent) from time to time, the Borrowers shall promptly compensate such Lender for and hold such Lender harmless from any loss, cost or expense incurred by it as a result of:

(a) any continuation, conversion, payment or prepayment of any Loan other than a Base Rate Loan on a day other than the last day of the Interest Period for such Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);

(b) any failure by the Borrowers (for a reason other than the failure of such Lender to make a Loan) to prepay, borrow, continue or convert any Loan other than a Base Rate Loan on the date or in the amount notified by the Borrowers; or

(c) any assignment of a LIBOR Loan on a day other than the last day of the Interest Period therefor as a result of (i) a request by the Borrowers pursuant to <u>Section 12.15</u> or (ii) an assignment by any Lender that is a Lender on the Closing Date pursuant to <u>Section 12.07(b)</u> as part of the primary syndication of the Commitments and Loans following the Closing Date;

including any loss of anticipated profits and any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Loan or from fees payable to terminate the deposits from which such funds were obtained. Lenders shall provide Borrower Representative with a notice (with a copy to the Administrative Agent) setting forth in reasonable detail the basis for Lenders demand, which shall be conclusive absent manifest error. Borrowers shall pay such amount within ten (10) days after receipt of such notice.

For purposes of calculating amounts payable by the Borrowers to the Lenders under this <u>Section 3.05</u>, each Lender shall be deemed to have funded each LIBOR Loan made by it at the LIBO Base Rate used in determining the LIBO Rate for such Loan by a matching deposit or other borrowing in the London interbank market for a comparable amount and for a comparable period, whether or not such LIBOR Loan was in fact so funded.

#### 3.06 <u>Matters Applicable to all Requests for Compensation</u>.

(a) A certificate of the Administrative Agent or any Lender, Support Provider, or L/C Issuer claiming compensation under this <u>Article 3</u> and setting forth the additional amount or amounts to be paid to it hereunder and the calculation thereof in reasonable detail shall be conclusive in the absence of manifest error. In determining such amount, the Administrative Agent or such Lender, Support Provider or L/C Issuer may use any reasonable averaging and attribution methods. The Borrowers shall pay the Administrative agent, Lender, Support Provider or L/C Issuer the amount shown as due on any such certificate within ten (10) days of receipt thereof.

(b) Upon any Lender's making a claim for compensation under <u>Section 3.01</u> or <u>3.04</u>, the Borrowers may replace such Lender in accordance with <u>Section 12.15</u>.

3.07 <u>Survival</u>. All of the Borrowers' obligations under this Article 3 shall survive the Termination Date.

# **ARTICLE 4**

#### CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

**4.01** <u>Conditions of Initial Credit Extension</u> The obligation of each Lender to make its initial Credit Extension hereunder is subject to satisfaction of the following conditions precedent, except as provided under <u>Section 6.18</u>:

(a) <u>Loan Documents</u> Receipt by the Administrative Agent of executed counterparts of this Agreement, the Security Agreement, the Fee Letter, the other Loan Documents to be executed as of the Closing Date and the agreements and the other documents executed in connection herewith and therewith and the Notes (if requested), each properly executed by a Responsible Officer of the signing Loan Party and each other Person a party thereto.

(b) <u>Organization Documents, Resolutions, Etc</u> Receipt by the Administrative Agent of the following, each of which shall be originals or facsimiles (followed promptly by originals), dated as of a recent date before the Closing Date and in form and substance satisfactory to the Administrative Agent and its legal counsel:

(i) copies of the Organization Documents of each Loan Party certified to be true and complete as of a recent date by the appropriate Governmental Authority of the state or

other jurisdiction of its incorporation or organization, where applicable, and certified by a secretary or assistant secretary of such Loan Party to be true and correct as of the Closing Date;

(ii) such resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of each Loan Party as the Administrative Agent may require evidencing the identity, authority and capacity of each Responsible Officer thereof (A) executing any agreement, certificate or other document required to be delivered hereby or (B) authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Documents to which such Loan Party is a party; and

(iii) such documents and certifications as the Administrative Agent may reasonably require to evidence that each Loan Party is duly organized or formed, and is validly existing, in good standing and qualified to engage in business in its state of organization or formation, in the state in which its principal place of business is located, and in each other state in which a failure to be so qualified would have a Material Adverse Effect.

(c) <u>Filings, Registrations and Recordings</u> Receipt by the Administrative Agent of each document (including any UCC financing statements) required by the Collateral Documents or under Law or reasonably requested by the Administrative Agent and an intellectual property security agreement as to all intellectual property rights of a Loan Party registered with the United States Patent and Trademark Office or the United States Copyright Office to be filed, registered or recorded in order to create in favor of the Administrative Agent, for its benefit and the benefit of the Secured Parties, a perfected Lien on the Collateral described therein, prior and superior in right to any other Person (other than Permitted Liens), which shall be in proper form for filing, registration or recordation.

(d) <u>Pledged Stock; Stock Powers; Pledged Notes</u>. Receipt by the Administrative Agent of (i) any certificates representing the shares of Capital Stock pledged pursuant to the Security Agreement, together with an undated stock (or analogous) power for each such certificate executed in blank by a duly authorized officer of the pledgor thereof or, with respect to any uncertificated security, an Uncertificated Security Control Agreement executed by a pledgor and acknowledged by the applicable Loan Parties, and (ii) each promissory note (if any) pledged to the Administrative Agent pursuant to the Security Agreement endorsed (without recourse) in blank (or accompanied by an executed transfer form in blank) by the pledgor thereof.

(e) <u>Real Estate Documents</u>. Receipt by the Administrative Agent, with respect to each Mortgaged Property owned, if any, by a Loan Party as of the Closing Date, of all Mortgage Instruments and Mortgage Supporting Documents relating thereto.

(f) <u>Opinions of Counsel</u> Receipt by the Administrative Agent of favorable opinions of K&L Gates LLP, counsel to the Loan Parties, addressed to the Administrative Agent and each Secured Party, dated as of the Closing Date, and in form and substance satisfactory to the Administrative Agent.

(g) <u>Evidence of Insurance</u>. Receipt by the Administrative Agent of ACORD insurance evidencing insurance coverages and amounts satisfactory to the Administrative Agent and appropriate endorsements in favor of the Administrative Agent with respect thereto.

(h) <u>Liens Searches</u>. Receipt by the Administrative Agent of UCC and other Lien searches considered necessary by the Administrative Agent and other evidence as requested by Administrative Agent that no Liens exist other than Permitted Liens.

(i) <u>Transaction Documents</u>. The Transaction Documents shall be in form and substance reasonably satisfactory to the Administrative Agent and shall be in full force and effect. Concurrently with the initial funding of the Loans on the Closing Date, the Transactions shall have been consummated in accordance with the terms of the Transaction Documents.

#### (j) [Reserved].

(k) <u>Third Party Consents; Change of Ownership</u>. Receipt by the Administrative Agent of evidence reasonably satisfactory to the Administrative Agent that the Loan Parties have obtained all required consents and approvals of all Persons including all requisite Governmental Authorities and counterparties to Material Contracts, to the execution, delivery and performance of the Loan Documents.

(1) <u>Fees</u>. Receipt by the Administrative Agent and the Lenders of any fees required to be paid on or before the Closing Date under this Agreement and the Fee Letter.

(m) <u>Attorney Costs</u>. The Loan Parties shall have paid all reasonable Attorney Costs of the Administrative Agent, plus such additional amounts of Attorney Costs as shall constitute its reasonable estimate of Attorney Costs incurred or to be incurred by it through the closing proceedings (<u>provided</u> that such estimate shall not thereafter preclude a final settling of accounts between the Loan Parties and the Administrative Agent).

(n) <u>Compliance with Laws</u> Each Loan Party shall be in compliance in all material respects with all applicable Law, and shall have provided the Administrative Agent with true and correct copies of each of the accreditations, license and, certifications required by <u>Section 5.01</u> below.

(o) <u>Compliance with Agreements</u>. Each Loan Party shall be in compliance in all material respects with all<del>material agreements, and shall have provided the Administrative Agent with true and correct copies of each Material Contracts.</del>

(p) <u>No Litigation</u> There exists no pending or threatened Proceeding against the Loan Parties or any of their respective Affiliates or respective assets in any court or administrative forum, (i) which could reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect or (ii) that involves this Agreement or any other Loan Document.

- (q) <u>Financial Statements; Projections</u>. Receipt by the Administrative Agent of
  - (i) the Audited Pre-Closing Financial Statements;
  - (ii) the Interim Pre-Closing Financial Statements;
  - (iii) satisfactory projections through December 31, 2023;

(iv) pro forma financial statements for the twelve-month period ending no more than thirty (30) days prior to the Closing Date for the Consolidated Group on a consolidated basis; and request.

- (v) such other information as the Administrative Agent may reasonably
- (r) [Reserved].

(s) <u>No Material Adverse Change</u> There shall not have occurred since December 31, 2017 any developments or events which individually or in the aggregate with other such circumstances has had or could reasonably be expected to have a Material Adverse Effect with respect to the Loan Parties and their respective Subsidiaries, taken as a whole, or any of their respective assets.

(t) <u>Closing Certificate</u>. Receipt by the Administrative Agent of a certificate executed by a Responsible Officer of the Borrower Representative certifying that the conditions specified in <u>Sections 4.01(s)</u> and <u>(v)-(dd)</u> and <u>Sections 4.02(a)</u>, <u>(b)</u>, <u>(c)</u> and <u>(d)</u> have been satisfied and that the representations and warranties contained in <u>Article 5</u> are true and correct in all material respects (provided, that if any representation or warranty is by its terms qualified by concepts of materiality, such representation and warranty shall be true and correct in all respects) as of the Closing Date.

(u) [Reserved].

(v) <u>Investment Documents</u>. The Administrative Agent shall have received confirmation of ownership and capital structure of the Loan Parties and be reasonably satisfied with the constituent documents of the Loan Parties and related investment agreements. All preferred equity securities shall have a maturity or redemption date at least six (6) months after the Term Loan Maturity Date and shall otherwise not constitute Disqualified Capital Stock.

(w) <u>Patriot Act</u>.

(i) Receipt by the Administrative Agent and Lenders, at least ten (10) days prior to the Closing Date, of all documentation and other information about the Loan Parties required under applicable "know your customer" and anti-money laundering rules and regulations, including the PATRIOT Act, that has been requested by the Administrative Agent and the Lenders at least ten (10) Business Days prior to the Closing Date.

(ii) Receipt by the Administrative Agent and Lenders, at least five (5) days prior to the Closing Date, a Beneficial Ownership Certification in relation to any Borrower that qualifies as a "legal entity customer" under the Beneficial Ownership Regulation.

(x) [<u>Reserved</u>].

(y) <u>Solvency</u>. Receipt by the Administrative Agent of a Solvency Certificate from the Chief Financial Officer of Intermediate Holdings, substantially in the form of <u>Exhibit E</u> hereto.

(z) [<u>Reserved</u>].

(a a) <u>Minimum EBITDA</u> Receipt by the Administrative Agent of a certificate of the Borrower Representative's chief financial officer, in form, substance and detail satisfactory to the Administrative Agent demonstrating that the Consolidated EBITDA of the Consolidated Group (as adjusted in a manner satisfactory to the Administrative Agent) for the twelve (12) month period ended February 28, 2019 is equal to or greater than \$119,794,000.00.

(bb) <u>Maximum Closing Date Indebtedness</u> The Consolidated Group shall have maximum Indebtedness outstanding after giving effect to initial funding of the Loans on the Closing Date in an amount not in excess of fifty percent (50.0%) of the total capitalization of Borrower.

(cc) <u>Maximum Closing Date Leverage</u> Receipt by the Administrative Agent of a certificate of the Borrower Representative's chief financial officer, in form, substance and detail

satisfactory to the Administrative Agent, demonstrating that the Consolidated Total Leverage Ratio for the twelve (12) month period ended February 28, 2019, does not exceed 2.70:1.00 calculated on a Pro Forma Basis after giving effect to the initial funding of the Loans (and the application of the proceeds thereof) and based on Consolidated EBITDA determined pursuant to clause (aa) above.

(dd) <u>Other</u>. Receipt by the Administrative Agent and the Lenders of such other documents, instruments, agreements and information as reasonably requested by the Administrative Agent or any Lender, including, but not limited to, information regarding litigation, tax, accounting, labor, insurance, pension liabilities (actual or contingent), real estate leases, environmental matters, <u>material contracts</u><u>Material</u><u>Contracts</u>, debt agreements, property ownership, contingent liabilities, employment agreements, non-compete agreements and management of the Loan Parties and their respective Subsidiaries.

**4.02** <u>Conditions to all Credit Extensions</u>. The obligation of each Lender and the L/C Issuer and Support Provider to honor any Request for Credit Extension (or provide a Support Agreement), whether on the Closing Date or at any time thereafter, is subject to the following conditions precedent:

(a) The representations and warranties of each Loan Party contained in <u>Article 5</u> or any other Loan Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct (i) if such date is the Closing Date, on and as of such date and (ii) otherwise, in all material respects (provided, that if any representation or warranty is by its terms qualified by concepts of materiality, such representation and warranty shall be true and correct in all respects) on and as of such date, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct as of such earlier date, and except that for purposes of this <u>Section 4.02</u>, the representations and warranties contained in subsections (a) and (b) of <u>Section 5.05</u> shall be deemed to refer to the most recent statements furnished pursuant to clauses (a) and (b), respectively, of <u>Section 6.01</u>.

(b) No Default or Event of Default shall exist, or would result from such proposed Credit Extension.

(c) After giving effect to such Credit Extension, (i) the total Revolving Exposures shall not exceed the total Revolving Commitments and (ii) the Loan Parties shall be in compliance on a Pro Forma Basis with the financial covenants set forth in <u>Article 8</u> computed using the covenant levels and financial information for the most recently ended quarter for which information is available.

(d) The Administrative Agent and, if applicable, the applicable L/C Issuer (or the Support Provider, as the case may be) shall have received a Request for Credit Extension in accordance with the requirements hereof.

(e) The obligation of each Revolving Lender holding a Revolving Commitment to honor a Loan Notice for a Borrowing of a Revolving Loan to finance a Permitted Acquisition or fees and transaction costs associated with such Permitted Acquisitions is further subject to receipt by the Administrative Agent of a certificate of the Borrower Representative's chief financial officer, in form, substance and detail satisfactory to Administrative Agent, demonstrating that the Consolidated Total Leverage Ratio on the date of such Borrowing does not exceed 2.703.00 to 1.00 calculated on a Pro Forma Basis after giving effect to such Borrowing and related Permitted Acquisition and using the financial information for the most recently ended quarter for which information is available.

(f) The obligation of each Term Loan Lender holding a Delayed – Draw Term Loan Commitment to honor a Loan Notice for a Borrowing of a Delayed – Draw Term Loan is further subject to:

(i) No Loan Notice for a Borrowing of a Delayed – Draw Term Loan may be requested at any time after the second anniversary of the ClosingSecond Amendment Effective Date, and no Lender shall have any obligation to honor a request for such Borrowing after such date.

satisfied.

(ii)

Evidence satisfactory to Administrative Agent that the requirements for a "Permitted Acquisition" have been

(iii) Receipt by the Administrative Agent of a certificate of the Borrower Representative's chief financial officer, in form, substance and detail satisfactory to Administrative Agent, demonstrating that the Consolidated Total Leverage Ratio on the date of such Borrowing does not exceed 2.703.00 to 1.00 calculated on a Pro Forma Basis after giving effect to such Borrowing and related Permitted Acquisition and using the financial information for the most recently ended quarter for which information is available.

Each Request for Credit Extension submitted by the Borrower Representative shall be deemed to be a representation and warranty by the Loan Parties that the conditions specified in <u>Section 4.02</u> have been satisfied on and as of the date of the applicable Credit Extension.

**4.03** Satisfaction of Conditions. In determining the satisfaction of the conditions specified in this Article 4, to the extent any item is required to be satisfactory to (a) any individual Lender, such item shall be deemed satisfactory to each Lender which has not notified the Administrative Agent in writing (with reasonable detail relating to why such Lender views a condition as not having been satisfied) prior to the Closing Date or the date on which a Credit Extension has been requested to be made, as applicable, that the respective item or matter does not meet its satisfaction or (b) the Required Lenders, such item shall be deemed satisfactory to the Required Lenders unless the Required Lenders have notified the Administrative Agent in writing (with reasonable detail relating to why the Required Lenders view a condition as not having been satisfied) prior to the Closing Date or the date on which a Credit Extension has been requested to be made, as applicable, that the respective item or matter does not meet their satisfaction, in each case, regardless of whether Administrative Agent has knowledge that such condition is satisfied, provided, that the occurrence of the Closing Date or the making of a Credit Extension, the execution and delivery to the Administrative Agent by a Lender of a counterpart of this Agreement (or if applicable, an Assignment and Assumption) shall be deemed confirmation by such Lender that (a) the decision of such Lender to execute and deliver to the Administrative Agent an executed counterpart of this Agreement (or become a Lender hereunder) was made by such Lender independently and without reliance on Administrative Agent or any other Lender and (b) all documents made available to such Lender were acceptable to such Lender. No Credit Extension shall release any Loan Party from any liability for failure to satisfy one or more of the applicable conditions contained in this <u>Article 4</u>.

#### **ARTICLE 5**

#### **REPRESENTATIONS AND WARRANTIES**

The Loan Parties hereby represent and warrant to the Administrative Agent and the Lenders that, both immediately before and after giving effect to any Permitted Acquisition, any Restricted Payment and any Credit Extension:

**5.01** <u>Existence, Qualification and Power</u> Each Loan Party (a) is a corporation, partnership or limited liability company duly organized or formed, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization, (b) has all requisite power and authority and all requisite Permits to (i) own its assets and carry on its business and (ii) execute, deliver and perform its

obligations under the Loan Documents to which it is a party, except where the failure to have such Permits, either singularly or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, and (c) is duly qualified and is licensed and in good standing under the Laws of each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification or license, except to the extent that the failure to do so could not reasonably be expected to result in a Material Adverse Effect.

**5.02** <u>Authorization; No Contravention</u>. The execution, delivery and performance by each Loan Party of each Loan Document to which such Person is party, have been duly authorized by all necessary corporate or other organizational action, and do not (a) contravene the terms of any Loan Party's Organization Documents; (b) conflict with or result in any breach or contravention of, or the creation of any Lien under, (i) any Contractual Obligation under any Material Contract to which any Loan Party is a party or (ii) any order, injunction, writ or decree of any Governmental Authority or any arbitral award to which any Loan Party or the Property of any Loan Party is subject; (c) violate any Law (including Regulation U or Regulation X issued by the FRB); or (d) result in a limitation on any material licenses, permits or other Governmental Approvals applicable to the business, operations or properties of any Loan Party.

**5.03** <u>Governmental Authorization; Other Consents</u> No approval, consent, exemption, authorization, or other action by, or notice to, or filing with, any Governmental Authority or any other Person is necessary or required in connection with the execution, delivery or performance by, or enforcement against, any Loan Party of this Agreement or any other Loan Document, other than (i) those that have already been obtained and are in full force and effect and (ii) filings to perfect the Liens created by the Collateral Documents.

**5.04** <u>Binding Effect</u>. Each Loan Document has been duly executed and delivered by each Loan Party that is party thereto. Each Loan Document constitutes a legal, valid and binding obligation of each Loan Party that is party thereto, enforceable against each such Loan Party in accordance with its terms, except as enforceability may be limited by applicable Debtor Relief Laws or by equitable principles relating to enforceability.

#### 5.05 Financial Statements; No Material Adverse Effect.

(a) The Audited Pre-Closing Financial Statements (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; (ii) fairly present in all material respects the financial condition of the Loan Parties and their Subsidiaries as of the date thereof and their results of operations for the period covered thereby in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; and (iii) show all material Indebtedness and other liabilities, direct or contingent, of the Loan Parties and their Subsidiaries as of the date thereof, including liabilities for taxes, commitments and Indebtedness.

(b) The Interim Pre-Closing Financial Statements (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; (ii) fairly present in all material respects the financial condition of the Loan Parties and their Subsidiaries as of the date thereof and their results of operations for the period covered thereby, subject, in the case of clauses (i) and (ii), to the absence of footnotes and to normal year-end audit adjustments; and (iii) show all material Indebtedness and other liabilities, direct or contingent (to the extent required by GAAP), of the Loan Parties and their Subsidiaries as of the date thereof, including liabilities for taxes, material commitments and Indebtedness.

(c) From the date of the Audited Pre-Closing Financial Statements and the Interim Pre-Closing Financial Statements to and including the Closing Date, there has been no Disposition by the Loan Parties and their Subsidiaries, or any Involuntary Disposition, of any material part of the business or Property of the Loan Parties and their Subsidiaries, taken as a whole, and no purchase or other acquisition by any of them of any business or property (including any Capital Stock of any other Person) material in relation to the consolidated financial condition of the Loan Parties and their Subsidiaries, taken as a whole, in each case, which is not reflected in the foregoing financial statements or in the notes thereto and has not otherwise been disclosed in writing to the Lenders on or prior to the Closing Date.

(d) The financial statements delivered pursuant to Sections 6.01(a) and 6.01(b) have been prepared in accordance with GAAP (except as may otherwise be permitted under Sections 6.01(a) and 6.01(b)) and present fairly (on the basis disclosed in the footnotes to such financial statements) in all material respects the consolidated financial condition, results of operations and cash flows of the Loan Parties and their respective Subsidiaries as of the dates thereof and for the periods covered thereby.

(e) Since December 31, 2017, there has been no event or circumstance that has had or could reasonably be expected to have a Material Adverse Effect.

**5.06** <u>Litigation</u>. There are no Proceedings pending or, to the knowledge of the Loan Parties after due investigation, threatened in writing, at law, in equity, in arbitration or before any Governmental Authority, by or against any Loan Party or any of its Subsidiaries or against any of their properties or revenues that (a) purport to affect or pertain to this Agreement or any other Loan Document, or (b) could reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

#### 5.07 <u>No Default</u>.

(a) No Loan Party is (i) in breach of or in default under any Material Contract, or (ii) in breach of or in default under any Contractual Obligation that could reasonably be expected to have a Material Adverse Effect.

(b) No Default or Event of Default has occurred and is continuing.

5.08 <u>Ownership of Property: Liens</u> Each of the Loan Parties and its Subsidiaries has good record and marketable title in fee simple to, or valid leasehold interests in, all Real Property necessary or used in the ordinary conduct of its business. No Property of the Loan Parties and their Subsidiaries is subject to any Liens, other than Permitted Liens.

5.09 <u>Environmental Compliance</u>. Except as could not reasonably be expected to have a Material Adverse Effect:

(a) Each of the Facilities and all operations at the Facilities are in compliance with all applicable Environmental Laws, and there is no violation of any Environmental Law with respect to the Facilities or the Businesses, and there are no conditions relating to the Facilities or the Businesses that could give rise to liability under any applicable Environmental Laws.

(b) None of the Facilities contains, or has previously contained, any Hazardous Materials at, on or under the Facilities in amounts or concentrations that constitute or constituted a violation of, or could give rise to liability under, Environmental Laws.

(c) Neither any Loan Party nor any Subsidiary has received any written or verbal notice of, or inquiry from any Governmental Authority regarding, any violation, alleged violation, non-compliance, liability or potential liability regarding environmental matters or compliance with Environmental Laws with regard to any of the Facilities or the Businesses, nor does any Responsible Officer of any Loan Party have knowledge or reason to believe that any such notice will be received or is being threatened.

(d) Hazardous Materials have not been transported or disposed of from the Facilities, or generated, treated, stored or disposed of at, on or under any of the Facilities or any other location, in each case by or on behalf of any Loan Party or any Subsidiary in violation of, or in a manner that would be reasonably likely to give rise to liability under, any applicable Environmental Law.

(e) No judicial proceeding or governmental or administrative action is pending or, to the knowledge of the Responsible Officers of the Loan Parties, threatened, under any Environmental Law to which any Loan Party or any Subsidiary is or will be named as a party, nor are there any consent decrees or other decrees, consent orders, administrative orders or other orders, or other administrative or judicial requirements outstanding under any Environmental Law with respect to any Loan Party, any Subsidiary, the Facilities or the Businesses.

(f) There has been no release or, threat of release of Hazardous Materials at or from the Facilities, or arising from or related to the operations (including disposal) of any Loan Party or any Subsidiary in connection with the Facilities or otherwise in connection with the Businesses, in violation of or in amounts or in a manner that could give rise to liability under Environmental Laws.

**5.10** <u>Insurance</u>. The properties of the Loan Parties and their Subsidiaries are insured with financially sound and reputable insurance companies (none of which are Affiliates of the Loan Parties), in such amounts, with such deductibles and covering such risks as are customarily carried by companies engaged in similar businesses and owning similar properties in localities where the applicable Loan Party or the applicable Subsidiary operates. The insurance coverage of the Loan Parties complies with the requirements of <u>Section 6.07</u> and the insurance coverage in effect on the <u>ClosingSecond Amendment Effective</u> Date is outlined as to carrier, policy number, expiration date, type, amount and deductibles on <u>Schedule 5.10</u>.

**5.11** <u>Taxes</u>. The Loan Parties and their respective Subsidiaries have filed all federal, state and other material tax returns and reports required to be filed, and have paid all federal, state and other material taxes, assessments, fees and other governmental charges levied or imposed upon them or their properties, income or assets otherwise due and payable, except those which are being Properly Contested. There is no material proposed tax assessment against any Loan Party or Subsidiary. No Loan Party nor any Subsidiary thereof is party to any tax sharing agreement.

# 5.12 ERISA Compliance.

(a) Each ERISA Plan and each Loan Party is in compliance in all material respects with the applicable provisions of ERISA, the Internal Revenue Code, the Consolidated Omnibus Budget Reconciliation Act of 1985 <u>COBRA</u>") and the regulations and published interpretations thereunder, and other federal or state Laws. Each ERISA Plan that is intended to qualify under Section 401(a) of the Internal Revenue Code has a currently effective favorable determination letter (or in the case of a volume submitter or prototype plan is the subject of a currently effective favorable opinion letter) from the IRS or an application for such letter is currently being processed by the IRS with respect thereto (and each ERISA Plan has been timely amended to reflect changes in the applicable qualification requirements under Section

401(a) of the Internal Revenue Code and any applicable IRS guidance issued thereunder) and, to the best knowledge of the Loan Parties, nothing has occurred which would prevent, or cause the loss of, such qualification. Each Loan Party and each ERISA Affiliate has made all required contributions to each ERISA Plan subject to Sections 412 and 430 of the Internal Revenue Code, and no application for a funding waiver or an extension of any amortization period pursuant to Section 412 or 430 of the Internal Revenue Code has been made with respect to any ERISA Plan. Each Loan Party and each ERISA Affiliate has performed all their obligations under each ERISA Plan according to their terms, including filing or furnishing to the IRS, Department of Labor or other Governmental Authority, or to participants or beneficiaries of each ERISA Plan, any reports, returns, notices and other documentation required to be filed or furnished.

(b) There are no pending or, to the knowledge of the Loan Parties, threatened claims, actions or lawsuits, or action by any Governmental Authority, with respect to any Plan. There has been no prohibited transaction or violation of the fiduciary responsibility rules with respect to any Plan.

(c) (i) No ERISA Event has occurred or is reasonably expected to occur; (ii) no Pension Plan has any Unfunded Pension Liability and no other ERISA Plan providing retiree welfare benefits has any unfunded liability for benefits; (iii) no Loan Party or any ERISA Affiliate has incurred, or reasonably expects to incur, any liability under Title IV of ERISA with respect to any Pension Plan (other than premiums due and not delinquent under Section 4007 of ERISA); (iv) no Loan Party or any ERISA Affiliate has incurred, or reasonably expects to incur, any liability (and no event has occurred which, with the giving of notice under Section 4219 of ERISA, would result in such liability) under Sections 4201 or 4243 of ERISA with respect to a Multiemployer Plan; and (v) no Loan Party or any ERISA Affiliate has engaged in a transaction that could be subject to Section 4069 or 4212(c) of ERISA.

**5.13** <u>Subsidiaries</u>. Set forth on <u>Schedule 5.13</u> is a complete and accurate list of the name and jurisdiction of organization of each Loan Party and each Subsidiary (including each Excluded Subsidiary which is identified as such therein) as of the <u>ClosingSecond Amendment</u> <u>Effective</u> Date, (or following the delivery of the first Compliance Certificate hereunder, as of the date of the most recently delivered Compliance Certificate), together with (a) the number of shares of each class of Capital Stock of any Loan Party outstanding as of the <u>ClosingSecond Amendment Effective</u> Date and (b) the number and percentage of outstanding shares of each class owned (directly or indirectly) by any Loan Party or any Subsidiary as of the <u>ClosingSecond Amendment Effective</u> Date. None of the shares of Capital Stock of any Subsidiary is subject to any outstanding options, warrants, rights of conversion or purchase and all other similar rights with respect thereto. The outstanding Capital Stock of each Loan Party and each Subsidiary is validly issued, and, in the case of any Loan Party that is a corporation, fully paid and non-assessable. No Subsidiary of Intermediate Holdings has outstanding any shares of Disqualified Capital Stock.

# 5.14 Margin Regulations; Investment Company Act, Use of Proceeds.

(a) The Loan Parties are not engaged and will not engage, principally or as one of its important activities, in the business of purchasing or carrying margin stock (within the meaning of Regulation U issued by the FRB), or extending credit for the purpose of purchasing or carrying margin stock. No proceeds of any Borrowing shall be used for the purpose of purchasing or carrying margin stock or to extend credit to others for the purpose of purchasing or carrying margin stock.

(b) None of the Loan Parties, any Person Controlling any Loan Party or any Subsidiary is or is required to be registered as an "investment company" under the Investment Company Act of 1940.

(c) No proceeds of any Borrowing will be used in violation of <u>Section 6.11</u>.

**5.15** <u>Disclosure</u>. No report, financial statement, certificate or other information furnished (whether in writing or orally) by or on behalf of any Loan Party to the Administrative Agent or any Lender in connection with the transactions contemplated hereby and the negotiation of this Agreement or delivered hereunder (as modified or supplemented by other information so furnished) contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading, provided that, with respect to any projected financial information, the Loan Parties represent that such information was prepared in good faith based upon assumptions believed to be reasonable at the time and at the time made available to Administrative Agent and the Lenders. Notwithstanding the foregoing, Administrative Agent and Lenders acknowledge that the pro forma financial statements and other economic forecasts and information of a general industry nature delivered by Borrowers hereunder are not factual representations and that the actual financial results of the Borrowers may differ from the pro forma financial statements and other economic forecasts submitted from time to time. As of the <u>ClosingSecond Amendment Effective</u> Date, the information included in the Beneficial Ownership Certificate is true and correct in all respects.

**5.16** <u>Compliance with Laws</u>. Each of the Loan Parties and each Subsidiary has operated at all times in compliance in all material respects with the requirements of all Laws and all orders, writs, conditions of participation, contracts, standards, policies, injunctions, decrees, and Governmental Approvals applicable to it, its properties or the Facilities. Without limiting the generality of the foregoing:

(i) neither any Loan Party nor any Subsidiary thereof is in receipt of any written notice of any material violation of any Law, statute, rule, regulation, ordinance, code, judgment, order writ, decree, permit, concession, franchise or other governmental approval applicable to it or any of its property, which notice, individually or in the aggregate could reasonably be expected to result in a Material Adverse Effect; and

(ii) neither any Loan party nor any Subsidiary or any Affiliate thereof is in violation of and shall not violate any of the country or list based economic and trade sanctions administered and enforced by OFAC that are described or referenced at http://ustreas.gov/offices/enforcement/ofac/ or as otherwise published from time to time.

**5.17** Intellectual Property; Licenses, Etc. The Loan Parties and their Subsidiaries own, or possess the legal right to use, all of the trademarks, service marks, trade names, copyrights, patents, patent rights, franchises, licenses and other intellectual property rights (collectively, "IP Rights") that are necessary for the operation of their respective businesses without conflict with the rights of any other Person. Set forth on <u>Schedule 5.17</u> is a list of all IP Rights registered or pending registration with the United States Copyright Office or the United States Patent and Trademark Office and owned by any Loan Party, or that any Loan Party has the right to use, as of the Closing Date (or following the delivery of the first Compliance Certificate hereunder, as of the date of the most recently delivered Compliance Certificate). No claim has been asserted and is pending by any Person challenging or questioning the use of any IP Rights or the validity or effectiveness of any IP Rights by any Loan Party know of any such claim, and, to the knowledge of the Responsible Officers of the Loan Party or any Subsidiary or the granting of a right or a license in respect of any IP Rights from any Loan Party or any Subsidiary or the granting of a right or a license in respect of any IP Rights from any Loan Party or any Subsidiary does not infringe on the rights of any Person. As of the Closing Date, (or following the delivery of the first Compliance Certificate hereunder, as of the date of the most recently delivered Compliance Certificate), none of the IP Rights owned by any of the Loan Party or any Subsidiary does not infringe on the rights of any Person. As of the Closing Date, (or following the delivery of the first Compliance Certificate hereunder, as of the date of the most recently delivered Compliance Certificate), none of the IP Rights owned by any of the Loan Parties is subject to any licensing agreement or similar arrangement except as set forth on <u>Schedule 5.17</u>.

**5.18** <u>Broker's Fees</u>. Neither any Loan Party nor any Subsidiary has any obligation to any Person in respect of any finder's, broker's, investment banking or other similar fee in connection with any of the transactions contemplated under the Loan.

**5.19** <u>Labor Matters</u>. There are no collective bargaining agreements or Multiemployer Plans covering the employees of any Loan Party or any Subsidiary as of the <u>ClosingSecond Amendment Effective</u> Date, and neither any Loan Party nor any Subsidiary has suffered any strikes, walkouts, work stoppages or other material labor difficulty within the last five (5) years.

**5.20 Business Locations.** Set forth on <u>Schedule 5.20(a)</u> is a list of all Real Property located in the United States that is owned or leased by any Loan Party as of the <u>Closing</u>Second Amendment Effective Date (or following the delivery of the first Compliance Certificate hereunder as of the date of the most recently delivered Compliance Certificate). Set forth on <u>Schedule 5.20(b)</u> is a list of all locations where any tangible personal property of any Loan Party (excluding the location of moveable assets, which are moved from location to location in the ordinary course of business) with a value in excess of \$100,0001,000,000 is located as of the <u>ClosingSecond Amendment Effective</u> Date (or following the delivery of the first Compliance Certificate hereunder, as of the date of the most recently delivered Compliance Certificate). Set forth on <u>Schedule 5.20(c)</u> is the state of organization, chief executive office and tax payer identification number of each Loan Party as of the <u>ClosingSecond Amendment Effective</u> Date.

**5.21** <u>Perfection of Security Interests in the Collateral</u> The Collateral Documents create valid security interests in, and Liens on, the Collateral purported to be covered thereby, which security interests and Liens are currently perfected security interests and Liens (except to the extent perfection is deferred pursuant to <u>Schedule 6.18</u>), prior to all other Liens other than Permitted Liens as of the <u>ClosingSecond Amendment Effective</u> Date. The exact legal name of each Loan Party is as set forth on the signature pages hereto as of the <u>ClosingSecond Amendment Effective</u> Date.

**5.22** Solvency. Both before and after giving effect to (a) the Loans to be made or extended on the ClosingSecond Amendment Effective Date or such other date as Loans requested hereunder are made or extended, the issuance of the guaranties of the Obligations and the pledge of assets as security therefor by all of the Loan Parties, (b) the disbursement of the proceeds of such Loans pursuant to the instructions of the Loan Parties, (c) the consummation of the transactions contemplated in the Loan Documents, (d) the payment of any Restricted Payment and (e) the payment and accrual of all transaction costs in connection with the foregoing, the Loan Parties taken as a whole are Solvent.

# 5.23 [Reserved].

5.24 <u>Material Contracts [Reserved</u>. <u>Schedule 5.24 contains a true, correct and complete list of all Material Contracts in effect as of the Closing Date, and except as described thereon, all such Material Contracts are in full force and effect and no material breaches, defaults or events of default currently exist thereunder.].</u>

**5.25** <u>Accounts</u>. <u>Schedule 5.25</u> sets forth a complete and accurate list as of the <u>ClosingSecond Amendment Effective</u> Date of all deposit accounts and all securities accounts maintained by each Loan Party, together with a description thereof and such Schedule correctly identifies the name and address of each depository or broker dealer where the account is maintained, the name in which the account is held, the purpose of the account, and the complete account number thereof.

5.26 <u>Holding Company Status</u> Neither Holdings nor Intermediate Holdings is engaged in any trade or business in violation of <u>Section 7.15</u>.

5.27 <u>Inventory Suppliers</u>. <u>Schedule 1.01(c)</u> sets forth a complete and accurate list of all Inventory Suppliers.

**5.28** <u>Patriot Act</u>. Each Loan Party and its Subsidiaries are in compliance with the (i) Trading with the Enemy Act, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V) and any other enabling legislation or executive order relating thereto, and (ii) Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA Patriot Act of 2001) (the <u>'Patriot Act</u>'). No part of the proceeds of the Loans will be used, directly or indirectly, for any payments to any Person, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

**5.29 <u>Regulatory Matters</u>**. Without limiting the generality of any other representation or warranty made in this Agreement, each Loan Party hereby represents and warrants that, except as disclosed in <u>Schedule 5.29(a)</u>, the following statements are true, complete and correct, and the Loan Parties hereby covenant and agree to notify the Administrative Agent within three (3) Business Days (but in any event prior to the Loan Parties submitting any requests for advances of reserves or escrows) following the occurrence of any facts, events or circumstances, whether threatened, existing or pending, that would make any of the following representations and warranties untrue, incomplete or incorrect (together with such supporting data and information as shall be necessary to fully explain to the Administrative Agent the scope and nature of the fact, event or circumstance), and shall provide to the Administrative Agent within two (2) Business Days of the Administrative Agent's request, such additional information as the Administrative Agent shall reasonably request regarding such disclosure:

(a) <u>Permits</u>. Each Loan Party has (i) each Permit and other rights from, and have made all declarations and filings with, all applicable Governmental Authorities, all self-regulatory authorities and all courts and other tribunals necessary to engage in the Business conducted by the Loan Parties and the ownership, if any, and operation of the Facilities, and (ii) no knowledge that any Governmental Authority is considering limiting, suspending or revoking any such Permit. All such Permits are valid and in full force and effect and each Loan Party is in material compliance with the terms and conditions of all such Permits except where failure to be in such compliance or for a Permit to be valid and in full force and effect could not reasonably be expected to have a Material Adverse Effect.

(b) **Specific Licensing** Each Loan Party is duly licensed under the applicable Laws of each state where such Loan Party conducts business and is required to be licensed to conduct business. Each Loan Party owns, leases or operates a health care business and/or provides health care goods and services and (i) if it maintains Medicare and Medicaid provider status, is the holder of the provider identification numbers as identified on <u>Schedule 5.29(b)</u> hereto, all of which are current and valid and such Loan Party has not allowed, permitted, authorized or caused any other Person to use any such provider identification number, and (ii) has obtained all material Permits necessary for any Borrower to own its assets, to carry on its business, to execute, deliver and perform the Loan Documents, and to receive payments from the Payors and, if organized as a not-for-profit entity, has and maintains its status, if any, as an organization exempt from federal taxation under Section 501(c)(3) of the Internal Revenue Code. No Loan Party has been notified by any such Governmental Authority or other Person during the immediately preceding twenty-four

(24) month period that such party has rescinded, limited or not renewed, or intends to rescind, limit or not renew, any such license or approval.

(c) <u>Physician Ownership</u> Except as set forth on <u>Schedule 5.29(c)</u>, no physician has any direct or indirect "financial relationship" that constitutes an "ownership or investment interest" (each as defined in 42 CFR 411.354(b)) in any Loan Party or any Affiliate of any Loan Party.

#### (d) <u>Participation Agreements/Provider Status/Cost Reports</u>.

(i) Each Loan Party has the requisite participation agreement or provider number or other Permit to bill Medicare and/or Medicaid program in the state or states in which such Loan Party operates (to the extent such Loan Party participates in Medicare or Medicaid in such state or states) and all other Third Party Payor Programs which have historically accounted for any portion of the revenues of the Business or such Loan Party the loss of which could not reasonably be expected to have a Material Adverse Effect.

(ii) There is no investigation, audit, claim review, or other action pending or, to the knowledge of any Loan Party, threatened which could reasonably be expected to result in a revocation, suspension, termination, probation, restriction, limitation, or non-renewal of any Participation Agreement or provider number or result in a Loan Party's exclusion from any Third Party Payor Program, nor has any Third Party Payor Program made any decision not to renew any Participation Agreement or provider agreement related to the Business that could reasonably be expected to have a Material Adverse Effect, nor has any Loan Party made any decision not to renew any Participation Agreement or provider agreement that could reasonably be expected to have a Material Adverse Effect, nor is there any action pending or threatened to impose material sanctions with respect to the Business or penalty, sanction, fine or imposition upon any Loan Party.

(iii) To the knowledge of each Loan Party, each Loan Party and its contractors, have properly and legally billed all Third Party Payors (or intermediaries of Third Party Payors, as applicable) for goods and services rendered with respect to the Business and have maintained their records to reflect such billing practices. No funds relating to any Borrower are now, or, to the knowledge of any Loan Party will be, withheld by any Third Party Payor.

(iv) All Medicare, Medicaid, and private insurance cost reports and financial reports submitted by each Loan Party are and will be materially accurate and complete and have not been and will not be misleading in any material respects. No cost reports for any goods and services rendered with respect to the Business remain "open" or unsettled. There are no current, pending or outstanding Medicare, Medicaid or Third Party Payor Program reimbursement audits or appeals pending with respect to the Business or any Loan Party that could reasonably be expected to have a Material Adverse Effect and no Loan Party has any knowledge that any cost reports or financial reports previously given or declared are inaccurate.

(v) No Loan Party has any obligation (whether or not currently due) under or pursuant to any agreement, instrument or applicable Law to reimburse, repay or make payment to any Governmental Authority for any loans, advances, grants or monies given or paid to any Loan Party or any Affiliate of any Loan Party and no Person asserts a claim that any such Loan Party has any such obligation to reimburse, repay or make payment to any Governmental Authority for any loans, advances, advances, grants or monies given or paid to any Loan Party or any Affiliate of any Loan Party and no Person asserts a claim that any such advances, grants or monies given or paid to any Loan Party or any Affiliate of any Loan Party.

#### (e) <u>No Violation of Healthcare Laws</u>.

(i) No Loan Party is in violation of any Healthcare Laws, except where any such violation could not reasonably be expected to have a Material Adverse Effect. With respect to any existing Healthcare Laws not currently effective (any "Future Effective")

<u>Healthcare Law</u>"), no Loan Party is aware of any fact, circumstance or condition that exists that if not cured or corrected would constitute a violation of any Future Effective Healthcare Law when the obligation of compliance under such Future Effective Healthcare Law becomes effective, except where any such violation would not have a Material Adverse Effect.

#### (ii) Each Loan Party is HIPAA Compliant.

(f) **Proceedings**. No Loan Party is subject to any proceeding, suit or, to any Loan Party's knowledge, investigation by any federal, state or local government or quasi-governmental body, agency, board or authority or any other administrative or investigative body (including the Office of the Inspector General of the United States Department of Health and Human Services): (i) which could reasonably be expected to result in the imposition of a fine, sanction, or lower reimbursement rate for products or services rendered to eligible patients which has not been provided for on their respective financial statements, or which could reasonably be expected to result in the revocation, transfer, surrender, suspension or other impairment of the Permits of the Business; (ii) which pertains to any state or federal Medicare or Medicaid cost reports or claims filed by any Loan Party (including, but not limited to, any reimbursement audits), or any disallowance by any commission, board or agency in connection with any audit of such cost reports, other than as set forth on <u>Schedule 5.29(f)(iii)</u>; or (iv) which pertains to or requests any voluntary disclosure pertaining to a potential overpayment matter involving the submission of claims to such payor by any Loan Party.

#### (g) Fraud & Abuse.

No Loan Party has, or to its knowledge has been threatened to have, and no owner, officer, manager, (i) employee or person with a "direct or indirect ownership interest" (as that phrase is defined in 42 C.F.R. §420.201) in any Loan Party has, engaged in any of the following: (A) knowingly and willfully making or causing to be made any false statement or representation of a material fact in any application for any benefit or payment under any Healthcare Laws; (B) knowingly and willfully making or causing to be made any false statement or representation of a material fact for use in determining rights to any benefit or payment under any Healthcare Laws; (C) failing to disclose knowledge by a claimant of the occurrence of any event affecting the initial or continued right to any benefit or payment under any Healthcare Laws on its own behalf or on behalf of another, with intent to secure such benefit or payment fraudulently; (D) knowingly and willfully soliciting or receiving any remuneration (including any kickback, bribe or rebate), directly or indirectly, overtly or covertly, in cash or in kind or offering to pay such remuneration (1) in return for referring an individual to a person for the furnishing or arranging for the furnishing of any item or service for which payment may be made in whole or in part by any Healthcare Laws, or (2) in return for purchasing, leasing or ordering or arranging for or recommending the purchasing, leasing or ordering of any good, facility, service, or item for which payment may be made in whole or in part by Medicare, Medicaid, or any other governmental payor; (E) presenting or causing to be presented a claim for reimbursement for services that is for an item or services that was known or should have been known to be (1) not provided as claimed, or (2) false or fraudulent; or (F) knowingly and willfully making or causing to be made or inducing or seeking to induce the making of any false statement or representation (or omitting to state a fact required to be stated therein or necessary to make the statements contained therein not misleading) of a material fact with

respect to (1) a facility in order that the facility may qualify for Governmental Authority certification, or (2) information required to be provided under 42 U.S.C. § 1320a-3.

No Loan Party has been, or to its knowledge has been threatened to be, and no owner, officer, manager, (ii) employee or person with a "direct or indirect ownership interest" (as that phrase is defined in 42 C.F.R. §420.201) in any Loan Party: (A) has had a civil monetary penalty assessed against him or her pursuant to 42 U.S.C. §1320a-7a or is the subject of a proceeding seeking to assess such penalty; (B) has been excluded from participation in a Federal Health Care Program (as that term is defined in 42 U.S.C. §1320a-7b) or is the subject of a proceeding seeking to assess such penalty, or has been "suspended" or "debarred" from selling products to the U.S. government or its agencies pursuant to the Federal Acquisition Regulation, relating to debarment and suspension applicable to federal government agencies generally (48 C.F.R. Subpart 8.4), or other applicable Laws or regulations; (C) has been convicted (as that term is defined in 42 C.F.R. §1001.2) of any of those offenses described in 42 U.S.C. §1320a-7b or 18 U.S.C. §§669, 1035, 1347, 1518 or is the subject of a proceeding seeking to assess such penalty; (D) has been involved or named in a U.S. Attorney complaint made or any other action taken pursuant to the False Claims Act under 31 U.S.C. §§3729-3731 or qui tam action brought pursuant to 31 U.S.C. §3729 et seq.; (E) has been made a party to any other action by any Governmental Authority that may prohibit it from selling products to any governmental or other purchaser pursuant to any law; or (F) was or has become subject to any federal, state, local governmental or private payor civil or criminal investigations or inquiries, proceedings, validation review, program integrity review or statement of charges involving and/or related to its compliance with Healthcare Laws or involving or threatening its participation in Medicare, Medicaid or other Third Party Payor Programs or its billing practices with respect thereto.

#### 5.30 Compliance of Products.

(a) Each Loan Party:

(i) has obtained all Required Permits, or has contracted with third parties holding Required Permits, necessary for compliance with all applicable Laws including the Specified Laws, and all such Required Permits are current and each holder of such Required Permits is in material compliance with the terms and conditions of all such Required Permits;

(ii) except as set forth on <u>Schedule 5.30(a)</u>, has been operating in compliance in all material respects with all reporting and regulatory requirements imposed upon it as well as the Specified Laws, including reporting to FDA and other agencies, to include state government agencies of product deviations, contamination or of device malfunctions and/or device-related serious injuries or deaths and reporting to FDA of Corrections or Removals, when and as required under the FDCA;

(iii) has not, and none of its officers, directors, employees, shareholders, their agents or affiliates have, made an untrue statement of material fact or fraudulent statement to the FDA or failed to disclose a material fact required to be disclosed to the FDA, committed an act, made a statement, or failed to make a statement that could reasonably be expected to provide a basis for the FDA to invoke its policy respecting "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities," set forth in 56 Fed. Regulation 46191 (September 10, 1991);

(iv) has not received any notice that any Governmental Authority, including the FDA, the Office of the Inspector General of HHS, the United States Department of Justice or any equivalent foreign agency, has commenced or threatened to initiate any action to enjoin a Loan Party, or any of its officers, directors, employees, shareholders, agents or Affiliates, from conducting their respective businesses at any facility owned or used by any of them, or for any material civil penalty, injunction, seizure or criminal action, or which would result in the revocation, transfer, surrender, suspension or other material impairment of any Required Permit;

(v) is not a participant in any federal program whereby any federal, state or local government or quasigovernmental body, agency, board or other authority may have the right to recover funds by reason of the advance of federal funds, including those authorized under the Hill-Burton Act (42 U.S.C. 291, et seq.);

(vi) has not been threatened to be (A) excluded from United States health care programs pursuant to 42 U.S.C. §1320a7 and related regulations, (B) "suspended" or "debarred" from selling products to the United States government or its agencies pursuant to the Federal Acquisition Regulation, relating to debarment and suspension applicable to federal government agencies generally (48 C.F.R. Subpart 9.4), or other applicable Laws or regulations, or (C) made a party to any other action by any Governmental Authority that may prohibit it from selling products to any governmental or other purchaser pursuant to any Law;

(vii) is in compliance with all Environmental Laws;

(viii) maintains or causes to be maintained a standard of care in the storage, use, transportation and disposal of all Products, medical equipment, medical supplies, medical products and medical waste, of any kind and in any form, that is at least comparable to that which exists on the date of this Agreement and that is in conformity in all material respects with all applicable regulations and Laws;

(ix) maintains or causes to be maintained corporate regulatory compliance program ("<u>CCP</u>") in accordance with Specified Laws, applicable regulatory guidance, and customary business practices for each Loan Party which includes at least the following components: (A) specific officer within high-level personnel identified as having overall responsibility for regulatory compliance (B) training and education programs which effectively communicate the compliance standards and procedures to employees and agents;(C) policies and procedures to allow employees and other agents to anonymously report criminal or suspect conduct and potential compliance problems; (D) consistent enforcement of compliance policies including discipline of individuals responsible for the failure to detect violations of the CCP; and (E) mechanisms to immediately respond to detected violations of the CCP;

(x) except as set forth on <u>Schedule 5.30(a)</u>, has not received from the FDA at any time warning letters, Form FDA-483 inspection reports, "Untitled Letter", other correspondence or notice setting forth allegedly objectionable observations or alleged violations of laws and regulations enforced by the FDA, including the FDCA, or any comparable correspondence from any state or local authority responsible for regulating medical device products and establishments, or any comparable correspondence from any foreign counterpart of the FDA, or any comparable correspondence from any foreign counterpart of any state or local authority with regard to any Product or the manufacture, processing, packing, or holding thereof;

(xi) [reserved];

(xii) has entered into a Transport and Disposal Agreement with a reputable and qualified Person for the transport and disposal of hazardous wastes pursuant to which such Person has agreed to provide such transport and disposal services at all facilities at which such biomedical wastes and hazardous wastes are generated in conformity in all material respects with all applicable regulations and Laws and such Transport and Disposal Agreement remains in full force and effect; and

(xiii) except as set forth on <u>Schedule 5.30(a)</u>, maintains or causes to be maintained a policy that prevents the exposure of employees or contractors to bloodborne pathogens by prohibiting staff from handling returned or used Product if such Product is not received in a decontaminated manner.

#### (b) With respect to Products:

(i) No Loan Party has acquired, received, or otherwise transferred any human tissue or organs for valuable consideration for use in human transplantation, in violation of any Law.

(ii) <u>Schedule 5.30(b)</u>, hereto lists all Required Permits issued by a Governmental Authority relating to a Product and/or the applicable Loan Party's manufacture, sale, development, testing or marketing thereof maintained by Loan Parties as of the <u>ClosingSecond Amendment Effective</u> Date, together with the applicable Product category corresponding to the Required Permits. Loan Parties have delivered to the Administrative Agent on or prior to the <u>ClosingSecond Amendment Effective</u> Date copies of all such Required Permits. If, after the <u>ClosingSecond Amendment Effective</u> Date, any Loan Party acquires or renews any Required Permit issued by a Governmental Authority relating to a Product and/or the applicable Loan Party's manufacture, sale, development, testing or marketing thereof issued, Loan Parties shall promptly deliver a copy of such new or renewed Required Permit along with a copy of an update to <u>Schedule 5.30(b)</u>;

(iii) Except as set forth on <u>Schedule 5.30(b)</u>, each Product has been and/or shall be manufactured, imported, possessed, owned, warehoused, marketed, promoted, sold, labeled, furnished, distributed and marketed in accordance with all applicable Permits and Laws, including but not limited to the FDCA;

(iv) Without limiting the generality of <u>Section 5.30(a)(i)</u> above, with respect to any Product being tested or manufactured by any Loan Party or any Subsidiary of any Loan Party, such Person has received, and such Product shall be the subject of, all Required Permits needed in connection with the testing or manufacture of such Product as such testing is currently being conducted by or on behalf of such Person, and such Person has not received any notice from any applicable Governmental Authority, including the FDA, that such Governmental Authority is conducting an investigation or review of (A) such Person's manufacturing facilities and processes for such Product which have disclosed any material deficiencies or violations of applicable Laws (including Healthcare Laws) and/or the Required Permits related to the manufacture of such Product, or (B) any such Required Permit or that any such Required Permit has been revoked or withdrawn, nor has any such Governmental Authority issued any order or recommendation stating that the development, testing and/or manufacturing of such Product by such Person should cease;

(v) Without limiting the generality of <u>Section 5.30(a)(i)</u> above, with respect to any Product marketed, leased, rented, or sold by any Loan Party or any Subsidiary of any Loan Party, such Person shall have received, and such Product shall be the subject of, all Required

Permits needed in connection with the marketing and sales of such Product as currently being marketed, leased, rented, or sold by such Person, and such Person has not received any notice from any applicable Governmental Authority, including the FDA, that such Governmental Authority is conducting an investigation or review of any such Required Permit or approval or that any such Required Permit has been revoked or withdrawn, nor has any such Governmental Authority issued any order or recommendation stating that such marketing or sales of such Product cease or that such Product be withdrawn from the marketplace; and

(vi) the Loan Parties and their Subsidiaries have not experienced any significant failures in their manufacturing of any Product such that the amount of such Product successfully manufactured by them in accordance with all specifications thereof and any required payments related thereto in any month shall decrease significantly with respect to the quantities of such Product produced in the prior month.

(c) Neither the execution nor performance by any Loan Party of any Loan Documents, nor the exercise of any remedies by any party thereunder, will adversely affect any of the Required Permits.

**5.31** OFAC. Neither any Loan Party nor any Subsidiary or any Affiliate thereof is in violation of any of the Sanctions. Neither any Loan Party nor any Subsidiary thereof, nor to the knowledge of such Loan Party or any of its Subsidiaries, any director, officer, employee, agent, Affiliate or representative thereof (a) is a Sanctioned Person or a Sanctioned Entity, (b) has its assets located in a Sanctioned Entity, (c) derives revenues from investments in, or transactions with a Sanctioned Person or a Sanctioned Entity or (d) is owned or controlled by a Sanctioned Entity or a Sanctioned Person. No part of the proceeds of the Loans will be used, directly or indirectly, (i) to fund any activities or business of or with any Person, or in any country or territory, that, at the time of such funding, is, or whose government is, the subject of Sanctions, or (ii) in any other manner that would result in a violation of Sanctions by any Person (including any Person participating in the Loans, whether as underwriter, advisor, investor, or otherwise).

#### **ARTICLE 6**

#### **AFFIRMATIVE COVENANTS**

On the Closing Date and at all times thereafter until and including the Termination Date, the Loan Parties shall and shall cause each Subsidiary to:

6.01 <u>Financial Statements</u>. Deliver to the Administrative Agent for the benefit of each Lender:

(a) <u>Annual Financial Statements</u> As soon as available, but in any event within <u>one hundred and twentyninety</u> (12090) days after the end of each Fiscal Year (commencing with the Fiscal Year ending December 31, 2018) of the Loan Parties and their <u>SubsidiariesAdaptHealth Corp.</u>, consolidated balance sheets of <u>AdaptHealth Corp.</u>, Holdings, andthe Loan Parties and their Subsidiaries as at the end of such Fiscal Year, and the related consolidated statements of income or operations, retained earnings, shareholders' equity and cash flows for such Fiscal Year, setting forth in each case in comparative form the figures as of the end of and for the previous Fiscal Year, all in reasonable detail and prepared in accordance with GAAP, and in the case of the consolidated financial statements audited and accompanied by a report and opinion of KPMG LLP or other independent certified public accountants of nationally recognized standing reasonably acceptable to the Administrative Agent, which report and opinion shall be prepared in accordance with generally accepted auditing standards and shall not be subject to any "going concern" or like qualification, explanation or exception or any qualification, explanation or exception as to the scope of such audit;

(b) Quarterly Financial Statements. As soon as available, but in any event within forty-five (45) days after the end of each of the first three (3) Fiscal Quarters (commencing with the Fiscal Quarter ending March 31, 2019) of each Fiscal Year of A daptHealth Corp., Holdings, and the Loan Parties and their Subsidiaries, consolidated balance sheets of A daptHealth Corp., Holdings, and the Loan Parties and their Subsidiaries, consolidated balance sheets of A daptHealth Corp., Holdings, and the Loan Parties and their Subsidiaries, consolidated balance sheets of A daptHealth Corp., Holdings, and the Loan Parties and their Subsidiaries as at the end of such Fiscal Quarter, and the related consolidated statements of income or operations, retained earnings, shareholders' equity and cash flows for such Fiscal Quarter and for the portion of the Fiscal Year then ended, setting forth in each case in comparative form the figures as of the end of and for the corresponding Fiscal Quarter of the previous Fiscal Year and the corresponding portion of the previous Fiscal Year, all in reasonable detail and certified by a Responsible Officer of the Borrower Representative as fairly presenting in all material respects the financial condition, results of operations, shareholders' equity and cash flows of AdaptHealth Corp., Holdings, and the Loan Parties and their Subsidiaries in accordance with GAAP, subject only to normal year-end audit adjustments and the absence of footnotes; and

(c) <u>Management Discussion and Analysis</u> Concurrently with any delivery under clause (a) or (db) of this <u>Section 6.01</u> (with respect to clause (d), solely in the case of financial statements with respect to a month that coincides with the end of any Fiscal Quarter), a management discussion and analysis describing any differences in the reported financial results as between the periods covered and that in the same periods during the immediately preceding Fiscal Year, and as between such periods and the same periods included in the budget delivered pursuant to <u>Section 6.02(c)</u> below, which shall include, among any other information or explanation reasonably requested by the Administrative Agent, an explanation of any revenues, Consolidated EBITDA, Consolidated Capital Expenditures and new or lost customers that would assist the Lenders to better understand the results being reported; and.

(d) <u>Monthly Reports</u>. As soon as available, but in any event within thirty (30) days after the end of each calendar month of the Loan Parties and their Subsidiaries, consolidated balance sheets of the Loan Parties and their Subsidiaries as at the end of such month, and the related consolidated statements of income or operations, retained earnings, shareholders' equity and eash flows for such month, setting forth in each ease in comparative form the figures as of the end of and for the corresponding month of the previous Fiseal Year and to any budget provided pursuant to <u>Section 6.02(c)</u>, all in reasonable detail and certified by a Responsible Officer of the Borrower Representative as fairly presenting in all material respects the financial condition, results of operations, shareholders' equity and eash flows of the Loan Parties and their Subsidiaries in accordance with GAAP, subject only to normal year-end audit adjustments and the absence of footnotes

Notwithstanding the foregoing, documents required to be delivered pursuant to clauses (a) and (b) of this Section 6.01 (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and, if so delivered, shall be deemed to have been delivered on the date (i) on which such materials are publicly available as posted on the Electronic Data Gathering, Analysis and Retrieval system (EDGAR); or (ii) on which such documents are posted on the Loan Parties' behalf on an Internet or intranet website, if any, to which each Lender and the Administrative Agent have access (whether a commercial, third-party website or whether made available by the Administrative Agent). Each Lender shall be solely responsible for timely accessing posted documents and maintaining its copies of such documents; provided, that, the Borrower Representative shall promptly notify (which may be by electronic mail) the Administrative Agent of the filing and availability of any such item and provide to the Administrative Agent by electronic mail a link thereto.

6.02 <u>Certificates: Other Information</u> Deliver to the Administrative Agent for the benefit of each Lender, in form and detail satisfactory to the Administrative Agent:

(a) <u>Accountant Certification</u>. Concurrently with the delivery of the financial statements referred to in <u>Section 6.01(a)</u>, a certificate of its independent certified public accountants certifying such financial statements;

(b) <u>Compliance Certificate</u>. Concurrently with the delivery of the financial statements referred to in <u>Sections 6.01(a)</u> and <u>6.01(b)</u>, a duly completed Pro Forma Compliance Certificate executed by a Responsible Officer of the Borrower Representative, and such Pro Forma Compliance Certificate shall: (i) include such supplements to <u>Schedules 5.13</u>, <u>5.17</u>, <u>5.20(a)</u>, <u>5.20(b)</u>, and <u>5.20(c)</u> as are necessary such that, as supplemented, such Schedules would be accurate and complete as of the date of such Pro Forma Compliance Certificate <u>provided</u>, <u>however</u>, that in the event that any item included in such supplement shall constitute, result in or disclose a Default or an Event of Default hereunder, in no event shall the delivery of such supplement constitute a waiver of such Default or Event of Default by the Administrative Agent or any Lender, (ii) specify any information required to be delivered pursuant to the Security Agreement that has not already been identified in a written notice delivered to the Administrative Agent in accordance with the Security Agreement, and (iii) either confirm that there has been no material change in the Loan Parties' insurance coverage since delivery of the immediately prior Pro Forma Compliance Certificate or identify any such change thereto;

(c) <u>Annual Budget</u> Within sixty (60) days after the end of each Fiscal Year, the annual business plan and budget of <u>AdaptHealth Corp.</u>, <u>Holdings</u>, and the Loan Parties and their respective Subsidiaries containing, among other things, projected financial statements (including, without limitation, consolidated balance sheets of <u>AdaptHealth Corp.</u>, <u>Holdings</u>, and the Loan Parties and their Subsidiaries as at the end of each such Fiscal Quarter, and the related consolidated statements of income or operations, retained earnings, shareholders' equity and cash flows for each such Fiscal Quarter) for each Fiscal Quarter through the Term Loan Maturity Date;

(d) <u>Audit Letters</u>. Copies of any detailed audit reports, management letters or recommendations submitted to the board of directors (or the audit committee of the board of directors) of <u>AdaptHealth Corp.</u>, <u>Holdings</u>, and the Loan Parties and their Subsidiaries by independent accountants in connection with the accounts or books of the Loan Parties and their Subsidiaries, or any audit of any of them;

(e) <u>Public Company Reporting</u>. (i) to the extent that any Loan Party is a public company, promptly Promptly after the same are available (and in any event within ten (10) days thereof), copies of each annual report, proxy or financial statement or other report or communication sent to the stockholders of any Loan Party, AdaptHealth Corp. and copies of all annual, regular, periodic and special reports and registration statements which any Loan Party may file or be required to file with the SEC under Section 13 or 15(d) of the Exchange Act or to a holder of any Indebtedness owed by AdaptHealth Corp., Holdings, Intermediate Holdings any Loan Party, or any Subsidiary in its capacity as such a holder and not otherwise required to be delivered to the Administrative Agent pursuant hereto, (ii) all reports and written information to and from the United States Environmental Protection Agency, or any state or local agency responsible for environmental, health or safety matters, and (iii) all material reports and written information to and from any state or local agency responsible for health and safety matters, or any successor agencies or authorities concerning environmental, health or safety matters;

(f) <u>Insurance Report</u>. By the last day of each Fiscal Year, a report in form and substance reasonably satisfactory to Administrative Agent outlining all material insurance coverage maintained as of the date of such report by the Loan Parties and their Subsidiaries and all material insurance coverage planned to be maintained by the Loan Parties and their Subsidiaries in the immediately succeeding Fiscal Year;

(g) <u>Annual Collateral Verification</u>. Concurrently with the delivery of the financial statements referred to in <u>Section 6.01(a)</u>, an officer's certificate (i) either confirming that there has been no change in the information set forth in the then existing schedules to the Security Agreement or identifying any such changes thereto and (ii) certifying that all UCC financing statements (including fixtures filings, as applicable) or other appropriate filings, recordings or registrations, have been filed of record in each governmental, municipal or other appropriate office in each jurisdiction identified pursuant to clause (i) above to the extent necessary to protect and perfect the security interests under the Loan Documents for a period of not less than eighteen (18) months after the date of such certificate (except as noted therein with respect to any continuation statements to be filed within such period);

(h) <u>Real Property Leases. Within forty-five (45) days after the end of each Fiseal Quarter, a report in form and substance</u> satisfactory to the Administrative Agent identifying: (i) all Real Property lease agreements to which any Loan Party is a party, (ii) the property governed by such Real Property lease agreements, (iii) the expiration dates of such lease agreements, and (iv) the amount of rental payments due under each lease agreement and such other information as the Administrative Agent may reasonably request; and [Reserved]; and

(i) <u>Additional Information</u> Promptly (and in any event within two (2) days after a request therefor), such additional information (including Medicare and Medicaid cost reports and audits) regarding the business, financial or corporate affairs of any Loan Party or any Subsidiary, or compliance with the terms of the Loan Documents, as the Administrative Agent or any Lender through the Administrative Agent may from time to time reasonably request (including as required under the Patriot Act and the Beneficial Ownership Regulation).

# 6.03 <u>Notices</u>.

(a) Promptly (and in any event within two (2) Business Days) notify the Administrative Agent and each Lender in writing of the occurrence of any Default or Event of Default.

(b) Promptly (and in any event within two (2) Business Days) notify the Administrative Agent and each Lender in writing of any matter that has resulted or could reasonably be expected to result in a Material Adverse Effect.

(c) Promptly (and in any event within two (2) Business Days) notify the Administrative Agent and each Lender in writing of the occurrence of any ERISA Event.

(d) Promptly (and in any event within two (2) Business Days) notify the Administrative Agent and each Lender in writing of any material change in accounting policies or financial reporting practices by any Loan Party or any Subsidiary.

(e) Promptly (and in any event within two (2) Business Days), notify the Administrative Agent and each Lender, in writing, of the threat or institution of, or any material development in, any Proceeding against or affecting any Loan Party (i) in which the amount involved or relief sought is in excess of \$1,000,000, (ii) which could reasonably be expected to have a Material Adverse Effect, (iii) which seeks injunctive relief, (iv) which alleges criminal misconduct by any Loan Party, (v) which alleges material violations of any Laws or Governmental Approvals, or (vi) which alleges the violation of any law regarding, or seeks remedies in connection with, any Environmental Liability.

(f) Promptly (and in any event within two (2) Business Days of such event), notify the Administrative Agent and each Lender, in writing, of (i) any loss, damage or destruction to the Collateral in

the amount of \$1,000,000 or more individually, whether or not covered by insurance and (ii) any change in the information which would require a material correction or material addition to <u>Schedule 1</u> to the Security Agreement.

(g) Immediately notify the Administrative Agent and each Lender, in writing, upon the occurrence of, upon becoming aware of, or upon receipt of notice from a third party of, (i) any Loan Party's default pursuant to the terms of any Material Contractor lease to which such Loan Party is a party or (i) the termination of the intervention of the in

(ii) the termination of, or the intent or threat to terminate, any such Material Contract or lease.

(h) Upon the written request of the Administrative Agent following the occurrence of any event or the discovery of any condition which the Administrative Agent or the Required Lenders reasonably believe has caused (or could be reasonably expected to cause) the representations and warranties set forth in <u>Section 5.09</u> to be untrue in any material respect, furnish or cause to be furnished to the Administrative Agent, at the Loan Parties' expense, a report of an environmental assessment of reasonable scope, form and depth, (including, where appropriate, invasive soil or groundwater sampling) by a consultant reasonably acceptable to the Administrative Agent as to the nature and extent of the presence of any Hazardous Materials on any Facilities and as to the compliance by any Loan Party or any of its Subsidiaries with Environmental Laws at such Facilities. If the Loan Parties fail to deliver such an environmental report within forty-five (45) days after receipt of such written request then the Administrative Agent may arrange for same, and the Loan Parties hereby grant to the Administrative Agent and its representatives access to the Facilities to reasonably undertake such an assessment (including, where appropriate, invasive soil or groundwater sampling). The reasonable cost of any assessment arranged for by the Administrative Agent pursuant to this provision will be payable by the Loan Parties on demand and added to the obligations secured by the Collateral Documents.

(i) Promptly notify the Administrative Agent and each Lender of any Loan Party's receipt of notice of any citation, any investigation or audit, or pending or threatened proceedings relating to, any material violation by any Loan Party of any Healthcare Law, including, without regard to materiality, (x) any investigation or audit or proceeding involving violation of any of the Medicare and/or Medicaid fraud and abuse provisions and (y) any criminal or civil investigation initiated, claim filed or disclosure required by the Office of Inspector General, the Department of Justice, CMS (formerly HCFA), or any other Governmental Authority.

(j) Promptly notify the Administrative Agent and each Lender of any Loan Party's receipt of a written recommendation from any Governmental Authority or other regulatory body that such Loan Party should have its licensure, provider or supplier number or accreditation suspended, revoked, or limited in any material way, or have its eligibility to participate in Medicare, Medicaid or any other government program to accept assignments or rights to reimbursement under Medicaid, Medicare, or any other government program regulations suspended, revoked, or limited in any material way.

(k) Any change in the information provided in the Beneficial Ownership Certification that would result in a change to the list of beneficial owners identified in such certification.

(1) In addition to and not in limitation of any other provision of this Agreement, it is understood and agreed that Borrower Representative shall (a) within two (2) Business Days notify the Administrative Agent and each Lender of any material adverse developments related to that certain subpoena dated July 11, 2017 and issued to Borrower Representative or any other Loan Party by the United States Attorneys' Office for the Eastern District of Pennsylvania pursuant to 18 U.S.C. § 3486 (the <u>Subpoena</u>"), (b) shall participate in a phone call with the Administrative Agent and the Lenders quarterly, at the request of the Administrative Agent, in order to provide an update on the status of the Subpoena, relevant related details and what action the Loan Parties have taken over the course of the previous quarter

and propose to take with respect thereto, and (c) shall provide such other documents, instruments, agreements and information related to the Subpoena as reasonably requested by the Administrative Agent and each Lender.

Each notice pursuant to this <u>Section 6.03</u> shall be accompanied by a statement of a Responsible Officer of the Borrower Representative setting forth details of the occurrence referred to therein and stating what action the Borrowers have taken and propose to take with respect thereto. Each notice pursuant to <u>Section 6.03(a)</u> shall describe with particularity any and all provisions of this Agreement and any other Loan Document that have been breached.

#### 6.04 <u>Payment of Obligations: Tax Returns.</u>

(a) Pay and discharge as the same shall become due and payable, all its obligations and liabilities, including (a) all federal, state and other material tax liabilities, assessments and governmental charges or levies upon it or its properties or assets, unless the same are being Properly Contested; and (b) all lawful claims which, if unpaid, would by law become a Lien upon its Property unless the same are being Properly Contested.

(b) Timely file or cause to be timely filed all federal, state and other material tax returns required to be filed.

#### 6.05 <u>Preservation of Existence, Material Contracts, Etc.</u>

(a) Preserve, renew and maintain in full force and effect its legal existence under the Laws of the jurisdiction of its organization except in a transaction permitted by <u>Section 7.04</u> or <u>Section 7.05</u>.

(b) Preserve, renew and maintain in full force and effect its good standing and qualification to do business under the Laws of the jurisdiction of its organization and in any other jurisdiction where failure to so maintain good standing or qualification would have or would constitute a Material Adverse Effect.

(c) Preserve, renew and maintain all Governmental Approvals as are necessary for the conduct of its business as currently conducted and herein contemplated.

(d) Preserve, register and renew whenever applicable all of its material registered patents, copyrights, trademarks, trade names and service marks as are necessary for the conduct of its business as currently conducted and herein contemplated.

(e) Maintain all Material Contracts to which it is a party without default or right of any counterparty thereto to terminate or accelerate thereunder unless replaced with one or more alternative contracts of substantially equivalent value.

#### 6.06 Maintenance of Properties.

(a) Maintain, preserve and protect all of its property owned or used in the operation of its business in good working order and condition, ordinary wear and tear excepted.

(b) Make all necessary repairs thereto and renewals and replacements thereof.

#### (c) Use the standard of care typical in the industry in the operation and maintenance of its Facilities.

Maintenance of Insurance. Maintain or cause to be maintained, with financially sound and reputable insurers rated not less 6.07 than A-, Class VII by Best's, commercial general liability insurance, professional liability insurance, product liability insurance, business interruption insurance, pollution liability insurance and all risk property insurance, in each case with respect to liabilities, losses or damage in respect of the assets, properties and businesses of each Loan Party as may customarily be carried or maintained under similar circumstances by Persons of established reputation engaged in similar businesses, in each case in such amounts (giving effect to self-insurance), with such deductibles, covering such risks, and in amounts and otherwise on such terms and conditions as shall be customary for such Persons (including perils of flood, quake and/or windstorm, as applicable) and reasonably acceptable to the Administrative Agent. Without limiting the generality of the foregoing, if the Property or any part thereof is identified by the Secretary of Housing and Urban Development as being situated in an area now or subsequently designated as having special flood hazards (including, without limitation, those areas designated as Zone A or Zone V), flood insurance in compliance with applicable flood insurance Laws and in an amount equal to the lesser of: (x) the minimum amount required, under the terms of coverage, to compensate for any damage or loss on a replacement basis (or the unpaid balance of the Obligations if replacement cost coverage is not available for the type of building insured); or (y) such lesser amount as may be required by Administrative Agent; provided, such amount not to be less than the amount required under any applicable flood insurance Laws. Each such policy of insurance shall (i) name Administrative Agent, on behalf of each Lender as an additional insured by endorsement thereunder as its interests may appear, (ii) in the case of each property insurance policy, contain a lender's loss payable clause or endorsement, satisfactory in form and substance to Administrative Agent, that names Administrative Agent, on behalf of Lenders, as the lender's loss payee thereunder and (iii) provide for at least thirty (30) days' prior written notice to Administrative Agent of any modification or cancellation of such policy. The Administrative Agent and Secured Parties have no responsibility for premiums, warranties or representations to underwriters. The Loan Parties or their insurance broker shall provide a certificate of insurance upon each policy renewal or replacement. In the event Borrowers fail within ten (10) Business Days after Administrative Agent's request to provide Administrative Agent with evidence of the insurance coverage required by this Agreement, Administrative Agent may purchase insurance at Borrowers' expense to protect the Administrative Agent's interests in the Collateral. This insurance may, but need not, protect Borrowers' interests. The coverage purchased by Administrative Agent may, but need not, pay any claim made by any Loan Party or any claim that is made against any Loan Party in connection with the Collateral. Loan Parties may later cancel any insurance purchased by Administrative Agent, but only after providing Administrative Agent with evidence that Loan Parties have obtained insurance as required by this Agreement. If Administrative Agent purchases insurance for the Collateral, to the fullest extent provided by law, Loan Parties will be responsible for the costs of that insurance, including interest and other charges imposed by Administrative Agent in connection with the placement of the insurance, until the effective date of the cancellation or expiration of the insurance. The costs of the insurance may be added to the Obligations. The costs of the insurance may be more than the cost of insurance Loan Parties are able to obtain on their own.

**6.08** <u>Compliance with Laws</u>. Comply in all material respects with the requirements of all Laws and Governmental Approvals applicable to it (including Environmental Laws) and all orders, writs, injunctions and decrees applicable to it or to its business or Property, except in such instances in which such requirement of Law or order, writ, injunction or decree is being Properly Contested or where failure to comply could not reasonably be expected to have a Material Adverse Effect. The Loan Parties will (i) other than with respect to a Voluntary Termination, maintain in full force and effect, and free from restrictions, probations, conditions or known conflicts which would materially impair the use or operation of any healthcare or other business conducted or hereafter conducted or of the Business as it is currently

conducted, all Permits necessary under Healthcare Laws to continue to bill or receive payment or reimbursement under all Third Party Payor Programs in which any Loan Party or the Business participates as of the date of this Agreement or any time hereafter, and (ii) provide to the Administrative Agent upon request, an accurate, complete and current list of all participation agreements with Third Party Payors with respect to the Business and the business of each Loan Party (collectively, '<u>Participation Agreements</u>'). Other than with respect to a Voluntary Termination, the Loan Parties will at all times comply in all material respects with all requirements, contracts, conditions and stipulations applicable to such Loan Party necessary in order to maintain in good standing and without default or limitation under all such Participation Agreements.

#### 6.09 Books and Records.

(a) Maintain proper books of record and account, in which full, true and correct entries shall be made of all financial transactions and matters involving the assets and business of such Loan Party or such Subsidiary, as the case may be, in each case in accordance with GAAP.

(b) Maintain such books of record and account in material conformity with all applicable requirements of any Governmental Authority having regulatory jurisdiction over such Loan Party or such Subsidiary, as the case may be.

**6.10 Inspection Rights.** Permit (a) representatives and independent contractors of the Administrative Agent to visit and inspect any of its properties, to examine its corporate, financial and operating records, and conduct audits and make copies thereof or abstracts therefrom, and to discuss its affairs, finances and accounts with its directors, officers, and independent public accountants, all at the expense of the Loan Parties and at such reasonable times during normal business hours and as often as may be reasonably desired, upon reasonable advance notice to the Loan Parties; provided, however, that absent the existence of a Default or Event of Default, the Loan Parties shall be liable for no more than one (1) such inspection in any Fiscal Year, and (b) representatives and independent contractors of the Administrative Agent to conduct an annual audit of the Collateral at the expense of the Loan Parties and upon reasonable advance notice to the Loan Parties, provided, that when a Default or an Event of Default exists the Administrative Agent (or any of its respective representatives or independent contractors) may do any of the foregoing at the expense of the Loan Parties at any time during normal business hours and without advance notice and, in the case of Collateral audits, as frequently as deemed necessary by the Administrative Agent. A representative of each Lender shall have the right to accompany the Administrative Agent in connection with all such inspections, audits and examinations (at such Lender's sole cost and expense if a Default or an Event of Default has not occurred and is continuing).

### 6.11 <u>Use of Proceeds</u>.

(a) (i) Use the proceeds of the Initial Term Loan to (x) refinance Indebtedness of the Loan Parties and their Subsidiaries under the Original Loan Agreement, (y) pay transaction costs, fees and expenses related to the consummation of the transactions contemplated under this Agreement, and (z) make distributions in an aggregate amount equal to \$142,736,452 to holders of Capital Stock of Holdings as set forth in Section 2.5(ii) of the Note and Unit Purchase Agreement (the "<u>Closing Date Distributions</u>").

(ii) Use the proceeds of the Delayed – Draw Term Loans solely to finance Permitted Acquisitions and to pay fees and transaction costs associated with such Permitted Acquisitions.

(iii) Use the proceeds of each Incremental Facility consisting of Term Loans solely to finance Permitted Acquisitions and to pay fees and transaction costs associated with such Permitted Acquisitions.

(b) Use proceeds of the Revolving Loans and Swingline Loans (i) in an amount not to exceed \$25,000,000 to finance working capital, make Capital Expenditures, and for other general corporate purposes; and (ii) in an amount not to exceed \$50,000,000 solely to finance Permitted Acquisitions and to pay fees and transaction costs associated with such Permitted Acquisitions.

(c) Notwithstanding the foregoing, in no event shall the proceeds of the Credit Extensions be used in contravention of any Law or of any Loan Document.

6.12 <u>Additional Subsidiaries</u>. Simultaneously with (or such longer period as the Administrative Agent may provide at its sole option) any Acquisition or the formation of any Subsidiary:

(a) notify the Administrative Agent thereof in writing, together with (i) jurisdiction of formation, (ii) number of shares of each class of Capital Stock outstanding, (iii) number and percentage of outstanding shares of each class owned (directly or indirectly) by any Loan Party or any Subsidiary and (iv) number and effect, if exercised, of all outstanding options, warrants, rights of conversion or purchase and all other similar rights with respect thereto; and

(b) simultaneously with a Permitted Acquisition, or within thirty (30) days (in each case, or such later date as may be approved in writing by the Administrative Agent at its sole option) of any Permitted Acquisition or any Subsidiary being formed, cause each Subsidiary formed or acquired in connection with such Permitted Acquisition, other than an Excluded Subsidiary or an Excluded Foreign Subsidiary, to (A) become a Borrower or Guarantor (to be determined by the Administrative Agent absent the prior direction of the Required Lenders in their sole discretion) by executing and delivering to the Administrative Agent a joinder agreement or such other document as the Administrative Agent may reasonably request for such purpose (including as required under the Security Agreement), and (B) deliver to the Administrative Agent documents of the types referred to in Sections 4.01(b), (c), and (d) and take any actions required under <u>Section 6.14</u>, and, if requested by the Administrative Agent, favorable opinions of counsel to such Person (which shall cover, among other things, the legality, validity, binding effect and enforceability of the documentation referred to in clause (A)), all in form, content and scope reasonably satisfactory to the Administrative Agent.

6.13 <u>ERISA Compliance</u>. Do, and cause each of its ERISA Affiliates to do, each of the following: (a) maintain each ERISA Plan in compliance in all material respects with the applicable provisions of ERISA, the Internal Revenue Code and other federal or state Law; (b) cause each ERISA Plan that is qualified under<u>Section 401(a)</u> of the Internal Revenue Code to maintain such qualification; and (c) make all required contributions to any ERISA Plan subject to <u>Section 412</u>, <u>Section 430</u> or <u>Section 431</u> of the Internal Revenue Code.

#### 6.14 <u>Further Assurances</u>.

(a) To the extent not delivered to the Administrative Agent on or before the Closing Date (including in respect of afteracquired property and Persons that become Subsidiaries of any Loan Party after the Closing Date), the Loan Parties will promptly (and in any event within thirty (30) days) deliver to the Administrative Agent such modifications to the terms of the Loan Documents (or, to the extent applicable as determined by the Administrative Agent, such other documents, including, within thirty (30) days of such acquisition, Deposit Account Control Agreements), in each case in form and substance reasonably satisfactory to the Administrative Agent and as the Administrative Agent reasonably deems necessary or advisable in order to ensure that each Loan Party (including any Person required to become a Guarantor or Borrower pursuant to <u>Section 6.12</u> hereof) shall effectively grant to the Administrative Agent, for the benefit of the Secured Parties, a valid and enforceable security interest in all

of its property, including all of its Capital Stock (other than Excluded Property), as security for the Obligations of such Loan Party.

(b) Without limiting the generality of the above, the Loan Parties will cause (a) 100% of the issued and outstanding Capital Stock owned by the Loan Parties of each Domestic Subsidiary and (b) 65% (or such greater percentage that could not reasonably be expected to cause any material adverse tax consequences) of the issued and outstanding Capital Stock owned by the Loan Parties entitled to vote (within the meaning of Treas. Reg. Section 1.956-2(c)(2)) and 100% of the issued and outstanding Capital Stock not entitled to vote (within the meaning of Treas. Reg. Section 1.956-2(c)(2)) in each Foreign Subsidiary directly owned by a Loan Party or any Foreign Subsidiary Holdco owned by a Loan Party to be subject at all times to a first priority, perfected Lien in favor of the Administrative Agent pursuant to the terms and conditions of the Collateral Documents or such other security documents as the Administrative Agent shall request; provided, however, that the Loan Parties shall not be required to pledge the Capital Stock of any non-Wholly Owned Subsidiary to the extent (a) consent of the minority investor is required pursuant to the Organizational Documents of such Excluded Subsidiary (and such consent was not required in connection with or in contemplation or anticipation of such pledge) and not obtained after using commercially reasonable efforts and (b) such non-Wholly Owned Subsidiary is an Excluded Subsidiary.

(c) With respect to each Account for which either the perfection, enforceability, or validity of the Administrative Agent's Liens in such Account, or the Administrative Agent's right or ability to obtain direct payment to the Administrative Agent of the proceeds of such Account, is governed by any federal, state, or local statutory requirements other than those of the UCC, the Loan Parties will take such steps as the Administrative Agent may from time to time reasonably request to ensure such perfection, enforceability, validity of the Lien or right to obtain direct payment, including compliance with the Federal Assignment of Claims Act of 1940.

(d) If an Event of Default exists, each Loan Party shall grant (and in any event, shall be deemed to grant) a first priority, perfected Lien (except for Permitted Liens) on any additional Property of the Loan Parties (including Excluded Property other than as described in clauses (e), (h) and (j) of the Excluded Property definition) in favor of Administrative Agent for the benefit of itself, the other Lenders and the other Secured Parties to secure the Obligations pursuant to the terms and conditions of the Collateral Documents as requested by Administrative Agent in its sole discretion.

(e) Each Loan Party will, and will cause each Subsidiary to, at its own cost and expense, cause to be promptly and duly taken, executed, acknowledged and delivered all such further acts, documents and assurances as may from time to time be necessary or as the Administrative Agent or the Required Lenders may from time to time reasonably request in order to carry out the intent and purposes of the Loan Documents and the transactions contemplated thereby, including all such actions to establish, create, preserve, protect and perfect a first priority Lien (subject only to Permitted Liens) in favor of the Administrative Agent for the benefit of the Lenders on the Collateral (including Collateral acquired after the Closing Date), including on any and all assets of each Loan Party (other than Excluded Property), whether now owned or hereafter acquired.

(f) The Loan Parties shall deliver to the Administrative Agent fully-executed agreements, in form and substance satisfactory to the Administrative Agent, between the Loan Parties, the Administrative Agent and the Loan Parties' vendors (including the Brightree Collateral Assignment Agreement to be delivered in accordance with <u>Section 4.01(ji</u>) that provide off-site data storage, network hosting or management, shared application services or other "cloud-based" services to any Loan Party, which shall provide the Administrative Agent with such access to the Loan Parties' billing and Accounts records as the Administrative Agent deems necessary or helpful in connection with the Administrative

Agent's and Lenders' audit and inspection rights under the Loan Documents and the collection or realization of or on the Collateral

# 6.15 <u>Covenant with Respect to Environmental Matters</u>. In respect of all environmental matters:

(a) comply in all material respects with the requirements of all federal, state, and local Environmental Laws applicable to the Loan Parties or their Property; notify the Administrative Agent promptly in the event of any spill, release or disposal of Hazardous Material on, or hazardous waste pollution or contamination affecting, the Facilities in material violation of applicable Environmental Laws of which a Loan Party has actual knowledge; forward to the Administrative Agent promptly any written notices relating to such matters received from any Governmental Authority; and pay when due any fine or assessment against the Facilities, <u>provided</u>, that the Loan Parties shall not be required to pay any such fine or assessment so long as the validity thereof shall be Properly Contested; and <u>provided further</u> that, in any event, payment of any such fine or assessment shall be made before any of their Property shall be subjected to a Lien or be seized or sold in satisfaction thereof;

(b) promptly notify the Administrative Agent upon becoming aware of any fact or change in circumstances that would be expected to cause any of the representations and warranties contained in <u>Section 5.09</u> to cease to be true in all material respects (without duplication of any materiality qualifier therein) for any time before the Termination Date;

(c) not become involved, and will not knowingly permit any tenant of the Facilities to become involved, in any operations at the Facilities generating, storing, disposing, or handling Hazardous Materials in material violation of applicable Environmental Laws or any other activity that could lead to the imposition on any Lender or the Administrative Agent of any liability, or the imposition on the Loan Parties or the Facilities of any material liability or any lien under any Environmental Laws;

(d) promptly contain or remove any Hazardous Materials found on the Facilities in violation of any applicable Environmental Law, which containment or removal must be done in compliance with applicable Environmental Laws and at the Loan Parties' expense; and the Loan Parties agree that the Administrative Agent has the right, at its sole option but at the Loan Parties' expense, to have an environmental engineer or other representative review the work being done; and

(e) indemnify, protect, defend and hold harmless each Indemnitee from and against and all liabilities, obligations, losses, damages (including, consequential damages), penalties, actions, judgments, suits, claims, costs, expenses and disbursements of any kind or nature whatsoever (including, the reasonable fees and disbursements of counsel for and consultants of such Indemnitees in connection with any investigative, administrative or judicial proceeding, whether or not such Indemnitees shall be designated a party thereto), which may be imposed on, incurred by, or asserted against such Indemnitees (whether direct, indirect, or consequential) now or hereafter arising as a result of any claim for environmental cleanup costs, any resulting damage to the environment and any other environmental claims against any Loan Party, any Lender, the Administrative Agent, any other Indemnitee or the Facilities. The provisions of this Section 6.15(e) shall continue in effect and shall survive the Termination Date.

6.16 <u>Covenants with Respect to Real Property</u> With respect to Mortgaged Property acquired by any Loan Party after the Closing Date, deliver to the Administrative Agent within sixty (60) days of the date such Mortgaged Property was acquired, as the case may be, each in form and substance reasonably satisfactory to the Administrative Agent:

(a) a Mortgage Instrument encumbering the fee interest of any Loan Party in such Mortgaged Property;

(b) all other Mortgage Supporting Documents required by Administrative Agent with respect to such Mortgaged Property;

(c) a customary legal opinion of special local counsel for the Loan Parties for the state in which such Mortgaged Property is located in form and substance reasonably acceptable to the Administrative Agent;

(d) evidence reasonably satisfactory to the Administrative Agent of comprehensive "all risk" insurance with respect to such Mortgaged Property in form and substance satisfactory to the Administrative Agent; and

(e) an environmental site assessment of such Mortgaged Property from an environmental consulting firm acceptable to the Administrative Agent.

**6.17** <u>Lenders Meetings</u>. The Loan Parties will, upon the request of the Administrative Agent, participate in a meeting of the Administrative Agent and Lenders once during each Fiscal Year to be held at the Borrowers' corporate offices (or at such other location as may be agreed to by the Borrowers and the Administrative Agent) at such time as may be agreed to by the Borrowers and the Administrative Agent, provided that during the existence of an Event of Default, meetings may be held more frequently than once per Fiscal Year.

6.18 <u>Post-Closing Covenants</u>. The Loan Parties shall satisfy the requirements and/or provide to the Administrative Agent each of the documents, instruments, agreements and information set forth on <u>Schedule 6.18</u>, in form and substance acceptable to the Administrative Agent, on or before the date specified for such requirement in such Schedule or such later date to be determined by the Administrative Agent, at its sole option, each of which shall be completed or <u>provided</u> in form and substance satisfactory to the Administrative Agent.

**6.19 Qualified ECP Guarantors** Each Qualified ECP Guarantor hereby jointly and severally, absolutely, unconditionally and irrevocably undertakes to provide such funds or other support as may be needed from time to time by each other Loan Party to honor all of its obligations under this Agreement in respect of Hedge Obligations (provided, however, that each Qualified ECP Guarantor shall only be liable under this <u>Section 6.19</u> for the maximum amount of such liability that can be hereby incurred without rendering its obligations under this <u>Section 6.19</u> or otherwise under this Agreement, as it relates to such other Loan Party, voidable under applicable Law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amount). The obligations of each Qualified ECP Guarantor under this Section shall remain in full force and effect until each Loan and all other Obligations (other than contingent indemnification obligations to the extent no claim giving rise thereto has been asserted) have been paid in full and all Commitments and Letters of Credit have been terminated. Each Qualified ECP Guarantor intends that this Section constitute, and this Section shall be deemed to constitute, a "keepwell, support, or other agreement" for the benefit of each other Loan Party for all purposes of Section 1a(18)(A)(v)(II) of the Commodity Exchange Act.

**6.20** <u>Interest Rate Protection</u> Within ninety (90) days after the Closing Date, the Borrowers shall enter into, and thereafter maintain in full force and effect, Secured Hedge Agreements with coverage in a notional amount of not less than 50% of the outstanding principal amount of the Term Loans and having a term of at least three (3) years commencing after the Closing Date, and otherwise on other terms

and conditions reasonably satisfactory to the Administrative Agent. If the Borrowers are required to maintain such Secured Hedge Agreements and an Incremental Term Loan or a Delayed – Draw Term Loan is subsequently funded, the Borrowers shall have a period of thirty (30) days following such funding to comply with the requirements of this <u>Section 6.20</u> as to entering into Secured Hedge Agreements with coverage in a notional amount of not less than 50% of the outstanding principal amount of the Term Loan (after taking into account the additional Term Loans funded).

# 6.21 Covenants Regarding Products and Compliance with Required Permits.

(a) Without limiting the generality of <u>Section 6.08</u>, each Loan Party and its Subsidiaries shall comply fully and completely in all material respects with all Required Permits at all times issued by any Governmental Authority, including the FDA, with respect to such development, testing, manufacture, marketing, sales, or leasing of such Product by such Person as such activities are at any such time being conducted by such Person, including the timely filing (after giving effect to any extension duly obtained) of all notifications, reports, submissions, Required Permit renewals, cost reports and other reports of every kind whatsoever required by applicable Laws (which reports shall be materially accurate and complete in all material respects and not misleading in any material respect and shall not remain open or unsettled) and shall operate in a manner such that the Required Permits remain in full force and effect.

(b) Loan Parties and their Subsidiaries shall maintain in full force and effect the Transport and Disposal Agreement or such other agreement in form and substance for the transport and disposal of hazardous wastes with respect to all facilities at which such waste is generated.

6.22 <u>Healthcare Operations</u>. Without limiting the generality of the foregoing covenants and to induce the Administrative Agent and the Lenders to enter into this Agreement and to make the Loans and other credit accommodations contemplated hereby, the Loan Parties hereby covenant that the Loan Parties will:

(a) timely file or caused to be timely filed (after giving effect to any extension duly obtained), all notifications, reports, submissions, Permit renewals, cost reports and other reports or documents of every kind whatsoever required by Healthcare Laws (which reports will be materially accurate and complete in all material respects and not misleading in any material respect and shall not remain open or unsettled);

(b) maintain in full force and effect, and free from restrictions, probations, conditions or known conflicts that would materially impair the use or operation of any healthcare or other business conducted or hereafter conducted by any Loan Party, all Permits necessary under Healthcare Laws to carry on the Business of the Loan Parties as it is conducted on the Closing Date or substantially similar thereto; and

(c) at all times be HIPAA Compliant.

**6.23** <u>Patriot Act; OFAC</u>. (a) Comply with the Patriot Act, (b) use no part of the proceeds of the Loans, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended, and (c) comply with the Sanctions.

## ARTICLE 7

#### **NEGATIVE COVENANTS**

On the Closing Date and at all times thereafter until and including the Termination Date, no Loan Party shall, nor shall it permit any Subsidiary to, directly or indirectly:

- 7.01 Indebtedness. Create, incur, assume or suffer to exist any Indebtedness (including any Earn-Out Obligations), except:
  - (a) Indebtedness under the Loan Documents (including Banking Services Obligations);

(b) Indebtedness of the Loan Parties and their Subsidiaries existing on the <u>ClosingSecond Amendment Effective</u> Date and set forth in <u>Schedule 7.01</u> and refinancings and extensions of any such Indebtedness if the representations, warranties, covenants, events of default and other material terms and conditions thereof are not materially less favorable to the obligor thereon or to the Lenders than the Indebtedness being refinanced or extended, and the average life to maturity thereof is greater than or equal to that of the Indebtedness being refinanced or extended, such Indebtedness permitted under the immediately preceding clause shall not (i) include Indebtedness of an obligor that was not an obligor with respect to the Indebtedness being extended, renewed or refinanced or (ii) exceed in a principal amount the Indebtedness being renewed, extended or refinanced;

(c) purchase money Indebtedness (including obligations in respect of Capital Leases but excluding Synthetic Leases) hereafter incurred by the Loan Parties or any of their Subsidiaries to finance the purchase of fixed assets, provided that (i) such Indebtedness when incurred shall not exceed the purchase price of the asset(s) financed, (ii) no such Indebtedness shall be refinanced for a principal amount in excess of the principal balance outstanding thereon at the time of such refinancing and (iii) the total amount of all such Indebtedness at any time outstanding shall not exceed \$40,000,000;

(d) obligations (contingent or otherwise) of any Loan Party or any Subsidiary existing or arising under any Hedge Agreement to the extent required by Section 6.20;

(e) intercompany Indebtedness permitted under <u>Section 7.03(h)</u>, <u>provided</u> that such intercompany Indebtedness is evidenced by a demand note (which may cover all such intercompany Indebtedness) in form and substance satisfactory to the Administrative Agent and pledged and delivered to the Administrative Agent pursuant to the Security Agreement as additional collateral security for the Obligations, and the obligations under such demand note shall be subordinated to the Obligations in a manner satisfactory to the Administrative Agent;

(f) Indebtedness assumed in connection with (or attaching to assets of a Person that becomes a Subsidiary in connection with) a Permitted Acquisition, provided that (i) the aggregate amount of such Indebtedness shall not exceed  $\frac{6,000,0007,500,000}{6,000,0007,500,000}$  in the aggregate at any time and (ii) such Indebtedness exists at the time such Person becomes a Subsidiary or such Permitted Acquisition occurs and is not created in contemplation of or in connection therewith;

(g) Guarantees with respect to Indebtedness permitted under this <u>Section 7.01</u>;

(h) current Indebtedness maturing in less than one (1) year and incurred in the ordinary course of business for raw materials, supplies, equipment, services Taxes or labor;

(i) so long as in the aggregate such Indebtedness does not exceed \$4,000,000, Indebtedness consisting of the financing of insurance premiums in the ordinary course of business or consistent with past practice;

(j) Indebtedness in respect of performance bonds, bid bonds, appeal bonds, surety bonds and completion guarantees and similar obligations (including, in each case, letters of credit issued to provide such bonds, guaranties and similar obligations), in each case provided in the ordinary course of business, including those incurred to secure health, safety and environmental obligations in the ordinary course of business;

(k) Indebtedness arising from overdraft facilities and/or the honoring by a bank or other financial institution of a check, draft or similar instrument drawn against insufficient funds in the ordinary course of business;

(1) Indebtedness due to sellers in connection with Permitted Acquisitions, including any Earn-Out Obligations incurred in connection thereto, so long as (a) with respect to the Verus Merger and subject to clause (d) hereof, the aggregate principal amount of such Indebtedness shall not exceed \$12,000,000 at any time outstanding, (b) with respect to the Gould's Acquisition and subject to clause (d) hereof, the aggregate principal amount of such Indebtedness shall not exceed \$1,500,000 at any time outstanding, (c) with respect to the Med Way Acquisition and subject to clause (d) hereof, the aggregate principal amount of such Indebtedness shall not exceed \$5,000,000 at any time outstanding, (d) each of the Earn-Out Obligations with respect to the Verus Merger, Gould's Acquisition and Med Way Acquisition shall be subordinated to the Obligations in a manner satisfactory to the Administrative Agent and prior to any payments (other than payments solely in Capital Stock (which may not be Disqualified Capital Stock) of Holdings) of Earn-Out Obligations with respect to each of the Verus Merger, Gould's Acquisition and Med Way Acquisition, Administrative Agent shall have received a Pro Forma Compliance Certificate demonstrating that, upon giving effect to such payment on a Pro Forma Basis, (i) the Loan Parties would be in compliance with the financial covenants set forth in Article 8 as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to Section 6.01(a) or Section 6.01(b), as applicable, (ii) the Consolidated Total Leverage Ratio does not exceed 3.00:1.00 as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to Section 6.01(a) or Section 6.01(b), as applicable, and (iii) Excess Liquidity as of the date of the payment shall be at least \$10,000,000, and (e) for all other Permitted Acquisitions, the aggregate principal amount of such Indebtedness shall not exceed \$8,500,00020,000,000 at any time outstanding and shall, at the request of the Administrative Agent, be subject to a Subordination Agreement acceptable to the Administrative Agent in its sole discretion;

(m) [reserved];

(n) the Gould's Subordinated Note in an aggregate principal amount not to exceed \$2,000,000 so long as the obligations thereunder are subject to a Subordination Agreement in form and substance satisfactory to the Administrative Agent (provided, prior to any payments (other than payments solely in Capital Stock (which may not be Disqualified Capital Stock) of Holdings) of principal under the Gould's Subordinated Note, Administrative Agent shall have received a Pro Forma Compliance Certificate demonstrating that, upon giving effect to such payment on a Pro Forma Basis, (i) the Loan Parties would be in compliance with the financial covenants set forth in<u>Article 8</u> as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to<u>Section 6.01(a)</u> or <u>Section 6.01(b)</u>, as applicable, (ii) the Consolidated Total Leverage Ratio does not exceed 3.00:1.00 as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to <u>Section 6.01(a)</u> or <u>Section 6.01(b)</u>, as applicable, and (iii) Excess Liquidity as of the date of the payment shall be at least \$10,000,000);

(o) Incremental Equivalent Debt; and

(p) (o) other unsecured Indebtedness in an aggregate principal amount not exceeding \$5,000,00010,000,000 at any time

outstanding.

7.02 Liens. Create, incur, assume or suffer to exist any Lien upon any of its Property, whether now owned or hereafter acquired, other than the following:

(a) Liens pursuant to any Loan Document;

(b) Liens existing on the ClosingSecond Amendment Effective Date and listed on Schedule 7.02;

(c) Liens (other than Liens imposed under ERISA) for taxes, assessments or governmental charges or levies not yet due or which are being Properly Contested;

(d) statutory Liens of landlords and Liens of carriers, warehousemen, mechanics, materialmen and suppliers and other Liens imposed by law or pursuant to customary reservations or retentions of title arising in the ordinary course of business, <u>provided</u> that such Liens secure only amounts not yet due and payable or, if due and payable, (i) are unfiled and no other action has been taken to enforce the same or (ii) are being Properly Contested;

(e) segregated cash pledges or deposits in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, other than any Lien imposed by ERISA;

(f) segregated cash deposits to secure the performance of bids, trade contracts, licenses and leases, statutory obligations, surety bonds (other than bonds related to judgments or litigation), performance bonds and other obligations of a like nature (other than Indebtedness) incurred in the ordinary course of business;

(g) easements, rights-of-way, restrictions and other similar encumbrances affecting Real Property which, in the aggregate, are not substantial in amount, and which do not in any case materially detract from the value or marketability of the property subject thereto or materially interfere with the ordinary conduct of the business of the applicable Person;

(h) Liens securing Indebtedness permitted under <u>Section 7.01(c)</u>, <u>provided</u> that (i) such Liens do not at any time encumber any Property other than the Property financed by such Indebtedness, (ii) the Indebtedness secured thereby does not exceed the cost or fair market value, whichever is lower, of the Property being acquired on the date of acquisition and (iii) such Liens (unless such Lien is a refinancing a pre-existing Lien) attached to such Property concurrently with or within thirty (30) days after the acquisition thereof;

(i) leases, non-exclusive licenses or subleases granted to others not interfering in any material respect with the business of any Loan Party or any Subsidiary and not adverse to the interests of the Administrative Agent or the Lenders in any material respect;

(j) any interest of title of a lessor under, and Liens arising from precautionary UCC financing statements (or equivalent filings, registrations or agreements in foreign jurisdictions) solely evidencing such lessor's interest under, leases permitted by this Agreement;

(k) normal and customary rights of setoff upon deposits of cash in favor of banks or other depository institutions holding such deposits;

(I) Liens of a collection bank arising under Section 4-210 of the UCC on items in the course of collection;

(m) Liens solely on any cash earnest money deposits made by any Loan Party or any Subsidiary in connection with any letter of intent or purchase agreement in respect of any Investment permitted hereunder;

(n) all bonds, deposits and security instruments or other Liens required or imposed by any Governmental Authority or Third Party Payor in connection with the Business of the Loan Parties in the ordinary course of business;

(o) Liens in favor of the Administrative Agent, L/C Issuer, Support Provider or Swingline Lender on cash collateral securing the obligations of a Defaulting Lender to fund risk participations hereunder;

(p) Liens consisting of judgment, appeal bonds, judicial attachment liens or other similar Liens arising in connection with court proceedings, provided that the enforcement of such Liens is effectively stayed and all such Liens secure judgments the existence of which do not constitute an Event of Default under Section 9.01(h);

(q) Liens (junior in priority and subordinated in all respects to the Liens securing the Obligations in a manner satisfactory to the Administrative Agent through at least one hundred and eighty
 (180) days following the Term Loan Maturity Date (as such date may be extended through any amendment, restatement, supplement or other modification to this Agreement)) securing Indebtedness permitted under <u>Section 7.01(1);</u>

(r) Liens securing Indebtedness permitted under Section 7.01(o); and

(s) (t) Liens (junior in priority to the Liens securing the Obligations) not otherwise permitted hereunder so long as neither (i) the aggregate outstanding principal amount of the obligations secured thereby nor (ii) the aggregate fair market value (determined as the date such Lien is incurred) of the assets subject thereto exceeds (as to the Borrowers and all Subsidiaries) \$4,000,0007,500,000 at any one time.

7.03 Investments. Make any Investments, except:

(a) cash or Cash Equivalents;

(b) accounts receivable created, acquired or made and trade credit extended in the ordinary course of business and payable or dischargeable in accordance with customary trade terms;

(c) Investments consisting of stock, obligations, securities or other property received in settlement of accounts receivable (created in the ordinary course of business) from bankrupt obligors;

(d) Investments existing as of the ClosingSecond Amendment Effective Date and set forth in Schedule 7.03; provided that the amount of such Investment is not increased after the ClosingSecond Amendment Effective Date except in accordance with this Section 7.03;

(e) Guarantees permitted by <u>Section 7.01;</u>

(f) Permitted Acquisitions and Capital Expenditures otherwise permitted hereunder;

(g) loans or advances to officers, directors, managers, consultants and employees of any Loan Party for travel, entertainment, relocation and analogous ordinary business purposes in the ordinary course of business or constituting advances in payroll payments and expenses or relating to indemnification or reimbursement in respect of liability relating to their serving in any capacity in the ordinary course of business; provided, such loans and advances at any time outstanding shall not in the aggregate exceed \$1,000,0002,000,000;

(h) intercompany Investments by any Loan Party in any other wholly owned Loan Party (excluding Intermediate Holdings);

(i) non-cash purchase money loans and advances to employees, directors and officers to finance the purchase by such Persons of Capital Stock issued by Intermediate Holdings;provided, such loans and advances at any time outstanding shall not in the aggregate exceed \$4,000,000;

(j) to the extent constituting an Investment, any Secured Hedge Agreement entered into for risk management purposes and not speculative purposes;

(k) Investments constituting deposits made in connection with the purchase of goods or services in the ordinary course of business or otherwise constituting deposits permitted pursuant to Section 7.02(f), (k) or (n);

(l) Investments in the ordinary course of business and consistent with past practice, consisting of (i) endorsements for collection or deposit, (ii) customary trade arrangements with customers,

(iii) loans or advances made to distributors not to exceed in the aggregate at any time \$250,000 so long as no Default or Event of Default exists at the time such loan or advance is made or would result from the making of such loan or advance, (iv) advances of payroll payments to employees or other advances of salaries or compensation (including advances against commissions) to employees and sales representatives not to exceed in the aggregate at any time \$500,000 so long as no Default or Event of Default exists at the time such loan or advance is made or would result from the making of such loan or advance and (v) Investments maintained in connection with any Loan Party's deferred compensation plan approved by such Loan Party's board of directors or managers, as applicable;

(m) Investments in Excluded Subsidiaries and non-Wholly Owned Subsidiaries in an aggregate amount (as of the date such Investment is made) not to exceed \$3,000,000 prior to the Termination Date;

(n) Investments (other than Acquisitions) to the extent that payment for such Investments is made solely from Net Cash Proceeds from the issuance or sale of Qualified Capital Stock of AdaptHealth Corp. (other than Cure Amounts); and

(0) (n) so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom, other Investments by the Loan Parties and their Subsidiaries (other than Investments in Persons who are not Loan Parties) not exceeding in the aggregate the sum of (i) \$3,000,000 plus (ii) any returns (including dividends, interest, distributions and returns of principal) actually received by the Loan Parties in cash in respect of Investments made under Section 7.03(k).

**7.04 Fundamental Changes**. Merge, dissolve, liquidate, consolidate with or into another Person; <u>provided</u>, <u>however</u>, that, subject to the terms of <u>Sections 6.12</u> and <u>6.14</u> (a) any Loan Party other than Intermediate Holdings may merge or consolidate with any other Loan Party other than Intermediate Holdings, <u>provided</u> that, if such transaction involves a Borrower, such Borrower is the surviving entity, (b) any Subsidiary may merge with any Person that is not a Loan Party in connection with a Disposition permitted under<u>Section 7.05</u>, and (c) any Wholly Owned Subsidiary may dissolve, liquidate or wind up its affairs at any time provided that such dissolution, liquidation or winding up, as applicable, could not reasonably be expected to have a Material Adverse Effect and substantially all of its assets and business is transferred to a Loan Party in any manner satisfactory to the Administrative Agent. Notwithstanding anything in the foregoing to the contrary, no merger or consolidation otherwise permitted under any of the foregoing provisos shall be permitted if any Loan Party would not be Solvent after giving effect to such merger or consolidation.

7.05 **<u>Dispositions</u>**. Make any Disposition other than:

(a) sales of inventory in the ordinary course of business of the Loan Parties and their Subsidiaries;

(b) any sale, transfer or other disposition of Property by any Loan Party to any other Loan Party so long as no Loan Party would fail to be Solvent after giving effect to such sale, transfer or other disposition;

(c) so long as no Default or Event of Default exists or would result therefrom, (i) sell, transfer or otherwise dispose of equipment for fair market value not in excess of \$100,000 per transaction (or series of related transactions) and \$500,000 in any Fiscal Year of the Loan Parties, and (ii) sell or trade-in equipment in connection with the acquisition of replacement equipment;

(d) the sale, transfer or other disposition of obsolete or worn out tangible Property which is no longer used or useful in the conduct of business of the Loan Parties and their Subsidiaries;

(e) any Involuntary Disposition by any Loan Party or any Subsidiary;

(f) any non-exclusive license of any IP Rights by any Loan Party or any Subsidiary in the ordinary course of business, <u>provided</u> that such license does not materially impair the value of any such IP Right as Collateral or the security interest of the Administrative Agent granted under the Loan Documents;

(g) the sale, transfer or other disposition of accounts receivable constituting bad debts in connection with the compromise, settlement or collection thereof in the ordinary course of business (and not as part of a bulk sale or receivables financing); and

(h) other Dispositions by the Borrowers and their Subsidiaries the proceeds (valued at the principal amount in the case of non-cash proceeds consisting of notes or other evidences of Indebtedness and valued at fair market value in the case of all other non-cash proceeds) of which do not exceed \$2,000,0003,000,000 in the aggregate in any Fiscal Year, provided that (i) no such Disposition involves the sale or other disposition of a minority equity interest in the any Borrower or any other Loan Party (other than Intermediate Holdings to the extent permitted hereunder), (ii) no less than ninety percent (90%) of the proceeds of such Disposition consist of cash or Cash Equivalents received contemporaneously with the consummation of such Disposition, (iii) the proceeds received in consideration for such Disposition shall not be less than the fair market value of the Property disposed of, (iv) the Net Cash

Proceeds of any such Disposition are applied in the manner specified in <u>Section 2.05(b)(ii)</u> hereof and (v) both immediately before and after giving effect to any such Disposition, no Default or Event of Default exists or would result therefrom.

Notwithstanding the foregoing, nothing in this <u>Section 7.05</u> shall directly or indirectly permit any Disposition that would result in a Change of Control.

7.06 <u>Restricted Payments</u>. Directly or indirectly declare or make any Restricted Payment or incur any obligation to do so, except that:

(a) each Subsidiary of the Loan Parties may make Restricted Payments (directly or indirectly) to any Loan Party (other than Intermediate Holdings);

(b) each Loan Party and each Subsidiary may declare and make non-cash dividend payments or other distributions payable solely in the Capital Stock (other than Disqualified Capital Stock) of the Person making such dividend or distribution;

(c) the Borrowers may pay expense reimbursement obligations in respect of reasonable and documented third-party outof-pocket costs incurred by the Sponsor that are reimbursable to the Sponsor by the Borrowers pursuant to management agreements entered into following the Closing Date, so long as (i) such reimbursed costs do not exceed \$500,000 in the aggregate in any Fiscal Year, (ii) immediately before and after giving effect to such payment no Default or Event of Default then exists or would result therefrom and (iii) the Loan Parties will be in compliance on a Pro Forma Basis with the financial covenants set forth in <u>Article 8</u> for the most recently completed four (4) Fiscal Quarter period, after giving effect to such payment;

(d) the Loan Parties may make Restricted Payments to Holdings to pay for the repurchase, retirement or other acquisition or retirement for value of Capital Stock of Holdings held by any present or former employee or director of Holdings or any of its Subsidiaries pursuant to any employee or director equity plan, employee or director stock option plan or any other employee or director benefit plan or any agreement (including any stock subscription or shareholder agreement) upon such Person's death, disability, retirement or termination of employment or under the terms of any such benefit plan or agreement, <u>provided</u> that (i) the aggregate amount of any such purchases or redemptions shall not exceed \$1,750,000 in any Fiscal Year, (ii) no Default or Event of Default then exists or would result therefrom, and (iii) the Loan Parties shall have Excess Liquidity in an amount equal to or greater than \$10,000,000;

(e) the Loan Parties may make Restricted Payments to Holdings, and Holdings may in turn distribute Tax Distributions (which may only be paid annually based on HoldingsAdaptHealth Corp.'s audited financial statements or, so long as no Event of Default is then outstanding, in multiple installments, based on Borrower's good-faith and reasonable estimate of income to be generated by HoldingsHolding's and its Subsidiaries' business in such year) to allow ParentsAdaptHealth Corp. to meet their tax obligations on such income in a timely manner;

(f) the Loan Parties may make dividend payments and distributions as required pursuant to (i) Section 14 of that certain operating agreement for Royal Homestar LLC, dated as of March 1, 2013, in the form provided to the Administrative Agent on the Closing Date, (ii) Section 4.02 of that certain operating agreement for Southeast Sleep Holdings, LLC, in the form provided to the Administrative Agent on May 17, 2018, (iii) Section 4.02 of that certain operating agreement for Sleep Therapy, LLC, in the form provided to the Administrative Agent on December 20, 2018, subject to <u>Section 7.10</u> hereof, and (iv) Section 4.02 of that certain operating agreement for Clean General Healthcare Systems, LLC, in the form provided to the Administrative Agent on May 17, 2018<u>provided</u>, in each case, that (1) no Default or Event

of Default then exists or would result therefrom and (2) the Loan Parties will be in compliance on a Pro Forma Basis with the financial covenants set forth in <u>Article 8</u> for the most recently completed four (4) Fiscal Quarter period, after giving effect to such dividend payment or distribution;

(g) the Loan Parties may make Restricted Payments to Holdings (i) to make Preferred Note Cash Interest Payments due and owing at such time under the Preferred Note (without giving effect to any amendment, restatements, supplement or any other modification to such Preferred Note not approved in writing by the Administrative Agent) and (ii) AHYDO Catch-Up Payments, in each case so long as (A) immediately before and after giving effect to such payment no Default or Event of Default then exists or would result therefrom and (B) the Loan Parties will be in compliance on a Pro Forma Basis with the financial covenants set forth in<u>Article 8</u> for the most recently completed four (4) Fiscal Quarter period, after giving effect to such payment;

(h) the Loan Parties may make Restricted Payments to Holdings to make Preferred Note PIK Interest Payments due and owing at such time under the Preferred Note (without giving effect to any amendment, restatements, supplement or any other modification to such Preferred Note not approved in writing by the Administrative Agent) so long as (i) immediately before and after giving effect to such payment no Default or Event of Default then exists or would result therefrom, (ii) the Loan Parties will be in compliance on a Pro Forma Basis with the financial covenants set forth in <u>Article 8</u> (other than with respect to the Consolidated Total Leverage Ratio which is subject to the immediately following clause (iii)) for the most recently completed four (4) Fiscal Quarter period, after giving effect to such payment, and (iii) the Loan Parties will be in compliance on a Pro Forma Basis with a Consolidated Total Leverage Ratio of less than 3.00 to 1.00 for the most recently completed four (4) Fiscal Quarter period, after giving effect to such payment;

(i) the Loan Parties may make Restricted Payments not otherwise permitted by this <u>Section 7.06</u> so long as (i) immediately before and after giving effect to such payment no Default or Event of Default then exists or would result therefrom and (ii) the Loan Parties will be in compliance on a Pro Forma Basis with a Consolidated Total Leverage Ratio of less than 1.75 to 1.00 for the most recently completed four (4) Fiscal Quarter period, after giving effect to such payment; and

(j) the Loan Parties may make Restricted Payments constituting Closing Date Distributions.

7.07 <u>Change in Nature of Business; Cloud-Based Billing</u> Engage to any material extent in any business different from the business conducted by the Loan Parties and their Subsidiaries on the Closing Date and businesses reasonably related thereto or reasonably ancillary, complementary or a reasonable extension, development or expansion of such business. No Loan Party will utilize or enter into any agreement with any remote access or internet-based/cloud-based billing services or records repository provider without first entering into such agreements with the Administrative Agent and such service provider, in form and substance satisfactory to the Administrative Agent, as the Administrative Agent shall deem necessary or helpful in the collection or realization of or on its Collateral.

**7.08** Transactions with Affiliates and Insiders. Enter into or permit to exist any transaction or series of transactions with (a) any Excluded Subsidiary or (b) any officer, director or Affiliate of such Person other than (i) transactions between the Loan Parties, (ii) intercompany transactions expressly permitted by Sections 7.01, 7.03, 7.04, 7.05 or 7.06, (iii) customary and reasonable compensation and reimbursement of expenses of officers and directors, (iv) to the extent permitted by Section 7.06, payment of the Sponsor expense reimbursements, and (v) except as otherwise specifically limited in this Agreement, other transactions which are entered into in the ordinary course of such Person's business on terms and

conditions substantially as favorable to such Person as would be obtainable by it in a comparable arm's length transaction with a Person other than an Excluded Subsidiary, officer, director or Affiliate.

**7.09 Burdensome Agreements**. Enter into or permit to exist any Contractual Obligation that encumbers or restricts the ability of any Loan Party or any Subsidiary to (i) pay dividends or make any other distributions to any Loan Party on its Capital Stock or with respect to any other interest or participation in, or measured by, its profits, (ii) pay any Indebtedness or other obligation owed to any Loan Party, (iii) make loans or advances to any Loan Party, (iv) sell, lease or transfer any of its Property to any Loan Party, (v) grant any Lien on any of its Property to secure the Obligations pursuant to the Loan Documents or any renewals, refinancings, exchanges, refundings or extension thereof or (vi) act as a Loan Party pursuant to the Loan Documents or any renewals, refinancings, exchanges, refundings or extension thereof, except (in respect of any of the matters referred to in clauses (i)-(vi) above) for (A) this Agreement and the other Loan Documents, (B) any document or instrument governing Indebtedness incurred pursuant to <u>Section 7.01(c)</u>, provided that any such restriction contained therein relates only to the asset or assets constructed or acquired in connection therewith, (C) any Permitted Lien or any document or instrument governing any Permitted Lien, provided that any such restriction contained therein relates only to the asset or assets subject to such Permitted Lien, (D) customary restrictions and conditions contained in any agreement relating to the sale of any Property permitted under <u>Section 7.05</u> pending the consummation of such sale, and (E) customary restrictions on assignment contained in leases, licenses and other contracts entered into in the ordinary course of business with third parties and not for the purpose of circumventing any provision of this Agreement.

**7.10** <u>Use of Proceeds</u>. Use the proceeds of any Credit Extension, whether directly or indirectly, and whether immediately, incidentally or ultimately, to purchase or carry margin stock (within the meaning of Regulation U of the FRB) or to extend credit to others for the purpose of purchasing or carrying margin stock or to refund Indebtedness originally incurred for such purpose; <u>provided</u>, <u>however</u>, that no proceeds of any Credit Extension shall be received in any manner by any Excluded Subsidiary, including by way of any Investment or distribution of any Loan Party other than as expressly permitted herein.

## 7.11 Amendments to Certain Agreements.

(a) Pay any Earn-Out Obligations or obligations under the Gould's Subordinated Note, in each case, prior to the due date for such obligations or when a Default or an Event of Default is continuing.

(b) Amend or modify, or waive any rights under any Material Contract if, in any case, such amendment, modification or waiver could reasonably be expected to be materially adverse to the interests of the Administrative Agent or the Lenders.

## 7.12 Organization Documents; Fiscal Year; Legal Name, State of Formation and Form of Entity.

(a) Amend, modify or change its Organization Documents in a manner which is materially adverse to the interests of the Administrative Agent or the Lenders.

(b) Change its Fiscal Year.

(c) Change its name or its state of formation or form of organization without providing thirty (30) days prior written notice to the Administrative Agent, provided that nothing herein shall permit any Loan Party to change its state of formation in a state or jurisdiction outside the United States.

(d) Make any significant change in accounting treatment or reporting practices, except as required by GAAP.

**7.13** <u>Ownership of Subsidiaries</u>. Notwithstanding any other provisions of this Agreement to the contrary, (i) permit any Person (other than any Borrower or any Wholly Owned Subsidiary) to own any Capital Stock of any Subsidiary other than owned by any holders (other than Loan Parties) of Capital Stock of any non-Wholly Owned Subsidiary permitted unde<u>clauses (d)</u>, (f), and/or (m) of Section 7.03, (ii) permit any Subsidiary to issue or have outstanding any shares of Disqualified Capital Stock, (iii) create, incur, assume or suffer to exist any Lien on any Capital Stock of any Subsidiary other than pursuant to the Loan Documents.

# 7.14 Sale and Leaseback Transactions. Enter into any Sale and Leaseback Transaction.

**7.15** Limitations on Holdings. Permit AdaptHealth Corp., Intermediate Holdings or Holdings, directly or indirectly, to (a) incur, directly or indirectly, any Indebtedness or any other obligation or liability whatsoever other than (i) Holdings' Obligations under the Preferred Note and (ii) Intermediate Holdings' Obligations hereunder, (b) create or suffer to exist any Lien upon any property or assets now owned or hereafter acquired by Holdings or Intermediate Holdings other than, with respect to Intermediate Holdings only, the Liens created under the Loan Documents to which it is a party, (c) engage in any business or activity or own any assets (including, without limitation, cash and Cash Equivalents) other than (i) holding one hundred percent (100%) of the Capital Stock of Holdings. Intermediate Holdings, performing its obligations and activities incidental thereto and (ii) with respect to Intermediate Holdings, performing its obligations under the Loan Documents, (d) consolidate with or merge with or into, or convey, transfer or lease all or substantially all its assets to, any Person, or (e) fail to hold itself out to the public as a legal entity separate and distinct from all other Persons.

7.16 <u>Account Control Agreements: Bank Accounts</u>. Open, maintain or otherwise have any bank account, other than (a) deposit accounts with CIT Bank, N.A., (b) deposit accounts consented to by the Administrative Agent that are subject to a Deposit Account Control Agreement, and (c) Excluded Accounts.

# 7.17 <u>Permits</u>.

(a) (i) Suffer or permit to occur (A) any transfer of a Required Permit or rights thereunder to any Person (other than a Loan Party or the Administrative Agent) other than with respect to a Voluntary Termination; or (B) any rescission, withdrawal, revocation, termination, amendment or modification of or other alteration to the nature, tenor or scope of any Required Permit except for any such amendment, modification or other alteration which does not have a Material Adverse Effect and does not materially adversely affect the Administrative Agent's rights and remedies with respect to the Collateral; or

(ii) rescind, withdraw, revoke, amend, modify, supplement, or otherwise alter the nature, tenor or scope of the Required Permits in any material respect, other than with respect to a Voluntary Termination.

(b) Terminate, surrender, modify, limit, withdraw or rescind any Participation Agreement or participation in any other Third Party Payor Program.

7.18 <u>Covenants Relating to Excluded Subsidiaries</u>. Notwithstanding anything in this Agreement to the contrary:

(a) no Loan Party shall provide any guarantee of, or any credit support for, any Indebtedness or other obligation (contingent or otherwise) of an Excluded Subsidiary or Excluded Foreign

Subsidiary, or otherwise be directly or indirectly liable for any Indebtedness or other obligation (contingent or otherwise) of such Excluded Subsidiary or Excluded Foreign Subsidiary, (i) have any direct or indirect obligation to maintain or preserve the financial condition of such Excluded Subsidiary or Excluded Foreign Subsidiary or to cause any such Excluded Subsidiary or Excluded Foreign Subsidiary to achieve any specified level of operating results, and (ii) permit a Lien on any of its property to secure, or permit any of its property to be otherwise subject (directly or indirectly) to the satisfaction of, any Indebtedness or other obligation (contingent or otherwise), of any Excluded Subsidiary or Excluded Foreign Subsidiary; and

(b) no Loan Party shall permit an Excluded Subsidiary to (i) own any Capital Stock issued by a Loan Party, (ii) hold any Indebtedness of any Loan Party, or (iii) hold any Lien on property of any Loan Party.

## **ARTICLE 8**

# FINANCIAL COVENANTS

**8.01** Financial Covenants. So long as any Lender shall have any Commitment hereunder or any Loan or other Obligation hereunder shall remain unpaid or unsatisfied, or any Letter of Credit shall remain outstanding, the Loan Parties agree that:

(a) <u>Consolidated Total Leverage Ratio</u> The Consolidated Total Leverage Ratio as of the end of any Fiscal Quarter set forth below shall not be greater than the ratio set forth opposite such Fiscal Quarter:

Each Fiscal Quarter	Consolidated Total
Ending	Leverage Ratio
June 30, 2019 through and	3.75 to 1.00
including September 30,	
2020	
December 31, 2020 through the	3.25 to 1.00
Termination Date	

(b) <u>Consolidated Fixed Charges Coverage Ratio</u>. Beginning with the Fiscal Quarter ending June 30, 2019, the Consolidated Fixed Charges Coverage Ratio as of the end of each Fiscal Quarter shall not be less than 1.25 to 1.00.

## **ARTICLE 9**

# EVENTS OF DEFAULT AND REMEDIES

9.01 <u>Events of Default</u>. Any of the following shall constitute an Event of Default:

(a) <u>Non-Payment</u>. Any Borrower or any other Loan Party fails to pay when and as required to be paid pursuant to this Agreement or any other Loan Document, (i) any amount of principal of any Loan or any Letter of Credit Liabilities, or (ii) within one (1) Business Day after the same becomes due, any interest on any Loan or any Letter of Credit Liabilities, or any commitment fee, utilization fee or other fee due hereunder, or (iii) within five (5) Business Day after the same becomes due, any other amount payable hereunder or under any other Loan Document; or

(b) <u>Specific Covenants</u>. Any Loan Party fails to perform or observe any term, covenant or agreement contained in any of <u>Sections 6.01</u>, <u>6.02</u>, <u>6.03</u>, <u>6.05</u> (with respect to each Borrower's existence), <u>6.07</u>, <u>6.10</u>, <u>6.11</u>, <u>6.12</u>, <u>6.14</u>, <u>6.18</u>, <u>6.19</u>, <u>Article 7</u>, <u>Article 8</u> or <u>Article 10</u>; or

(c) <u>Other Defaults</u>. (i) An event of default has occurred under any other Loan Document, or (ii) any Loan Party fails to perform, observe or comply with any other covenant or agreement (not specified in subsection (a) or (b) above) contained in any Loan Document and such failure continues for thirty (30) days after the earlier of (x) a Responsible Officer of any Loan Party becoming aware of such failure or (y) notice thereof to any Loan Party by the Administrative Agent or any Lender; or

(d) <u>Representations and Warranties</u>. Any representation, warranty, certification or statement of fact made or deemed made by or on behalf of any Borrower or any other Loan Party herein, in any other Loan Document, or in any document delivered in connection herewith or therewith shall be incorrect or misleading in any material respect when made or deemed made (except to the extent already qualified by knowledge, materiality or Material Adverse Effect, in which case it shall be true and correct in all respects and shall not be incorrect or misleading in any respect); or

(e) <u>Cross-Default</u>. (i) Any Loan Party or any Subsidiary fails to make any payment when due (whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise) in respect of one or more items of Indebtedness (other than Indebtedness hereunder or under any Hedge Agreement) having an aggregate principal amount (including undrawn committed or available amounts) of more than \$2,500,000; (ii) any Loan Party or any Subsidiary fails to observe or perform any other agreement or condition relating to any such Indebtedness or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event occurs, the effect of which default or other event is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, with the giving of notice if required, such Indebtedness to be demanded or to become due or to be repurchased, prepaid, defeased or redeemed (automatically or otherwise), or an offer to repurchase, prepay, defease or redeem such Indebtedness to be made, prior to its stated maturity; or (iii) there occurs under any Hedge Agreement an Early Termination Date (as defined in such Hedge Agreement) resulting from (A) any event of default under such Hedge Agreement as to which any Loan Party or any Subsidiary is an Affected Party (as so defined); or

(f) <u>Insolvency Proceedings, Etc</u> Any Loan Party or any of its Subsidiaries institutes or consents to the institution of any proceeding under any Debtor Relief Law, or makes an assignment for the benefit of creditors, or applies for or consents to the appointment of any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer for it or for all or any material part of its Property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of such Person and the appointment continues undischarged or unstayed for sixty

(60) calendar days; or any proceeding under any Debtor Relief Law relating to any such Person or to all or any material part of its Property is instituted without the consent of such Person and such Person fails to challenge such Proceeding or such Proceeding is challenged but continues undismissed or unstayed for sixty (60) calendar days, or an order for relief is entered in any such proceeding; or

(g) <u>Inability to Pay Debts; Attachment</u> (i) Any Loan Party or any of its Subsidiaries becomes unable to or admits in writing its inability or fails generally to pay its debts as they become due, or (ii) any writ or warrant of attachment or execution or similar process is issued or levied against all or any material part of the Property of any such Loan Party or any Loan Party otherwise becomes insolvent; or

(h) <u>Judgments</u>. There is entered against any Loan Party or any of its Subsidiaries (i) one or more final judgments or orders for the payment of money (including a disgorgement order issued by a Governmental Authority) in an aggregate amount exceeding \$2,500,000 (to the extent not covered by independent third-party insurance as to which the insurer has not disclaimed its obligation to cover), or (ii) one or more non-monetary final judgments that have, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect and, in either case, (A) enforcement proceedings are commenced by any creditor upon such judgment or order, or (B) there is a period of thirty (30) consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, is not in effect; or

(i) <u>ERISA</u>. (i) An ERISA Event occurs which has resulted or could reasonably be expected to result in liability of any Loan Party under Title IV of ERISA to the Pension Plan, Multiemployer Plan or the PBGC in an aggregate amount in excess of \$2,500,000, or (ii) any Loan Party or any ERISA Affiliate fails to pay when due, after the expiration of any applicable grace period, any installment payment with respect to its withdrawal liability under Section 4201 of ERISA under a Multiemployer Plan in an aggregate amount in excess of \$2,500,000; or

(j) Invalidity of Loan Documents Any Loan Document, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect or ceases to give the Administrative Agent, for the benefit of the Lenders, a valid and perfected Lien in any material portion of the Collateral purported to be covered by the Loan Documents with the priority required by the relevant Loan Document (except to the extent perfection is deferred pursuant to the terms of <u>Schedule 6.18</u>); or any Loan Party or any other Affiliate of a Loan Party contests in any manner the validity or enforceability of any Loan Document; or any Loan Party denies that it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind any Loan Document; or

- (k) <u>Change of Control</u>. There occurs any Change of Control; or
- (l) [Reserved]; or
- (m) [<u>Reserved</u>]; or

(n) <u>Healthcare Proceedings</u> The institution of any proceeding by FDA or similar Governmental Authority to order the withdrawal of any Product or Product category that is material to any Loan Parties' business, taken as a whole, from the market or to enjoin any such Person or any representative of any such Person from manufacturing, marketing, selling or distributing any Product or Product category that is material to such Persons' business, taken as a whole where such proceeding could reasonably be expected to have a Material Adverse Effect; or

(o) <u>Required Permits</u> The institution of any action or proceeding by the FDA or any other Governmental Authority to revoke, suspend, reject, withdraw, limit, or restrict any Required Permit held by any Loan Party or any representative of such Person if the same could reasonably be expected to result in a Material Adverse Effect or a material adverse change in, or a material adverse effect upon, the prospects of the Loan Parties taken as a whole; or

(p) <u>Enforcement Actions</u>. The commencement of any enforcement action against any Loan Party by the FDA or any other Governmental Authority if such enforcement action could reasonably be expected to result in a Material Adverse Effect; or

(q) <u>Product Recall</u>. The Recall of any Product from the market, the voluntary withdrawal of any Product from the market, or actions to discontinue the sale of any Product, if the same could reasonably be expected to result in a Material Adverse Effect; or

(r) <u>Change in Law</u>. A Change in Law, including a change in FDA policies or procedures or state government agency policies or procedures, occurs which could reasonably be expected to have a Material Adverse Effect; or

(s) <u>Inventory Supplier Agreements</u>. The termination of any agreements with manufacturers that supply any Product or any components of any Product or any changes to any agreements with manufacturers that supply any Product or any components of any Product that could reasonably be expected to have a Material Adverse Effect; or

(t) <u>Earn-Out Obligations in connection with Verus Merger, Gould's Acquisition, and Med Way Acquisition</u> Holdings, Intermediate Holdings or any Loan Party makes a payment (other than a payment solely in Capital Stock of Holdings) with respect to the Earn-Out Obligations in connection with the Verus Merger, Gould's Acquisition or Med Way Acquisition without the Administrative Agent having received a Pro Forma Compliance Certificate demonstrating that, upon giving effect to such payment on a Pro Forma Basis, (i) the Loan Parties would be in compliance with the financial covenants set forth in <u>Article 8</u> as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to <u>Section 6.01(a)</u> or <u>Section 6.01(b)</u>, as applicable, (ii) the Consolidated Total Leverage Ratio does not exceed 3.00:1.00 as of the most recent Fiscal Quarter for which the Loan Parties have delivered financial statements pursuant to <u>Section 6.01(b)</u>, as applicable, and (iii) Excess Liquidity as of the date of the payment shall be at least \$10,000,000.

9.02 <u>Remedies upon Event of Default</u> Upon the occurrence and during the continuance of any Event of Default, the Administrative Agent may, and upon the direction of the Required Lenders shall, take any or all of the following actions:

(a) declare the commitment of each Lender to make Loans and any obligation of the L/C Issuers or Support Providers to make L/C Credit Extensions to be terminated, whereupon such commitments and obligation shall be terminated;

(b) declare the unpaid principal amount of all outstanding Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document other than a Secured Hedge Agreement which shall be governed by its own provisions to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Loan Parties;

(c) require that the Borrowers Cash Collateralize the total Letter of Credit Liabilities (in an amount equal to 105% of the total Letter of Credit Liabilities as of such date) and prepay Letter of Credit Fees; and

(d) exercise on behalf of itself and the other Secured Parties any and all rights and remedies available to it and the Lenders under the Loan Documents or applicable Law;

provided, however, that upon the occurrence of any Event of Default described in Section 9.01(f) or 9.01(g), the obligation of each Lender to make Loans and any obligation of the L/C Issuers to make L/C Credit Extensions and Support Providers to issue Support Agreements shall automatically terminate, the unpaid principal amount of all outstanding Loans and all interest and other amounts as aforesaid shall automatically become due and payable, and the obligation of the Loan Parties to Cash Collateralize the total

Letter of Credit Liabilities as aforesaid shall automatically become effective, in each case without further act of the Administrative Agent or any Lender. Except as expressly provided for herein, presentment, demand, protest and all other notices (including notice of acceleration and notice of intent to accelerate) of any kind are hereby waived by the Borrowers.

# 9.03 [<u>Reserved</u>].

**9.04** <u>Application of Funds</u>. Upon the occurrence and during the continuance of an Event of Default (or after the Loans have otherwise become due and payable and the Letter of Credit Liabilities have been required to be Cash Collateralized as set forth in the proviso to <u>Section 9.02</u>), any amounts received on account of the Obligations shall be applied by the Administrative Agent in the following order upon the Administrative Agent's election or the direction of the Required Lenders:

(a) <u>First</u>, to payment of that portion of the Obligations constituting fees, indemnities, expenses, protective advances and other amounts (including reasonable Attorney Costs and amounts payable under <u>Article 3</u>) payable or reimbursable to the Administrative Agent in its capacity as such;

(b) <u>Second</u>, to payment of that portion of the Obligations constituting fees payable to the Lenders, ratably among them in proportion to the fees payable to them;

(c) <u>Third</u>, to payment of that portion of the Obligations constituting accrued and unpaid interest on the Loans and Unreimbursed Amounts and fees, premiums and scheduled periodic payments, and any interest accrued thereon, due under any Secured Hedge Agreement, ratably among the Secured Parties in proportion to the respective amounts described in this clause (c) held by them;

(d) <u>Fourth</u>, to payment of that portion of the Obligations constituting unpaid principal of the Loans and Unreimbursed Amounts, payments of amounts due in respect of any Banking Services Obligations, breakage, termination or other payments, and any interest accrued thereon, due under any Secured Hedge Agreement, to Cash Collateralize that portion of Letter of Credit Liabilities comprised of the aggregate undrawn amount of Letters of Credit and Support Agreements (together with aggregate facing and similar fees and expenses of the issuers of the Letters of Credit that will accrue on such Letters of Credit through the stated expiry of such Letters of Credit (assuming no drawings thereon before stated expiry date)), ratably among the Secured Parties in proportion to the respective amounts described in this clause (d) held by them;

(e) <u>Fifth</u>, to the payment of any other unpaid Obligations and to Cash Collateralize any other contingent Obligations which are not yet due and payable but with respect to which a claim has been or may reasonably be expected to be asserted by the applicable Secured Party in an amount estimated by Administrative Agent to be the amount of related costs, expenses and indemnification Obligations that may become due and payable; and

(f) <u>Last</u>, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrowers or as otherwise required by Law.

All amounts owing under this Agreement in respect of such Obligations, including fees, interest, default interest, interest on interests, expense reimbursement and indemnities, shall be payable in accordance with the foregoing waterfall provisions irrespective of whether a claim in respect of such amounts is allowed or allowable in any insolvency Proceeding under any Debtor Relief Law to the fullest extent permitted by Law.

Subject to <u>Section 2.03(c)</u>, amounts used to Cash Collateralize the aggregate undrawn amount of Letters of Credit and Support Agreements pursuant to <u>clause (d)</u> above shall be applied to satisfy drawings under such Letters of Credit and Support Agreements if and as they occur. If any amount remains on deposit as cash collateral after all Letters of Credit and Support Agreements have either been fully drawn or expired, and all Unreimbursed Amounts have been paid, such remaining amount shall be applied to other Obligations, if any, in the order set forth above.

In connection with any distribution of payments and collections pursuant to this <u>Section 9.04</u>, the Administrative Agent may in its discretion assume that no amounts are due to any Secured Hedge Provider or any provider of Banking Services or are due on account of clause (e) unless such Secured Hedge Provider or other applicable Secured Party has notified the Administrative Agent of the amount of any such liability owed to it prior to such distribution.

Notwithstanding anything to the contrary set forth above, Excluded Hedge Obligations with respect to any Guarantor shall not be paid with amounts received from such Guarantor or its assets, but appropriate adjustments shall be made with respect to payments from other Loan Parties to preserve the allocation to Obligations otherwise set forth above in this Section.

# 9.05 Loan Parties Right to Cure.

(a) Notwithstanding anything to the contrary contained in this <u>Article 9</u>, for purposes of determining whether the Loan Parties have failed to comply with any covenant contained in <u>Section 8.01</u> that is calculated by using Consolidated EBITDA, the Borrowers shall have the right (the "<u>Cure Right</u>") to increase Consolidated EBITDA for any Fiscal Quarter by the amount of Net Cash Proceeds actually received by Intermediate Holdings from the <u>Sponsor</u> (by way of common equityissuance of Qualified Capital Stock for cash by AdaptHealth Corp. or from cash contributions or in return forto the issuance capital of <u>Capital Stock</u>, in each case AdaptHealth Corp. as cash common equity or other than Disqualified Qualified Capital Stock) (any such equity contribution Net Cash Proceeds to included in the calculation of Consolidated EBITDA, a <u>Specified Equity Contribution</u>" and the amount of any such Specified Equity Contribution, the '<u>Cure Amount</u>") after the end of such Fiscal Quarter and on or prior to the date that is ten (10) <u>daysBusiness Days</u> after the date on which financial statements with respect to such Fiscal Quarter are required to be delivered pursuant to <u>Section 6.01(b)</u> (the "<u>Cure Date</u>"), in each case so long as, and to the extent that, such amounts are then contributed by Intermediate Holdings in cash to the common equity of the Borrowers, and that, on or prior to the Cure Date, the Borrowers shall inform the Administrative Agent of the Specified Equity Contribution, whereupon the covenants contained in <u>Section 8.01</u> shall be recalculated giving effect to the following pro forma adjustments:

(i) Consolidated EBITDA shall be increased, solely for the purpose of determining the existence of a Default or an Event of Default under the covenants contained in Section 8.01 with respect to any period of four (4) consecutive Fiscal Quarters that includes the Fiscal Quarter for which the Cure Right was exercised and not for any other purpose under this Agreement, by an amount equal to the Net Cash Proceeds of the Specified Equity Contribution; it being understood that for purposes of calculating the Consolidated Total Leverage Ratio as of the end of the Fiscal Quarter for which the Cure Right was exercised and the immediately following three (3) Fiscal Quarters, Consolidated Funded Indebtedness shall not be reduced by the amount of any prepayment or repayment of Indebtedness made with the proceeds of the Specified Equity Contribution; and

(ii) if, after giving effect to the foregoing recalculations, the Borrowers shall then be in compliance with the requirements of all covenants contained in <u>Section 8.01</u>, the Borrowers shall be deemed to have satisfied the requirements of the financial performance

covenants contained in <u>Section 8.01</u> as of the relevant date of determination with the same effect as though there had been no failure to comply therewith at such date, and the applicable breach or default of the financial performance covenants contained in <u>Section 8.01</u> that had occurred shall be deemed cured for the purposes of this Agreement, <u>provided</u>, that in any event, neither Administrative Agent nor any Lender shall be required to provide any Credit Extension following any Default or Event of Default under the covenants contained in <u>Section 8.01</u> for which a Cure Right has been exercised but for which the Specified Equity Contribution has not yet been made to the Borrowers and applied in accordance with this <u>Section 9.05(a)</u>.

Notwithstanding anything herein to the contrary, (a) there shall be no more than two (2) Fiscal Quarters in each four (4) Fiscal Quarter period in which a Cure Right is exercised; (b) during the term of this Agreement, there shall be no more than five (5) Fiscal Quarters in which a Cure Right is exercised; (c) there shall not be two (2) consecutive Fiscal Quarters in which a Cure Right is exercised; (d) the Cure Amount for any Fiscal Quarter shall be no greater than and not less than the amount required to cause the Borrowers to be in compliance with the financial covenants contained in Section 8.01 calculated by using Consolidated EBITDA for such Fiscal Quarter; (e) any and all proceeds of a Specified Equity Contribution shall be immediately due and payable as a mandatory prepayment of the Obligations in accordance with Section 2.05(b)(iii) and applied in accordance with Section 2.05(b)(vii)(B); (f) all Cure Amounts will be disregarded in calculating Consolidated EBITDA for all purposes (other than those set forth in this Section 9.05) or for any other purpose under this Agreement or the other Loan Documents (including in respect of the making of any Restricted Payment or the determination of the Applicable Margin); and (g) any Obligations prepaid with Cure Amounts shall be deemed outstanding for purposes of determining compliance with the financial covenants contained in Section 8.01 for the current Fiscal Quarter and the next three (3) Fiscal Quarters thereafter.

## **ARTICLE 10**

#### **GUARANTY**

**10.01** The Guaranty. Each Guarantor hereby guarantees to each Secured Party and the Administrative Agent as hereinafter provided, as primary obligor and not as surety, the prompt payment and performance of the Obligations in full when due (whether at stated maturity, as a mandatory prepayment, by acceleration or otherwise) strictly in accordance with the terms thereof. Each Guarantor hereby further agrees that if any of the Obligations are not paid in full when due (whether at stated maturity, as a mandatory prepayment, by acceleration or otherwise), each Guarantor will promptly pay the same, without any demand or notice whatsoever, and that in the case of any extension of time of payment or renewal of any of the Obligations, the same will be promptly paid in full when due (whether at extended maturity, as a mandatory prepayment, by acceleration or otherwise) in accordance with the terms of such extension or renewal (collectively, the "Guaranteed Obligations").

Subject to Section 10.06 and the last sentence of this Section 10.01 below, the Guarantors hereby jointly and severally agree, in furtherance of the foregoing and not in limitation of any other right which the Administrative Agent or any Secured Party may have at law or in equity against any Guarantor by virtue hereof, that upon the failure of any Guaranteed Obligations to be paid when and as the same shall become due, whether at stated maturity, by required prepayment, declaration, acceleration, demand or otherwise (including amounts that would become due but for the operation of the automatic stay under Section 362(a) of the Bankruptcy Code), the Guarantors will, upon demand pay, or cause to be paid, in cash, to the Administrative Agent for the ratable benefit of Secured Parties, an amount equal to the sum of the unpaid principal amount of all Guaranteed Obligations then due as aforesaid, accrued and unpaid interest on such Guaranteed Obligations (including interest which, but for any Borrower's becoming the subject of a case under the Bankruptcy Code, would have accrued on such Guaranteed Obligations, whether or not a claim is

allowed against any Borrower for such interest in the related bankruptcy case) and all other Guaranteed Obligations then owed to the Secured Parties as aforesaid.

Notwithstanding any provision to the contrary contained herein or in any other of the Loan Documents, the Guaranteed Obligations of each Guarantor under this Agreement and the other Loan Documents shall be limited to an aggregate amount equal to the largest amount that would not render such obligations subject to avoidance under the Debtor Relief Laws.

**10.02 Obligations Unconditional** The Guaranteed Obligations of each Guarantor under<u>Section 10.01</u> are joint and several and absolute and unconditional, irrespective of the value, genuineness, validity, regularity or enforceability of any of the Loan Documents or any other agreement or instrument referred to therein, or any substitution, release, impairment or exchange of any other guarantee of or security for any of the Obligations, and, to the fullest extent permitted by applicable Law, irrespective of any other circumstance whatsoever which might otherwise constitute a legal or equitable discharge or defense of a surety or guarantor, it being the intent of this <u>Section 10.02</u> that the obligations of each Guarantor hereunder shall be absolute and unconditional under any and all circumstances. Each Guarantor agrees that such Guarantor shall have no right of subrogation, indemnity, reimbursement or contribution against the Borrowers or any other Guarantor for amounts paid under this <u>Article 10</u> until the Termination Date. Without limiting the generality of the foregoing, it is agreed that, to the fullest extent permitted by law, the occurrence of any one or more of the following shall not alter or impair the liability of any Guarantor hereunder, which shall remain joint and several and absolute and unconditional as described above:

(a) at any time or from time to time, without notice to any Guarantor, the time for any performance of or compliance with any of the Obligations shall be extended, or such performance or compliance shall be waived;

(b) any of the acts mentioned in any of the provisions of any of the Loan Documents or any other agreement or instrument referred to in the Loan Documents shall be done or omitted;

(c) the maturity of any of the Obligations shall be accelerated, or any of the Obligations shall be modified, supplemented or amended in any respect, or any right under any of the Loan Documents or any other agreement or instrument referred to in the Loan Documents shall be waived or any other guarantee of any of the Obligations or any security therefor shall be released, impaired or exchanged in whole or in part or otherwise dealt with;

(d) any Lien granted to, or in favor of, the Administrative Agent or any Secured Party or Secured Parties as security for any of the Obligations shall fail to attach or be perfected;

(e) any of the Obligations shall be determined to be void or voidable (including for the benefit of any creditor of any Guarantor) or shall be subordinated to the claims of any Person (including any creditor of any Guarantor); or

(f) any other action or inaction shall occur that might constitute a surety defense.

**10.03** <u>Reinstatement</u>. The Guaranteed Obligations of any Guarantor under this Article 10 shall be automatically reinstated if and to the extent that for any reason any payment by or on behalf of any Person in respect of the Obligations is rescinded or must be otherwise restored by any holder of any of the Obligations, whether as a result of any proceedings in bankruptcy or reorganization or otherwise, and each Guarantor agrees that it will indemnify the Administrative Agent and each Secured Party on demand for all reasonable costs and expenses (including fees and expenses of counsel) incurred by the Administrative

Agent or such Secured Party in connection with such rescission or restoration, including any such costs and expenses incurred in defending against any claim alleging that such payment constituted a preference, fraudulent transfer or similar payment under any bankruptcy, insolvency or similar law. This paragraph shall survive any termination of this Agreement.

#### 10.04 <u>Waivers</u>.

Each Guarantor hereby waives, to the fullest extent permitted by Law, for the benefit of the Administrative Agent and (a) each Secured Party; (a) any right to require the Administrative Agent or any Secured Party, as a condition of payment or performance by such Guarantor, to (i) proceed against any Borrower, any other guarantor (including any other Guarantor) of the Guaranteed Obligations or any other Person, (ii) proceed against or exhaust any security held from any Borrower, any such other guarantor or any other Person, (iii) proceed against or have resort to any balance of any deposit account or credit on the books of the Administrative Agent and Secured Parties in favor of any Borrower or any other Person, or (iv) pursue any other remedy in the power of the Administrative Agent and the Secured Parties whatsoever; (b) any defense arising by reason of the incapacity, lack of authority or any disability or other defense of any Borrower or any other Guarantor including any defense based on or arising out of the lack of validity or the unenforceability of the Guaranteed Obligations or any agreement or instrument relating thereto or by reason of the cessation of the liability of any Borrower or any other Guarantor from any cause other than payment in full of the Guaranteed Obligations; (c) any defense based upon any statute or rule of law which provides that the obligation of a surety must be neither larger in amount nor in other respects more burdensome than that of the principal or any law, rule, regulation, or order of any jurisdiction affecting any term of the Guaranteed Obligations; (d) any defense based upon the Administrative Agent's or any Secured Party's errors or omissions in the administration of the Guaranteed Obligations, except behavior which amounts to bad faith; (e) (i) any principles or provisions of law, statutory or otherwise, which are or might be in conflict with the terms hereof and any legal or equitable discharge of such Guarantor's obligations hereunder, (ii) the benefit of any statute of limitations affecting such Guarantor's liability hereunder or the enforcement hereof, (iii) any rights to set offs, recoupments and counterclaims, and (iv) promptness, diligence and any requirement that the Administrative Agent and the Secured Parties protect, secure, perfect or insure any security interest or lien or any property subject thereto; (f) notices, demands, presentments, protests, notices of protest, notices of dishonor and notices of any action or inaction, including acceptance hereof, notices of default under any Loan Document, any Secured Hedge Agreement or any agreement or instrument related thereto, notices of any renewal, extension or modification of the Guaranteed Obligations or any agreement related thereto, notices of any extension of credit to any Borrower and notices of any of the matters referred to in Section 10.02 and any right to consent to any thereof; and (g) any defenses or benefits that may be derived from or afforded by law which limit the liability of or exonerate guarantors or sureties, or which may conflict with the terms hereof. Each Guarantor agrees that such Guarantor shall have no right of recourse to security for the Obligations, except through the exercise of rights of subrogation to the extent permitted by Section 10.02; provided, however, Guarantor agrees that such rights shall be automatically (and without any further action) irrevocably waived and released if such security is acquired by a Person as a result of the exercise of the remedies under the Loan Documents.

**10.05** <u>Remedies</u>. Each Guarantor agrees that, to the fullest extent permitted by Law, as between such Guarantor, on the one hand, and the Administrative Agent and the Secured Parties, on the other hand, the Obligations may be declared to be forthwith due and payable as provided in <u>Section 9.02</u> (and shall be deemed to have become automatically due and payable in the circumstances provided in <u>Section 9.02</u>) for purposes of <u>Section 10.01</u> notwithstanding any stay, injunction or other prohibition preventing such declaration (or preventing the Obligations from becoming automatically due and payable) as against any other Person and that, in the event of such declaration (or the Obligations being deemed to have become automatically due and payable), the Obligations (whether or not due and payable by any other Person) shall

forthwith become due and payable by each Guarantor for purposes of <u>Section 10.01</u>. Each Guarantor acknowledges and agrees that its Guaranteed Obligations hereunder are secured in accordance with the terms of the Collateral Documents and that the Secured Parties may exercise their remedies thereunder in accordance with the terms thereof.

10.06 Contribution by Guarantors. All Guarantors desire to allocate among themselves (collectively, the "Contributing Guarantors"), in a fair and equitable manner, their respective obligations arising under the Guaranty. Accordingly, in the event any payment or distribution is made on any date by a Guarantor (a "Funding Guarantor") under the Guaranty such that its Aggregate Payments exceeds its Fair Share as of such date, such Funding Guarantor shall be entitled to a contribution from each of the other Contributing Guarantors in an amount sufficient to cause each Contributing Guarantor's Aggregate Payments to equal its Fair Share as of such date. Fair Share' means, with respect to a Contributing Guarantor as of any date of determination, an amount equal to (a) the ratio of (i) the Fair Share Contribution Amount with respect to such Contributing Guarantor to (ii) the aggregate of the Fair Share Contribution Amounts with respect to all Contributing Guarantors multiplied by (b) the aggregate amount paid or distributed on or before such date by all Funding Guarantors under the Guaranty in respect of the Guaranteed Obligations. 'Fair Share Contribution Amount' means, with respect to a Contributing Guarantor as of any date of determination, the maximum aggregate amount of the obligations of such Contributing Guarantor under the Guaranty that would not render its obligations hereunder or thereunder subject to avoidance as a fraudulent transfer or conveyance under Section 548 of the Bankruptcy Code or any comparable applicable provisions of state law, provided, solely for purposes of calculating the "Fair Share Contribution Amount" with respect to any Contributing Guarantor for purposes of this Section 10.06, any assets or liabilities of such Contributing Guarantor arising by virtue of any rights to subrogation, reimbursement or indemnification or any rights to or obligations of contribution hereunder shall not be considered as assets or liabilities of such Contributing Guarantor. "Aggregate Payments" means, with respect to a Contributing Guarantor as of any date of determination, an amount equal to (1) the aggregate amount of all payments and distributions made on or before such date by such Contributing Guarantor in respect of the Guaranty (including in respect of this Section 10.06), minus (2) the aggregate amount of all payments received on or before such date by such Contributing Guarantor from the other Contributing Guarantors as contributions under this Section 10.06. The amounts payable as contributions hereunder shall be determined as of the date on which the related payment or distribution is made by the applicable Funding Guarantor. The allocation among Contributing Guarantors of their obligations as set forth in this Section 10.06 shall not be construed in any way to limit the liability of any Contributing Guarantor hereunder. Each Guarantor is a third party beneficiary to the contribution agreement set forth in this Section 10.06 and a right to receive any Fair Share Contribution Amount shall be deemed an asset of the Guarantor entitled to such amount.

**10.07** <u>Guarantee of Payment; Continuing Guarantee</u> The guarantee in this Article 10 is an absolute and unconditional guaranty of payment and not of collection, is a continuing and irrevocable guarantee, and shall apply to all Obligations whenever arising.

**10.08** <u>Subordination of Other Obligations</u> Any Indebtedness of any Borrower or any Guarantor or any Subsidiary now or hereafter owing to any Loan Party (the '<u>Obligee</u>'') shall be subordinated in right of payment to the Obligations (provided, such payments shall be permitted so long as no Event of Default has occurred and is continuing) and any Lien now or hereafter securing such Indebtedness is hereby subordinated in priority to the Liens of Administrative Agent now or hereafter securing any of the Obligations, and any such Indebtedness collected or received by the Obligee after an Event of Default has occurred and is continuing shall be held in trust for the Administrative Agent for its benefit and the benefit of the Secured Parties and shall forthwith be paid over to Administrative Agent for its benefit and the benefit of the Secured Parties to be credited and applied against the Obligations but without affecting, impairing or limiting in any manner the liability of the Obligee under any other provision

#### **ARTICLE 11**

# THE ADMINISTRATIVE AGENT

#### 11.01 Appointment and Authorization of Administrative Agent.

(a) Each Secured Party hereby irrevocably appoints, designates and authorizes the Administrative Agent to take such action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise such powers and perform such duties as are expressly delegated to it by the terms of this Agreement or any other Loan Document, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere herein or in any other Loan Document, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, nor shall the Administrative Agent have or be deemed to have any fiduciary relationship with any Secured Party or participant, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against the Administrative Agent. Without limiting the generality of the foregoing sentence, the use of the term "agent" herein and in the other Loan Documents with reference to the Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable Law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.

(b) Each L/C Issuer shall act on behalf of the Lenders with respect to any Letters of Credit issued by it and the documents associated therewith, and each Support Provider shall act on behalf of the Lenders with respect to any Support Agreements entered into by it and the documents associated therewith and each L/C Issuer and Support Provider, respectively, shall have all of the benefits and immunities (i) provided to the Administrative Agent in this <u>Article 11</u> with respect to any acts taken or omissions suffered by such L/C Issuer or Support Provider in connection with Letters of Credit and Support Agreements issued by it or proposed to be issued by it and the applications and agreements for letters of credit pertaining to such Letters of Credit and Support Agreements as fully as if the term "Administrative Agent" as used in this <u>Article 11</u> and in the definition of "Agent-Related Person" included such L/C Issuer or Support Provider with respect to such acts or omissions, and (ii) as additionally provided herein with respect to the L/C Issuers or Support Providers.

(c) Swingline Lender shall act on behalf of the Lenders with respect to any Swingline Loan made by it and the documents associated therewith, and the Swingline Lender shall have all of the benefits and immunities (i) provided to the Administrative Agent in this <u>Article 11</u> with respect to any acts taken or omissions suffered by the Swingline Lender in connection with the Swingline Loans made by it or proposed to be made by it as fully as if the term "Administrative Agent" as used in this <u>Article 11</u> and in the definition of "Agent-Related Person" included the Swingline Lender with respect to such acts or omissions, and (ii) as additionally provided herein with respect to the Swingline Lender.

**11.02** Delegation of Duties. The Administrative Agent may execute any of its duties under this Agreement or any other Loan Document by or through its, or its Affiliates', agents, employees or attorneys-in-fact and shall be entitled to obtain and rely upon the advice of counsel and other consultants or experts concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects in the absence of the Administrative Agent's gross negligence or willful misconduct (as finally determined in a non-appealable decision of a court of competent jurisdiction).

11.03 Liability of Administrative Agent. No Agent-Related Person shall be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Loan Document or the transactions contemplated hereby, and each Loan Party and Secured Party hereby waives and agrees not to assert any right, claim or cause of action based thereon, except to the extent of liabilities resulting primarily from its own gross negligence or willful misconduct in connection with its duties expressly set forth herein, as finally determined in a non-appealable decision of a court of competent jurisdiction. Without limiting the foregoing, no Agent-Related Person shall be: (i) responsible or have any obligation to any other Secured Party for the due execution, validity, genuineness, effectiveness, sufficiency, or enforceability of, or for any recital, statement, warranty or representation in, this Agreement, any other Loan Document or any related agreement, document or order; (ii) required or have any obligation to ascertain, monitor or enforce or to make any inquiry concerning the performance or observance by any Loan Party and any Lender of any of the terms, conditions, covenants, or agreements of this Agreement or any of the other Loan Documents (including, without limitation, any restrictions, limitations or qualifications related to any Disqualified Institution); (iii) responsible for or have any duty to ascertain or monitor or to inquire into whether a condition set forth in any Loan Document is satisfied, or waived, including any condition set forth in Article 4 hereof; (iv) responsible or have any obligation to any other Secured Party for the state or condition of any properties of the Loan Parties constituting Collateral for the Obligations or any information contained in the books or records of the Loan Parties; (v) responsible or have any obligation to any other Secured Party for the validity, enforceability, collectability, effectiveness or genuineness of this Agreement or any other Loan Document or any other certificate, document or instrument furnished in connection therewith; (vi) liable with respect to or arising out of any assignment or participation of the Obligations, or disclosure of any information, to any Secured Party or such Secured Party's representatives, Approved Funds or Affiliates; or (vii) responsible or have any obligation to any other Secured Party to assure that the Collateral exists or is owned by the Loan Parties or is cared for, protected, or insured or has been encumbered, or that the Liens granted to the Administrative Agent therein have been properly or sufficiently or lawfully created, perfected (or continue to be perfected), protected, or enforced or are entitled to any particular priority, or to exercise at all or in any particular manner or under any duty of care, disclosure, or fidelity, or to continue exercising, any of the rights, authorities, and powers granted or available to the Administrative Agent pursuant to any of the Loan Documents. In addition to and not in limitation of the foregoing, it is understood and agreed that in respect of the Collateral, or any act, omission, or event related thereto, the Administrative Agent may act in any manner it may deem appropriate, in its sole discretion, given the Administrative Agent's own interest in the Collateral in its capacity as one of the Secured Parties and that the Administrative Agent shall have no other duty or liability whatsoever to any Secured Party as to any of the foregoing, including, without limitation, the preparation, form or filing of any Uniform Commercial Code financing statement, amendment or continuation or of any other type of document related to the creation, perfection, continuation or priority of any Lien as to any property of the Loan Parties.

# 11.04 Reliance by Administrative Agent.

(a) The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, communication, signature, resolution, representation, notice, consent, certificate, affidavit, letter, telegram, facsimile, telex or telephone message, electronic mail message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to any Loan Party), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent shall be fully justified in failing or refusing to take any action under any Loan Document unless it shall first receive such advice or concurrence of the Required Lenders as it deems appropriate and, if it so requests, it shall first be indemnified to its satisfaction by the Secured Parties against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from

acting, under this Agreement or any other Loan Document in accordance with a request or consent of the Required Lenders (or such greater number of Lenders as may be expressly required hereby in any instance) and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Secured Parties. Notwithstanding the foregoing, the Administrative Agent shall not be required to take, or to omit to take, any action that is, in the opinion of the Administrative Agent or its counsel, contrary to any Loan Document or applicable Law.

(b) For purposes of determining compliance with the conditions specified in <u>Article 4</u>, each Lender that has signed this Agreement (or an addendum or joinder to this Agreement) shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required hereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

**11.05** Notice of Default The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default and/or Event of Default, unless the Administrative Agent shall have received written notice from a Lender or any Borrower referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default." The Administrative Agent will notify the Lenders of its receipt of any such notice. The Administrative Agent shall take such action with respect to such Default and/or Event of Default as may be directed by the Required Lenders in accordance with<u>Article 9</u>, provided, that unless and until the Administrative Agent has received any such direction in accordance with <u>Section 11.04</u>, the Administrative Agent may (but shall not be obligated to) take any action, or refrain from taking any action, with respect to such Default and/or Event of Default as it shall deem advisable or in the best interest of the Lenders.

11.06 Credit Decision; Disclosure of Information by Administrative Agent Each Lender acknowledges that no Agent-Related Person has made any representation or warranty to it, and that no act by any Agent-Related Person hereafter taken, including any consent to and acceptance of any assignment or review of the affairs of any Loan Party or any Affiliate thereof, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Lender as to any matter, including whether Agent-Related Persons have disclosed material information in their possession. Each Lender represents to the Administrative Agent that it has, independently and without reliance upon any Agent-Related Person and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of the Loan Parties and their respective Subsidiaries, and all applicable bank or other regulatory Laws relating to the transactions contemplated hereby, and made its own decision to enter into this Agreement and to extend credit to the Borrowers and the other Loan Parties hereunder. Each Lender also represents that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of the Borrowers and the other Loan Parties. Except for notices, reports and other documents expressly required to be furnished to the Lenders by the Administrative Agent herein, no Agent-Related Person shall have any duty or responsibility to provide any Lender with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of any of the Loan Parties or any of their respective Affiliates which may come into the possession of any Agent-Related Person.

11.07 <u>Indemnification of Administrative Agent</u>. The Lenders shall indemnify upon demand each Agent-Related Person (to the extent not reimbursed by or on behalf of any Loan Party and without limiting the obligation of any Loan Party to do so), severally in proportion to its Pro Rata Share, and hold

harmless each Agent-Related Person from and against any and all Indemnified Liabilities incurred by it;<u>provided</u>, <u>however</u>, that no Lender shall be liable for the payment to any Agent-Related Person of any portion of such Indemnified Liabilities to the extent determined in a final, non-appealable judgment by a court of competent jurisdiction to have resulted primarily from such Agent-Related Person's own gross negligence or willful misconduct; <u>provided</u>, <u>however</u>, that no action taken in furtherance of the directions of the Required Lenders shall be deemed to constitute gross negligence or willful misconduct for purposes of this <u>Section 11.07</u>. Without limitation of the foregoing, each Lender shall reimburse each Agent-Related Person upon demand for its ratable share of any costs or out-of-pocket expenses (including Attorney Costs) incurred by any Agent-Related Person in connection with the preparation, execution, delivery, administration, modification, amendment or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement, any other Loan Document, or any document contemplated by or referred to herein. The obligations of the Lenders hereunder shall not diminish the obligations of the Borrowers to indemnify and reimburse the Agent-Related Persons for such amounts. The Administrative Agent may in its discretion first seek payment from the Lenders hereunder before seeking payment from the Borrowers for such amounts or may seek payments first from the Borrowers. In any event, any amounts received from Borrowers as reimbursement for amounts already reimbursed by Lenders shall be paid to Lenders in accordance with the terms hereof. The undertaking in this<u>Section 11.07</u> shall survive the Termination Date and the resignation of the Administrative Agent.

**11.08** Administrative Agent in its Individual Capacity. Administrative Agent and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with each of the Loan Parties and their respective Affiliates as though Administrative Agent were not the Administrative Agent, Support Provider or an L/C Issuer hereunder and without notice to or consent of any Secured Party. The Secured Parties acknowledge that, pursuant to such activities, Administrative Agent or its Affiliates may receive information regarding any Loan Party or its Affiliates (including information that may be subject to confidentiality obligations in favor of such Loan Party or such Affiliate) and acknowledge that the Administrative Agent shall be under no obligation to provide such information to them. With respect to its Loans, Administrative Agent and/or its Affiliates (as applicable) shall have the same rights and powers under this Agreement as any other Secured Party and may exercise such rights and powers as though it were not the Administrative Agent, and the terms "Lender" and "Lenders" include the Administrative Agent and/or its Affiliates (as applicable) in its individual capacity.

**11.09** Successor Administrative Agent. The Administrative Agent may resign as the Administrative Agent upon ten (10) days' notice to the Lenders and the Borrower Representative, provided that any such resignation by the Administrative Agent may also constitute its resignation as the L/C Issuer, Support Provider and Swingline Lender (if applicable). If the Administrative Agent resigns under this Agreement, the Required Lenders shall appoint from among the Lenders (or the Affiliates thereof) a successor administrative agent for the Lenders, which successor administrative agent shall (unless an Event of Default has occurred and is continuing) be subject to the approval of the Borrower Representative (which approval shall not be unreasonably withheld or delayed). If no successor administrative agent is appointed prior to the effective date of the resignation of the Administrative Agent, the Administrative Agent may appoint, after consulting with the Lenders, a successor administrative agent from among the Lenders (or the Affiliates thereof). Upon the acceptance of its appointment as successor administrative agent hereunder, the Person acting as such successor administrative agent shall succeed to all the rights, powers and duties of the retiring Administrative Agent (and Support Provider, L/C Issuer and Swingline Lender (if the retiring Administrative Agent desires to resign therefrom)) and the respective terms "Administrative Agent", "Support Provider, "L/C Issuer" and "Swingline Lender" shall mean such successor administrative agent, Letter of Credit issuer, support Provider, and swingline lender, and the retiring Administrative Agent's appointment, powers and duties in such capacities shall be terminated without any other further act or deed

on its behalf. After any retiring Administrative Agent's resignation hereunder as the Administrative Agent, the provisions of this <u>Article 11</u> and <u>Sections 12.04</u> and <u>12.05</u> shall inure to its benefit as to any actions taken or omitted to be taken by it while it was the Administrative Agent under this Agreement or while it was actively engaged in transferring its rights and obligations as Administrative Agent to the successor administrative agent. If no successor administrative agent has accepted appointment as the Administrative Agent by the date ten (10) days following a retiring Administrative Agent's notice of resignation, the retiring Administrative Agent is resignation shall nevertheless thereupon become effective and the Required Lenders shall perform all of the duties of the Administrative Agent hereunder until such time, if any, as the Required Lenders appoint a successor administrative agent as provided for above. No Person may be removed from its capacity as Administrative Agent.

**11.10** <u>Administrative Agent May File Proofs of Claim</u> In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial proceeding relative to any Loan Party, the Administrative Agent (irrespective of whether the principal of any Loan, Letter of Credit Liabilities or Swingline Exposure shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Loan Parties) shall be entitled and empowered, by intervention in such proceeding or otherwise:

(a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, Letter of Credit Liabilities, Swingline Exposures and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Secured Parties and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Secured Parties and the Administrative Agent and their respective agents and counsel and all other amounts due the Secured Parties and the Administrative Agent under <u>Sections 2.09</u> and <u>12.04</u>) allowed in such judicial proceeding; and

same;

(b)

to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Secured Party to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Secured Parties, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.09 and 12.04.

11.11 <u>Collateral and Guaranty Matters</u>. The Secured Parties irrevocably authorize the Administrative Agent, at its option and in its discretion, to release any Guarantor and any Lien on any Collateral granted to or held by the Administrative Agent under any Loan Document (i) upon the Termination Date, (ii) that is transferred or to be transferred as part of or in connection with any Disposition permitted hereunder or under any other Loan Document, or (iii) as approved in accordance with <u>Section 12.01</u>, and to execute in connection with such events such payoff letters and related documentation in form and substance satisfactory to Administrative Agent, in its sole discretion, as shall in Administrative Agent's sole discretion be deemed advisable. In connection with any such release, each Lender, the L/C Issuer and the Support Providers hereby direct the Administrative Agent, and the Administrative Agent agrees that it shall, upon the reasonable request of the Borrower Representative (and except in the case where the Termination Date has actually occurred, so long as no Default or Event of Default then exists), to (i) promptly execute and deliver or file such documents and perform other actions reasonably requested to release the guaranties and the Liens and (ii) deliver to the Loan Parties any portion of such Collateral so released in the possession of the Administrative Agent or as otherwise required under any Loan Documents,

Subordination Agreement, Acceptable Intercreditor Agreement, or applicable Law, in each case without recourse, representation or warranty. Upon request by the Administrative Agent at any time, the Required Lenders will confirm in writing the Administrative Agent's authority to release or subordinate its interest in particular types or items of Property, pursuant to this <u>Section 11.11</u>.

The Secured Parties hereby irrevocably authorize Administrative Agent (absent, with respect to any particular transaction, Administrative Agent receiving contrary written bidding instructions from the Required Lenders before such transaction), to credit bid all or any portion of the Obligations (including, to credit bid all or any portion of the Obligations (including accepting some or all of the Collateral in satisfaction of some or all of the Obligations pursuant to a deed in lieu of foreclosure or otherwise)) and in such manner purchase (either directly or through one or more acquisition vehicles) all or any portion of the Collateral (a) at any sale thereof conducted under the provisions of the Bankruptcy Code of the United States, including under Section 363, 1123 or 1129 of the Bankruptcy Code of the United States, or any similar Laws in any other jurisdictions to which a Loan Party is subject, (b) at any other sale or foreclosure or acceptance of collateral in lieu of debt conducted by (or with the consent or at the direction of) Administrative Agent (whether by judicial action or otherwise) in accordance with any applicable Law. In connection with any such credit bid and purchase, the Obligations owed to the Secured Parties shall be entitled to be, and shall be, credit bid on a ratable basis (with Obligations with respect to contingent or unliquidated claims receiving contingent interests in the acquired assets on a ratable basis that would vest upon the liquidation of such claims in an amount proportional to the liquidated portion of the contingent claim amount used in allocating the contingent interests) in the asset or assets so purchased (or in the Capital Stock or debt instruments of the acquisition vehicle or vehicles that are used to consummate such purchase). In connection with any such bid Administrative Agent shall be authorized (i) to form one or more acquisition vehicles to make a bid, (ii) to adopt documents providing for the governance of the acquisition vehicle or vehicles (provided that any actions by Administrative Agent with respect to such acquisition vehicle or vehicles, including any disposition of the assets or Capital Stock thereof shall be governed, directly or indirectly, by the vote of the Required Lenders, irrespective of the termination of this Agreement and without giving effect to the limitations on actions by the Required Lenders contained in clauses (a) through (h) of Section 12.01 of this Agreement (provided that, in any event, the consent of each Lender shall be required for any amendment that would treat or attempts to treat a Lender or a class of Lenders in a manner different than all other Lenders)), and (iii) to the extent that Obligations that are assigned to an acquisition vehicle are not used to acquire Collateral for any reason (as a result of another bid being higher or better, because the amount of Obligations assigned to the acquisition vehicle exceeds the amount of debt credit bid by the acquisition vehicle or otherwise), such Obligations shall automatically be reassigned to the Lenders pro rata and the Capital Stock and/or debt instruments issued by any acquisition vehicle on account of the Obligations that had been assigned to the acquisition vehicle shall automatically be cancelled, without the need for any Secured Party or any acquisition vehicle to take any further action.

**11.12** Other Agents; Arrangers and Managers. None of the Lenders or other Persons identified on the facing page or signature pages of this Agreement as a "syndication agent," "documentation agent," "co-agent," "book manager," "lead manager," "arranger," "lead arranger" or "co-arranger" (each, an "<u>Additional Titled Agent</u>") shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than, in the case of such Lenders, those applicable to all Lenders as such. Without limiting the foregoing, no Additional Titled Agent shall have or be deemed to have any fiduciary relationship with any Lender. Each Lender acknowledges that it has not relied, and will not rely, on any Additional Titled Agent in deciding to enter into this Agreement or in taking or not taking action hereunder. At any time that any Lender serving as an Additional Titled Agent shall have transferred to any other Person (other than any Affiliates) all of its interests in the Loans, such Lender shall be deemed to have concurrently resigned as such Additional Titled Agent.

11.13 Additional Secured Parties. The benefit of the provisions of the Loan Documents directly relating to the Collateral or any Lien granted thereunder shall extend to and be available to any Secured Party that is not a Lender, L/C Issuer or Support Provider as long as, by accepting such benefits, such Secured Party agrees, as among the Administrative Agent and all other Secured Parties, that such Secured Party is bound by (and, if requested by the Administrative Agent, shall confirm such agreement in a writing in form and substance acceptable to the Administrative Agent) this Article 11, Section 2.13 (Sharing of Payments), Section 12.08 (Confidentiality) and Section 12.09 (Setoff) as each may be in effect from time to time, and the decisions and actions of the Administrative Agent and the Required Lenders (or, where expressly required by the terms of this Agreement, a greater proportion of the Lenders) to the same extent a Lender is bound provided, however, that, notwithstanding the foregoing, (a) such Secured Party shall be bound by Section 11.07 (Indemnification of Administrative Agent) only to the extent of liabilities, costs and expenses with respect to or otherwise relating to the Collateral held for the benefit of such Secured Party, in which case the obligations of such Secured Party thereunder shall not be limited by any concept of ratability or similar concept, (b) except as set forth specifically herein as to such Secured Party (rather than the Secured Parties generally), each of the Administrative Agent, the Lenders, the L/C Issuers and the Support Providers shall be entitled to act at its sole discretion, without regard to the interest of such Secured Party, regardless of whether any Obligation to such Secured Party thereafter remains outstanding, is deprived of the benefit of the Collateral, becomes unsecured or is otherwise affected or put in jeopardy thereby, and without any duty or liability to such Secured Party or any such Obligation and (c) except as set forth specifically herein as to such Secured Party (rather than the Secured Parties generally), such Secured Party shall not have any right to be notified of, consent to, direct, require or be heard with respect to, any action taken or omitted in respect of the Collateral or under any Loan Document.

11.14 **Exclusive Right to Enforce Rights and Remedies** Notwithstanding anything to the contrary contained herein or in any other Loan Document, the authority to enforce rights and remedies hereunder and under the other Loan Documents against the Loan Parties or any of them or in respect of the Collateral are hereby granted to, and shall be vested exclusively in, and all actions and Proceedings in connection with any such enforcement shall be instituted and maintained exclusively by, Administrative Agent (or its agents or designees) in accordance with the Loan Documents for the benefit of the applicable Secured Parties; provided that the foregoing shall not prohibit (i) the Administrative Agent from exercising on its own behalf the rights and remedies that inure to its benefit (solely in its capacity as Administrative Agent) hereunder and under the other Loan Documents, (ii) each of the L/C Issuer, any Support Provider and the Swingline Lender from exercising the rights and remedies that inure to its benefit (solely in its capacity as L/C Issuer, Support Provider or Swingline Lender, as the case may be) hereunder and under the other Loan Documents, (iii) any Lender or Participant from exercising setoff rights in accordance with Section 12.09, (iv) any Lender from filing proofs of claim or appearing and filing pleadings on its own behalf during the pendency of a Proceeding relative to any Loan Party under any Bankruptcy Code or other Debtor Relief Law or (v) any Lender or other Secured Party from exercising any express right or remedy of such Lender under the Loan Documents where the Administrative Agent does not have the power and authority under the Loan Documents to act on behalf of such Lender or other Secured Party; and provided, further, that if at any time there is no Person acting as the Administrative Agent hereunder and under the other Loan Documents, then (A) the Required Lenders shall have the rights otherwise ascribed to the Administrative Agent pursuant to Section 9.02 and (B) in addition to the matters set forth in Section 11.04 and Section 12.09, any Lender may, with the consent of the Required Lenders, enforce any rights and remedies available to it and as authorized by the Required Lenders.

11.15 <u>Flood Laws</u>. Administrative Agent has adopted internal policies and procedures that address requirements placed on federally regulated lenders under the National Flood Insurance Reform Act of 1994 and related legislation (the <u>Flood Law</u>"). Administrative Agent may post an E-System (or otherwise distribute to each Lender in the syndicate) documents that it receives in connection with the Flood Laws. However, Administrative Agent reminds each Lender and Participant in the facility that,

pursuant to the Flood Laws, each federally regulated Lender (whether acting as a Lender or Participant in the credit facility) is responsible for assuring its own compliance with the flood insurance requirements, and Administrative Agent disclaims any liability in connection with the failure of any such Lender or Participant to comply with flood insurance requirements.

**11.16 Banking Services Obligations/Hedging.** Except as otherwise expressly set forth herein, no Hedge Party or any party that provides Banking Services that obtains the benefit of the provisions of Section 9.04, the Guaranty or any Collateral by virtue of the provisions hereof or any Loan Document shall have any right to notice of any action or to consent to, direct or object to any action hereunder or under any other Loan Document or otherwise in respect of the Collateral (including the release or impairment of any Collateral) (or to notice of or to consent to any amendment, waiver or modification of the provisions hereof or of the Guaranty or any Loan Document) other than in its capacity as a Lender and, in such case, only to the extent expressly provided in the Loan Documents. Notwithstanding any other provision of this Article 11 to the contrary, the Administrative Agent shall not be required to verify the payment of, or that other satisfactory arrangements have been made with respect to, Obligations arising in connection with the Banking Services Obligations and Secured Hedge Agreements except to the extent expressly provided herein and unless the Administrative Agent has received a written designation notice, in form proscribed by the Administrative Agent, of such Obligations, together with such supporting documentation as the Administrative Agent may request, from the applicable Hedge Party or the party providing Banking Services, as the case may be (in each case, other than in the case where Administrative Agent or its Affiliates is the Hedge Party or such provider). Administrative Agent shall not be required to verify the payment of, or that other satisfactory arrangements have been made with respect to, Obligations arising in connection with Banking Services Obligations or Secured Hedge Party or the party providing Banking Services, as the case may be (in each case, other than in the case where Administrative Agent or its Affiliates is the Hedge Party or such provider). Administrative Agent shall n

## 11.17 Certain ERISA Matters.

(a) Each Lender (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, the Administrative Agent and the Arranger and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrowers or any other Loan Party, that at least one of the following is and will be true:

(i) Such Lender is not using "plan assets" (within the meaning of 29 CFR § 2510.3-101, as modified by Section 3(42) of ERISA) of one or more Benefit Plans in connection with the Loans, the Letters of Credit or the Commitments,

(ii) The transaction exemption set forth in one or more PTEs, such as PTE 84-14 (a class exemption for certain transactions determined by independent qualified professional asset managers), PTE 95-60 (a class exemption for certain transactions involving insurance company general accounts), PTE 90-1 (a class exemption for certain transactions involving bank collective investment funds) or PTE 96-23 (a class exemption for certain transactions determined by in-house asset managers), is applicable with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Letters of Credit, the Commitments and this Agreement,

(iii) (A) such Lender is an investment fund managed by a "Qualified Professional Asset Manager" (within the meaning of Part VI of PTE 84-14), (B) such Qualified Professional Asset Manager made the investment decision on behalf of such Lender to enter into, participate in, administer and perform the Loans, the Letters of Credit, the Commitments and this

Agreement, (C) the entrance into, participation in, administration of and performance of the Loans, the Letters of Credit, the Commitments and this Agreement satisfies the requirements of sub-sections (b) through (g) of Part I of PTE 84-14 and (D) to the best knowledge of such Lender, the requirements of subsection (a) of Part I of PTE 84-14 are satisfied with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Letters of Credit, the Commitments and this Agreement, or

(iv) Such other representation, warranty and covenant as may be agreed in writing between the Administrative Agent, in its sole discretion, and such Lender.

(b) In addition, unless sub-clause (i) in the immediately preceding clause (a) is true with respect to a Lender or such Lender has not provided another representation, warranty and covenant as provided in sub-clause (iv) in the immediately preceding clause (a), such Lender further (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto, for the benefit of, the Administrative Agent and the Arranger and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrowers or any other Loan Party, that:

(i) none of the Administrative Agent or the Arranger, or any of their respective Affiliates is a fiduciary with respect to the assets of such Lender (including in connection with the reservation or exercise of any rights by the Administrative Agent under this Agreement, any Loan Document or any documents related hereto or thereto),

(ii) the Person making the investment decision on behalf of such Lender with respect to the entrance into, participation in, administration of and performance of the Loans, the Letters of Credit, the Commitments and this Agreement is independent (within the meaning of 29 CFR § 2510.3-21) and is a bank, an insurance carrier, an investment adviser, a broker-dealer or other person that holds, or has under management or control, total assets of at least \$50,000,000, in each case as described in 29 CFR § 2510.3-21(c)(1)(i)(A)-(E).

(iii) the Person making the investment decision on behalf of such Lender with respect to the entrance into, participation in, administration of and performance of the Loans, the Letters of Credit, the Commitments and this Agreement is capable of evaluating investment risks independently, both in general and with regard to particular transactions and investment strategies (including in respect of the Obligations),

(iv) the Person making the investment decision on behalf of such Lender with respect to the entrance into, participation in, administration of and performance of the Loans, the Letters of Credit, the Commitments and this Agreement is a fiduciary under ERISA or the Internal Revenue Code, or both, with respect to the Loans, the Letters of Credit, the Commitments and this Agreement and is responsible for exercising independent judgment in evaluating the transactions hereunder, and

(v) no fee or other compensation is being paid directly to the Administrative Agent or the Arranger or any of their respective Affiliates for investment advice (as opposed to other services) in connection with the Loans, the Letters of Credit, the Commitments or this Agreement.

(c) The Administrative Agent and the Arranger hereby informs the Lenders that each such Person is not undertaking to provide impartial investment advice, or to give advice in a fiduciary capacity, in connection with the transactions contemplated hereby, and that such Person has a financial

interest in the transactions contemplated hereby in that such Person or an Affiliate thereof (i) may receive interest or other payments with respect to the Loans, the Letters of Credit, the Commitments and this Agreement, (ii) may recognize a gain if it extended the Loans, the Letters of Credit or the Commitments for an amount less than the amount being paid for an interest in the Loans, the Letters of Credit or the Commitments by such Lender or (iii) may receive fees or other payments in connection with the transactions contemplated hereby, the Loan Documents or otherwise, including structuring fees, commitment fees, arrangement fees, facility fees, upfront fees, underwriting fees, ticking fees, agency fees, administrative agent or collateral agent fees, utilization fees, minimum usage fees, letter of credit fees, fronting fees, deal-away or alternate transaction fees, amendment fees, processing fees, term out premiums, banker's acceptance fees, breakage or other early termination fees or fees similar to the foregoing.

## ARTICLE 12

## MISCELLANEOUS

**12.01** <u>Amendments, Etc.</u> Except as provided in <u>Sections 2.15</u>, <u>2.17</u>, <u>2.18</u>, and <u>3.03(b)</u> as in effect on the date hereof, no amendment or waiver of any provision of any Loan Document (other than the Fee Letter, <u>the Landlord Agreements</u>, the Deposit Account Control Agreements, the Issuer Documents, any Secured Hedge Agreement, agreements hereafter executed solely in respect of the Banking Services Obligations and <u>Schedule 6.18</u>), and no consent to any departure by any Borrower or any other Loan Party therefrom, shall be effective unless in writing executed by (1) in the case of any amendment, consent or waiver to cure any ambiguity, omission, defect or inconsistency or granting a new Lien for the benefit of the Secured Parties or extending any existing Lien over additional property or adding additional Subsidiaries of Intermediate Holdings or other pledgors as parties thereto, the Administrative Agent and the applicable Borrower or Loan Party, (2) in the case of any amendment necessary to implement the terms of a Term Loan Increase or a Revolving Commitment Increase in accordance with the terms hereof, the Administrative Agent, the Borrowers and the participating Lenders, and (3) in the case of any other amendment, consent or waiver to Agent with the consent of the Required Lenders) and the applicable Borrower or the applicable Loan Party, as the case may be, and acknowledged by the Administrative Agent, provided, that no such amendment, waiver or consent shall:

(a) extend the expiry or increase the amount of the Commitment of any Lender (or reinstate any Commitment terminated pursuant to <u>Section 9.02</u>) without the written consent of such Lender (it being understood and agreed that a waiver or amendment of any condition precedent set forth in <u>Section</u>

4.02 or of any Default or Event of Default shall not be considered an extension or increase in Commitments of any Lender);

(b) postpone any date fixed by this Agreement or any other Loan Document for any payment of principal (excluding mandatory prepayments), Unreimbursed Amounts, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby;

(c) reduce or forgive the principal of, or the rate of interest specified herein on, any Loan or Unreimbursed Amounts, or any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender directly affected thereby, <u>provided</u>, that only the consent of the Required Lenders shall be necessary to amend the definition of "Default Rate" or to waive any obligation of the Borrowers to pay interest at the Default Rate, and no change to the definition of "Consolidated Total Leverage Ratio" or in the component definitions thereto shall be considered to be a reduction or forgiveness of interest;

(d) change Section 2.13 or Section 9.04 or the definition of "Pro Rata Share" in a manner that would alter the sharing or application or priority of payments required thereby without the written consent of each Lender;

(e) change any provision of this <u>Section 12.01</u> or the definition of "Required Lenders", "Required Revolving Lenders" or "Required Term Loan Lenders" or any other provision of any Loan Document specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights thereunder or make any determination or grant any consent thereunder without the written consent of each Lender;

(f) release all or substantially all of the Collateral, consensually subordinate the Liens of Administrative Agent on the Collateral or consensually subordinate the Obligations to other indebtedness or liabilities (in accordance with this Agreement (as in effect on the date hereof) or in accordance with financing to one or more Loan Parties, pursuant to Section 364 of the Bankruptcy Code or any similar proceeding under any Debtor Relief Law) without the written consent of each Lender;

(g) change any provision that restricts the Sponsor or its Affiliates from holding an interest (whether through an assignment, a participation or otherwise) in the Obligations without the written consent of each Lender; or

(h) release any Borrower; or release all or substantially all of the Guarantors from their obligations under the Loan Documents (or otherwise limit such Guarantors' liability) without the written consent of each Lender directly affected thereby;

and, provided further, that (i) no amendment, waiver or consent shall, unless in writing and executed by the Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Loan Document; (ii) to the extent that an L/C Issuer or Support Provider (as the case may be) is a party to this Agreement, no amendment, waiver or consent shall, unless in writing and executed by the L/C Issuers or Support Providers in addition to the Lenders required above, affect the rights or duties of any L/C Issuers or Support Providers under this Agreement; (iii) no amendment, waiver or consent shall, unless in writing and executed by the Swingline Lender in addition to the Lenders required above, affect the rights or duties of the Swingline Lender under this Agreement; (iv) a Secured Hedge Agreement, agreements hereafter executed solely in respect of the Banking Services Obligations, the Fee Letterthe Landlord Agreements, the Deposit Account Control Agreements and the Issuer Documents may be amended, or rights or privileges thereunder waived, in a writing executed only by the required parties thereto; (v) no amendment, waiver or consent shall, unless in writing signed by Required Term Loan Lenders and Required Revolving Lenders, amend or modify the terms of Section 2.05(b)(vii) in a manner that would alter the application of mandatory prepayments; (vi) no amendment, waiver or consent shall unless in writing and signed by all Revolving Lenders, amend the percentage specified in the definition of Required Revolving Lenders; and (vii) no amendment, waiver or consent shall, unless in writing and signed by all Term Loan Lenders, amend the percentage specified in the definition of Required Term Loan Lenders. Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder, except that (i) the Revolving Commitment of such Lender may not be increased without the consent of such Defaulting Lender and (ii) any waiver, amendment or modification requiring the consent of each affected Lender which affects such Defaulting Lender more adversely than other affected Lenders shall require the consent of such Defaulting Lender.

Each waiver or consent under any Loan Document shall be effective only in the specific instance and for the specific purpose for which it was given.

### 12.02 Notices and Other Communications; Facsimile Copies.

(a) <u>General</u>. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (which includes messages sent by electronic mail or other electronic transmission). All such written notices shall be mailed, faxed or delivered to the applicable address, facsimile number or (subject to subsection (c) below) electronic mail address, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:

(i) if to the Borrower Representative or the Loan Parties, to the Borrower Representative, at the address, facsimile number, electronic mail address or telephone number specified for the Borrower Representative on <u>Schedule 12.02</u> or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the Administrative Agent;

(ii) if to the Administrative Agent, the L/C Issuer, the Support Provider, or Swingline Lender, to the address, facsimile number, electronic mail address or telephone number specified for such Person on <u>Schedule 12.02</u> or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the Borrower Representative and Administrative Agent; and

(iii) if to any other Lender, to the address, facsimile number, electronic mail address or telephone number specified in its Administrative Questionnaire or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the Borrower Representative and the Administrative Agent.

All such notices and other communications shall be deemed to be delivered or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when received by the relevant party hereto; (B) if delivered by mail, four (4) Business Days after deposit in the mails, postage prepaid; (C) if delivered by facsimile, when sent and receipt has been confirmed by telephone; and (D) if delivered by electronic mail (which form of delivery is subject to the provisions of subsection (c) below), when delivered, <u>provided</u>, <u>however</u>, that notices and other communications to the Administrative Agent pursuant to <u>Article 2</u> shall not be effective until actually received by such Person. In no event shall a voicemail message be effective as a notice, communication or confirmation hereunder.

(b) <u>Effectiveness of Facsimile Documents and Signatures</u> Loan Documents may be transmitted and/or executed by facsimile, or by electronic mail. The effectiveness of any such documents and signatures shall, subject to applicable Law, have the same force and effect as manually signed originals and shall be binding on all Loan Parties, the Administrative Agent and the Lenders. The Administrative Agent may also require that any such documents and signatures be confirmed by a manually signed original thereof, <u>provided</u>, that the failure to request or deliver the same shall not limit the effectiveness of any facsimile or other electronic document or signature.

(c) <u>Limited Use of Electronic Mail</u> Electronic mail and internet and intranet websites may be used only to distribute routine communications, such as financial statements and other information as provided in <u>Section 6.01</u> and <u>Section 6.02</u>, and to deliver any other notices and other communications and to distribute Loan Documents for execution by the parties thereto, and may not be used for any other purpose without the consent of the Administrative Agent.

(d) <u>Reliance by Administrative Agent and Lenders</u>. The Administrative Agent and the Lenders shall be entitled to rely and act upon, and shall not incur any liability for relying upon, any notices (including telephonic Loan Notices) purportedly delivered by or on behalf of any Loan Party or the Borrower Representative even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Borrowers shall indemnify each Agent-Related Person and each Lender from all losses, costs, expenses and liabilities resulting from the reliance, in good faith, by such Person on each notice purportedly delivered by or on behalf of any Loan Party or Borrower Representative. All telephonic notices to and other communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

Public Materials Borrowers hereby acknowledges that (a) the Administrative Agent will make available to the ( e ) Lenders materials and/or information provided by or on behalf of Loan Parties hereunder (collectively, 'Borrower Materials") by posting the Borrower Materials on any E-System and (b) certain of the Lenders (each, a **Public Lender**") may have personnel who do not wish to receive material non-public information with respect to Holdings or its Subsidiaries, or the respective securities of any of the foregoing, and who may be engaged in investment and other market-related activities with respect to such Persons' securities. Borrowers hereby agree that so long as a Loan Party is the issuer of any outstanding debt or equity securities that are registered or issued pursuant to a Rule 144A private offering or is actively contemplating issuing any such securities it will, if requested by the Administrative Agent, use commercially reasonable efforts to identify that portion of the Borrower Materials that may be distributed to the Public Lenders and that (w) all such Borrower Materials shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Loan Parties shall be deemed to have authorized the Administrative Agent and the Lenders to treat such Borrower Materials as not containing any material non-public information (although it may be sensitive and proprietary) with respect to any Loan Party or its securities for purposes of United States Federal and state securities laws provided, however, that to the extent such Borrower Materials constitute Information, they shall be treated as set forth inSection 12.08; (y) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of any E-System designated "Public Side Information"; and (z) the Administrative Agent shall treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of any E-System not designated "Public Side Information." Notwithstanding the foregoing, Borrowers shall be under no obligation to mark the Borrower Materials "PUBLIC."

**12.03** <u>No Waiver; Cumulative Remedies</u>. No failure by any Lender or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

**12.04** Attorney Costs, Expenses Any action taken by any Loan Party under or with respect to any Loan Document, even if required under any Loan Document or at the request of any Secured Party, shall be at the expense of such Loan Party, and no Secured Party shall be required under any Loan Document to reimburse any Loan Party therefor except as expressly provided therein. The Borrowers agree (a) to pay or reimburse the Administrative Agent for all reasonable, documented out-of-pocket costs and expenses incurred in connection with the development, preparation, negotiation and execution of this Agreement and the other Loan Documents and any amendment, waiver, consent or other modification of the provisions hereof and thereof (whether or not the transactions contemplated hereby or thereby are consummated), and the consummation, syndication and administration of the transactions contemplated

hereby and thereby, including all reasonable Attorney Costs and costs and expenses in connection with the use of IntraLinks, SyndTrak, StuckyNet, or other similar information transmission systems in connection with this Agreement (each being an 'E-System'), and (b) to pay or reimburse the Administrative Agent, the L/C Issuer and each Support Provider and each Lender for all documented out-of-pocket costs and expenses incurred in connection with the enforcement, attempted enforcement, or preservation of any rights following the occurrence and during the continuance of a Default or an Event of Default or the exercise of remedies under this Agreement or the other Loan Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), including all Attorney Costs. In the case of Attorney Costs otherwise payable or reimbursable under this paragraph (other than pursuant to clause (b) in the immediately preceding sentence, which Attorney Costs will not be so limited), Attorney Costs shall be limited to the reasonable fees, disbursements and other charges of one counsel to the Secured Parties and one separate counsel for the Administrative Agent and, if reasonably necessary, of one special counsel and one local counsel to the Secured Parties and another to the Administrative Agent in each and any relevant jurisdiction, and, solely in the case of any actual or perceived conflict of interest, one additional counsel in each relevant jurisdiction for each Secured Party similarly situated. The foregoing costs and expenses shall include all due diligence, search, filing, recording, title insurance and appraisal charges and fees and taxes related thereto, and other out-of-pocket expenses (including travel, courier, reproduction, printing and delivery expenses) incurred by the applicable Persons and the cost of independent public accountants, consultants and other outside experts retained by the Administrative Agent or any Lender. All amounts due under this Section 12.04 shall be deemed part of the Obligations when incurred and shall be payable within ten (10) Business Days after demand therefor. The agreements in this Section 12.04 shall survive the Termination Date.

12.05 Indemnification by the Loan Parties. The Loan Parties agree jointly and severally to indemnify and hold harmless each Agent-Related Person, each Lender, the L/C Issuer, Support Providers, each Secured Party and the respective Affiliates of all such Persons, and the directors, officers, employees, counsel, trustees, advisors, agents, financing sources, managed funds, controlling persons, attorneys-infact, and members of all of the foregoing Persons (collectively, the "Indemnitees") from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including Attorney Costs incurred by each Indemnitee and other costs of investigation or defense, including those incurred upon any appeal), but in the case of Attorney Costs, limited to the reasonable fees, disbursements and other charges of one primary counsel to the Indemnitees and, if necessary, of special counsel and one local counsel to the Indemnitees in each and any relevant jurisdiction and in the case of any actual or perceived conflict of interest, one additional counsel to each relevant jurisdiction for the group of Indemnitees similarly situated, of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Indemnitee in any way relating to or arising out of or in connection with (a) the execution, delivery, enforcement, performance, syndication or administration of any Loan Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby, (b) any Commitment, Loan or Letter of Credit or Support Agreement or the use or proposed use of the proceeds therefrom (including any refusal by any L/C Issuer to honor a demand for payment under a Letter of Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Letter of Credit), (c) any actual or alleged presence or release of Hazardous Materials on or from any property currently or formerly owned or operated by any Borrower, any Subsidiary or any other Loan Party, or any Environmental Liability related in any way to any Borrower, any Subsidiary or any other Loan Party, or (d) any actual or prospective claim, litigation, investigation or Proceeding relating to any of the foregoing, whether based on contract, tort or any other theory (including any investigation of, preparation for, or defense of any pending or threatened claim, investigation, litigation or proceeding), whether brought by a third party or by any Loan Party, and regardless of whether any Indemnitee is a party thereto (all the foregoing, collectively, the "Indemnified Liabilities"), provided, that

such indemnity shall not, as to any Indemnitee, be available to the extent that such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses or disbursements are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence, bad faith or willful misconduct of such Indemnitee's officers, directors, employees or agents). No Indemnitee shall be liable for any damages arising from the use by others of any information or other materials obtained through the internet or any E-System in connection with this Agreement, nor shall any Indemnitee have any liability for any punitive, special, indirect or consequential damages relating to this Agreement or any other Loan Document or arising out of its activities in connection herewith or therewith (whether arising or occurring before or after the Closing Date). All amounts due under this Section 12.05 shall be payable within ten (10) Business Days after demand therefor. The agreements in this Section 12.05 shall survive the resignation of the Administrative Agent, the replacement of any Lender and the Termination Date. To the extent that the indemnification set forth in this Section 12.05 may be unenforceable, each Loan Party shall contribute the maximum portion which it is permitted to pay and satisfy under applicable Law to the payment and satisfaction of all such Indemnified Liabilities incurred by the Indemnitees or any of them. Without limiting the generality of any provision of this Section 12.05, to the fullest extent permitted by law, each Loan Party hereby waives all rights for contribution or any other rights of recovery with respect to liabilities, losses, damages, costs and expenses arising under or relating to Environmental Laws that it might have by statute or otherwise against any Indemnitee, except to the extent that such items are determined by a final and non-appealable decision of a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of such Indemnitee. No Loan Party shall, without the prior written consent of each applicable Indemnitee, effect any settlement of any pending or threatened proceedings in respect of which indemnity could have been sought hereunder by such Indemnitee unless such settlement (i) includes an unconditional release of such Indemnitee in form and substance reasonably satisfactory to such Indemnitee from all liability or claims that are the subject matter of such proceedings and (ii) does not include any statement as to or any admission of fault, culpability, wrong doing or a failure to act by or on behalf of any Indemnitee. This Section 12.05 shall not apply with respect to Taxes other than any Taxes that represent liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements arising from any non-Tax related Indemnified Liability.

**12.06** Payments Set Aside. To the extent that any payment by or on behalf of any Loan Party is made to the Administrative Agent or any Lender exercises its right of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent or such Lender in its discretion) to be repaid or turned-over to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied (along with any Lien previously terminated with respect to such obligation) shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred, the Termination Date had not occurred and such termination had not occurred, and (b) each Lender severally agrees to pay to the Administrative Agent upon demand its applicable share of any amount so recovered from or repaid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect. The provisions of this paragraph shall survive any termination of this Agreement.

### 12.07 Successors and Assigns.

(a) <u>Successors and Assigns Generally</u>. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, the Indemnitees and to the extent provided in <u>Section 11.13</u> each other Secured Party,

provided that (x) neither any Borrower nor any other Loan Party may assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent and each Lender and (y) no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of <u>subsection (b)</u> of this <u>Section 12.07</u>, (ii) by way of participation in accordance with the provisions of <u>subsection (d)</u> of this <u>Section 12.07</u>, (iii) by way of participation in accordance with the provisions of <u>subsection (d)</u> of this <u>Section 12.07</u> or (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this <u>Section 12.07</u> (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in <u>subsection 12.07</u>, the Indemnitees and to the extent provided in <u>Section 11.13</u> each other Secured Party) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans at the time owing to it), provided that any such assignment shall be subject to the following conditions:

(i) <u>Minimum Amounts</u>.

(A) in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund of a Lender, no minimum amount need be assigned; and

(B) in any case not described in subsection (b)(i)(A) of this Section 12.07, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the applicable Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of such Trade Date) shall not be less than \$1,000,000, in the case of any assignment of a Revolving Commitment, or \$1,000,000, in the case of any assignment of a Term Loan Commitment or Term Loan, unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower Representative, otherwise consents (each such consent not to be unreasonably withheld, conditioned or delayed).

(ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loan or the Commitment assigned, except that this clause (ii) shall not prohibit any Lender from assigning all or a portion of its rights and obligations among separate tranches or facilities on a non-pro rata basis.

(iii) <u>Required Consents</u> No consent shall be required for any assignment except to the extent required by <u>subsection (b)(i)(B)</u> of this <u>Section 12.07</u> and, in addition:

(A) the consent of the Borrower Representative (such consent not to be unreasonably withheld, conditioned or delayed) shall be required unless (x) an Event of Default under <u>clauses (a)</u>, (b) (with respect to <u>Article 8</u> only), (f), (g) or (k) of <u>Section 9.01</u> has occurred and is continuing at the time of such assignment, or (y) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund of a Lender, <u>provided</u> that the

Borrower Representative shall be deemed to have consented to any such assignment if the Administrative Agent has not received an objection thereto in writing within ten (10) Business Days after the Borrower Representative's receipt of notice of such assignment; and

(B) the consent of the Administrative Agent (and also the L/C Issuer or, if applicable, the Support Provider and the Swingline Lender, in the case of any assignment of Revolving Exposure), such consent not to be unreasonably withheld, conditioned or delayed, shall be required for assignments in respect of (i) any Revolving Exposure if such assignment is to a Person that is not a Lender with a Revolving Commitment or an Approved Fund with respect to such assigning Lender, <u>provided</u> that Administrative Agent's (and L/C Issuer's or, if applicable, Support Provider's and the Swingline Lender's) consent, shall be required if the proposed assignee Lender is a Defaulting Lender or (ii) a Term Loan to a Person who is not a Lender, an Affiliate of a Lender or an Approved Fund of a Lender.

(iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee of \$3,500 (unless waived by the Administrative Agent at its sole option), and the assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.

(v) <u>No Assignment to certain Persons</u> No such assignment may be made to the Sponsor, any Loan Party or any Defaulting Lender or any of their respective Affiliates or Subsidiaries reasonably identifiable on basis of such Person's name. No such assignment shall be made to a natural person (or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person).

(vi) <u>No Assignment to Prohibited Assignee</u>. No such assignment shall be made to any Disqualified Institution (so long as the list of Disqualified Institutions is available to the Lenders) unless an Event of Default described in<u>Section 9.01(a)</u>, <u>Section 9.01(f)</u> or <u>Section 9.01(g)</u> has occurred and is continuing at the time of such assignment.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to subsection (c) of this <u>Section 12.07</u>, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of <u>Sections 3.01</u>, <u>3.04</u>, <u>3.05</u>, <u>12.04</u> and <u>12.05</u> with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request of the assignee Lender made itself or through the Administrative Agent, the Borrowers (at their expense) shall execute and deliver a Note to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with subsection (d) of this <u>Section 12.07</u>.

For purposes of determining compliance with <u>Section 12.07(b)(v)</u> above, the Administrative Agent and assigning Lender may rely upon the representations and warranties of the proposed assignee

Lender; it being agreed that neither the Administrative Agent nor any assigning Lender shall have any duty of inquiry to determine such compliance.

Register. The Administrative Agent, acting solely for this purpose as a non-fiduciary agent of the Borrowers (the (c) "Registrar"), shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, the principal amounts of the Loans owing to, and the Letter of Credit Liabilities and Swingline Exposure held by, each Lender pursuant to the terms hereof from time to time (the <u>Register</u>"). No assignment or transfer of a Loan or a Commitment (other than a pledgee described in subsection (f) below) shall be effective unless and until registered in the Register. The entries in the Register shall be conclusive, and the Borrowers, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary, provided that, failure to make any such recordation, or any error in such recordation, shall not affect any Lender's Commitments or any Borrower's or other Loan Party's Obligations in respect of any Loan or Letter of Credit. The Register shall be available for inspection by any Borrower and any Lender (but, with respect to any Lender, solely as to the Loans and Commitments thereof), at any reasonable time and from time to time upon reasonable prior notice. Each Borrower hereby designates the entity serving as the Administrative Agent to serve as such Borrower's non-fiduciary agent solely for purposes of maintaining the Register as provided in this Section, and each Borrower hereby agrees that the entity serving as Registrar and its Affiliates, and their respective officers, directors, employees and agents shall constitute Indemnitees under Section 12.05. At the written request of the registered Lender, the Registrar shall note a collateral assignment of a Loan on the Register as described in subsection (f) below and, provided that the Registrar has received the name and address of such collateral assignee, the Registrar (i) shall not permit any further transfers of the Loan on the Register absent receipt of written consent to such transfers from such collateral assignee and (ii) shall record the transfer of the Loan on the Register to such collateral assignee (or such collateral assignee's designee, nominee or assignee) upon written request by such collateral assignee and compliance with the other provisions of this Agreement governing collateral assignments.

(d) <u>Participations</u>. Any Lender may at any time, without the consent of, or notice to, the Borrower Representative, any Loan Party, the Administrative Agent, the L/C Issuer, the Support Provider or the Swingline Lender, sell participations to any Person (other than a natural person) (or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person), a Defaulting Lender or its Affiliates, or any Loan Party, the Sponsor or any of their respective Affiliates or Subsidiaries) (each, a "<u>Participant</u>") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans owing to it (including such Lender's Letter of Credit Liabilities and Swingline Exposure))provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) each Borrower, each other Loan Party, the Administrative Agent and the Lenders, the L/C Issuer and Swingline Lender shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement, provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver described in the first proviso of <u>Section 12.01</u> (clauses (a) through (g)) that affects such Participant. Subject to <u>subsection (e)</u> of this Section, each Borrower and each other Loan Party agrees that each Participant shall be entitled to the benefits of <u>Sections 3.01</u>, <u>3.04</u> and <u>3.05</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this

Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of <u>Section 12.09</u> as though it were a Lender, <u>provided</u> such Participant agrees to be subject to <u>Section 2.13</u> as though it were a Lender.

Each Lender that sells a participation shall acting solely for this purpose as a non-fiduciary agent of the Borrowers, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under the Loan Documents (the 'Participant Register'); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any participant or any information relating to a Participant's interest in any Commitments, Loans, Letters of Credit or other Obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such Commitment, Loan, Letter of Credit or other Obligation is in registered form under Section 5f 103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) shall have no responsibility for maintaining a Participant Register.

(e) <u>Limitations upon Participant Rights</u> A Participant shall not be entitled to receive any greater payment under <u>Section</u> <u>3.01</u> or <u>3.04</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower Representative's prior written consent and such Participant shall in any event be subject to replacement pursuant to <u>Section 12.15</u> in the event it exercises such rights. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of <u>Section</u>

3.01 unless the Borrower Representative is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Borrowers, to comply with <u>Section 3.01</u> as though it were a Lender.

(f) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank and any pledge to a trustee as security for the benefit of the noteholders and other securityholders or creditors of a Lender, <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledge or assignment.

(g) Lender Securitization. In addition to any other assignment permitted pursuant to this Section, the Loan Parties hereby agree that (i) the Lenders, their Affiliates and Approved Funds (the <u>"Lender Parties</u>") may sell or securitize the Loans (a "<u>Lender Securitization</u>") through the pledge of the Loans as collateral security for loans to a Lender Party or the assignment or issuance of direct or indirect interests in the Loans (such as, for instance, collateralized loan obligations), and (ii) such Lender Securitization may be rated by a Rating Agency, provided that the consent of the Borrower Representative (such consent not to be unreasonably withheld, conditioned or delayed) to the assignment or issuance of direct or indirect interests in the Loans shall be required unless an Event of Default has occurred and is continuing at the time of such assignment or issuance, or such assignment or issuance is to a Lender Party or a Person Controlled by a Lender Party (<u>provided</u>, <u>further</u>, that such consent shall be deemed given if Administrative Agent has not received an objection to any such consent or issuance in writing within five (5) Business Days after the Borrower Representative's receipt of notice of such proposed assignment or issuance). The Loan Parties shall reasonably cooperate with the Lender Parties to effect the Lender Securitization, including providing such information as may be reasonably requested by the Lenders or Rating Agencies in connection with the rating of the Loans or the Lender Securitization shall be reimbursable by

the Loan Parties and (ii) the Loan Parties shall not be required to enter into any amendment or additional documentation that adversely affects the rights or the obligations, of Borrowers under the Loan Documents or changes or affects in a manner adverse to the Borrowers the financial terms of the Loans, in each case, as determined by the Borrower Representative in its sole discretion.

(h) <u>Consent to Disclosure: Cooperation of Loan Parties</u> Subject to the provisions of <u>Section 12.08</u>, each Loan Party authorizes each Lender to disclose to any prospective participant or assignee of a Commitment or Loan, any and all information in such Lender's possession concerning the Loan Parties and their financial affairs which has been delivered to such Lender by or on behalf of the Loan Parties pursuant to this Agreement, or which has been delivered to such Lender by or on behalf of the Loan Parties in connection with such Lender's credit evaluation of the Loan Parties prior to entering into this Agreement. If necessary, each Loan Party agrees to (i) execute any documents (including new Revolving Notes and/or Term Notes) reasonably required to effectuate and acknowledge each assignment of a Commitment or Loan to an assignee in accordance with <u>Section 12.07</u>, (ii) make the Loan Parties' management available to meet with the Administrative Agent and prospective participants and assignees of Commitments or Loans and (iii) assist the Administrative Agent or the Lenders in the preparation of information relating to the financial affairs of the Loan Parties as any prospective participant or assignee of a Commitment or Loan reasonably may request.

(i) <u>Assignments to Affiliated Lenders</u> Notwithstanding anything in this Agreement to the contrary, any Term Loan Lender may, at any time, assign all or a portion of its Term Loans on a non-pro rata basis to an Affiliated Lender through open-market purchases, subject to the following limitations:

(i) in connection with an assignment to an Affiliated Lender, (A) the Affiliated Lender shall have identified itself in writing as an Affiliated Lender to the assigning Term Loan Lender and the Administrative Agent prior to the execution of such assignment and (B) the Affiliated Lender shall be deemed to have represented and warranted to the assigning Term Loan Lender and the Administrative Agent that (x) the requirements set forth in this <u>Section 12.07(i)</u>, shall have been satisfied upon consummation of the applicable assignment; and (y) either (1) such Affiliated Lender is not in possession of material non-public information that has not been disclosed to the assigning Term Loan Lender or (2) such Affiliated Lender cannot make the representation set forth in <u>Section 12.07(i)</u> (i)(y) hereof;

(ii) the assigning Lender and the Affiliated Lender purchasing such Lender's portion of the Term Loan shall execute and deliver to the Administrative Agent an assignment agreement substantially in the form of Exhibit Z hereto (an "Affiliated Lender Assignment and Assumption");

(iii) Affiliated Lenders will not (A) have the right to receive information, reports or other materials provided solely to Lenders by the Administrative Agent or any other Lender, except to the extent made available to the Borrowers, (B) attend or participate in meetings attended solely by the Lenders and the Administrative Agent, or (C) access any electronic site established for the Lenders or confidential communications from counsel to or financial advisors of the Administrative Agent or the Lenders;

(iv) (A) for purposes of any consent to any amendment, waiver or modification of, or any action under, and for the purpose of any direction to the Administrative Agent or any Lender to undertake any action (or refrain from taking any action) under, this Agreement or any other Loan Document, each Affiliated Lender will be deemed to have consented in the same proportion as the Term Loan Lenders that are not Affiliated Lenders that have consented to such

matter, unless such matter requires the consent of all or all affected Lenders and adversely affects such Affiliated Lender more than other Term Loan Lenders in any material respect, provided, that in no event shall an amendment, waiver or modification cause the Commitments of any Affiliated Lender to increase, extend the due date for the payments of interest and scheduled amortization (including on the Term Loan Maturity Date) owing to any Affiliated Lender or reduce the amounts owing to any Affiliated Lender, in each case, without the consent of such Affiliated Lender, (B) for purposes of voting on any plan of reorganization or plan of liquidation pursuant to any Debtor Relief Laws (a 'Plan'), each Affiliated Lender hereby agrees (x) not to vote on such Plan, (y) if such Affiliated Lender does vote on such Plan notwithstanding the restriction in the foregoing clause (x), such vote will be deemed not to be in good faith and shall be "designated" pursuant to Section 1126(e) of the Bankruptcy Code (or any similar provision in any other Debtor Relief Laws), and such vote shall not be counted in determining whether the applicable class has accepted or rejected such Plan in accordance with Section 1126(c) of the Bankruptcy Code (or any similar provision in any other Debtor Relief Laws) and (z) not to contest any request by any party for a determination by the Bankruptcy Court (or other applicable court of competent jurisdiction) effectuating the foregoing clause (y), in each case under this clause (iv)(B) unless such Plan adversely affects such Affiliated Lender more than other Term Loan Lenders in any material respect, and (C) each Affiliated Lender hereby irrevocably appoints the Administrative Agent (such appointment being coupled with an interest) as such Affiliated Lender's attorney-in-fact, with full authority in the place and stead of such Affiliated Lender and in the name of such Affiliated Lender (solely in respect of Term Loans therein and not in respect of any other claim or status such Affiliated Lender may otherwise have), from time to time in the Administrative Agent's discretion to take any action and to execute any instrument that the Administrative Agent may deem reasonably necessary or appropriate to carry out the provisions of this clause (iv), including to ensure that any vote of such Affiliated Lender on any Plan is (at the Administrative Agent's sole option) withdrawn, otherwise not counted or made in accordance with the votes of the non-Affiliated Lenders in the same proportion as the non-Affiliated Lenders have voted;

(v) the aggregate principal amount of the Term Loans held at any one time by Affiliated Lenders may not exceed fifteen percent (15%) of the aggregate outstanding principal amount of the Term Loans;

(vi) Affiliated Lenders shall not constitute more than forty-nine percent (49%) of the total number of all Term Loan Lenders;

(vii) the Affiliated Lender will not be entitled to bring actions against the Administrative Agent, in its role as such (all of which claims each Affiliated Lender hereby releases and agrees to indemnify, defend and hold the Secured Parties harmless from) or receive advice of counsel or other advisors to the Administrative Agent or any other Lenders or challenge the attorney client privilege of their respective counsel;

(viii) the portion of any Loans held by Affiliated Lenders required to be held by Lenders in order for such Lenders to constitute "Required Lenders" or Required Term Loan Lenders shall be disregarded in determining Required Lenders or Required Term Loan Lender at any time or to meet any minimum head-count; and

(ix) to the extent that any Affiliated Lender receives an offer from the Borrowers to repurchase Term Loans on a pro rata basis, such Affiliated Lender shall be required to accept such offer.

Each Affiliated Lender that is a Term Loan Lender hereunder agrees to comply with the terms of this <u>Section 12.07(i)</u> (notwithstanding that it may be granted access to an electronic site established for the Lenders by the Administrative Agent), and agrees that in any subsequent assignment of all or any portion of its Term Loans it shall identify itself in writing to the assignee as an Affiliated Lender prior to the execution of such assignment.

For the avoidance of doubt, Lenders shall not be permitted to assign any other Term Loan (except in accordance with the foregoing provisions) or any Revolving Commitments or Revolving Loans (or grant any participation therein) to an Affiliated Lender and any purported assignment of or participation in each other Term Loan any Revolving Commitments or Revolving Loans to an Affiliated Lender shall be null and void (or at the Administrative Agent's election, any Commitments or Loans purported to be acquired by such Affiliated Lender shall be deemed permanently canceled and discharged).

(j) <u>Borrower Buybacks</u>. Notwithstanding anything in this Agreement to the contrary, any Term Loan Lender may, at any time, assign all or a portion of its Term Loans on a non-pro rata basis to the Borrowers in accordance with the procedures set forth on <u>Exhibit</u> <u>G</u>, pursuant to an offer made available to all Term Loan Lenders on a pro rata basis (a "<u>Dutch Auction</u>"), subject to the following limitations:

(i) the Borrowers shall represent and warrant, as of the date of the launch of the Dutch Auction and on the date of any such assignment, that neither the Borrower, their respective Affiliates nor any of their respective directors or officers has any material non-public information that has not been disclosed to the Term Loan Lenders generally (other than to the extent any such Term Loan Lender does not wish to receive material non-public information with respect to the Borrowers or its Subsidiaries or any of their respective securities) prior to such date or that the Borrowers cannot make such representation and warranty;

(ii) immediately and automatically, without any further action on the part of the Borrowers, any Lender, the Administrative Agent or any other Person, upon the effectiveness of such assignment of Term Loans from a Term Loan Lender to the Borrowers (or any of them), such Term Loans and any related Commitments and all rights and obligations as a Term Loan Lender related thereto shall, for all purposes under this Agreement, the other Loan Documents and otherwise, be deemed to be irrevocably prepaid, terminated, extinguished, cancelled and of no further force and effect, and the Borrowers shall neither obtain nor have any rights as a Term Loan Lender to under the other Loan Documents by virtue of such assignment

(iii) the Borrowers shall not use the proceeds of any Revolving Loans, Delayed-Draw Term Loans or other Loans for any such assignment; and

(iv) no Default or Event of Default shall have occurred and be continuing before or immediately after giving effect to such assignment.

# 12.08 Confidentiality.

(a) Each Loan Party acknowledges that (i) from time to time financial advisory, investment banking and other services may be offered or provided to it (in connection with this Agreement or otherwise) by each Lender or by one or more affiliates of such Lender and (ii) information delivered to each Lender by the Loan Parties may be provided to each such affiliate, it being understood that any such affiliate receiving such information shall be bound by the provisions of <u>Section 12.08(b)</u> as if it were a Lender under this Agreement.

Page 1	61
--------	----

Each of the Administrative Agent and each Lender severally (and not jointly) agrees to maintain the confidentiality of (b)the Information (as defined below), except that Information may be disclosed (i) to its Affiliates and Approved Funds and to its and its Affiliates' and Approved Funds' respective partners, directors, officers, employees, agents, consultants, counsel, accountants, advisors, controlling persons, managed funds, financing sources, actual and prospective investors, and other representatives (collectively, the "Representatives") (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), or to Rating Agencies, (ii) to the extent requested or required by any regulatory authority purporting to have jurisdiction over it (including any self-regulatory authority, such as the National Association of Insurance Commissioners), including, without limitation, any regulatory filings, (iii) to the extent required by applicable Laws or by any subpoena or similar judicial or legal process, (iv) to any other party to the Loan Documents, (v) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or Proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (vi) to (A) any actual or prospective assignee of a Lender, assignee or successor of Administrative Agent, or pledgee of or Participant in any of its rights or obligations under this Agreement, or (B) any actual or prospective counterparty to any swap or derivative transaction or Banking Services transactions relating to any Borrower or any other Loan Party and its obligations, or (C) any counterparty to any Subordination Agreement or Acceptable Intercreditor Agreement to the extent required by the provisions of such Subordination Agreement or Acceptable Intercreditor Agreement, provided that such parties agree to be bound by confidentiality provisions substantially similar to those hereunder, and to the Representatives of the foregoing parties in clauses (A) and (B) and (C), (vii) with the consent of the Borrower Representative, (viii) to the extent such Information (A) becomes publicly available other than as a result of a breach of this Section or (B) becomes available to the Administrative Agent, any Lender, or any of their respective Representatives on a non-confidential basis from a source other than the Loan Parties or (ix) for purposes of establishing a due diligence defense. The terms of this provision shall supersede and replace any previous agreement regarding the confidentiality of the Information and shall terminate upon the termination of the Commitments and the payment of the Obligations.

For purposes of this Section, (i) "Information" means, all information received from any Loan Party or any of its Subsidiaries relating to any Loan Party or any of its Subsidiaries or any of their respective businesses, other than any such information that is available to or in the possession of the Administrative Agent, any other Secured Party or their Representatives on a non-confidential basis prior to disclosure by any Loan Party or any of its Subsidiaries and (ii) "Rating Agencies" means Moody's, S&P, Fitch Ratings Ltd., or any other nationally recognized rating agency or service. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information. Notwithstanding anything herein to the contrary, "Information" shall not include, and the Administrative Agent and each other Secured Party may disclose without limitation of any kind, any information with respect to the "tax treatment" and "tax structure" (in each case, within the meaning of Treasury Regulation Section 1.6011-4) of the transactions contemplated hereby and all materials of any kind (including opinions or other tax analyses) that are provided to the Administrative Agent or such other Secured Party relating to such tax treatment or tax structure of the transaction as well as other information, this sentence shall only apply to such portions of the document or similar item that relate to the tax treatment or tax structure of the Loans and transactions contemplated hereby.

(c) No Borrower or Affiliate thereof will issue any press releases or other public disclosure using the name of the Administrative Agent or its Affiliates or any other Lender or its Affiliates or referring to this Agreement or the other Loan Documents without at least three (3) Business Days' prior

notice to the Administrative Agent or such Lender and without the prior written consent of the Administrative Agent or such Lender unless (and only to the extent that) such Borrower or Affiliate is required to do so under law and then, in any event, such Borrower or Affiliate will consult with such Lender before issuing such press release or other public disclosure. The Borrowers hereby consent to the publication by any Secured Party of a tombstone or similar advertising material relating to the financing transactions contemplated by this Agreement (such material may, without limitation, include a description of the Loan Parties and the use of any identifying trademark or other marks of a Loan Party). Each Secured Party reserves the right to provide to industry trade organizations information necessary and customary for inclusion in league table measurements.

**12.09** Set-off. In addition to any rights and remedies of the Lenders provided by law, upon the occurrence and during the continuance of any Event of Default, each Lender and any Affiliate of a Lender is authorized at any time and from time to time, with the prior written consent of the Administrative Agent, but without prior notice to the Borrowers or any other Loan Party (any such notice being waived by the Borrowers on their own behalf and on behalf of each Loan Party), to the fullest extent permitted by law, to set off and apply any and all deposits (general or special, time or demand, provisional or final) at any time held by, and other indebtedness at any time owing by, such Lender or such Affiliate to or for the credit or the account of the respective Loan Parties against any and all Obligations owing to such Lender hereunder or under any other Loan Document, now or hereafter existing, irrespective of whether or not the Administrative Agent or such Lender shall have made demand under this Agreement or any other Loan Document and although such Obligations may be contingent or unmatured or denominated in a currency different from that of the applicable deposit or indebtedness. Each Lender agrees promptly to notify the Borrowers and the Administrative Agent after any such set-off and application made by such Lender, provided that the failure to give such notice shall not affect the validity of such set-off and application. Any Lender exercising a right to set-off shall purchase for cash (and the other Lender's Pro Rata Share of the Obligations.

**12.10** Interest Rate Limitation Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable Law (the "<u>Maximum Rate</u>"). If the Administrative Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to the Borrowers. In determining whether the interest contracted for, charged, or received by the Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

12.11 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

**12.12** <u>Integration</u>. This Agreement, together with the other Loan Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter, including the Proposal Letter (other than in respect of any terms that expressly survive the termination or expiration thereof following the delivery of this Agreement), <u>provided</u> that the Fee Letter shall survive the effectiveness of this Agreement and the initial Credit Extensions hereunder and shall continue to be in full force and effect after the Closing Date in accordance with their terms. In the event of any conflict between the provisions of this Agreement and

those of any other Loan Document, the provisions of this Agreement shall control, <u>provided</u> that the inclusion of supplemental rights or remedies in favor of the Administrative Agent or the Lenders in any other Loan Document shall not be deemed a conflict with this Agreement. Each Loan Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.

**12.13** <u>Survival of Representations and Warranties</u>. All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent and each Lender, regardless of any investigation made by the Administrative Agent or any Lender or on their behalf and notwithstanding that the Administrative Agent or any Lender may have had notice or knowledge of any Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied.

**12.14** <u>Severability</u>. If any provision of this Agreement or any other Loan Document is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Loan Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

**12.15 Replacement of Lenders.** If (i) any Lender requests compensation under <u>Section 3.04</u>, (ii) the <u>Borrower isBorrowers are</u> required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to<u>Section 3.01</u>, unless such Lender has waived its right to such additional amount, (iii) a Lender (a '<u>Non-Consenting Lender</u>'') does not consent to a proposed change, waiver, discharge or termination with respect to any Loan Document that has been approved by the Required Lenders as provided in <u>Section 12.01</u> but requires unanimous consent of all Lenders or all Lenders directly affected thereby (as applicable), (iv) it is, and continues to be, unlawful for any Lender to fund or maintain LIBOR Loans, as provided in <u>Section 3.02</u>, or (v) any Lender is a Defaulting Lender, then Administrative Agent or the Borrower Representative may, at its sole option, expense and effort, and upon notice to such Lender and, in the case of an election made by the Borrower Representative, the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, <u>Section 12.07</u>), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), provided that:

(a) the Borrowers shall have paid to the Administrative Agent the assignment fee specified in <u>Section 12.07(b)</u>, if any;

(b) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans and Unreimbursed Amounts, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts under <u>Section 3.05</u>) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrowers (in the case of all other amounts);

(c) in the case of any such assignment resulting from a claim for compensation under <u>Section 3.04</u> or payments required to be made pursuant to <u>Section 3.01</u>, such assignment will result in a reduction in such compensation or payments thereafter;

(d) such assignment does not conflict with applicable Laws; and

(e) in the case of any such assignment resulting from a Non-Consenting Lender's failure to consent to a proposed change, waiver, discharge or termination with respect to any Loan Document, the applicable replacement Lender consents to the proposed change, waiver, discharge or termination, <u>provided</u> that the failure by such Non-Consenting Lender to execute and deliver an Assignment and Assumption shall not impair the validity of the removal of such Non-Consenting Lender and the mandatory assignment of such Non-Consenting Lender's Commitments and outstanding Loans and participations in outstanding Letters of Credit pursuant to this<u>Section 12.15</u> shall nevertheless be effective without the execution by such Non-Consenting Lender of an Assignment and Assumption.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrowers or Administrative Agent to require such assignment and delegation cease to apply.

# 12.16 Governing Law.

(a) THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HERETO AND THERETO SHALL BE CONSTRUED IN ACCORDANCE WITH AND GOVERNED BY THE LAW OF THE STATE OF NEW YORK WITHOUT REGARD TO ANY CONFLICTS OF LAWS PRINCIPLES THEREOF THAT WOULI CALL FOR THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.

(b) Each Loan Party irrevocably and unconditionally agrees that it will not commence any action, litigation or proceeding of any kind or description, whether in law or equity, whether in contract or in tort or otherwise, against any Secured Party of the foregoing in any way relating to this Agreement or any other Loan Document or the transactions relating hereto or thereto, in any forum other than the courts of the State of New York sitting in New York County, and of the United States District Court of the Southern District of New York, and any appellate court from any thereof, and each of the parties hereto irrevocably and unconditionally submits to the jurisdiction of such courts and agrees that all claims in respect of any such action, litigation or proceeding may be heard and determined in such New York State court or, to the fullest extent permitted by applicable Law, in such federal court. Each of the parties hereto agrees that a final judgment in any other manner provided by law. Nothing in this Agreement or in any other Loan Document shall affect any right that any Secured Party may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document spans any Loan Party or its properties in the courts of any jurisdiction. Each Loan Party hereby irrevocably waives, to the fullest extent not prohibited by applicable Law, any objection that it may now or hereafter have to the laying of the venue of any such proceeding brought in such a court and any claim that any such proceeding brought in such a court has been brought in an inconvenient forum.

**12.17** <u>Waiver of Right to Trial by Jury</u>. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY LOAN DOCUMENT OR IN ANY WAY CONNECTED WITH OF RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO ANY LOAN DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING O HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREB' AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED B' COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS

# AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

**12.18** <u>USA Patriot Act Notice</u> Each Lender and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Loan Parties that pursuant to the requirements of the Patriot Act, it is required to obtain, verify and record information that identifies the Loan Parties, which information includes the name and address of the Loan Parties and other information that will allow such Lender or the Administrative Agent, as applicable, to identify the Loan Parties in accordance with the Patriot Act.

12.19 Nonliability of Lenders. The relationship between the Loan Parties on the one hand and the Lenders and the Administrative Agent on the other hand shall be solely that of borrower or guarantor, as applicable, and lender. Neither the Administrative Agent nor any Lender or other Secured Party has any fiduciary relationship with or duty to any Loan Party arising out of or in connection with this Agreement or any of the other Loan Documents, and the relationship between the Loan Parties, on the one hand, and the Administrative Agent and the Lenders and other Secured Parties, on the other hand, in connection herewith or therewith is solely that of debtor and creditor. Neither the Administrative Agent nor any Lender or any other Secured Party undertakes any responsibility to any Loan Party to review or inform any Loan Party of any matter in connection with any phase of any Loan Party's business or operations. The Loan Parties agree that neither the Administrative Agent nor any Lender or other Secured Party shall have liability to any Loan Party (whether sounding in tort, contract or otherwise) for losses suffered by any Loan Party in connection with, arising out of, or in any way related to the transactions contemplated and the relationship established by the Loan Documents, or any act, omission or event occurring in connection therewith, unless it is determined in a final non-appealable judgment by a court of competent jurisdiction that such losses resulted from the gross negligence or willful misconduct of the party from which recovery is sought. NO SECURED PARTY SHALL BE LIABLE FOR ANY DAMAGES ARISING FROM THE USE BY OTHERS OF ANY INFORMATION OR OTHER MATERIALS OBTAINED THROUGH INTRALINKS OF OTHER SIMILAR INFORMATION TRANSMISSION SYSTEMS IN CONNECTION WITH THIS AGREEMENT, NO SHALL ANY SECURED PARTY HAVE ANY LIABILITY WITH RESPECT TO, AND THE BORROWER REPRESENTATIVE ON BEHALF OF ITSELF AND EACH OTHER LOAN PARTY, HEREBY WAIVES, RELEASES AND AGREES NOT TO SUE FOR ANY SPECIAL, PUNITIVE, INDIRECT OR CONSEQUENTIAL DAMAGES RELATING T( THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR ARISING OUT OF ITS ACTIVITIES IN CONNECTIO HEREWITH OR THEREWITH (WHETHER BEFORE OR AFTER THE CLOSING DIAGEL) oan Parties acknowledge that they have been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Loan Documents to which it is a party. No joint venture is created hereby or by the other Loan Documents or otherwise exists by virtue of the transactions contemplated hereby among the Administrative Agent, Lenders or among the Loan Parties and the Lenders and the Administrative Agent. The Loan Parties further acknowledge that each Lender or one or more of its affiliates may be a financial and securities firm and that such Lender or such affiliates may from time to time effect transactions, for its own or its affiliates' account or the account of customers, and hold positions in loans, securities or options on loans or securities of a Loan Party and its affiliates and of other companies that may be the subject of the transactions contemplated by this Agreement. The Loan Parties further acknowledge and agree that (a) no fiduciary, advisory or agency relationship between a Loan Party and the Secured Parties (or any of them) is intended to be or has been created in respect of any of the transactions contemplated by this Agreement, irrespective of whether any Secured Party or its affiliates has advised or is advising any Loan Party on other matters, (b) the Secured Parties, on the one hand, and the Loan Parties, on the other hand, have an arms-length business relationship that does not directly or indirectly give rise to, nor does any Loan Party rely on, any fiduciary duty on any Secured Party's part, (c) each Loan Party is capable of evaluating and understanding, and each Loan Party understands and accepts, the terms, risks and conditions of the transactions contemplated by this

Agreement and the other Loan Documents, (d) the Secured Parties have not provided any legal, accounting, regulatory or tax advice with respect to the transactions and each Loan Party has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate and it is not relying on CIT for such advice, (e) the Loan Parties have been advised that the Secured Parties and their respective affiliates are or may be engaged in a broad range of transactions that may involve interests that differ from any Loan Party's interests and that the Secured Parties and their respective affiliates have no obligation to disclose such interests and transactions to any Loan Party by virtue of any fiduciary, advisory or agency relationship, (f) the Loan Parties will not assert and waive, to the fullest extent not prohibited by law, any claims any Loan Party may have against any Secured Parties shall have no liability (whether direct or indirect) to any Loan Party in respect of such a fiduciary duty claim or to any Person asserting a fiduciary duty claim on behalf of or in right of any Loan Party, including a Loan Party's stockholders, employees or creditors, and (g) should the Secured Parties or their respective affiliates have no therwise diminish such Loan Party's or such Loan Party's affiliates' obligations thereunder or with respect thereto.

12.20 <u>Acknowledgment and Consent to Bail-In of EEA Financial Institutions</u> Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among the parties hereto, each party hereto acknowledges that any liability of any EEA Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of an EEA Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

(a) the application of any Write-Down and Conversion Powers by an EEA Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an EEA Financial Institution; and

- (b) the effects of any Bail-In Action on any such liability, including, if applicable:
  - (i) a reduction in full or in part or cancellation of any such liability;

(ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such EEA Financial Institution, its parent entity, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or

(iii) the variation of the terms of such liability in connection with the exercise of the Write-Down and Conversion Powers of any EEA Resolution Authority.

12.21 <u>Effect of Amendment and Restatement</u> Upon the effectiveness of this Agreement, the debt, obligations and other liabilities (including, without limitation, interest and fees accrued on or prior to the Closing Date) governed by the Original Loan Agreement shall continue to be in full force and effect, but shall be governed by the terms and conditions set forth in this Agreement. All Obligations evidenced by the Original Loan Agreement, together with any and all additional Obligations incurred by any Loan Party hereunder or under any of the other Loan Documents (whether executed and/or delivered in connection with the Original Loan Agreement, this Agreement or otherwise), shall continue to be secured by all of the pledges and grants of security interests and Liens provided in connection with the Original Loan Agreement (and, from and after the Closing Date, shall be secured by all of the pledges and grants of security interests and grants of security interests and Liens provided in connection with this Agreement). Each Loan Party hereby reaffirms its Obligations

under each Loan Document to which it is party (whether executed and/or delivered in connection with the Original Loan Agreement, this Agreement or otherwise), in each case, as amended, restated, supplemented or otherwise modified from time to time. Each Loan Party further agrees that all of the Loan Documents (whether executed and/or delivered in connection with the Original Loan Agreement, this Agreement or otherwise) remain in full force and effect, as amended, restated, supplemented or otherwise modified, following the execution and delivery of this Agreement and that all references to the "Loan Agreement" in all such Loan Documents shall be deemed to refer to this Agreement. The execution and delivery of this Agreement shall constitute an amendment and restatement, but not a novation, substitution or repayment, of any of the Obligations incurred prior to the Closing Date, whether pursuant to the Original Loan Agreement or other return of any principal, interest, fee or other payments made prior to the Closing Date pursuant to the Original Loan Agreement by any Loan Party to or for the benefit of the Administrative Agent or any Lender. Each Loan Party, jointly and severally, represents and warrants that, as of the Closing Date, there are no claims or offsets against, or defenses or counterclaims to, their Obligations (or the Obligations of any Loan Party) under the Original Loan Agreement or any of the Loan Documents (and hereby expressly waives any and all such claims, offsets, defenses and counterclaims).

# **ARTICLE 13**

### APPOINTMENT OF THE BORROWER REPRESENTATIVE; JOINT AND SEVERAL LIABILITY OF THE BORROWERS

**13.01 Borrower Representative**. Each Borrower hereby irrevocably appoints the Borrower Representative, as the agent for such Borrower on its behalf, to (i) request Loans from the Lenders, (ii) request L/C Issuer to issue Letters of Credit and Support Providers to issue Support Agreements, (iii) to give and receive notices under the Loan Documents and (iv) take all other action which the Borrower Representative or the Borrowers are permitted or required to take under this Agreement.

### 13.02 Joint and Several Liability of Borrowers.

(a) Joint and Several Liability Each Borrower hereby agrees that such Borrower is jointly and severally liable for, and hereby absolutely and unconditionally guarantees to the Administrative Agent and other Secured Parties and their respective successors and assigns, the full and prompt payment (whether at stated maturity, by acceleration or otherwise) and performance of, all Obligations owed or hereafter owing to the Administrative Agent and other Secured Parties by each other Borrower. Each Borrower agrees that its guaranty obligation hereunder is a continuing guaranty of payment and performance and not of collection, that its obligations under this Section 13.02 shall not be discharged until payment and performance, in full, of the Obligations has occurred, and that its obligations under this Section 13.02 shall be absolute, unconditional and irrevocable, irrespective of, and unaffected by, (i) the genuineness, validity, regularity, enforceability or any future amendment of, or change in, any Obligation or any agreement, document or instrument to which any Borrower is or may become a party; (ii) the absence of any action to enforce any Obligation or the waiver or consent by the Administrative Agent or any other Secured Party with respect to any of the provisions governing any Obligation; (iii) the insolvency of any Borrower or Subsidiary; and (iv) any other action or circumstances that might otherwise constitute a legal or equitable discharge or defense of a surety or guarantor. Each Borrower shall be regarded, and shall be in the same position, as principal debtor with respect to the Obligations guaranteed hereunder.

(b) <u>Waivers by Borrowers</u>. Each Borrower expressly waives all rights it may have now or in the future under any statute, or at common law, or at law or in equity, or otherwise, to compel the

Administrative Agent or other Secured Parties to marshal assets or to proceed in respect of the Obligations guaranteed hereunder against any other Borrower or Subsidiary, any other party or against any security for the payment and performance of the Obligations before proceeding against, or as a condition to proceeding against, such Borrower. Each Borrower consents and agrees that the Administrative Agent or the other Secured Parties may, at any time and from time to time, without notice or demand, whether before or after an actual or purported termination, repudiation or revocation of this Agreement by any Borrower, and without affecting the enforceability or continuing effectiveness hereof as to such Borrower: (i) with the consent of the other Borrowers, supplement, restate, modify, amend, increase, decrease, extent, renew or otherwise change the time for payment or the terms of this Agreement or any part thereof, including any increase or decrease of the rate(s) of interest thereon; (ii) with the consent of the other Borrowers, supplement, restate, modify, amend, increase, decrease, or enter into or give any agreement with respect to, this Agreement or any part thereof, or any of the Security Documents; (iii) waive, approve or consent to any action, condition, covenant, default, remedy, right, representation or term of this Agreement or any other Loan Document; (iv) accept partial payments; (v) release, reconvey, terminate, waive, abandon, fail to perfect, subordinate, exchange, substitute, transfer or enforce any security or guarantees, and apply any security and direct the order or manner of sale thereof as the Administrative Agent or Lenders in their sole and absolute discretion may determine; (vi) release any person from any personal liability with respect to this Agreement or any part thereof; (vii) settle, release on terms satisfactory to the Required Lenders or by operation of applicable Laws or otherwise liquidate or enforce any security or guaranty in any manner, consent to the transfer of any security and bid and purchase at any sale; or (viii) consent to the merger, change or any other restructuring or termination of the corporate or partnership existence of any Borrower or any other person, and correspondingly restructure the obligations evidenced hereby, and any such merger, change, restructuring or termination shall not affect the liability of any Borrower or the continuing effectiveness hereof, or the enforceability hereof with respect to all or any part of the obligations evidenced hereby. It is agreed among each Borrower, the Administrative Agent and Lenders that the foregoing consents and waivers are of the essence of the transaction contemplated by this Agreement and the other Loan Documents and that, but for the provisions of this Section 13.02 and such waivers, the Administrative Agent and Lenders would decline to enter into this Agreement.

(c) <u>Benefit of Guaranty</u>. Each Borrower agrees that the provisions of this <u>Section 13.02</u> are for the benefit of the Administrative Agent and the other Secured Parties and their respective successors, transferees, endorsees and assigns, and nothing herein contained shall impair, as between any other Borrower and the Administrative Agent or the other Secured Parties, the obligations of such other Borrower under the Loan Documents.

(d) <u>Waiver of Subrogation, Etc.</u> Notwithstanding anything to the contrary in this Agreement or in any other Loan Document, and except as set forth in <u>Section 13.02(g)</u>, each Borrower hereby expressly and irrevocably waives until after the Termination Date any and all rights at law or in equity to subrogation, reimbursement, exoneration, contribution, indemnification or set off and any and all defenses available to a surety, guarantor or accommodation co-obligor, including, without limitation, any defense based upon any statute or rule of law that provides that the obligation of a surety must be neither larger in amount nor in other respects more burdensome than that of the principal, and any defense of the statute of limitations in any action hereunder or in any action for the collection or performance of any obligations hereby guaranteed. Each Borrower acknowledges and agrees that this waiver is intended to benefit the Administrative Agent and Lenders and other Secured Parties and shall not limit or otherwise affect such Borrower's liability hereunder or the enforceability of this <u>Section 13.02</u>, and that the Administrative Agent, the Lenders and the other Secured Parties and their respective successors and assigns are intended third party beneficiaries of the waivers and agreements set forth in this <u>Section 13.02(d)</u>.

(e) <u>Election of Remedies</u>. If the Administrative Agent or any other Secured Party may, under applicable Law, proceed to realize its benefits under any of the Loan Documents, the

Administrative Agent or any other Secured Party may, at its sole option, determine which of its remedies or rights it may pursue without affecting any of its rights and remedies under this <u>Section 13.02</u>. If, in the exercise of any of its rights and remedies, the Administrative Agent or any other Secured Party shall forfeit any of its rights or remedies, including its right to enter a deficiency judgment against any Borrower or any other Person, whether because of any applicable Laws pertaining to "election of remedies" or the like, each Borrower hereby consents to such action by the Administrative Agent or such other Secured Party and waives any claim based upon such action, even if such action by the Administrative Agent or such other Secured Party shall result in a full or partial loss of any rights of subrogation that each Borrower might otherwise have had but for such action by the Administrative Agent or such other Secured Party to seek a deficiency judgment against any Borrower shall not impair any other Borrower's obligation to pay the full amount of the Obligations.

(f) <u>Limitation</u>. Notwithstanding any provision herein contained to the contrary, each Borrower's liability under this <u>Section</u> <u>13.02</u> (which liability is in any event in addition to amounts for which such Borrower is primarily liable under <u>Article 2</u>) shall be limited to an amount not to exceed as of any date of determination the greater of:

(i) the net amount of all Loans advanced to any other Borrower under this Agreement and then re-loaned or otherwise transferred to, or for the benefit of, such Borrower; and

(ii) the amount that could be claimed by the Administrative Agent and the other Secured Parties from such Borrower under this <u>Section 13.02</u> without rendering such claim voidable or avoidable under Section 548 of the Bankruptcy Code or under any applicable state Uniform Fraudulent Transfer Act, Uniform Fraudulent Conveyance Act or similar statute or common law after taking into account, among other things, such Borrower's right of contribution and indemnification from each other Borrower under <u>Section 13.02(h)</u>.

## (g) <u>Contribution with Respect to Guaranty Obligations</u>.

(i) To the extent that any Borrower shall make a payment under this <u>Section 13.02</u> of all or any of the Obligations (other than Obligations related to Loans and other extensions of credit made directly or indirectly to that Borrower, or on such Borrower's behalf, in which case such Borrower shall be primarily liable) (a "<u>Guarantor Paymenf</u>") that, taking into account all other Guarantor Payments then previously or concurrently made by any other Borrower, exceeds the amount that such Borrower would otherwise have paid if each Borrower had paid the aggregate Obligations satisfied by such Guarantor Payment in the same proportion that such Borrower's "Allocable Amount" (as defined below) (as determined immediately prior to such Guarantor Payment) bore to the aggregate Allocable Amounts of each of the Borrowers as determined immediately prior to the making of such Guarantor Payment, then, following indefeasible payment in full in cash of the Obligations and termination of the Commitments, such Borrower shall be entitled to receive contribution and indemnification payments from, and be reimbursed by, each other Borrower for the amount of such excess, pro rata based upon their respective Allocable Amounts in effect immediately prior to such Guarantor Payment.

(ii) As of any date of determination, the "<u>Allocable Amount</u>" of any Borrower shall be equal to the maximum amount of the claim that could then be recovered from such Borrower under this <u>Section 13.02</u> without rendering such claim voidable or avoidable under Section 548 of the Bankruptcy Code or under any applicable state Uniform Fraudulent Transfer Act, Uniform Fraudulent Conveyance Act or similar statute or common law.

(iii) This <u>Section 13.02(g)</u> is intended only to define the relative rights of Borrowers and nothing set forth in this <u>Section 13.02(h)</u> is intended to or shall impair the obligations of Borrowers, jointly and severally, to pay any amounts as and when the same shall become due and payable in accordance with the terms of this Agreement, including <u>Section 13.02(a)</u>. Nothing contained in this <u>Section 13.02(h)</u> shall limit the liability of any Borrower to pay the Loans made directly or indirectly to that Borrower, or on such Borrower's behalf, and accrued interest, fees and expenses with respect thereto for which such Borrower shall be primarily liable.

(iv) The parties hereto acknowledge that the rights of contribution and indemnification hereunder shall constitute assets of the BorrowerBorrowers to which such contribution and indemnification is owing.

(v) The rights of the indemnifying Borrowers against other Borrowers under this <u>Section 13.02(h)</u> shall be exercisable on or after the Termination Date, but shall in all respects be subordinate to any Obligations owing to any Secured Party.

(h) <u>Liability Cumulative</u>. The liability of Borrowers under this <u>Section 13.02</u> is in addition to and shall be cumulative with all liabilities of each Borrower to the Administrative Agent and Lenders under this Agreement and the other Loan Documents to which such Borrower is a party or in respect of any Obligations or obligation of the other Borrower, without any limitation as to amount, unless the instrument or agreement evidencing or creating such other liability specifically provides to the contrary.

(i) <u>Stay of Acceleration</u> If acceleration of the time for payment of any amount payable by the Borrowers under this Agreement is stayed upon the insolvency, bankruptcy or reorganization of any of the Borrowers, all such amounts otherwise subject to acceleration under the terms of this Agreement shall nonetheless be payable jointly and severally by the <u>BorrowerBorrowers</u> hereunder forthwith on demand by the Administrative Agent made at the request of the Required Lenders.

(j) <u>Benefit to Borrowers</u>. All of the Borrowers and their Subsidiaries are engaged in related businesses and integrated to such an extent that the financial strength and flexibility of each such Person has a direct impact on the success of each other Person. Each Borrower and each Subsidiary will derive substantial direct and indirect benefit from the extension of credit hereunder.

Signature Pages Follow

**BORROWERS:** 

# ADAPTHEALTH LLC

By: Name: Luke McGee Title: Chief Executive Officer

ADAPTHEALTH INTERMEDIATE HOLDCO LLC AIRCARE HOME RESPIRATORY, LLC AMERICOAST MARYLAND LLC ASSOCIATED HEALTHCARE SYSTEMS, INC. BENNETT MEDICAL SERVICES LLC **BRADEN PARTNERS, L.P. CLEARVIEW MEDICAL INCORPORATED CPAP SOLUTIONS, LLC** CPAP2ME, INC. FAMILY HOME MEDICAL SUPPLY LLC FIRST CHOICE DME LLC FIRST CHOICE HOME MEDICAL EQUIPMENT, LLC GOULD'S DISCOUNT MEDICAL, LLC HEALTH SOLUTIONS LLC HOME MEDICAL EXPRESS, INC. HOME MEDISERVICE, LLC HOMETOWN HOME HEALTH, LLC LMI DME HOLDINGS LLC **MED STAR SURGICAL & BREATHING** EQUIPMENT INC. MED WAY MEDICAL, INC. MEDBRIDGE HOME MEDICAL LLC **MED-EQUIP, INC.** MEDSTAR HOLDINGS LLC **OCEAN HOME HEALTH OF PA LLC** OCEAN HOME HEALTH SUPPLY LLC OGLES OXYGEN, LLC **ORBIT MEDICAL OF PORTLAND, INC.** PALMETTO OXYGEN, LLC PPS HME HOLDINGS LLC **PPS HME LLC ROBERTS HOME MEDICAL, LLC ROYAL DME LLC ROYAL MEDICAL SUPPLY INC. SLEEPEASY THERAPEUTICS, INC.** SOUND OXYGEN SERVICE LLC TOTAL RESPIRATORY, LLC TRICOUNTY MEDICAL EQUIPMENT AND SUPPLY, LLC VERUS HEALTHCARE, INC. **VERUS HEALTHCARE, LLC** 

By:

Name: Luke McGee Title: Chief Executive Officer

CIT FINANCE LLC, as Administrative Agent

By: Name: Andrew C. Sepe

Title: Director

 $\underline{ Third\ Amended\ and\ Restated\ Credit\ and\ Guaranty\ Agreement-Signature\ Page}$ 

CIT FINANCE LLC, as a Lender

By: Name: Andrew C. Sepe Title: Director

**REGIONS BANK**, as a Lender

By: Name: Title:

SUNTRUST BANK, as a Lender

By: Name: Title:

CITIZENS BANK, N.A., as a Lender

By: Name: Title:

BANKUNITED, N.A., as a Lender

By: Name: Title:

FIRST MIDWEST BANK, as a Lender

By: Name: Title:

WEBSTER BANK, NATIONAL ASSOCIATION, as a Lender

By: Name: Title:

BANCALLIANCE INC., as a Lender

By: Alliance Partners LLC, its: Attorney-in Fact

By:

Name: Title:

WILMINGTON SAVINGS FUND SOCIETY, FSB, as a Lender

By: Name: Title:

CAPSTAR BANK, as a Lender

By: Name: Title:

# PROVIDENT BANK, as a Lender

<del>By:</del> <del>Name:</del> <del>Title:</del>

#### CERTIFICATION PURSUANT TO RULES 13A-14 AND 15D-14 UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED

I, Luke McGee, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of AdaptHealth Corp.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements and other financial information included in this report, fairly present, in all material respects, the financial condition, results of operations, and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of the financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize, and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

May 8, 2020

/s/ Luke McGee

Chief Executive Officer and Director (Principal Executive Officer)

#### CERTIFICATION PURSUANT TO RULES 13A-14 AND 15D-14 UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED

I, Gregg Holst, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of AdaptHealth Corp.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements and other financial information included in this report, fairly present, in all material respects, the financial condition, results of operations, and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of the financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize, and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

May 8, 2020

/s/ Gregg Holst

Gregg Holst Chief Financial Officer (Principal Financial and Accounting Officer)

#### CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS REQUIRED BY SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of AdaptHealth Corp. (the "Company") on Form 10-Q for the quarter ended March 31, 2020, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned hereby certify that to the best of our knowledge:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

May 8, 2020	/s/ Luke McGee Luke McGee	Chief Executive Officer and Director (Principal Executive Officer)
May 8, 2020	/s/ Gregg Holst Gregg Holst	Chief Financial Officer (Principal Financial and Accounting Officer)